Genetic Determinants of Cognitive Function and Age-Related Brain Changes Maaike Schuur

Genetic Determinants of Cognitive Function and Age-Related Brain Changes <u>Maaike Schuur</u>

zafing FRASMUS UNIVERS

TEIT ROTTERDAM



ISBN: Cover & Layout: Printed by: Copyright: 978-90-8559-993-7 S. P. Schuur voor CIVIL-X, New York, USA Optima Grafische Communicatie, Rotterdam M. Schuur, Rotterdam, the Netherlands, 2010

All rights reserved. No part of this book may be reproduced, stored in a retrieval system, or transmitted, in any form or by any means, electronic, mechanical, photocopying, recording, or otherwise, without the prior written permission of the holder of the copyright.

Titel proefschrift

Genetic Determinants of Cognitive Function and Age-Related Brain Changes

Vertaling van de titel

Genetische determinanten van cognitief functioneren en aan veroudering gerelateerde hersenveranderingen

Proefschrift

ter verkrijging van de graad van doctor aan de Erasmus Universiteit Rotterdam op gezag van de rector magnificus Prof.dr. H.G. Schmidt en volgens besluit van het College voor Promoties.

De openbare verdediging zal plaatsvinden op woensdag 12 mei 2010 om 15:30 uur

door

Maaike Schuur

geboren te Anloo

Frafino ERASMUS UNIVERSITEIT ROTTERDAM

Promotiecommissie

Promotoren:	Prof.dr.ir. C.M. van Duijn		
	Prof.dr. B.A. Oostra		
Overige leden:	Dr. G.J. Biessels		
	Dr. V. Bonifati		
	Prof.dr. D.I. Boomsma		
Copromotor:	Dr. J.C. van Swieten		

Table of contents

Part I	Introduction	
Chapter 01	General introduction, outline, and scope of the thesis	13
Part II	New determants of cognitive function	
Chapter 02	Insulin-resistance and metabolic syndrome are related to executive function in women in a large family-based study	25
Chapter 03	The apolipoprotein E gene and its age-specific effects on cognitive function	43
Chapter 04	Genome-wide linkage screen of cognitive function identifies susceptible chromosomal regions	59
Chapter 05	Genome-wide association study of cognitive executive functions: Meta-analysis of the CHARGE consortium	77
PART III	Genetic determants of Alzheimer's disease	
Chapter 06	Cathepsin D gene and the risk of Alzheimer's disease: A population-based study and meta-analysis	99
Chapter 07	Transferrin and HFE genes interact in Alzheimer's disease risk: the Epistasis Project	115
PART IV	Genetic determinants of age-related brain pathology in hypertensive patients	
Chapter 08	Genetic risk factors for cerebral small vessel disease in hypertensive patients from a genetically isolated population	137
Chapter 09	The relation of the sortilin-related receptor gene to hippocampal volume and plasma amyloid beta levels in a family-based study of hypertensive patients	147
Chapter 10	Polymorphisms of the renin-angiotensin system and alfa-adducin in associate to circulating amyloid beta levels	163

Part IV	General discussion and summary	
Chapter 11	Findings of this thesis	179
Chapter 12	Grading the credibility of genetic associations in Alzheimer's disease using the Venice criteria: practical considerations following from the Alzgene database	189
Chapter 13	Future research	205
Chapter 14	Nederlandse samenvatting en conclusies	213

Dankwoord	221
Curriculum Vitae	227
List of Publications	231
Portfolio Summary	235

Publications and manuscripts based on the studies described in this thesis

- Schuur M, Henneman P, van Swieten JC, M.C. Zillikens, de Koning I, Chapter 02 Janssens ACJW, Witteman JCM, Aulchenko YS, Frants RR, Oostra BA, Willems van Dijk K, van Duijn CM. Insulin-resistance and the metabolic syndrome are related to executive function in women in a large family-based study. European Journal of Epidemiology, In press Chapter 03 ILiu F, Pardo LM, Schuur M, Sanchez-Juan P, Isaacs A, Sleegers K, de Koning I, Zorkoltseva IV, Axenovich TI, Witteman JC, Janssens AC, van Swieten JC, Aulchenko YS, Oostra BA, van Duijn CM The apolipoprotein E gene and its age-specific effects on cognitive function. Neurobiol Aging Epub 2008 Nov 10 Chapter 04 Schuur M, Amin N, Ikram MA, Hommel D, Janssens ACJW, Zorkoltseva IV, Kirichenko A, de Koning I, Axenovich TI, Hofman A, Aulchenko YS, Breteler MMB, Oostra BA, van Swieten JC, van Duijn CM. Genome-wide linkage screen of cognitive function identifies susceptible chromosomal regions. Submitted Chapter 05 Schuur M, Bressler J, Debette S, Fitzpatrick A, Vernon Smith A, Petrovic K, Bish J, Li G, Qiong Yang, Ikram MA, de Koning I, van Swieten JC, Oostra BA, Hofman A, Schmidt R, Seshadri S, Lenore Launer, Breteler MMB, van Duijn CM, Mosley T. Genome-wide association study of cognitive executive functions: Meta-analysis of the CHARGE consortium. In preparation Chapter 06 Schuur M, Ikram MA, van Swieten JC, Isaacs A, Vergeer-Drop JM, Hofman A, Oostra BA, Breteler MMB, van Duijn CM. Cathepsin D gene and the risk of Alzheimer's disease: A population-based
 - study and meta-analysis. Neurobiol Aging Epub 2009 Nov 17

Chapter 07	Lehmann DJ, Schuur M, Warden DR, Hammond N, Belbin O, Kölsch H,			
	Lehmann MG, Wilcock GK, Brown K, Kehoe PG, Morris CM, Barker R,			
	Coto E, Alvarez V, Deloukas P, Mateo I, Gwilliam R, Combarros O,			
	Arias-Vásquez A, Ikram MA, Aulchenko YS, Breteler MMB, van Duijn CM,			
	Heun R, Cortina-Borja M, Morgan K, Robson K, Smith AD.			
	Transferrin and HFE genes interact in Alzheimer's disease risk: the Epistasis			
	Project. Submitted			

Chapter 08 Schuur M, van Swieten JC, Schol-Gelok S, Ikram MA, Vernooij MW, Liu F, Isaacs A, de Boer R, de Koning I, Niessen WJ, Vrooman H, Oostra BA, van der Lugt A, Breteler MMB, van Duijn CM.
Genetic risk factors for cerebral small vessel disease in hypertensive patients from a genetically isolated population. Journal of Neurology, Neurosurgery and Psychiatry, *In Press*

- Chapter 09 Schuur M, van der Lijn F, Heijer T, Verbeek M, Aulchenko YS, Vrooman HA, Niessen W, Oostra BA, Breteler MMB, van Duijn CM, van der Lugt A, van Swieten JC.
 The relation of the sortilin-related receptor gene to hippocampal volume and plasma amyloid beta levels in a family-based study of hypertensive patients. *Submitted*
- Chapter 10 Schuur M, Amin N, Coppus T, Breteler MMB, Oostra BA, van Swieten JC, Janssens ACJW, Verbeek M, van Duijn CM.
 Polymorphisms of the renin-angiotensin system and alfa-adducin in associate to circulating amyloid beta levels. *Submitted*
- Chapter 12 Schuur M, Broer L, van Duijn CM, Janssens ACJW. Grading the credibility of genetic associations in Alzheimer's disease using the Venice criteria: practical considerations following from the Alzgene database. *Submitted*

Part I

Introduction

01 General introduction, outline, and scope of the thesis

The brain is by far the most complicated structure of the human being, and its malfunction is characterized by various degrees and types of morbidity. Several brain functions deteriorate with increasing age during life. Cognitive decline and age-related brain pathology are common in the elderly, but these changes may also become manifest early in life and preceding the onset of clinical symptoms of disease. The detection of early changes may be relevant for therapeutic interventions to prevent disease, and are therefore also increasingly targeted in genetic research as endophenotypes. Endophenotypes are defined as heritable phenotypes that are related to the disease of interest, and are typically approached as quantitative outcomes, i.e., instead of hypertension, the endophenotype of interest is systolic or diastolic blood pressure. In contrast to classical risk factors in epidemiology, an endophenotype is by definition not uniquely associated to a single disease. Blood pressure for example, is consistently associated to various clinically relevant outcomes such as stroke, myocardial infarction and heart failure. There is an increasing interest in the genetic research of endophenotypes, and genome-wide association studies of endophenotypes have been very successful [1,2]. In this thesis I focus on cognitive function and age-related brain changes early in life as endophenotypes for late-life brain disease and as targets for early prevention.

Cognitive deterioration can be seen in pre-clinical stages of neurodegenerative and neuropsychiatric disorders like dementia, schizophrenia, bipolar disorder and attention deficit hyperactivity disorder (ADHD) [3-5]. Cognitive function is a broad concept referring to multiple cognitive domains, among which memory, language, executive function and visuospatial ability. Although the domains are highly correlated, it is known that specific domains are related to specific diseases. Cognitive function is in part determined by our genetic make-up. The heritability is estimated to around 40% [6] and there have been various studies that have tried to identify genes explaining the heritability of cognitive functions. These included candidate gene studies [7,8], linkage studies [9-13] and genome-wide association studies [14-17]. The genes and chromosomal regions that have been found so far are partly explained by genes related to neuropsychiatric disease, and partly by genes related to dementia and Alzheimer's disease (AD) with the Apolipoprotein E gene as genetic factor with one of the strongest effects. In the studies presented here, we will focus on a cognitive test battery targeting AD [18]. Dementia is one of the most common causes of morbidity and mortality in the Western society (prevalence of 25 million cases worldwide), in which Alzheimer disease accounts for over 70% of cases [19,20]. Regarding the high prevalence and major impact of these diseases, early diagnosis and treatment strategies have a high priority in neuroscience. Identifying risk factors for cognitive decline would benefit our increasingly elderly population.

At postmortem, AD is characterized by neurofibrillary tangles and amyloid plaques [21]. The major components of the amyloid plaques are amyloid beta ($A\beta$) proteins, which are formed after

proteolytic processing of the amyloid precursor protein (APP). A β pathology is also found in brains of cognitively healthy elderly who may develop the disease later. A β levels can be measured in plasma probably due to leakage through the blood brain barrier. It has been suggested that the plasma A β 42/A β 40 ratio may be a suitable marker for early AD pathology [22,23]. Various epidemiological studies have show that low A β 42/A β 40 ratios in plasma are associated with an increased risk of dementia [22,23]. Since also plasma A β levels are heritable with estimates ranging from 54 to 73% [24], they are interesting endophenotypes for genetic research.

Another approach to study early brain changes is magnetic resonance imaging (MRI), which is a sensitive non-invasive imaging technique to visualize brain pathology. Hippocampal atrophy, white matter lesions, microbleeds and lacunar infarcts are the most common age-related brain changes visible on MRI, and have been associated with hypertension, stroke, dementia and cognitive impairment [23-28]. The mechanisms underlying the pathogenesis of these changes are largely unknown. Several studies suggest that structural changes in blood vessels in the brain lead to ischemic damage, causing white matter lesions and lacunar infarcts, and that leakage of red blood cells might lead to microbleeds and subsequent hemosiderin depositions [29,30]. Hypertension and atherosclerosis most likely cause damage to blood vessel walls, but also amyloid angiopathy plays an important role in vascular related events. [31]. The heritability estimates for age-related brain changes are high, ranging from 55 to 71% for white matter lesions [32-35] and ranging from 40 to 69% for medial temporal lobe atrophy [36-40].

The rationale behind the studies presented in this thesis is to use cognitive function, plasma Aβ levels and MRI changes as endophenotypes in the search for new determinants of neurodegenerative and neuropsychiatric diseases. We followed different approaches for our genetic studies. We used a candidate-gene approach by studying genes that have previously been reported in these traits or were involved in a plausible pathway for the disease. In the search for unknown genes, we conducted both genome-wide association studies as well as linkage studies. The underlying hypothesis for the genome-wide association approach is the common-disease common-variant hypothesis. Genome-wide linkage is considered in particular valuable in identifying rare variants with large effects. There is an increasing awareness that a large part of the heritability remains unexplained despite genome-wide association studies. Family-data may contribute to unravelling the genetics of complex traits and add to the data derived from current genome-wide associations studies [43].

The chapters of this thesis are divided in 4 main parts.

Part II focuses on determinants of cognitive function. Chapter 2 describes an epidemiological study on the relation between common cardiovascular risk factors and cognitive function.

Chapter 3 presents a candidate gene study of APOE. Chapters 4 and 5 present genome-wide approaches: a linkage study in chapter 4 and a genome-wide association study in chapter 5. Part III focuses on genetic determinants of AD. Chapter 6 presents a candidate gene study and meta-analysis of the Cathepsin D gene and chapter 7 describes a collaborative study considering interactive effects of the transferrin and HFE genes. Part IV focuses on determinants of age-related brain changes in hypertensive patients. Here, the results of three candidate gene studies are presented in chapters 8, 9 and 10. Finally, a short overview of all findings, followed by a discussion on the rating of genetic association studies is given in part V.

References

- Aulchenko YS, Ripatti S, Lindqvist I, Boomsma D, Heid IM, Pramstaller PP, Penninx BW, Janssens AC, Wilson JF, Spector T, Martin NG, Pedersen NL, Kyvik KO, Kaprio J, Hofman A, Freimer NB, Jarvelin MR, Gyllensten U, Campbell H, Rudan I, Johansson A, Marroni F, Hayward C, Vitart V, Jonasson I, Pattaro C, Wright A, Hastie N, Pichler I, Hicks AA, Falchi M, Willemsen G, Hottenga JJ, de Geus EJ, Montgomery GW, Whitfield J, Magnusson P, Saharinen J, Perola M, Silander K, Isaacs A, Sijbrands EJ, Uitterlinden AG, Witteman JC, Oostra BA, Elliott P, Ruokonen A, Sabatti C, Gieger C, Meitinger T, Kronenberg F, Doring A, Wichmann HE, Smit JH, McCarthy MI, van Duijn CM, Peltonen L. Loci influencing lipid levels and coronary heart disease risk in 16 European population cohorts. Nat Genet 2009;41:47-55.
- Levy D, Ehret GB, Rice K, Verwoert GC, Launer LJ, Dehghan A, Glazer NL, Morrison AC, Johnson AD, Aspelund T, Aulchenko Y, Lumley T, Kottgen A, Vasan RS, Rivadeneira F, Eiriksdottir G, Guo X, Arking DE, Mitchell GF, Mattace-Raso FU, Smith AV, Taylor K, Scharpf RB, Hwang SJ, Sijbrands EJ, Bis J, Harris TB, Ganesh SK, O'Donnell CJ, Hofman A, Rotter JI, Coresh J, Benjamin EJ, Uitterlinden AG, Heiss G, Fox CS, Witteman JC, Boerwinkle E, Wang TJ, Gudnason V, Larson MG, Chakravarti A, Psaty BM, van Duijn CM. Genome-wide association study of blood pressure and hypertension. Nat Genet 2009.
- 3. Green MF. Cognitive impairment and functional outcome in schizophrenia and bipolar disorder. J Clin Psychiatry 2006;67 Suppl 9:3-8; discussion 36-42.
- 4. Seidman LJ, Biederman J, Weber W, Hatch M, Faraone SV. Neuropsychological function in adults with attention-deficit hyperactivity disorder. Biol Psychiatry 1998;44:260-8.
- 5. Huntley JD, Howard RJ. Working memory in early Alzheimer's disease: a neuropsychological review. Int J Geriatr Psychiatry 2009; 25:121-32
- 6. Butcher LM, Kennedy JK, Plomin R. Generalist genes and cognitive neuroscience. Curr Opin Neurobiol 2006;16:145-51.
- 7. Deary IJ, Johnson W, Houlihan LM. Genetic foundations of human intelligence. Hum Genet 2009;126:215-32.
- 8. Payton A. The impact of genetic research on our understanding of normal cognitive ageing: 1995 to 2009. Neuropsychol Rev 2009.
- Paunio T, Tuulio-Henriksson A, Hiekkalinna T, Perola M, Varilo T, Partonen T, Cannon TD, Lonnqvist J, Peltonen L. Search for cognitive trait components of schizophrenia reveals a locus for verbal learning and memory on 4q and for visual working memory on 2q. Hum Mol Genet 2004;13:1693-702.
- 10. Buyske S, Bates ME, Gharani N, Matise TC, Tischfield JA, Manowitz P. Cognitive traits link to human chromosomal regions. Behav Genet 2006;36:65-76.
- Rommelse NN, Arias-Vasquez A, Altink ME, Buschgens CJ, Fliers E, Asherson P, Faraone SV, Buitelaar JK, Sergeant JA, Oosterlaan J, Franke B. Neuropsychological endophenotype approach to genome-wide linkage analysis identifies susceptibility loci for ADHD on 2q21.1 and 13q12.11. Am J Hum Genet 2008;83:99-105.

- 12. Luciano M, Wright MJ, Duffy DL, Wainwright MA, Zhu G, Evans DM, Geffen GM, Montgomery GW, Martin NG. Genome-wide scan of IQ finds significant linkage to a quantitative trait locus on 2q. Behav Genet 2006;36:45-55.
- 13. Almasy L, Gur RC, Haack K, Cole SA, Calkins ME, Peralta JM, Hare E, Prasad K, Pogue-Geile MF, Nimgaonkar V, Gur RE. A genome screen for quantitative trait loci influencing schizophrenia and neurocognitive phenotypes. Am J Psychiatry 2008;165:1185-92.
- 14. Papassotiropoulos A, Henke K, Stefanova E, Aerni A, Muller A, Demougin P, Vogler C, Sigmund JC, Gschwind L, Huynh KD, Coluccia D, Mondadori CR, Hanggi J, Buchmann A, Kostic V, Novakovic I, van den Bussche H, Kaduszkiewicz H, Weyerer S, Bickel H, Riedel-Heller S, Pentzek M, Wiese B, Dichgans M, Wagner M, Jessen F, Maier W, de Quervain DJ. A genome-wide survey of human short-term memory. Mol Psychiatry 2009.
- 15. Papassotiropoulos A, Stephan DA, Huentelman MJ, Hoerndli FJ, Craig DW, Pearson JV, Huynh KD, Brunner F, Corneveaux J, Osborne D, Wollmer MA, Aerni A, Coluccia D, Hanggi J, Mondadori CR, Buchmann A, Reiman EM, Caselli RJ, Henke K, de Quervain DJ. Common Kibra alleles are associated with human memory performance. Science 2006;314:475-8.
- 16. Cirulli ET, Kasperaviciute D, Attix DK, Need AC, Ge D, Gibson G, Goldstein DB. Common genetic variation and performance on standardized cognitive tests. Eur J Hum Genet 2010; Epub.
- 17. Seshadri S, DeStefano AL, Au R, Massaro JM, Beiser AS, Kelly-Hayes M, Kase CS, D'Agostino RB, Sr., Decarli C, Atwood LD, Wolf PA. Genetic correlates of brain aging on MRI and cognitive test measures: a genome-wide association and linkage analysis in the Framingham Study. BMC Med Genet 2007;8 Suppl 1:S15.
- 18. Sleegers K, de Koning I, Aulchenko YS, van Rijn MJ, Houben MP, Croes EA, van Swieten JC, Oostra BA, van Duijn CM. Cerebrovascular risk factors do not contribute to genetic variance of cognitive function: the ERF study. Neurobiol Aging 2007;28:735-41.
- 19. Bonita R. Epidemiology of stroke. Lancet 1992;339:342-4.
- 20. Cummings JL. Alzheimer's disease. N Engl J Med 2004;351:56-67.
- 21. Braak H, Braak E. Neuropathological stageing of Alzheimer-related changes. Acta Neuropathol 1991;82:239-59.
- 22. van Oijen M, Hofman A, Soares HD, Koudstaal PJ, Breteler MM. Plasma Abeta(1-40) and Abeta(1-42) and the risk of dementia: a prospective case-cohort study. Lancet Neurol 2006;5:655-60.
- 23. Lambert JC, Schraen-Maschke S, Richard F, Fievet N, Rouaud O, Berr C, Dartigues JF, Tzourio C, Alperovitch A, Buee L, Amouyel P. Association of plasma amyloid beta with risk of dementia: the prospective Three-City Study. Neurology 2009;73:847-53.
- 24. Ertekin-Taner N, Graff-Radford N, Younkin LH, Eckman C, Adamson J, Schaid DJ, Blangero J, Hutton M, Younkin SG. Heritability of plasma amyloid beta in typical late-onset Alzheimer's disease pedigrees. Genet Epidemiol 2001;21:19-30.
- 25. Werring DJ, Frazer DW, Coward LJ, Losseff NA, Watt H, Cipolotti L, Brown MM, Jager HR. Cognitive dysfunction in patients with cerebral microbleeds on T2*-weighted gradient-echo MRI. Brain 2004;127:2265-75.

- Kato H, Izumiyama M, Izumiyama K, Takahashi A, Itoyama Y. Silent cerebral microbleeds on T2*-weighted MRI: correlation with stroke subtype, stroke recurrence, and leukoaraiosis. Stroke 2002;33:1536-40.
- Kuller LH, Shemanski L, Manolio T, Haan M, Fried L, Bryan N, Burke GL, Tracy R, Bhadelia R. Relationship between ApoE, MRI findings, and cognitive function in the Cardiovascular Health Study. Stroke 1998;29:388-98.
- 28. Vermeer SE, Longstreth WT, Jr., Koudstaal PJ. Silent brain infarcts: a systematic review. Lancet Neurol 2007;6:611-9.
- 29. van Dijk EJ, Prins ND, Vrooman HA, Hofman A, Koudstaal PJ, Breteler MM. Progression of cerebral small vessel disease in relation to risk factors and cognitive consequences: Rotterdam Scan study. Stroke 2008;39:2712-9.
- Goos JD, Kester MI, Barkhof F, Klein M, Blankenstein MA, Scheltens P, van der Flier WM. Patients with Alzheimer disease with multiple microbleeds: relation with cerebrospinal fluid biomarkers and cognition. Stroke 2009;40:3455-60.
- Jagust WJ, Zheng L, Harvey DJ, Mack WJ, Vinters HV, Weiner MW, Ellis WG, Zarow C, Mungas D, Reed BR, Kramer JH, Schuff N, DeCarli C, Chui HC. Neuropathological basis of magnetic resonance images in aging and dementia. Ann Neurol 2008;63:72-80.
- 32. Fazekas F, Kleinert R, Roob G, Kleinert G, Kapeller P, Schmidt R, Hartung HP. Histopathologic analysis of foci of signal loss on gradient-echo T2*-weighted MR images in patients with spontaneous intracerebral hemorrhage: evidence of microangiopathy-related microbleeds. AJNR Am J Neuroradiol 1999;20:637-42.
- 33. Chen YW, Gurol ME, Rosand J, Viswanathan A, Rakich SM, Groover TR, Greenberg SM, Smith EE. Progression of white matter lesions and hemorrhages in cerebral amyloid angiopathy. Neurology 2006;67:83-7.
- 34. Carmelli D, Reed T, DeCarli C. A bivariate genetic analysis of cerebral white matter hyperintensities and cognitive performance in elderly male twins. Neurobiol Aging 2002;23:413-20.
- 35. Carmelli D, DeCarli C, Swan GE, Jack LM, Reed T, Wolf PA, Miller BL. Evidence for genetic variance in white matter hyperintensity volume in normal elderly male twins. Stroke 1998;29:1177-81.
- 36. Jeerakathil T, Wolf PA, Beiser A, Massaro J, Seshadri S, D'Agostino RB, DeCarli C. Stroke risk profile predicts white matter hyperintensity volume: the Framingham Study. Stroke 2004;35:1857-61.
- Atwood LD, Wolf PA, Heard-Costa NL, Massaro JM, Beiser A, D'Agostino RB, DeCarli C. Genetic variation in white matter hyperintensity volume in the Framingham Study. Stroke 2004;35:1609-13.
- Turner ST, Fornage M, Jack CR, Jr., Mosley TH, Knopman DS, Kardia SL, Boerwinkle E, de Andrade M. Genomic susceptibility loci for brain atrophy, ventricular volume, and leukoaraiosis in hypertensive sibships. Arch Neurol 2009;66:847-57.
- Sullivan EV, Pfefferbaum A, Swan GE, Carmelli D. Heritability of hippocampal size in elderly twin men: equivalent influence from genes and environment. Hippocampus 2001;11:754-62.

- 40. Wright IC, Sham P, Murray RM, Weinberger DR, Bullmore ET. Genetic contributions to regional variability in human brain structure: methods and preliminary results. Neuroimage 2002;17:256-71.
- 41. Hulshoff Pol HE, Schnack HG, Posthuma D, Mandl RC, Baare WF, van Oel C, van Haren NE, Collins DL, Evans AC, Amunts K, Burgel U, Zilles K, de Geus E, Boomsma Dl, Kahn RS. Genetic contributions to human brain morphology and intelligence. J Neurosci 2006;26:10235-42.
- 42. Lunetta KL, Erlich PM, Cuenco KT, Cupples LA, Green RC, Farrer LA, Decarli C, for the MSG. Heritability of magnetic resonance imaging (MRI) traits in Alzheimer disease cases and their siblings in the MIRAGE study. Alzheimer Dis Assoc Disord 2007;21:85-91.
- 43. Axenovich TI, Zorkoltseva IV, Belonogova NM, Struchalin MV, Kirichenko AV, Kayser M, Oostra BA, van Duijn CM, Aulchenko YS. Linkage analysis of adult height in a large pedigree from a Dutch genetically isolated population. Hum Genet 2009.

01 General Introduction, outline and scope of the thesis

Part II

New determinants of cognitive function

02 Insulin-resistance and metabolic syndrome are related to executive function in women in a large familybased study

Abstract

Background While type 2 diabetes is well-known to be associated with poorer cognitive performance, few studies have reported on the association of metabolic syndrome (MetS) and contributing factors, such as insulin-resistance (HOMA-IR), low adiponectin-, and high C-reactive protein (CRP)- levels. We studied whether these factors are related to cognitive function and which of the MetS components are independently associated.

Methods The study was embedded in an ongoing family-based cohort study in a Dutch population. All participants underwent physical examinations, biomedical measurements, and neuropsychological testing. Linear regression models were used to determine the association between MetS, HOMA-IR, adiponectin levels, CRP, and cognitive test scores. Cross-sectional analyses were performed in 1898 subjects (mean age 48 years, 43% men).

Results People with MetS had significantly higher HOMA-IR scores, lower adiponectin levels, and higher CRP levels. MetS and high HOMA-IR were associated with poorer executive function in women (p=0.03 and p=0.009). The most consistent individual component of MetS, contributing to the association with executive test scores was systolic blood pressure.

Conclusions MetS and HOMA-IR are associated with poorer executive function in women. Of the MetS components, systolic blood pressure is independently associated with executive function.

Introduction

Cognitive impairment is a common problem in the elderly and an important predictor of dementia (DSM-IV). There are multiple risk factors contributing to cognitive decline. For long, type 2 diabetes has been recognized as a major risk factor of dementia [1]. Epidemiological studies indicate that hyperinsulinemia and insulin-resistance, which characterize type 2 diabetes, may cause cognitive dysfunction [2,3]. Insulin-resistance is also a feature of the metabolic syndrome (MetS) and there is increasing interest in the role of MetS in cognition. The syndrome is defined as the presence of at least 3 out of 5 factors, which include central obesity, hyperglycemia, hypertension, hypertriglyceridemia, and low high-density lipoprotein cholesterol (HDL-c) levels [4]. It has a high prevalence of over 22% in Western societies [5], and is an important risk factor of cardiovascular disease [6,7]. Potentially, MetS is an important, modifiable, determinant of cognitive pathology [8-10]. Previous studies have shown associations between cognitive dysfunction and MetS, but results are not consistent and predominantly concern elderly people [11]. Of its individual components, hypertension and hyperglycemia have proven to be the most consistent determinants of cognitive function in a recent review of published studies [12].

Adipose tissue also plays an important role in the etiology of MetS. Adipose tissue functions as endocrine tissue, among which secretion of adipocytokines and cytokines, which indirectly affect C-reactive protein (CRP) levels [13-15]. The visceral fat depot is thought to play an important role in adipocytokine secretion and MetS [16]. Several groups have suggested that in particular adiponectin

might be a promising biomarker for MetS [17-19]. Whether it is related to cognitive function in humans is not known, but animal studies indicate that adiponectin may influence brain metabolism [20]. Regarding CRP levels, it has been reported that the association of MetS with cognitive function is stronger in those with high levels of CRP [21,22].

In this study, we examined whether MetS, insulin-resistance, plasma adiponectin and CRP- levels are related to cognitive function. In addition, we investigated which MetS components attributed independently to the association.

Methods

Study population

The study was performed in the Erasmus Rucphen Family Study (ERF), which is an ongoing familybased cohort study in a genetically isolated population [23]. The ERF population includes around 3000 individuals, all living descendents of 22 couples who, at the end of the 19th century had at least six children baptized in the community church. Extensive data on cardiovascular risk factors, body composition, cognitive functioning, and blood chemistry are available. These data were collected between 2002 and 2005. All participants gave informed consent and the study was approved by the Medical Ethics Committee at ErasmusMC.

Clinical and laboratory assessments

To define MetS, we used the definition proposed by the National Cholesterol Education Program Adult Treatment Panel (ATPIII) [4]. Waist circumference was measured exactly halfway the distance between the lower costal margin and the iliac crest [24]. Blood pressure was measured twice with an automatic device in a sitting position [25].

Blood was taken after fasting of at least 8 hours and serum triglycerides (TG), serum HDL-c, and plasma glucose (FPG) were assessed. Fasting plasma insulin was analyzed with the INS-Irma kit of Biosource (cat.#: KIP1254). Insulin-resistance was determined as homeostasis model assessment insulin-resistance (HOMA-IR) [26], which was computed by multiplying fasting plasma insulin (micro-international units per milliliter) and fasting plasma glucose (millimoles per liter)/22.5. Total plasma adiponectin was analyzed with the Human adiponectin RIA kit (cat.#: HADP-61HK) of Linco Research. To measure total plasma CRP, we used the US C-reactive protein ELISA (cat.# DSL-10-42100) of Diagnostic Systems Laboratories, Inc (expected range within 90th percentile: 254-16104 ng/ml).

We considered age, gender, smoking, alcohol use, education, depression, and Apolipoprotein E genotype (APOE) as possible confounders. Use of alcohol was defined as drinking at least 1 glass each day. All subjects were categorized into four groups with respect to level of education: 1. primary education; 2. lower vocational education; 3. intermediate vocational education and secondary education; and 4. higher vocational education and university training. The years of schooling in the first category corresponded to 6 years; in the second to 7-10 years; in the third to 11-14 years; and

in the fourth to 15-18 years.

Depressive symptoms were measured with the Hospital Anxiety and Depression Scale (HADS-D). Genomic DNA was extracted from whole blood samples, utilizing the salting out method. Samples were genotyped for APOE with a Taqman allelic discrimination Assay-By-Design (Applied Biosystems, Foster City, CA). The measurements were performed conform the manufacturers protocols. Clinical and laboratory data was not available for 721 participants.

Cognitive function

The battery of neuropsychological tests measured different cognitive domains and a Dutch validated version was available [27,28]. General cognitive ability was measured with the Dutch Adult Reading Test (DART). Memory function was measured with a word learning test with 5 exposure trials from which four scores were derived: immediate memory, defined as the number of correctly recalled words in trial 1: learning, defined as the total number of recalled words in trials 2 to 5: delayed recall. defined as the number of correctly recalled words after 20 minutes; and recognition, defined as the number of correctly recognized and correctly rejected words. Executive function was assessed with the Trail Making Test parts A and B (TMT), the Stroop Color and Word Test cards I, II and III, and verbal fluency tests. TMT and Stroop are time-demanding tasks in which participants had to complete the test as quickly as possible either by connecting letters and numbers (TMT) or by naming words and colors (Stroop) [29,30]. For analyses, we used time in seconds depicted as the ratios of TMT-B/TMT-A and Stroop-III/Stroop-II. These ratios were used to adjust for mental slowness, which is mainly reflected in TMT-A and Stroop-II. When participants timed out, scores were imputed for TMT-B and Stroop-III (3-8%). The imputation was performed using missing value analysis in SPSS through expectation maximization algorithms using age, sex, education and either score on TMT-A or Stroop-I and Stroop-II as predictor variables. For the verbal fluency tests, participants had to name as many animals and as many words starting with the same letter (D, A and T), each in one minute. The score was defined as the total number of correct words on both tasks. Lastly, visuo-spatial ability was assessed with the WAISIII block-design subtest.

In addition to analyzing individual test scores, three composite scores were used: 1. memory (z-memory); 2. executive function (z-executive); 3. global cognition (z-global). These scores were computed by taking the average of z-scores of the individual tests, which were computed by subtracting the mean from the test score divided by the standard deviation [31]. For time-demanding tasks, the test score was subtracted from the mean. The composite scores were only computed when at least 3 memory scores (z-memory), 2 executive scores (z-executive) and 6 total test scores (z-global) were available. In the current study, we excluded participants who had a history of stroke or dementia (N=21) or were physically not able to perform a neuropsychological test (e.g. sensory handicaps (N=6), illiteracy (N=28), other (N=10)). There were 10 participants with missing test scores due to technical problems.

28

Cognitive function

The battery of neuropsychological tests measured different cognitive domains and a Dutch validated version was available [27,28]. General cognitive ability was measured with the Dutch Adult Reading Test (DART). Memory function was measured with a word learning test with 5 exposure trials from which four scores were derived: immediate memory, defined as the number of correctly recalled words in trial 1; learning, defined as the total number of recalled words in trials 2 to 5; delayed recall, defined as the number of correctly recalled words after 20 minutes; and recognition, defined as the number of correctly recognized and correctly rejected words. Executive function was assessed with the Trail Making Test parts A and B (TMT), the Stroop Color and Word Test cards I, II and III, and verbal fluency tests. TMT and Stroop are time-demanding tasks in which participants had to complete the test as guickly as possible either by connecting letters and numbers (TMT) or by naming words and colors (Stroop) [29.30]. For analyses, we used time in seconds depicted as the ratios of TMT-B/TMT-A and Stroop-III/Stroop-II. These ratios were used to adjust for mental slowness, which is mainly reflected in TMT-A and Stroop-II. When participants timed out, scores were imputed for TMT-B and Stroop-III (3-8%). The imputation was performed using missing value analysis in SPSS through expectation maximization algorithms using age, sex, education and either score on TMT-A or Stroop-I and Stroop-II as predictor variables. For the verbal fluency tests, participants had to name as many animals and as many words starting with the same letter (D, A and T), each in one minute. The score was defined as the total number of correct words on both tasks. Lastly, visuo-spatial ability was assessed with the WAISIII block-design subtest.

In addition to analyzing individual test scores, three composite scores were used: 1. memory (z-memory); 2. executive function (z-executive); 3. global cognition (z-global). These scores were computed by taking the average of z-scores of the individual tests, which were computed by subtracting the mean from the test score divided by the standard deviation [31]. For time-demanding tasks, the test score was subtracted from the mean. The composite scores were only computed when at least 3 memory scores (z-memory), 2 executive scores (z-executive) and 6 total test scores (z-global) were available. In the current study, we excluded participants who had a history of stroke or dementia (N=21) or were physically not able to perform a neuropsychological test (e.g. sensory handicaps (N=6), illiteracy (N=28), other (N=10)). There were 10 participants with missing test scores due to technical problems.

Statistical analysis

One large pedigree was extracted from the genealogical database for eligible participants. The pedigree was cut into smaller pedigrees before analyses with the software PedCut [32]. In total, 1898 people could be linked to pedigrees with a maximum bit size of 18 and these were included in the analyses.

Baseline descriptive analyses were performed with SPSS version 15.0, using the independentsamples T-test (continuous variables) and Chi-square statistics (categorical data). Deviations from normality were checked using P-P plots and histograms. C-reactive protein levels were transformed by taking the common logarithm. Because high levels of CRP could have been caused by acute inflammation, all values of the log-transformed variable greater than three times the standard deviation above the mean were excluded.

Possible confounders were first analyzed as single covariates in two regression models with either MetS or cognitive test scores as dependent variables. Also, interaction terms of MetS, HOMA-IR, adiponectin and CRP with age and sex were analyzed in these models. Covariates that were significantly associated were then entered simultaneously into a model using backward regression. We removed all covariates with a p-value > 0.10. The final model included age, level of education, HADS-D score, alcohol use and inbreeding coefficient. The analyses were stratified by sex, because the interaction terms of MetS with sex were significant.

To identify associations between cognitive test scores and MetS, HOMA-IR, adiponectin, and CRP and to adjust for family relationships, we performed regression analyses using the SOLAR software version 4.1.0 [33]. The MetS was included in the model as a binomial variable, and additionally as a continuous variable ranging from 0 to 5 to test the effect of increasing number of MetS components. To determine to what extend the association of MetS with cognitive test scores was accounted for by its individual components, all individual MetS components were analyzed simultaneously in one regression model. The components were entered in the model as continuous variables. We used quartiles of HOMA-IR, adiponectin and CRP with the lowest quartile as a reference category to estimate the association with adipose tissue endocrine function.

Results

The mean age of the study population was 47.7 (\pm 14.3) years in women and 49.2 (\pm 14.1) years in men, with a range of 18 to 86 years and less than 10% older than 67. Metabolic syndrome was present in 227 (21%) women and in 207 (25%) men. Of these, 155 (68%) women and 150 (73%) men fulfilled three of the MetS diagnostic criteria, 63 (28%) and 54 (26%) fulfilled four and 9 (4%) and 3 (1%) fulfilled all five. Null criteria were fulfilled by 282 women and 153 men. One and two criteria were fulfilled by respectively 385 and 305 women and 182 and 157 men. With regard to the individual MetS components in the total population, central obesity was present in 28% of women and 24% of men, hypertension in 49% and 70%, dyslipidemia in 47% and 43%, and finally increased glucose levels in 5% and 7%. A comparison of the individual MetS components is given in Table 1.

Table 2 shows the baseline characteristics of men and women with and without MetS regarding determinants of cognitive function and vascular and endocrine factors. Individuals with MetS were older, had lower education and had higher scores on the depression scale. Men with MetS drank alcohol more frequently, whereas women with MetS drank alcohol less frequently. The endocrine

Metabolic Syndrome in the Erasmus Rucphen Family study (n=1898)

	Men		Women		
	MetS absent (n=615)	MetS present (n=207)	MetS absent (n=849)	MetS present (n=227)	
MetS components					
Waist circumference (cm)	90.1 (9.5)	105.1 (9.3)	78.2 (9.5)	94.5 (10.2)	
SBP (mm Hg)	140.7 (17.2)	151.2 (17.3)	131.2 (18.4)	152.9 (21.0)	
DBP (mm Hg)	80.3 (9.5)	85.5 (9.4)	76.8 (9.3)	84.0 (10.0)	
Use of antihypertensive medication (%)	11.7	34.3	9.0	38.3	
Glucose (mmol/L, FPG)	4.6 (0.8)	5.2 (1.4)	4.3 (0.7)	5.1 (1.1)	
Use of antidiabetic medication (%)	0.7	9.7	0.2	8.8	
HDL-cholesterol (mmol/L)	1.2 (0.3)	0.9 (0.2)	1.5 (0.4)	1.2 (0.3)	
Triglycerides (mmol/L)	1.2 (0.6)	2.3 (1.0)	1.1 (0.5)	1.8 (0.8)	
Use of statins (%)	11.4	25.6	5.9	25.6	

Values presented as mean (standard deviation). MetS= metabolic syndrome

Table 1

factors were also significantly different between individuals with and without MetS. MetS was associated with higher levels of insulin, higher HOMA-IR scores, lower adiponectin levels and higher CRP levels.

Next, we performed regression analysis to study the associations between MetS and cognitive test scores. Although all mean test scores which are shown in Table 3 were lower in individuals with MetS, the adjusted analysis was significant only for Stroop-ratio and z-executive in women (p=0.03 and p=0.02). Increasing number of MetS components was associated to lower verbal fluency and z-executive scores (p = 0.04 and 0.02) in women. The same trend, however non-significant (p=0.10), was seen for Stroop-ratio.

	Men		Women		
	MetS absent (n=615)	MetS present (n=207)	MetS absent (n=849)	MetS present (n=227)	
Vascular and endocrine factors					
Smoking (%)	32.7	33.5	46.5	45.1	
Insulin (µU/ml, FPI)	11.8 (5.6)	19.1 (11.1) **	11.5 (4.7)	18.0 (9.2) **	
HOMA-IR	2.4 (1.4)	4.5 (2.9) **	2.2 (1.0)	4.1 (2.4) **	
Adiponectin (mg/L)	8.5 (4.2)	6.7 (3.7) **	13.1 (5.8)	9.5 (4.6) **	
CRP (mg/L)	3.1 (6.2)	3.5 (5.8) **	3.4 (5.4)	5.1 (7.4) **	
Determinants of cognitive functio	n				
Age (year)	47.4 (14.3)	54.4 (11.9) **	45.3 (13.7)	56.7 (12.6) **	
Education level (%)		**		**	
Low	30.0	37.7	26.2	46.9	
Low-intermediate	37.9	44.4	42.0	40.7	
High-intermediate	24.1	14.5	27.3	11.5	
High	8.0	3.4	4.5	3.7	
Alcohol (% frequent users) §	45.0	36.6 *	19.9	12.9 *	
HADS-D score	5.7 (4.0)	6.4 (3.8) *	5.4 (4.2)	7.6 (4.6) **	
Apoe 4 allele carriers (%)	36.3	42.6	35.5	40.3	

Values presented as mean (standard deviation).

WetS = metabolic syndrome, FPG = fasting plasma glucose, FPI= fasting plasma insulin, HOMA-IR = homeostatic assessment --insulin resistant model. HADS-D: Hospital Anxiety and Depression Scale. § defined as drinking at least one unit of alcohol per day * unadjusted p-value <0.05; ** unadjusted p-value <0.01.

Table 2

We followed up on these associations to see which individual MetS components attributed independently to our findings (Table 4). Higher systolic blood pressure was related to lower scores on Stroop-ratio, verbal fluency and z-executive. Lower HDL-c levels were also associated with lower verbal fluency scores. Diastolic blood pressure showed an opposite effect with Stroop-ratio

and z-executive, but the association with z-executive disappeared when excluding persons using antihypertensive medication. The other associations remained unaltered when excluding people using medication.

Association metabolic syndrome and cognitive function in men and women						
Cognitive test	Men			Women		
	n	MetS absent	MetS present	n	MetS absent	MetS present
General cognitive ability						
Dutch Adult Reading Test	750	62.1 (19.0)	58.3 (18.5)	993	60.4 (18.2)	52.6 (18.1)
Memory						
AVLT – Working memory	757	4.3 (1.7)	3.9 (1.5)	999	4.6 (1.7)	4.0 (1.6)
AVLT – Learning	757	32.1 (8.9)	29.2 (8.3)	999	35.5 (8.9)	31.8 (8.9)
AVLT – Recognition	759	27.7 (2.2)	27.1 (2.5)	1000	28.3 (2.2)	27.7 (2.4)
AVLT - Delayed recall	759	7.1 (2.9)	6.4 (2.8)	1000	8.3 (2.9)	7.2 (2.9)
Fuendation						
Ratio TMT-B / TMT-A †	752	2.7 (1.0)	2.8 (1.0)	996	2.6 (1.0)	2.9 (1.1)
Batio Stroop III / Stroop II +	734	17(04)	18(04)	996	17(03)	19(06)*
Verbal fluency	758	63.7 (18.0)	60.0 (17.1)	998	63.0 (17.9)	53.5 (18.7)
Visuospatial						
Block design	756	30.8 (15.6)	25.7 (13.9)	998	28.7 (15.1)	20.8 (11.2)
Composite scores						
Memory function	757	0.1 (0.8)	-0.2 (0.8)	999	0.1 (0.8)	-0.3 (0.9)
Executive function	759	0.1 (0.7)	-0.1 (0.7)	998	0.1 (0.6)	-0.4 (0.8) *
Global cognitive function	758	0.1 (0.7)	-0.2 (0.6)	998	0.1 (0.7)	-0.3 (0.7)

Values presented as mean (standard deviation). AVLT=Adult Verbal Learning Test, TMT=Trail Making Test,

Stroop-Stroop Color and Wood Test; † time demanding task: high values represent low test scores. p=p-value derived from regression analysis adjusted for age, education, depressive score, alcohol use and family-relationship. * p < 0.05

Table 3

Contribution of individual metabolic syndrome components to executive function in women							
	Ratio TMT-B / TMT-A †	Ratio Stroop III / Stroop II †	Verbal fluency	z-executive			
Waist circumference	0.006 (0.003)	0.005 (0.004)	0.009 (0.049)	-0.002 (0.002)			
SBP	0.002 (0.002)	0.009 (0.003) **	-0.120 (0.035) **	-0.005 (0.001) **			
DBP	-0.001 (0.004)	-0.016 (0.005) **	0.081 (0.063)	0.005 (0.002) *			
TG	-0.001 (0.048)	0.050 (0.066)	1.342 (0.807)	0.012 (0.029)			
HDL-c	0.048 (0.080)	0.005 (0.110)	3.277 (1.423) *	0.064 (0.051)			
FPG	0.005 (0.036)	0.024 (0.054)	0.765 (0.630)	0.009 (0.023)			

All individual MetS components were entered as continuous variables to the regression model. Results are presented as betas (standard errors). TMT=Trail Making Test; Stroop II=Stroop Color Card; Stroop III=Stroop Color and Word card; z-executive=composite score of executive tests;

SBP=systolic blood pressure; DBP=diastolic blood pressure; TG=Triglycerides; HDL-c=High Density Lipoprotein Cholesterol; FPG=Fasting Plasma Glucose. † time-demanding task: positive beta represents negative association. * p-value < 0.0; * p-value < 0.0;

Table 4

The findings related to adipose tissue endocrine function are given in Figure 1 with analyses limited to executive function. Women with high HOMA-IR scores had lower scores on TMT-ratio, Stroop-ratio and z-executive than women with low HOMA-IR scores. Higher adiponectin levels were associated with higher scores on z-executive in both men and women and with Stroop-ratio in women only. The associations of HOMA-IR with Stroop-ratio and z-executive were more significant (p=0.004 and 0.007) when analyzing HOMA-IR as continuous variables instead of guartiles. The other associations, however, became non-significant. No association between CRP and executive function was found in this population.

Associations of HOMA-IR (A), adiponectin (B), and CRP (C) with cognitive function in men and women



Results are presented as mean differences compared to quartile 1.

The differences are adjusted for age, education, depressive score, alcohol use and family-relationship.

HOMA-IR=homeostatis model assessment insulin-resistance, CRP=C-reactive protein

* p-value < 0.05; ** p-value < 0.01

Figure 1

chapter 02

Discussion

The main finding of the current study is that MetS and HOMA-IR is associated with poorer executive performance in women. Of the MetS components, systolic blood pressure is independently associated with executive function.

Our findings of association of MetS with cognitive performance confirms previous cross-sectional and longitudinal studies [22,34,35]. Although we studied different cognitive domains, mainly executive function was associated with MetS. This is in line with two recent studies that found significant differences in processing speed, semantic fluency and executive function between persons with and without MetS [36,37]. Associations of the MetS with fronto-subcortical syndrome and with frontal white matter changes were found by others [10,38]. Damage to cerebral small vessels could be one of the explanations for the observed associations in the current study [39-41], which is supported by the independent association of systolic blood pressure with cognitive function [12,34,42]. Whereas diabetes has been reported to be associated with cognitive function [12], the low prevalence of hypergelycemia in our population could explain the lack of association with cognitive tests. Insulin-resistance as an early sign of diabetes pathology was more common in our mainly middle aged population and was associated with cognitive tests. Although findings on the individual MetS components vary, a number of studies find that an increasing number of MetS components is associated with lower cognitive scores, which is also confirmed by our observations [10,35,42-44].

An effect of MetS components on the early development of Alzheimer's disease (AD) pathology may be another possible mechanism underlying our observations, since the insulin-degrading enzyme (IDE) functions less well in the presence of insulin-resistance, resulting in higher levels of amyloid β -peptide (A β) [45,46]. Long-term exposure to insulin due to insulin-resistance, may by itself also have a direct damaging effect on neurons [47]. Our observed association of HOMA-IR and executive function is in line with this. Previous studies on insulin-resistance measured as HOMA-IR have been inconsistent [3,48,49].

Despite possible pathways through which adiponectin could be linked to cognition, such as improvement of insulin-sensitivity, anti-inflammatory effects and effects on brain metabolism and the vascular system, our results do not show a consistent association of adiponectin with cognition [50-55]. Adiponectin levels are however, lower in individuals with MetS in our study and a trend was seen with executive function in women, which supports further study on the role of adiponectin in cognition.

Levels of CRP were not related to cognition in the present study, however, data on other markers, such as interleukin-6 (II-6) and a1-antichymotrypsin (ACT) were not available. These markers have been related to cognitive decline and increased systemic inflammation has been proposed as a possible modifier between MetS and cognition [22,54]. We could not confirm this, which may also be due to a more pronounced effect of inflammation on cognition in the elderly than in middle aged individuals.

34

Our study shows that there are gender differences in the association of MetS with cognitive function. Gender-specific effects of metabolic components on cognitive function have not extensively been addressed in previous studies [56], while there is evidence that MetS has greater effect on cardiovascular and cerebrovascular disease in women than in men, indirectly supporting our observed differences [57-60]. Furthermore, there is evidence that men are treated more aggressively for cardiovascular risk factors [61]. As a consequence, men may be less likely to experience the adverse effects of MetS. Another factor that could explain gender differences is depression, which is highly correlated to cognition and more common in women. Since our analyses were adjusted for depressive symptoms, however, we do not think that depression explains the observed differences in our study.

The strength of this study is the population-based design in which participants were not selected on the disease of interest. Also, the inclusion of a large sample size and a range of cognitive tests allowed detailed study of cognition. The cross-sectional design is a limitation of our study, and it does not allow claiming causality. Previous longitudinal studies, however, suggest that MetS might be a causative factor in the relationship with cognition [62]. Another limitation is that some of the significant p-values could be false-positive findings due to the large number of tests that we performed. However, the consistent associations argue against this.

In summation, the most important findings of our study are that MetS and HOMA-IR are associated with executive function in women. Of the MetS components, systolic blood pressure is independently associated with executive function.
References

- 1. Ott A, Stolk RP, van Harskamp F, Pols HA, Hofman A, Breteler MM. Diabetes mellitus and the risk of dementia: The Rotterdam Study. Neurology 1999;53:1937-42.
- Geroldi C, Frisoni GB, Paolisso G, Bandinelli S, Lamponi M, Abbatecola AM, Zanetti O, Guralnik JM, Ferrucci L. Insulin resistance in cognitive impairment: the InCHIANTI study. Archives of Neurology 2005;62:1067-72.
- 3. Young SE, Mainous AG, 3rd, Carnemolla M. Hyperinsulinemia and cognitive decline in a middle-aged cohort. Diabetes Care 2006;29:2688-93.
- 4. Grundy SM, Cleeman JI, Daniels SR, Donato KA, Eckel RH, Franklin BA, Gordon DJ, Krauss RM, Savage PJ, Smith SC, Jr., Spertus JA, Costa F. Diagnosis and management of the metabolic syndrome: an American Heart Association/National Heart, Lung, and Blood Institute Scientific Statement. Circulation 2005;112:2735-52.
- 5. Ford ES, Giles WH, Dietz WH. Prevalence of the metabolic syndrome among US adults: findings from the third National Health and Nutrition Examination Survey. The Journal of the American Medical Association 2002;287:356-9.
- 6. Alberti KG, Zimmet P, Shaw J, Group IDFETFC. The metabolic syndrome--a new worldwide definition. Lancet 2005;366:1059-62.
- 7. Olijhoek JK, van der Graaf Y, Banga JD, Algra A, Rabelink TJ, Visseren FL, the SSG. The metabolic syndrome is associated with advanced vascular damage in patients with coronary heart disease, stroke, peripheral arterial disease or abdominal aortic aneurysm. European Heart Journal 2004;25:342-8.
- 8. Kwon HM, Kim BJ, Lee SH, Choi SH, Oh BH, Yoon BW. Metabolic syndrome as an independent risk factor of silent brain infarction in healthy people. Stroke 2006;37:466-70.
- 9. Kumari M, Brunner E, Fuhrer R. Minireview: mechanisms by which the metabolic syndrome and diabetes impair memory. Journals of Gerontology Series A: Biological Sciences and Medical Sciences 2000;55:B228-32.
- Roriz-Cruz M, Rosset I, Wada T, Sakagami T, Ishine M, De Sa Roriz-Filho J, Cruz TR, Hosseinkhani M, Rodrigues RP, Sudoh S, Arai H, Wakatsuki Y, Souza AC, Nakagawa M, Kita T, Matsubayashi K. Cognitive impairment and frontal-subcortical geriatric syndrome are associated with metabolic syndrome in a stroke-free population. Neurobiol Aging 2007;28:1723-36.
- 11. Yaffe K, Weston AL, Blackwell T, Krueger KA. The metabolic syndrome and development of cognitive impairment among older women. Archives of Neurology 2009;66:324-8.
- 12. van den Berg E, Kloppenborg RP, Kessels RP, Kappelle LJ, Biessels GJ. Type 2 diabetes mellitus, hypertension, dyslipidemia and obesity: A systematic comparison of their impact on cognition. Biochim Biophys Acta 2009;1792:470-81.
- 13. Devaraj S, Torok N, Dasu MR, Samols D, Jialal I. Adiponectin decreases C-reactive protein synthesis and secretion from endothelial cells: evidence for an adipose tissue-vascular loop. Arteriosclerosis, Thrombosis, and Vascular Biology 2008;28:1368-74.
- 14. Guerre-Millo M. Adipose tissue and adipokines: for better or worse. Diabetes Metab 2004;30:13-9.

- 15. Rader DJ. Inflammatory markers of coronary risk. The New England Journal of Medicine 2000;343:1179-82.
- 16. Bacha F, Saad R, Gungor N, Arslanian SA. Adiponectin in youth: relationship to visceral adiposity, insulin sensitivity, and beta-cell function. Diabetes Care 2004;27:547-52.
- 17. Gilardini L, McTernan PG, Girola A, da Silva NF, Alberti L, Kumar S, Invitti C. Adiponectin is a candidate marker of metabolic syndrome in obese children and adolescents. Atherosclerosis 2006;189:401-7.
- 18. Luchsinger JA. Adiposity, hyperinsulinemia, diabetes and Alzheimer's disease An epidemiological perspective. European Journal of Pharmacology 2008.
- 19. Lara-Castro C, Fu Y, Chung BH, Garvey WT. Adiponectin and the metabolic syndrome: mechanisms mediating risk for metabolic and cardiovascular disease. Current Opinion in Lipidology 2007;18:263-70.
- 20. Kubota N, Yano W, Kubota T, Yamauchi T, Itoh S, Kumagai H, Kozono H, Takamoto I, Okamoto S, Shiuchi T, Suzuki R, Satoh H, Tsuchida A, Moroi M, Sugi K, Noda T, Ebinuma H, Ueta Y, Kondo T, Araki E, Ezaki O, Nagai R, Tobe K, Terauchi Y, Ueki K, Minokoshi Y, Kadowaki T. Adiponectin stimulates AMP-activated protein kinase in the hypothalamus and increases food intake. Cell Metab 2007;6:55-68.
- 21. Yaffe K. Metabolic syndrome and cognitive disorders: is the sum greater than its parts? Alzheimer Disease and Associated Disorders 2007;21:167-71.
- 22. Dik MG, Jonker C, Comijs HC, Deeg DJ, Kok A, Yaffe K, Penninx BW. Contribution of metabolic syndrome components to cognition in older individuals. Diabetes Care 2007;30:2655-60.
- 23. Aulchenko YS, Heutink P, Mackay I, Bertoli-Avella AM, Pullen J, Vaessen N, Rademaker TA, Sandkuijl LA, Cardon L, Oostra B, van Duijn CM. Linkage disequilibrium in young genetically isolated Dutch population. European Journal of Human Genetics 2004;12:527-34.
- 24. Santos RL, Zillikens MC, Rivadeneira FR, Pols HA, Oostra BA, van Duijn CM, Aulchenko YS. Heritability of fasting glucose levels in a young genetically isolated population. Diabetologia 2006;49:667-72.
- 25. Sleegers K, de Koning I, Aulchenko YS, van Rijn MJ, Houben MP, Croes EA, van Swieten JC, Oostra BA, van Duijn CM. Cerebrovascular risk factors do not contribute to genetic variance of cognitive function: the ERF study. Neurobiology of Aging 2007;28:735-41.
- 26. Matthews DR, Hosker JP, Rudenski AS, Naylor BA, Treacher DF, Turner RC. Homeostasis model assessment: insulin resistance and beta-cell function from fasting plasma glucose and insulin concentrations in man. Diabetologia 1985;28:412-9.
- Sleegers K, Roks G, Theuns J, Aulchenko YS, Rademakers R, Cruts M, van Gool WA, Van Broeckhoven C, Heutink P, Oostra BA, van Swieten JC, van Duijn CM. Familial clustering and genetic risk for dementia in a genetically isolated Dutch population. Brain 2004;127:1641-9.

- 28. Liu F, Pardo LM, Schuur M, Sanchez-Juan P, Isaacs A, Sleegers K, de Koning I, Zorkoltseva IV, Axenovich TI, Witteman JC, Janssens AC, van Swieten JC, Aulchenko YS, Oostra BA, van Duijn CM. The apolipoprotein E gene and its age-specific effects on cognitive function. Neurobiol Aging 2008.
- 29. Reitan RM. The relation of the trail making test to organic brain damage. Journal of Consulting and Clinical Psychology 1955;19:393-4.
- 30. Hammes J. Stroop Kleur-woord Test: Dutch Manual. Lisse, The Netherlands: Swets and Zeitlinger BV, 1978
- Prins ND, van Dijk EJ, den Heijer T, Vermeer SE, Jolles J, Koudstaal PJ, Hofman A, Breteler MM. Cerebral small-vessel disease and decline in information processing speed, executive function and memory. Brain 2005;128:2034-41.
- 32. Liu F, Kirichenko A, Axenovich TI, van Duijn CM, Aulchenko YS. An approach for cutting large and complex pedigrees for linkage analysis. European Journal of Human Genetics 2008;16:854-60.
- 33. Almasy L, Blangero J. Multipoint quantitative-trait linkage analysis in general pedigrees. The American Journal of Human Genetics 1998;62:1198-211.
- 34. Komulainen P, Lakka TA, Kivipelto M, Hassinen M, Helkala EL, Haapala I, Nissinen A, Rauramaa R. Metabolic syndrome and cognitive function: a population-based follow-up study in elderly women. Dementia and Geriatric Cognitive Disorders 2007;23:29-34.
- 35. Yaffe K, Haan M, Blackwell T, Cherkasova E, Whitmer RA, West N. Metabolic syndrome and cognitive decline in elderly Latinos: findings from the Sacramento Area Latino Study of Aging study. Journal of the American Geriatrics Society 2007;55:758-62.
- 36. van den Berg E, Dekker JM, Nijpels G, Kessels RP, Kappelle LJ, de Haan EH, Heine RJ, Stehouwer CD, Biessels GJ. Cognitive functioning in elderly persons with type 2 diabetes and metabolic syndrome: the Hoorn study. Dement Geriatr Cogn Disord 2008;26:261-9.
- 37. Segura B, Jurado MA, Freixenet N, Albuin C, Muniesa J, Junque C. Mental slowness and executive dysfunctions in patients with metabolic syndrome. Neurosci Lett 2009;462:49-53.
- 38. Segura B, Jurado MA, Freixenet N, Falcon C, Junque C, Arboix A. Microstructural white matter changes in metabolic syndrome: a diffusion tensor imaging study. Neurology 2009;73:438-44.
- 39. O'Brien JT, Wiseman R, Burton EJ, Barber B, Wesnes K, Saxby B, Ford GA. Cognitive associations of subcortical white matter lesions in older people. Annals of the New York Academy of Sciences 2002;977:436-44.
- 40. van Dijk EJ, Breteler MM, Schmidt R, Berger K, Nilsson LG, Oudkerk M, Pajak A, Sans S, de Ridder M, Dufouil C, Fuhrer R, Giampaoli S, Launer LJ, Hofman A, Consortium C. The association between blood pressure, hypertension, and cerebral white matter lesions: cardiovascular determinants of dementia study. Hypertension 2004;44:625-30.
- Pa J, Boxer A, Chao LL, Gazzaley A, Freeman K, Kramer J, Miller BL, Weiner MW, Neuhaus J, Johnson JK. Clinical-neuroimaging characteristics of dysexecutive mild cognitive impairment. Annals of Neurology 2009;65:414-23.

- 42. Gatto NM, Henderson VW, St John JA, McCleary C, Hodis HN, Mack WJ. Metabolic syndrome and cognitive function in healthy middle-aged and older adults without diabetes. Neuropsychol Dev Cogn B Aging Neuropsychol Cogn 2008;15:627-41.
- 43. Pavlik VN, Hyman DJ, Doody R. Cardiovascular risk factors and cognitive function in adults 30-59 years of age (NHANES III). Neuroepidemiology 2005;24:42-50.
- 44. Waldstein SR, Katzel LI. Interactive relations of central versus total obesity and blood pressure to cognitive function. Int J Obes (Lond) 2006;30:201-7.
- 45. Farris W, Mansourian S, Leissring MA, Eckman EA, Bertram L, Eckman CB, Tanzi RE, Selkoe DJ. Partial loss-of-function mutations in insulin-degrading enzyme that induce diabetes also impair degradation of amyloid beta-protein. American Journal of Pathology 2004;164:1425-34.
- 46. Craft S. Insulin resistance syndrome and Alzheimer's disease: age- and obesity-related effects on memory, amyloid, and inflammation. Neurobiology of Aging 2005;26 Suppl 1:65-9.
- 47. Li ZG, Zhang W, Grunberger G, Sima AA. Hippocampal neuronal apoptosis in type 1 diabetes. Brain Research 2002;946:221-31.
- 48. Abbatecola AM, Paolisso G, Lamponi M, Bandinelli S, Lauretani F, Launer L, Ferrucci L. Insulin resistance and executive dysfunction in older persons. Journal of the American Geriatrics Society 2004;52:1713-8.
- 49. Isik AT, Cankurtaran M, Bozoglu E, Comert B, Doruk H, Mas MR. Is there any relation between insulin resistance and cognitive function in the elderly? International Psychogeriatrics 2007;19:745-56.
- 50. Hara K, Yamauchi T, Imai Y, Manabe I, Nagai R, Kadowaki T. Reduced adiponectin level is associated with severity of coronary artery disease. International Heart Journal 2007;48:149-53.
- Roberts RO, Geda YE, Knopman DS, Boeve BF, Christianson TJ, Pankratz VS, Kullo IJ, Tangalos EG, Ivnik RJ, Petersen RC. Association of C-reactive protein with mild cognitive impairment. Alzheimer's & Dementia 2009;5:398-405.
- Han SH, Sakuma I, Shin EK, Koh KK. Antiatherosclerotic and anti-insulin resistance effects of adiponectin: basic and clinical studies. Progress in Cardiovascular Diseases 2009;52:126-40.
- 53. Yamauchi T, Kamon J, Waki H, Terauchi Y, Kubota N, Hara K, Mori Y, Ide T, Murakami K, Tsuboyama-Kasaoka N, Ezaki O, Akanuma Y, Gavrilova O, Vinson C, Reitman ML, Kagechika H, Shudo K, Yoda M, Nakano Y, Tobe K, Nagai R, Kimura S, Tomita M, Froguel P, Kadowaki T. The fat-derived hormone adiponectin reverses insulin resistance associated with both lipoatrophy and obesity. Nature Medicine 2001;7:941-6.
- 54. Yaffe K, Lindquist K, Penninx BW, Simonsick EM, Pahor M, Kritchevsky S, Launer L, Kuller L, Rubin S, Harris T. Inflammatory markers and cognition in well-functioning African-American and white elders. Neurology 2003;61:76-80.

- 55. Kubota N, Yano W, Kubota T, Yamauchi T, Itoh S, Kumagai H, Kozono H, Takamoto I, Okamoto S, Shiuchi T, Suzuki R, Satoh H, Tsuchida A, Moroi M, Sugi K, Noda T, Ebinuma H, Ueta Y, Kondo T, Araki E, Ezaki O, Nagai R, Tobe K, Terauchi Y, Ueki K, Minokoshi Y, Kadowaki T. Adiponectin stimulates AMP-activated protein kinase in the hypothalamus and increases food intake. Cell Metabolism 2007;6:55-68.
- 56. Martin B, Pearson M, Kebejian L, Golden E, Keselman A, Bender M, Carlson O, Egan J, Ladenheim B, Cadet JL, Becker KG, Wood W, Duffy K, Vinayakumar P, Maudsley S, Mattson MP. Sex-dependent metabolic, neuroendocrine, and cognitive responses to dietary energy restriction and excess. Endocrinology 2007;148:4318-33.
- 57. Pischon T, Hu FB, Rexrode KM, Girman CJ, Manson JE, Rimm EB. Inflammation, the metabolic syndrome, and risk of coronary heart disease in women and men. Atherosclerosis 2008;197:392-9.
- 58. Arenillas JF, Sandoval P, Perez de la Ossa N, Millan M, Guerrero C, Escudero D, Dorado L, Lopez-Cancio E, Castillo J, Davalos A. The metabolic syndrome is associated with a higher resistance to intravenous thrombolysis for acute ischemic stroke in women than in men. Stroke 2009;40:344-9.
- 59. McNeill AM, Rosamond WD, Girman CJ, Golden SH, Schmidt MI, East HE, Ballantyne CM, Heiss G. The metabolic syndrome and 11-year risk of incident cardiovascular disease in the atherosclerosis risk in communities study. Diabetes Care 2005;28:385-90.
- 60. Malik S, Wong ND, Franklin SS, Kamath TV, L'Italien GJ, Pio JR, Williams GR. Impact of the metabolic syndrome on mortality from coronary heart disease, cardiovascular disease, and all causes in United States adults. Circulation 2004;110:1245-50.
- 61. Gouni-Berthold I, Berthold HK, Mantzoros CS, Bohm M, Krone W. Sex disparities in the treatment and control of cardiovascular risk factors in type 2 diabetes. Diabetes Care 2008;31:1389-91.
- 62. Yaffe K, Kanaya A, Lindquist K, Simonsick EM, Harris T, Shorr RI, Tylavsky FA, Newman AB. The metabolic syndrome, inflammation, and risk of cognitive decline. The Journal of the American Medical Association 2004;292:2237-42.

02 Insulin-resistance and metabolic syndrome are related to executive function in women in a large familybased study

03 The apolipoprotein E gene and its agespecific effects on cogntive function

Abstract

The $\varepsilon 4$ allele of the apolipoprotein E gene (APOE) is a well-established determinant of Alzheimer's disease and cognitive function. We studied the age-specific effects of the APOE* $\varepsilon 4$ allele on cognitive function in a series of 2208 related individuals from a family-based study conducted in an isolated population in the Southwest part of The Netherlands. The effect of the $\varepsilon 4$ allele on cognitive function was evaluated using standard quantitative genetic analysis under a polygenic model, adjusted for cardiovascular risk factors. We found a significant association between the APOE* $\varepsilon 4$ allele and reduced scores on the Adult Verbal Learning Test (AVLT) in persons aged 50 years and older (AVLT short-term memory P = 0.01, AVLT learning P = 0.001, AVLT delayed recall P = 0.01 and memory compound score P = 0.001). The effect of APOE* $\varepsilon 4$ is most pronounced on learning ability, starting as early 40 years. The APOE* $\varepsilon 4$ allele is also strongly associated to cholesterol levels and atherosclerosis. This association did not explain the effect of APOE on cognitive function. Our study suggests that APOE* $\varepsilon 4$ is an important determinant of vascular and neurological pathology at late age.

Introduction

The epsilon4 allele of the apolipoprotein E gene (APOE*ɛ4) is the most important genetic risk factor for Alzheimer's disease (AD) [1]. Although its role in AD has long been known, recently a commercial genetic test for APOE*ɛ4 was launched to predict the risk of AD (http://www.labtestsonline.org) [2]. However, APOE*ɛ4 has also an established effect on lipid levels and through this on the risk and progression of atherosclerosis. Atherosclerosis and hypertension have been implicated in the AD and may partly explain the effects of APOE*ɛ4 on cognitive function [1]. If the effect of APOE*ɛ4 on neurodegeneration is in part caused by vascular pathology, this pathway opens the opportunity of clinical counseling of carriers by screening for vascular pathology. A crucial question in this respect is at which age pathology starts. It has been suggested that APOE*ɛ4 has clinically important effects on cognition in those who do not have signs or symptoms clinical AD. An extensive meta-analysis of all studies conducted in the period 1993-2004 showed evidence for a role of APOE*ɛ4 in cognitive function in non-demented people over 50 years [3]. APOE*ɛ4 was significantly related to reduced global cognitive functioning, episodic memory and executive function in a dose-dependent way, whereas no significant effects were seen for primary memory [3]. Although most studies focused on individuals aged 50 years and over, there is some evidence that with increasing age the effect of APOE*ɛ4 on cognition decreased. However, this trend was far from statistically significant in the meta-analysis. Animal studies provided significant evidence that apolipoprotein E has effects on early brain development [4], suggesting that APOE^{*}²⁴ may impact early cognitive reserve. For humans, the evidence supporting early effects of APOE on cognitive function is scarce and findings have been contradictory.

In the present study, we evaluated the effects of the APOE*ɛ4 allele on specific cognitive domains and vascular pathology over a wide age-range in a 3 generation family-based study. This design provides a powerful setting to address age specific effects of APOE*ɛ4 in a genetically and environmentally homogeneous background.

Description of the cognitive te	sts derived from the neurops	ychological battery		
Neuropsychological test	Cognitive domain	Task description	Score definition	Reference
Individual tests				
AVLT Short-term memory	Short-term memory	Recall immediately after presentation of 15 words	Number of correctly recalled words	Saan and Deelman 1986
AVLT Learning	Learning	Recall after 2nd to 5th presentation	Total number of correctly recalled words	Saan and Deelman 1986
AVLT Delay	Delayed recall	Recall after 30 minutes	Number of correctly recalled words	Saan and Deelman 1986
AVLT Recognition	Recognition	Recognize words from a list	Number of correctly recognized and rejected words	Saan and Deelman 1986
WAIS Verbal fluency	Semantic fluency and phonological fluency	Mention words fiting a frame (semantic & phonological)	Number of correctly mentioned words	Wechsler 2000
Trail-making test (TMT)	Cognitive flexibility	Connect numbers (A) and together with letters in accending order (B)	Ratio of time in seconds to complete part B over part A	Reitan 1955
Stroop	Susceptibility to interference	Read colors (card 2) which are wrongly named (card 3)	Ratio of time in seconds to complete the card 3 over card 2	Hammes 1978
WAIS Block design	Visuoconstructive abilites	Place blocks according to reference	Number of replicated blocks	Wechsler 2000
Compound scores				
Memory performance			Average of z-transformation of AVLT short, learning, delay, recognition	
Executive function			Average of z-transformation of WAIS verbal fluency, TMT, Stroop	
Overall cognitive function			Average of z-transformation of all tests	

AVLT: Auditory Verbal Learning Test WAIS: Weschler Adult Intelligence Scale

Table 1

chapter 03

Materials and Methods

Study population

The Erasmus Rucphen Family (ERF) cohort, which is part of the Genetic Research in Isolated Population (GRIP) program, is a family-based study that includes inhabitants of a genetically isolated community in the south-western area of the Netherlands [5]. ERF aims to investigate the genetic origins of complex disorders and traits. The study population essentially consists of one extended family of descendants from 20 related couples that lived in the isolate between 1850 and 1900 and had at least 6 children. With relatively limited migration until the last few decades, the isolate now includes approximately 20,000 inhabitants. All data were collected between 2002 and 2005. The Medical Ethical Committee of the Erasmus Medical Center Rotterdam approved the study and informed consent was obtained from all participants.

Data collection

Participants underwent extensive medical and neuropsychological examinations at the ERF research centre. The examinations included the determination of cardiovascular risk factors, such as serum total cholesterol, high-density lipoprotein (HDL) cholesterol, triglycerides, systolic and diastolic blood pressure, and common carotid intima media thickness (IMT). Serum markers were determined using an automated enzymatic procedure (Boehringer Mannheim System). Blood pressure was measured twice on the right arm in a sitting position after at least five minutes rest, using an automated device (OMRON 711); the average of the two values was used for analysis. IMT was evaluated using ultrasonography according to previously applied protocols [6-8]. The outcome variable was defined as the mean IMT of the near and far wall of both common carotid arteries. The battery of neuropsychological tests included the Dutch version of the Auditory Verbal Learning Test (AVLT) [9], the Trail Making Test (TMT) [10], the Stroop colourword test [11], the verbal fluency test [10] and the block design subtest of the Weschler Adult Intelligence Scale (WAIS) [12]. These tests were chosen to screen for cognitive deficits related to AD and other dementias [13] and cover different cognitive domains (Table 1). We assessed the general reading ability of the participants with the Dutch Adult Reading Test (DART) [14]. We also computed compound scores for memory performance, executive function and over-all cognitive function (Table 1), by averaging the z-transformed scores of several cognitive tests [15]. The z-scores were calculated based on the direction of the measurement of test performance. For tests where higher scores indicate better performance (AVLT and WAIS tests), $z = (x - \overline{x}) / sd;$ otherwise (TMT and Stroop), $z = (\overline{x} - x) / sd$. In this way, higher compound scores indicate better performance.

Finally, the education level attained by the subjects within the Dutch educational system was determined according to eight ordinal categories from primary school to university [16].

Genotyping

Genomic DNA was extracted from whole blood samples using the salting out method [17]. Samples were genotyped for the APOE C112R (ɛ4 allele) and APOE R158C (ɛ2 allele) polymorphisms with a Taqman allelic discrimination Assay-By-Design (Applied Biosystems, Foster City, CA). The assays utilized 5 nanograms of genomic DNA and 2 microliter reaction volumes. The amplification and extension protocol included an initial activation step of 10 min at 95 degrees, which preceded 40 cycles of denaturation at 95 degrees for 15 seconds and annealing and extension at 50 degrees for 60 seconds. Allele-specific fluorescence was analysed on an ABI Prism 7900HT Sequence Detection System with SDS v 2.1 (Applied Biosystems, Foster City, CA).

Statistical analysis

A considerable proportion of participants failed to complete the TMT part B test (N = 171, 7.9%),

while some failed to complete the Stroop card III (N = 16, 0.7%) and WAIS block design tests (N = 64, 2.9%) within the time limit. We imputed their scores based on correlations between sex, age, and education level. We grouped the APOE genotypes based on the number of $\varepsilon 4$ alleles in a dose-dependent manner [3], zero ($\varepsilon 2/\varepsilon 2$, $\varepsilon 2/\varepsilon 3$ and $\varepsilon 3/\varepsilon 3$ genotypes), one ($\varepsilon 2/\varepsilon 4$ and $\varepsilon 3/\varepsilon 4$ genotypes) and two copies ($\varepsilon 4/\varepsilon 4$ genotype). General characteristics of the study population among the genotypic groups were compared using the one-way ANOVA test for continuous variables and the chi-square test for dichotomous variables as implemented in SPSS V.11.0 (SPSS Inc. Chicago IL). The observed frequencies of the APOE genotypes were tested for deviations from Hardy-Weinberg equilibrium using the exact test for multiple alleles [18].

To evaluate the effect of the ε 4 allele on cognitive functioning and adjust for family relationships, we performed the variable screening analysis under the polygenic model using the SOLAR software package version 4.1.0 [19]. SOLAR was chosen for its power in discriminating the genetic and environmental effects by utilizing all of the information that is provided by large, complex pedigrees. The effect of APOE genotype on cognitive tests was estimated by including APOE genotype (0, 1, or 2 number of ε 4 alleles) as a covariate in the model, adjusted for other covariates including age, age-squared, sex, education, inbreeding, DART score, and cardiovascular risk factors (total cholesterol, triglycerides, IMT, and systolic and diastolic blood pressure). Inbreeding coefficients were computed based on all available genealogical information for the GRIP population (N = 107,091) using PEDIG software [20]. In addition, we investigated the interaction between APOE and age using a multiplicative model. Before SOLAR analyses, scores from the cognitive tests were normalized using a general rank-transformation [21].

To illustrate the age-specific effect of APOE on cognitive function, we smoothed the distribution of cross-sectional test scores across age, using locally weighted regression, or the LOESS smoother, implemented in the software package SigmaPlot version 8.02 [22].

Results

Information on both APOE genotype and cognitive tests is available for 2208 ERF participants in our study. We excluded 65 individuals who were illiterate, blind, deaf, retarded or who reported having a brain tumour, stroke or severe brain damage. The frequencies of APOE alleles were 4.8% for the ε 2 allele, 74.1% for the ε 3 allele and 21.1% for the ε 4 allele. The allele and genotype distributions were in Hardy-Weinberg equilibrium (P = 0.64). There were no significant differences in age, sex, education level and blood pressure between APOE genotype groups (Table 2). Heterozygous and homozygous APOE* ε 4 carriers had thicker IMT compared to non-carriers (P = 0.05, Table 2). Serum levels of total cholesterol (P = 1.75×10-7) and triglycerides (P = 4.80×10-5) significantly increased and serum HDL levels significantly decreased (P = 4.81×10-5) with an increasing number of APOE* ε 4 alleles.

We studied the relationship between cardiovascular factors and cognitive function (Table 3). Serum levels of triglycerides were significantly associated with AVLT recognition (P = 0.04). There was significant and consistent evidence for association between IMT and multiple cognitive domains (AVLT learning, P = 0.01; AVLT recognition, P = 0.01; WAIS verbal fluency, P = 0.0001; memory compound score, P = 0.01; and over-all cognitive function compound score, P = 0.01). Systolic blood pressure was significantly associated with the Stroop test (P = 0.00001) and executive function compound score (P = 0.01). Adjustment for APOE status had little influence on the relationship between vascular risk factors and cognitive function.

Characteristics per APOE genotype

			AP	DE*4			
Characteristics	0 (n=1	342)	1 (n=6	i99)	2 (n=1	02)	P-value
Age (years)	49.0	14.9	49.3	14.4	48.8	13.7	0.85
Gender (% male)	42.9		43.9		46.2		0.76
Body mass index (kg/m2)	27.0	4.6	26.8	4.7	27.5	4.8	0.40
Education	3.24	0.05	3.08	0.07	3.00	0.18	0.07
IMT(mm)	0.81	0.21	0.84	0.21	0.83	0.17	0.05
Systolic blood pressure (mmHg/cm)	140.27	20.46	141.08	21.02	141.44	18.25	0.42
Diastoic blood pressure (mmHg/cm)	80.08	10.44	80.30	10.39	82.24	9.67	0.13
Fasting glucose (mmol/l)	4.62	1.03	4.56	0.87	4.62	0.88	0.35
Serum cholesterol (mmol/l)	5.49	1.07	5.70	1.10	5.87	1.22	1.75E-07
Serum Triglycerides (mmol/l)	1.30	0.75	1.41	0.82	1.62	0.95	4.80E-05
Serum HDL (mmol/l)	1.30	0.36	1.25	0.35	1.21	0.34	4.81E-05

Values presented are means and standard deviations or percentages

Table 2

Cardivasular factors and cognitive function

	Cholest	terol	Triglyce	rides	HDI	L.	IMT	r	SB	Р	DB	P
Cognitive domain	beta	se	beta	se	beta	se	beta	se	beta	se	beta	se
Individual tests												
AVLT Short-term memory	0.05	0.03	-0.03	0.04	0.17	0.10	-0.40	0.23	-0.002	0.001	-0.003	0.002
AVLT Learning	0.19	0.14	0.17	0.20	0.47	0.45	-3.52	1.09 **	-0.013	0.006	-0.008	0.007
AVLT Delay	0.08	0.05	0.11	0.07	0.07	0.16	-0.65	0.39	0.001	0.002	-0.002	0.003
AVLT Recognition	0.09	0.04	0.10	0.06 *	-0.13	0.14	-1.01	0.32 **	0.002	0.002	0.006	0.002
WAIS Verbal fluency	0.44	0.29	0.07	0.40	0.36	0.93	-10.03	2.26 ***	-0.045	0.013	-0.005	0.015
TMT	0.04	0.02	-0.01	0.03	-0.01	0.06	0.00	0.16	-0.001	0.001	-0.003	0.002
Stroop	0.00	0.01	0.00	0.01	-0.03	0.03	0.03	0.06	0.002	0.000 ****	0.000	0.001
WAIS Block design	0.12	0.23	-0.05	0.32	0.03	0.73	-1.94	1.76	-0.008	0.010	-0.013	0.012
Compound scores												
Memory performance	0.03	0.01	0.02	0.02	0.03	0.04	-0.32	0.10 **	0.000	0.001	0.000	0.001
Executive function	0.00	0.01	0.00	0.02	0.04	0.04	-0.21	0.09	-0.002	0.000 **	0.001	0.001
Overall cognitive function	0.01	0.01	0.01	0.01	0.03	0.03	-0.26	0.07 **	-0.001	0.000	0.000	0.000

se: standard errors

Except TMT and Stroop, higher absolute values indicate better performance

P values adjusted for age, sex, inbreeding, education, and family relationship

* P value < 0.05

** P value < 0.001

Table 3

Table 4 presents the effect of the APOE* ϵ 4 allele on cognitive tests. There was a borderline significant association between APOE* ϵ 4 and AVLT learning (P = 0.07), which became significant (P = 0.05) when adjusting for cardiovascular factors. Test scores generally showed a non-significant trend of poorer performance with an increasing number of ϵ 4 alleles. Adjusting for cardiovascular factors had little influence on these results.

When studying cognitive function, there was significant evidence for interaction between APOE* ε 4 and age. The interaction term of age and APOE* ε 4 was significant for AVLT short-term memory (Pinteraction = 0.01), AVLT learning (Pinteraction = 0.05), and memory compound score (Pinteraction = 0.01), while for AVLT delayed recall (Pinteraction = 0.09) and AVLT recognition (Pinteraction = 0.07), the evidence was borderline significant.

			APOE	*E4				
Cognitive domain	0 (n=1	342)	1 (n=6	699)	2 (n=1	02)		
	mean	se	mean	se	mean	se	P1	P2
Individual tests								
AVLT Short-term memory	4.3	0.05	4.3	0.07	4.1	0.17	0.23	0.19
AVLT Learning	33.0	0.25	32.8	0.35	30.9	0.99	0.07	0.05
AVLT Delay	7.5	0.08	7.5	0.11	6.8	0.29	0.19	0.19
AVLT Recognition	27.8	0.06	27.8	0.09	27.7	0.20	0.68	0.96
WAIS Verbal fluency	61.5	0.51	61.0	0.71	60.8	1.83	0.80	0.98
ТМТ	2.7	0.03	2.7	0.04	2.7	0.10	0.64	0.64
Stroop	1.7	0.01	1.7	0.02	1.7	0.07	0.31	0.31
WAIS Block design	27.6	0.41	27.5	0.58	27.0	1.52	0.84	0.91
Compound scores								
Memory performance	0.00	0.02	-0.01	0.03	-0.16	0.08	0.14	0.15
Executive function	-0.01	0.02	-0.04	0.03	-0.02	0.08	0.71	0.92
Overall cognitive function	-0.01	0.02	-0.02	0.03	-0.10	0.07	0.23	0.31

Effect of APOE genotype on cognitive tests

se: standard errors

Except TMT and Stroop, higher absolute values indicate better performance

P1: adjusted for age, sex, inbreeding, education, DART, and family relationship

P2: additionally adjusted for total cholesterol, triglycerides, IMT, and systolic and diastolic blood presure

Table 4

When stratifying the data by age (Table 5), the ε 4 allele was significantly associated with poorer memory performance in those over 50 years of age (AVLT short-term memory P = 0.01, AVLT learning P = 0.001, AVLT delayed recall P = 0.01 and memory compound score P = 0.001). Adjusting for cardiovascular factors had little influence on these effects (Table 5). In younger subjects (\leq 50 years of age), none of the tests were significantly associated to cognitive function (Table 5).

To illustrate the age-specific effect of APOE on memory performance, we plotted the smoothed distribution of test scores across age (Figure 1). Figure 1A shows that APOE has little influence on AVLT short-term memory. Only after the age of 65 years does some effect of the genotype

become apparent. APOE genotype seems to have the most pronounced effect on AVLT learning. The effect starts around age 40 years (Figure 1B). The effects on AVLT delayed recall and memory compound score are less pronounced, but there is a trend towards poorer cognitive performance with increasing number of APOE*ɛ4 alleles (Figure 1C and 1D).

				<=50 y	ears					>50 years						
Cognitive domain	0 (n=709)		1 (n=3	862)	2 (n=	2 (n=51)			0 (n=6	0 (n=633)		1 (n=337)		2 (n=51)		
	mean	se	mean	se	mean	se	P1	P2	mean	se	mean	se	mean	se	P1	P2
Individual tests																
AVLT Short-term memory	4.99	0.06	5.00	0.09	4.84	0.23	0.64	0.59	3.55	0.06	3.46	0.08	3.30	0.20	0.01	0.01
AVLT Learning	37.22	0.30	37.62	0.40	35.47	1.33	0.72	0.89	28.34	0.32	27.63	0.45	26.28	1.15	0.001	0.003
AVLT Delay	8.66	0.10	8.87	0.14	8.18	0.40	0.63	0.99	6.12	0.11	6.09	0.15	5.38	0.33	0.01	0.04
AVLT Recognition	28.58	0.06	28.77	0.08	28.53	0.22	0.22	0.10	26.96	0.11	26.75	0.16	26.90	0.30	0.07	0.18
WAIS Verbal fluency	69.19	0.64	68.30	0.86	65.35	2.63	0.55	0.57	52.72	0.66	52.86	0.98	56.16	2.39	0.99	0.57
TMT	2.51	0.04	2.52	0.05	2.47	0.12	0.34	0.60	2.87	0.04	2.90	0.06	2.87	0.16	0.56	0.70
Stroop	1.59	0.01	1.59	0.01	1.66	0.12	0.99	0.47	1.88	0.02	1.91	0.03	1.82	0.05	0.45	0.24
WAIS Block design	35.65	0.56	35.14	0.82	33.47	2.35	0.97	0.73	18.70	0.37	19.23	0.52	20.44	1.42	0.55	0.36
Compound scores																
Memory performance	0.40	0.03	0.45	0.04	0.28	0.10	0.43	0.54	-0.44	0.03	-0.50	0.04	-0.61	0.10	0.001	0.01
Executive function	0.30	0.02	0.28	0.03	0.19	0.13	0.74	0.75	-0.35	0.03	-0.39	0.04	-0.23	0.09	0.66	0.77
Overall cognitive function	0.37	0.02	0.39	0.03	0.26	0.10	0.66	0.83	-0.44	0.02	-0.47	0.04	-0.47	0.07	0.01	0.07

Effect of APOE genotype on cognitive tests by age category

se: standard errors

Except TMT and Stroop, higher absolute values indicate better performance

P1: adjusted for age, sex, inbreeding, education, DART, and family relationship

P2: additionally adjusted for total cholesterol, triglycerides, IMT, and systolic and diastolic blood presure

Table 5

Because a considerable number of people could not complete the TMT-B test (N = 141), we investigated the distribution of these missing scores. The E4 allele was significantly associated with the proportion of people who could not complete the TMT-B test (none ϵ 4 = 5.4%, one ϵ 4 = 8.5%, and two ϵ 4 = 9.2%, P = 0.02). This effect was more pronounced in women (P = 0.004).

Age specific effect of the APOE E4 allele on memory performance



The distribution of test scores (y-axis) was smoothed across age (x-axis) using the LOESS smoother implemented in the software package SigmaPlot version 8.02. [22]. A: AVLT short-term memory; B: AVLT learning; C: AVLT delayed recall; D: memory compound score.

Figure 1

chapter 03

Discussion

In this study, we found a significant association between the APOE*ɛ4 allele and reduced memory performance in persons aged 50 years and older. This effect is independent of the effect of APOE on cardiovascular factors. In our analyses of cognitive function there is significant evidence for interaction between APOE*ɛ4 and age. The effect of APOE*ɛ4 increases significantly with age, particularly in terms of learning ability. As expected APOE*ɛ4 was strongly related to lipid levels and atherosclerosis, while serum levels of triglycerides, blood pressure and atherosclerosis were significantly associated to cognitive function. Additional adjustment for APOE status had little influence on the relationship between vascular risk factors and cognitive function.

The extensive meta-analysis of all studies conducted in the period 1993-2004 showed that APOE*ɛ4 was significantly related to reduced global cognitive functioning, episodic memory and executive function in a dose-dependent way, whereas no significant effects were seen for primary memory [3]. In contrast, in the present study we see a consistent and significant association to memory. Our findings are in agreement with a recent prospective, population-based study in 5804 subjects aged 70-80 years. That study showed that ɛ4 carriers had significantly poorer performance in immediate and delayed recall at baseline as well as greater decline during the

51

3.2 years of follow up [23]. The effect of APOE was less pronounced on attention and processing cognitive domains. Another study of 611 elderly clergymen showed that the APOE*ɛ4 allele had a pronounced influence on declines in episodic memory [24]. This study also used a compound score, including word list memory, recall, recognition, immediate and delayed recall, which is comparable to our compound score for memory. A number of smaller studies found a relation between APOE*ɛ4 and memory performance [25,26]. Furthermore, a family-based study of relatives of AD patients showed an effect of APOE*ɛ4 on memory in those not yet affected [27]. Finally, episodic memory loss is a key characteristic of AD [28-30] and several epidemiological studies found that measures of delayed recall and learning are predictive of the risk for developing dementia [31-33].

Most studies on APOE and cognition in humans have focused on the elderly. Animal studies, however, have demonstrated that apolipoprotein E has a role in early brain development [4]. In our study, the effect of APOE was not significant in people younger than 50 years of age. Of interest, APOE genotype showed some evidence for an early effect on learning ability. The effect starts in early middle age, at around 40 years.

Cardiovascular factors may potentially be an intermediate feature explaining part of the association between APOE and cognitive function. As expected, we observed a strong association between APOE and serum levels of total cholesterol, triglycerides, and HDL. Systolic blood pressure and the presence of atherosclerosis as measure by IMT were significantly and consistently associated with multiple cognitive domains. Although the relationship between blood pressure and Alzheimer's disease is only observed in prospective studies[34,35], also other studies have found a strong relationship between cognitive function to blood pressure [36] and atherosclerosis [37,38]. In line with the studies on Alzheimer's disease that suggest the effect of APOE on the risk of disease is determined primarily by the effect on lipid metabolism with the brain, the association between APOE and memory performance remained significant after adjusting for serum levels of total cholesterol and triglycerides, IMT, and systolic and diastolic blood pressure. Also other estimates for the relation between APOE*ɛ4 and cognitive function did not change when adjusting for vascular pathology. This indicates that the effect of APOE on cognitive functioning is not likely determined by the effect of APOE on cardiovascular factors. At the same time, the additional adjustment for APOE status had little influence on the relationship between vascular risk factors and cognitive function. The finding implies that measuring APOE will not be clinically relevant for preventive strategies targeting the relationship between vascular risk factors and cognitive function.

In summary, APOE*ɛ4 is associated with poorer memory performance in older people. The effect of APOE*ɛ4 increases significantly with age and is independent of vascular pathology. The effect of APOE*ɛ4 on learning ability starts as early as the age of 40 years. In light of the commercial test recently made available for APOE genotyping, our findings suggest that those who take

52

the test should be informed not only about the risk of AD but also about the effect of APOE genotype on cognitive function and vascular pathology. Whether or not the test is clinically useful remains to be determined in further studies [2]. Our findings clearly show that independent of the test outcome, management of vascular problems will be crucial for maintenance of cognitive function.

References

- Sleegers K, Roks G, Theuns J, Aulchenko YS, Rademakers R, Cruts M, van Gool WA, Van Broeckhoven C, Heutink P, Oostra BA, van Swieten JC, van Duijn CM. Familial clustering and genetic risk for dementia in a genetically isolated Dutch population. Brain 2004;127:1641-9.
- Patterson C, Feightner JW, Garcia A, Hsiung GY, MacKnight C, Sadovnick AD. Diagnosis and treatment of dementia: 1. Risk assessment and primary prevention of Alzheimer disease. Cmaj 2008;178:548-56.
- 3. Small BJ, Rosnick CB, Fratiglioni L, Backman L. Apolipoprotein E and cognitive performance: a meta-analysis. Psychol Aging 2004;19:592-600.
- 4. Kitamura HW, Hamanaka H, Watanabe M, Wada K, Yamazaki C, Fujita SC, Manabe T, Nukina N. Age-dependent enhancement of hippocampal long-term potentiation in knock-in mice expressing human apolipoprotein E4 instead of mouse apolipoprotein E. Neurosci Lett 2004;369:173-8.
- Aulchenko YS, Heutink P, Mackay I, Bertoli-Avella AM, Pullen J, Vaessen N, Rademaker TA, Sandkuijl LA, Cardon L, Oostra B, van Duijn CM. Linkage disequilibrium in young genetically isolated Dutch population. Eur J Hum Genet 2004;12:527-34.
- 6. Gozna ER, Marble AE, Shaw A, Holland JG. Age-related changes in the mechanics of the aorta and pulmonary artery of man. J Appl Physiol 1974;36:407-11.
- 7. Nikol S, Isner JM, Pickering JG, Kearney M, Leclerc G, Weir L. Expression of transforming growth factor-beta 1 is increased in human vascular restenosis lesions. J Clin Invest 1992;90:1582-92.
- 8. Ohji T, Urano H, Shirahata A, Yamagishi M, Higashi K, Gotoh S, Karasaki Y. Transforming growth factor beta 1 and beta 2 induce down-modulation of thrombomodulin in human umbilical vein endothelial cells. Thromb Haemost 1995;73:812-8.
- 9. Saan R, Deelman B. De 15-Woordentests A en B. (Een voorlopige bandleiding) [Internal publication]. Groningen: Section neuropsychology, Academic Hospital Groningen. 1986.
- 10. Reitan RM. The relation of the Trail Making Test to organic brain damage. Journal of Consulting Psychology 1955;19:393-394.
- 11. Hammes J. Stroop Kleur-woord Test: Dutch Manual. Swets and Zeitlinger BV: Lisse, The Netherlands 1978.
- 12. Wechsler D. Wechsler adult intelligence scale 3rd (WAIS-III): test Manual (Dutch version). New York: Psychological Corporation 2000.
- 13. Estevez-Gonzalez A, Kulisevsky J, Boltes A, Otermin P, Garcia-Sanchez C. Rey verbal learning test is a useful tool for differential diagnosis in the preclinical phase of Alzheimer's disease: comparison with mild cognitive impairment and normal aging. Int J Geriatr Psychiatry 2003;18:1021-8.
- 14. Schmand B, Lindeboom J, Van Harskamp F. De Nederlandse Leestest voor Volwassenen. [The Dutch adult reading test] Lisse: Swets and Zeitlinger. 1992.

- 15. Prins ND, Den Heijer T, Hofman A, Koudstaal PJ, Jolles J, Clarke R, Breteler MM. Homocysteine and cognitive function in the elderly: the Rotterdam Scan Study. Neurology 2002;59:1375-80.
- 16. Van der Elst W, van Boxtel MP, van Breukelen GJ, Jolles J. Rey's verbal learning test: normative data for 1855 healthy participants aged 24-81 years and the influence of age, sex, education, and mode of presentation. J Int Neuropsychol Soc 2005;11:290-302.
- 17. Miller SA, Dykes DD, Polesky HF. A simple salting out procedure for extracting DNA from human nucleated cells. Nucleic Acids Res 1988;16:1215.
- 18. Guo SW, Thompson EA. Performing the exact test of Hardy-Weinberg proportion for multiple alleles. Biometrics 1992;48:361-72.
- 19. Almasy L, Blangero J. Multipoint quantitative-trait linkage analysis in general pedigrees. Am J Hum Genet 1998;62:1198-211.
- 20. Boichard d. PEDIG: a FORTRAN package for pedigree analysis studied for large populations. Proceeding of the 7th Word Congress of Genet Appl Livest Prod 2002:28.
- 21. Huberty C. Applied Discriminant Analysis. Wiley, New York 1994.
- 22. Dagum E, Luati A. A study of the asymmetric and symmetric weights of Kernel smoothers and their spectral properties. Estadistica J. Interamerican Statist. Inst. 2001;53:215-258.
- Packard CJ, Westendorp RG, Stott DJ, Caslake MJ, Murray HM, Shepherd J, Blauw GJ, Murphy MB, Bollen EL, Buckley BM, Cobbe SM, Ford I, Gaw A, Hyland M, Jukema JW, Kamper AM, Macfarlane PW, Jolles J, Perry IJ, Sweeney BJ, Twomey C. Association between apolipoprotein E4 and cognitive decline in elderly adults. J Am Geriatr Soc 2007;55:1777-85.
- Wilson RS, Schneider JA, Barnes LL, Beckett LA, Aggarwal NT, Cochran EJ, Berry-Kravis E, Bach J, Fox JH, Evans DA, Bennett DA. The apolipoprotein E epsilon 4 allele and decline in different cognitive systems during a 6-year period. Arch Neurol 2002;59:1154-60.
- 25. Bondi MW, Salmon DP, Galasko D, Thomas RG, Thal LJ. Neuropsychological function and apolipoprotein E genotype in the preclinical detection of Alzheimer's disease. Psychol Aging 1999;14:295-303.
- 26. Wehling E, Lundervold AJ, Standnes B, Gjerstad L, Reinvang I. APOE status and its association to learning and memory performance in middle aged and older Norwegians seeking assessment for memory deficits. Behav Brain Funct 2007;3:57.
- 27. Lee JH, Flaquer A, Stern Y, Tycko B, Mayeux R. Genetic influences on memory performance in familial Alzheimer disease. Neurology 2004;62:414-21.
- Smith JD. Apolipoproteins and aging: emerging mechanisms. Ageing Res Rev 2002;1:345-65.
- 29. Huang Y. Apolipoprotein E and Alzheimer disease. Neurology 2006;66:S79-85.
- *30. Weisgraber KH, Mahley RW. Human apolipoprotein E: the Alzheimer's disease connection. Faseb J 1996;10:1485-94.*

- 31. Tierney MC, Yao C, Kiss A, McDowell I. Neuropsychological tests accurately predict incident Alzheimer disease after 5 and 10 years. Neurology 2005;64:1853-9.
- 32. Albert MS, Moss MB, Tanzi R, Jones K. Preclinical prediction of AD using neuropsychological tests. J Int Neuropsychol Soc 2001;7:631-9.
- *33.* Grober E, Kawas C. Learning and retention in preclinical and early Alzheimer's disease. Psychol Aging 1997;12:183-8.
- 34. Muller M, Tang MX, Schupf N, Manly JJ, Mayeux R, Luchsinger JA. Metabolic syndrome and dementia risk in a multiethnic elderly cohort. Dement Geriatr Cogn Disord 2007;24:185-92.
- Qiu C, von Strauss E, Fastbom J, Winblad B, Fratiglioni L. Low blood pressure and risk of dementia in the Kungsholmen project: a 6-year follow-up study. Arch Neurol 2003;60:223-8.
- 36. Knecht S, Wersching H, Lohmann H, Bruchmann M, Duning T, Dziewas R, Berger K, Ringelstein EB. High-normal blood pressure is associated with poor cognitive performance. Hypertension 2008;51:663-8.
- 37. Chang EH, Rigotti A, Huerta PT. Age-related influence of the HDL receptor SR-BI on synaptic plasticity and cognition. Neurobiol Aging 2007.
- 38. Komulainen P, Kivipelto M, Lakka TA, Hassinen M, Helkala EL, Patja K, Nissinen A, Rauramaa R. Carotid intima-media thickness and cognitive function in elderly women: a population-based study. Neuroepidemiology 2007;28:207-13.

03 The apolipoprotein E gene and its age-specific effects on cognitve function

04 Genome-wide linkage screen of cognitive function identifies susceptible chromosomal regions

Abstract

Cognitive function is a complex trait which involves multiple environmental and genetic factors. To localize genes involved in cognitive functioning, we conducted genome-wide linkage analyses in a large family from a genetically isolated population. A broad range of cognitive test measurements were available for 2882 participants. We performed non-parametric linkage analysis in participants with low cognitive test scores, defined as a score in the lowest 10% of the distribution. Genome-wide significant and suggestive thresholds for linkage were estimated empirically using simulation study. Fine-mapping was performed in significant linkage regions comparing participants with low cognitive scores to those with high cognitive scores in the same population. Replication analyses were performed in an independent study, the Rotterdam Study, which is an ongoing population-based cohort study with comparable cognitive estimates. We found significant linkage (LOD > 3.78) of cognition was found to chromosomes 1p13.1, 12q24.33, 19q13.43, 20p13, 21q22.13 and 21q22.3. Fine-mapping showed significant associations to chromosome 1 (p-value=0.03) and 21 (p-value=0.01) after multiple testing correction. The region on 21q22.13 was replicated in the Rotterdam Study (nominal p-value 0.003). The fine-mapping and replication results pointed to variants within the potassium inwardly-rectifying channel, subfamily J, member 6 gene (KCNJ6).

Introduction

Cognitive function is a broad concept referring to multiple cognitive domains, among which memory, language, executive function and visuospatial ability. Impairment of cognitive function is seen in patients with various diseases including dementia, bipolar disorder, schizophrenia and attention deficit hyperactivity disorder (ADHD) [1-3]. The heritability estimates for cognitive function vary between 20-79% [4,5], making cognition a potential endophenotype for the various neuropsychiatric outcomes [6,7]. The genes implicated in cognitive function are poorly understood. Many candidate genes have been studied [8,9], and the most consistently reported genetic factor affecting cognition is apolipoprotein E (APOE), the major genetic determinant of early- and late-onset Alzheimer's disease [10,11]. The effect of APOE on cognitive function is small, especially compared to its effect in Alzheimer's disease [12]. In search for unknown genes, both genome wide association studies as well as linkage studies have been conducted. The three genome wide association studies that have so far been conducted, were relatively small including 333 up to 700 persons in the discovery set and identified three genes: the sortilin-related receptor gene (SORL1) in relation to abstract reasoning[13], the WW and C2 domain containing 1 gene (KIBRA) in relation to delayed recall[14] and the sodium channel, voltage-gated, type I, alpha subunit gene (SCN1A) in relation to short-term memory[15]. In addition, several linkage studies have been conducted aiming to identify genes with a relatively large effect on cognition. There is evidence for linkage of memory to chromosomes 4 and 12, mental flexibility to chromosomes 5 and 11, IQ to chromosomes 2 and 6, motor timing to chromosomes 2 and 13, processing speed to chromosome 14, and reading ability to chromosome 18 04 Genome-wide linkage screen of cognitive function identifies susceptible chromosomal regions

[13,16-23] (Table 1). In contrast to dementia oriented studies of APOE, the linkage studies conducted were predominantly using cognitive function as an endophenotype for psychiatric outcomes including schizophrenia, alcohol dependence and ADHD. Two general issues of linkage studies is that the power of linkage analysis is low, requiring large sample sizes and that the regions of interest that are identified are large and single genes cannot be pinpointed. To overcome the first issue, we embedded our study within a large family-based program and to overcome the second issue, we combined our linkage studies with an association analysis in the area under the peak, allowing us to narrow down the region of interest [24].

First author	Year	Family	Trait	Chromosome	LOD
Paunio	2004	Schizophrenia	Verbal learning, memory	4q21	3.8
Posthuma	2005	General population	Performance IQ	2q24.1-31.1	4.4
			Verbal and full scale IQ	6p25.3-22.3	3.2
Buyske	2006	Alcohol dependence	Digit Span Test	11q25	3.1
			Digit Symbol Substitution Task	14q11	6.0
			Digit Symbol Substitution Task	14q24.2	3.9
Dick	2006	Alcohol dependence	Full scale IQ	6p	3.3
Luciano	2006	General population	Cambridge Contextual Reading Test, Performance IQ	2q24-31	4.2, 3.7
		Adolescent twins	Arithmetic-verbal subtest	6p	3.3
			Schonell reading test	14q13-21	3.2
			Arithmetic-verbal subtest	21g22	3.0
Singer	2006	Adult twins	Prospective memory	12g22	2.8
Seshadri	2007	General population	Reading Test, Native intelligence	18p11	5.1
Almasy	2008	Schizophrenia	Abstraction, mental flexibility	5q	3.4
Rommelse	2008	ADHD	Motor Timing	2q21.1	3.9
			Digit span	13012.11	4.0

Table 1

Here, we report the findings of a linkage analysis of various cognitive traits. The analyses were conducted in a large family-based study, the Erasmus Rucphen Family study (ERF) including 2882 persons derived from a genetically isolated population in the Netherlands. This population was not selected for any disease and participants were tested with an extensive neuropsychological test battery with exclusion of prevalent dementia. Since we are targeting genes with a major effect, we selected persons from the extremes of the trait distribution for the linkage analysis. For the fine-mapping of the region we used dense genotyping in the regions under linkage peak in ERF and replicated the findings in a large outbred, population-based cohort, the Rotterdam Study.

Materials and Methods

Study population

The study was conducted in a genetically isolated population in the South-West of the Netherlands. Participants were part of an ongoing family-based cohort study, the ERF-study, which is embedded in a program aiming to identify genetic risk factors of complex diseases. Participants are all descendents of a limited number of founders living in the 19th century. Extensive genealogical data is available for this population [25,26]. The study protocol included venous puncture for DNA isolation and chemistry, cognitive evaluation, cardiovascular examination, eye assessments and body composition measurements. All participants gave informed consent and the study was approved by the medical ethics committee at Erasmus MC University Medical Center. For the current study, participants with a history of cerebrovascular accidents, dementia, brain tumors or other conditions that could have influenced reliable neuropsychological assessment were excluded from analysis (N=80). Cognitive test data was available for 2882 participants.

Cognitive evaluation

A 50-minute test battery was used including tests that were applicable over a wide age-range and for which a Dutch validated version was available [27,28]. General cognitive ability was tested using the Dutch Adult Reading Test (DART), which measures the number of correctly pronounced words. Memory was tested with a word learning test (after Rev's Auditory Verbal Learning test) [27] consisting of five exposure trials measuring immediate memory, learning, delayed recall and recognition. Immediate memory was defined as the total number of words recalled at the first trial, learning was defined as the total number of correctly recalled words in trial 2 to 5, delayed recall was defined as the number of correctly recalled words after 20 minutes delay and recognition was defined as the sum of the number of correctly recognized and correctly rejected words. Executive function was assessed with the time-demanding Trail Making Test (part A and B), the Stroop Colour and Word Test (card I, II and III), and with the verbal fluency test. The first two tests are time-demanding tasks, in which subjects have to connect numbers and letters (TMT) or name correct words and colours (Stroop) as quickly as possible. Verbal fluency was assessed with an animal naming and letter naming subtask, in which subjects had to name as many items as possible of the same category with a one minute time limit. Visuospatial ability (VSA) was assessed with the WAIS III - Block Design Test. The test consisted of reproducing two-dimensional patterns using cubes that have red, white, and half-red-half-white faces. The total number of correctly reproduced blocks within a time limit was used as a score.

For analyses, we used the ratios of TMT-B and TMT-A, and of the Stroop Color-Word card III and card II. People with missing test scores on TMT-B due to exceeding the time limit or misunderstanding of test instructions, were given the lowest score of 300 seconds.

To derive more general measures of cognition for analyses, we also computed three composite scores.

These scores were based on z-score transformations of test values [29]. A memory composite score (zmem) was derived by taking the average of z-scores for immediate memory, learning, delayed recall and recognition. An executive composite score (zexec) was derived from average of z-scores for verbal fluency, stroop-ratio and TMT-ratio. Finally, a global cognitive function score (zglob) was computed by taking the average of z-scores for all tests, but the DART.

Participants were asked to report their highest level of education. Education was categorized in 8 categories: 1. primary education; 2. primary education plus a higher not completed education; 3. lower vocational education; 4. lower secondary education; 5. intermediate vocational education; 6. higher secondary education; 7. higher vocational education; 8. university training. Years of education varied from 6 years in the lowest category to at least 16 years in the highest.

Statistical analysis

General descriptive statistics were performed with SPSS for Windows (version 15.0) using logistic regression or Chi2-statistics for the comparison of cases and controls.

Linkage analysis

We performed linkage analysis of affected only, for which the cases were defined as individuals with the lowest 10% of the distribution of the residuals from the regression of the cognitive scores onto age, sex and education.

All study subjects were part of one large pedigree containing 23612 individuals spanning 18 generations. For analysis, we constructed smaller subpedigrees with a maximum bit size of 18 using the software PEDCUT [30] because of the linkage software restraints.

Samples were genotyped on the Illumina HumanHap 6k Beadchip linkage panel. Markers with a minor allele frequency greater than 5% and callrate higher than 95% were used in the analyses. Genotyping errors were checked with MERLIN and PEDCHECK [31,32]. Markers showing high Mendelian inconsistency rates were excluded from analysis. In case of sporadic errors, inconsistent variants were set to missing. For analyses, there were 5250 autosomal SNPs available. The linkage analysis included 233 to 260 cases depending on the cognitive trait that was analyzed.

Before running linkage analysis, the data was reformatted with the software MEGA2 to derive the correct input format [33]. We performed genome-wide nonparametric linkage analysis in MERLIN [31] using a pair-wise approach of estimating IBD allele sharing [34-36].

Thresholds for genome-wide significant and suggestive findings were estimated by performing 500 genome-wide simulations on the global cognitive trait (zglob) [37]. For these simulations, we used the complete pedigree. The typed marker set was used for simulation of the number of markers and intermarker distances using GENEDROP. Simulation linkage

analyses were done using the same files containing allele frequencies, pedigrees and genetic model as were used in the original linkage analysis. Per simulation, the highest log of odds (LOD) was extracted and combining the 500 simulations resulted in a LOD score of 3.79 corresponding to a genome-wide type 1 error rate of 5% (significant threshold) and of 2.72 corresponding to a type 1 error rate of 50% (suggestive threshold).

Association analysis

Significant regions were further studied in a denser genotype set. We selected SNPs within a region around the highest LOD-score minus 1 LOD. A random sample of the population was genotyped using the Illumina HumanHap 320 K chip. We performed logistic regression using the extremes of the distribution to define cases and controls. Cases were defined as described above; controls were participants within the highest 50% of the age, sex and education-adjusted trait. Family-relationship was taken into account by adjusting the p-values using genomic control [38]. The genomic control inflation factors ranged from 0.98 for immediate memory to 1.86 for DART with a mean of 1.17. Analyses were performed in GenABEL(R-library) [39] using the GRAMMAR method [40]. We corrected for multiple testing by performing 10,000 permutations.

Replication

We sought replication in the Rotterdam Study, which includes participants from a population-based cohort situated in a rural area of Rotterdam, The Netherlands [41]. The study includes 7983 elderly Caucasians aged 55 years or older who were invited for extensive examinations at baseline and three follow-up rounds. The study was approved by the local Medical Ethics Committee and all participants gave informed consent. Neuropsychological tests included the word learning test, the Stroop Color and Word Test, Verbal fluency (animal naming) and the Letter Digit Substitution Task. We computed composite scores for memory by using the z-scores of immediate memory and delayed recall. For executive, we used the z-scores of Stroop-ratio and verbal fluency test, and we used all tests for the estimation of the global cognition composite score (zglob). Cases and controls were defined similarly to the discovery study. In the current study, we did not include people who were demented or had a stroke prior to the neuropsychological testing. Genotyping was done on the Illumina Infinium HumanHap 550-schip v3.0 and the genotype data was used to impute to 2.5 million non-monomorphic, autosomal single nucleotide polymorphisms (SNPs) using release 22 HapMap CEU population as a reference. For the analyses, we selected SNPs within the linkage peak as described above. Analyses were performed in GenABEL (R-library) [39] using logistic regression models. To adjust for multiple testing, a Bonferroni corrected p-value <0.05 was considered significant.

Results

Baseline characteristics of the cases and controls are shown in Table 2. The descriptives of the specific cognitive tests can be found in the Supplementary Material (Supplementary Table 1). The mean age of the cases ranged from 48 to 51 years in the ERF Study and from 57 to 63 years in the Rotterdam Study.

Baseline characteristics of the study population by composite scores cognitive function

Linkage						Association										
					Family-based study					Population-based study						
zglob zmem zexec		zglob		zr	zmem		zexec		zglob		zmem		xec			
Case	Case	Case	Case	Control	Case	Control	Case	Control	Case	Control	Case	Control	Case	Control		
251	248	233	215	1101	224	1115	204	1171	191	955	315	1576	205	1026		
59.4%	57.3%	58.4%	59.1%	57.1%	55.8%	57.0%	58.8%	56.6%	60.2%	57.9%	57.6%	58.4%	56.2%	57.5%		
49.2 (17.6)	48.9 (19.1)	50.9 (16.6)	50.2 (17.8)	48.2 (13.7)	51.2 (16.7)	48.8 (13.8)	51.2(16.7)	49.5 (13.4)	63.4 (5.9)	63.7 (5.5)	62.9 (5.6)	63.7 (5.6)	66.2 (6.8)	65.4 (6.5)		
34.3%	36.3%	35.6%	34.0%	28.8%	34.8%	29.4%	37.3%	30.5%	15.2%	11.9%	13.7%	12.7%	17.8%	15.2%		
	zglob Case 251 59.4% 49.2 (17.6) 34.3%	zglob zmem Case Case 251 248 59.4% 57.3% 49.2 (17.6) 48.9 (19.1) 34.3% 36.3%	zijob zmem zesc Case Case Case 251 248 233 59.4% 57.3% 58.4% 492 (17.5) 48.9 (19.1) 50.9 (16.6) 34.3% 36.3% 35.6%	zglob zmem zexc zg Case Case Case Case 243 251 248 233 215 59.4% 59.1% 949.2 (17.5) 48.9 (19.1) 50.9 (16.5) 50.2 (17.8) 34.3% 35.6% 34.0%	zglob zemem zexec zglob Case Case Case Control 251 248 233 215 1101 949/4 57.3% 58.4% 59.1% 57.1% 49.2 (17.5) 48.9 (19.1) 50.9 (16.6) 50.2 (17.8) 48.2 (13.7) 34.3% 36.3% 35.6% 34.0% 28.8%	zglob zmem zexec zglob zmilly-b Case Case Case Control Case 23 251 248 233 215 1101 224 9.9% 57.3% 58.4% 50.1% 57.1% 55.6% 49.2 (17.6) 48.9 (19.1) 50.9 (16.5) 50.2 (17.8) 48.2 (13.7) 51.2 (16.7) 94.3% 35.6% 34.0% 28.8% 34.8% 54.8%	Zajob zemen zesco zagiob ramily-based study Case Case Case Control Case Control 251 248 233 215 1101 224 1115 94% 57.3% 58.4% 50.1% 55.8% 57.0% 58.9% 77.0% 942 (17.6) 48.9 (19.1) 50.9 (16.6) 50.2 (17.8) 48.2 (13.7) 51.2 (16.7) 48.8 (13.8) 34.3% 35.8% 35.8% 34.0% 28.8% 34.8% 29.4%	Zajob zenem zesco zgiob zemily-based study z Case Case Case Control Case 2 251 248 233 215 1101 224 1115 204 254 248 233 215 1101 224 1115 204 92/17.6) 48.9 (19.1) 50.9 (16.6) 50.2 (17.8) 48.2 (13.7) 51.2 (16.7) 48.8 % 94.3% 35.8% 35.8% 34.0% 28.8% 34.48% 29.4% 37.3%	zglob zmem zexc zglob zmem zexc Case Case Case Case Control Case	zglob zmem zexec zglob zglob zmem zexec zglob zglob zmem zglob <td>Linkays Zasociation Association zglob zmem zexec zglob zmem zexec zglob Case Case Case Control Case<td>Resolution zglob zmem zexec zeglob zemily-based study zexec zglob zglob zmem zexec zglob zglob zmem zexec zglob zglob zglob zglob zglob zmem zexec zglob <th< td=""><td>Linkage Family-based study Source Population-based study zglob zmem zglob zmem zglob zglob zglob zmem zglob zglob zglob zmem zglob zglob zglob zmem zglob zglob zglob zmem zglob zglob zmem zglob zglob zglob zmem zglob zglob</td><td>Linkaye Zesce Case Control Zesce Population-based study zado zado</td></th<></td></td>	Linkays Zasociation Association zglob zmem zexec zglob zmem zexec zglob Case Case Case Control Case <td>Resolution zglob zmem zexec zeglob zemily-based study zexec zglob zglob zmem zexec zglob zglob zmem zexec zglob zglob zglob zglob zglob zmem zexec zglob <th< td=""><td>Linkage Family-based study Source Population-based study zglob zmem zglob zmem zglob zglob zglob zmem zglob zglob zglob zmem zglob zglob zglob zmem zglob zglob zglob zmem zglob zglob zmem zglob zglob zglob zmem zglob zglob</td><td>Linkaye Zesce Case Control Zesce Population-based study zado zado</td></th<></td>	Resolution zglob zmem zexec zeglob zemily-based study zexec zglob zglob zmem zexec zglob zglob zmem zexec zglob zglob zglob zglob zglob zmem zexec zglob <th< td=""><td>Linkage Family-based study Source Population-based study zglob zmem zglob zmem zglob zglob zglob zmem zglob zglob zglob zmem zglob zglob zglob zmem zglob zglob zglob zmem zglob zglob zmem zglob zglob zglob zmem zglob zglob</td><td>Linkaye Zesce Case Control Zesce Population-based study zado zado</td></th<>	Linkage Family-based study Source Population-based study zglob zmem zglob zmem zglob zglob zglob zmem zglob zglob zglob zmem zglob zglob zglob zmem zglob zglob zglob zmem zglob zglob zmem zglob zglob zglob zmem zglob	Linkaye Zesce Case Control Zesce Population-based study zado		

Cases are defined as the lowest 10 % of the residuals of the trait regressed on sex, age and education. Controls are defined as highest 50% of the residuals. ** significant differences between cases and controls (pc-0.01) 1 use of antidepressive medication, HADS-D score and APOE 6+ carriership were known for less persons than were included in the genetic analysis. zmen: composite score for memory: zeec: composite score for executive function, zglob: composite score for global cognition.

Table 2

chapter 04

Chromosome	Variant	Physical Position	cM*	Trait	Domain	Non-parametric model	Gene	Candidate genes in region*
						LOD		
1p13.1	rs1555793	117358754	135 – 142	Stroop	Executive	4.21	IGSF2	PTGFRN, NGF
12q24.33	rs2270928	132238673	170 – 173	Zglob	Global	6.36	ZNF10	
				Zmem	Memory	5.70		
				WLTD	Memory	4.89		
				TMT	Executive	4.37		
				Recognition	Memory	4.09		
19q13.43	rs893186	63660991	110 - 111	DART	Global	5.26	ZNF324B	SLC27A5, HNG
				TMT	Executive	4.09		
				Fluency	Executive	3.99		
				Recognition	Memory	3.90		
20p13	rs751596	99330	0-4	VSA	Visuospatial	4.10	intergenic	TRIB3
				WLTI	Memory	3.80		
21q22.13	rs2835629	37443530	43 – 49	VSA	Visuospatial	4.26	TTC3	DYRK1A
				Zglob	Global	4.14		KCNJ6
				Stroop	Executive	3.92		
21q22.3	rs2256207	46886508	70 – 79	VSA	Visuospatial	4.52	PRMT2	S100B
				WF	Executive	3.78		PCNT

LOD scores in bold pass the significance threshold. * boudaries of the linkage peak defined as the highest LOD +/- 1LOD.

Stroop: stroop-ratio; zglob: composite score for global cognition; zmem: composite score for memory; WLTD: AVLT delayed recall; TMT: trail making test ratio; DART: Dutch Adult R VSA: Block Design Test; WLTI: AVLT immediate recall; WF: verbal fluency

Table 3

Table 3 shows the significant linkage regions with LOD scores greater than 3.79. Evidence for linkage to cognition was found on chromosomes 1p13.1, 12g24.33, 19g13.43, 20p13, 21g22.13 and 21g22.3. An overview of LOD scores per trait for these six regions is shown in Figure 1. The figure shows that the regions on chromosome 12, 19, 20 and 21 are significantly linked to at least 2 different cognitive tests. Table 4 gives the results of the fine-mapping analysis of the significant linkage regions in ERF, while Figure 2 shows the replication findings in the Rotterdam Study.

The most consistent findings of linkage and association was found on chromosome 21, where linkage was found to VSA, global cognition (zglob), Stroop and verbal fluency. The two regions showed significant evidence for association when adjusting for multiple testing by permutation testing (see Table 4). In the first region (21q22.13), significant association was seen with global cognition (zglob) and executive function (zexec, fluency) to variants within the potassium-inwardly-rectifying-channel-subfamily-J-member-6 gene (KCNJ6). This region was replicated in the outbred population (Rotterdam Study), where a variant (rs2836034) within KCNJ6 was significantly associated with memory (zmem) after Bonferonni correction (p=0.04; see Figure 2). In the second region on chromosome 21 (21q22.3), immediate memory (p=0.04), delayed recall (p=0.01) and recognition (p=0.04) were significantly associated to variants within the protein-arginine-methyltransferase-2 gene (PRMT2) and the minichromosome-maintenance-complex-component-3-associated-protein gene (MCM3AP). The highest LOD scores in the region for these traits were 3.07, 2.59 and 2.10 respectively. This region was not replicated in the Rotterdam study.



WLTI: AVLT immediate recall; Learn: word learning test trial 2-5; WLTD: AVLT delayed recall; Recogn: AVLT recognition; zmem: composite score for memory; Stroop: stroop-ratio; TMT: trail making test ratio; WF: verbal fluency: zexec: composite score for executive function: VSA: Block Design Test: zolob: composite score for olobal cognition;

Heatplot depicting the highest LOD score per trait and region in the ERF Study

DART: Dutch Adult Reading Test.

Figure 1

The peak under the LOD score on chromosome 1p13.1 included an interesting candidate gene, the nerve-growth-factor-beta gene (NGF) and significant association was seen with multiple cognitive traits after correction for multiple testing in the ERF Study. Although the findings were consistent when replicating the association in the Rotterdam Study for two variants (rs1555793 and rs10801929) in the sense that the lowest p-values were found for the same cognitive tests as in ERF (stroop nominal p=0.06 and zexec nominal p=0.03). However, these SNPs were not significant after Bonferroni correction.

The chromosome 12q24.33 region, was found to be significantly linked to memory (delayed recall, recognition and zmem), executive function (TMT) and global cognition (zglob) and fine-mapping pointed to the fibrosin-like-1 gene (FBRSL1) for delayed recall (rs2323982) and to an intergenic region between the pseudogene LOC647503 and the zinc-finger-protein-605 gene (ZNF605) for Stroop (rs1278607). In the replication sample, the smallest p-value for rs2323982 was also found with delayed recall, but was not significant (nominal p-value of 0.06).

At chromosome 19g13.43, we could not pinpoint a significant region with association. There was some evidence for association of memory to the zinc-finger-and-SCAN-domain-containing-18 gene (ZSCAN18), which was not significant after adjusting for multiple testing. In the Rotterdam Study we observed significant associations between executive function (stroop) and rs1051827 located in the zinc-finger-protein-606 gene (ZNF606), at 100 kb distance from ZSCAN18 (nominal p=0.001).

Results of the a	ssociation analy	ysis under the signifi	cant link	age peaks (high	nest LOD +/- 1 I	LOD) in the ERF Stu	dy		
Chromosome	Best Variant	Physical position	MAF	Trait	Domain	Nominal p-value	P -value†	Gene	Flanking genes
1p13.1	rs6691374	115444341	0.188	Stroop‡	Executive	5.34E-03	0.033	intergenic	TSPAN2, NGF
	rs1146179	115560916	0.218	TMT	Executive	5.65E-04	0.010	intergenic	TSPAN2, NGF
	rs699718	117047372	0.156	DART	Global	1.94E-04	0.023	intergenic	GAPDHL9, IGSF3
	rs6657718	114517886	0.156	Fluency	Executive	1.64E-03	0.023	intergenic	SYT6, MRP63P1
	rs10801929	117395987	0.329	Zexec	Executive	3.08E-03	0.044	intergenic	IGSF2, TTF2
	rs2582783	120126518	0.338	WLTD	Memory	2.45E-03	0.043	intergenic	REG4, HMGCS2
				Recogn		1.17E-03	0.032		
				Zmem		1.22E-03	0.027		
	rs10494201	118053780	0.315	VSA	Visuospatial	7.84E-04	0.009	intergenic	FAM46C, LOC100131261
12q24.33	rs10781655	131554987	0.384	Zglob‡	Global	0.02	0.223	MUC8	
			0.384	Zmem‡	Memory	3.51E-03	0.061	MUC8	
	rs2323982	131652594	0.324	WLTD‡	Memory	9.60E-04	0.010	FBRSL1	
	rs11246991	131254068	0.138	TMT‡	Executive	0.01	0.329	GALNT9	
	rs1132375	131841610	0.430	Recognition‡	Memory	0.10	0.918	ANKLE2	
	rs1882297	131579207	0.282	WLTI	Memory	6.29E-03	0.083	FBRSL1	
	rs1574157	131640365	0.037	Zexec	Executive	0.02	0.245	FBRSL1	
	rs1278607	132000536	0.448	Stroop	Executive	2.43E-03	0.038	intergenic	LOC647503, ZNF605
19q13.43	rs260470	63496625	0.267	DART‡	Global	0.02	0.367	ZNF8	
	rs260423	63420161	0.098	TMT‡	Executive	0.01	0.261	ZNF274	
	rs3915790	63507665	0.198	Fluency‡	Executive	0.04	0.729	intergenic	ZNF8, ZSCAN22
	rs8100801	63583872	0.049	Recognition‡	Memory	0.03	0.701	ZNF837	
	rs3810126	63302066	0.272	Zmem	Memory	7.97E-03	0.239	ZSCAN18	
20p13	rs3746807	693963	0.126	VSA±	Visuospatial	3.20E-03	0.185	C20orf54	
	rs1418258	11799	0.398	WLTI±	Memory	1.81E-03	0.019	intergenic	DEFB125
	rs6085394	608793	0.245	DART	Global	1.82E-03	0.070	intergenic	SCRT2, C20orf54
21022 13	rs1893654	37267597	0 489	VSA+	Visuospatial	1.08E-03	0.920	HLCS	
Erq22.10	rs1892682	38169935	0.256	Zalob±	Global	6.17E-03	0.011	KCN.I6	
				Zexec	Executive	3.88E-04	0.008		
	rs2835872	37949142	0.284	Stroop‡	Executive	6.24E-03	0.268	KCNJ6	
	rs2835933	38045624	0.266	Fluency	Executive	2.51E-04	0.025	KCNJ6	
21022.3	rs2298694	46474790	0.041	VSAt	Visuospatial	0.02	0.432	LSS	
	rs2070429	46786139	0.350	Fluencyt	Executive	0.04	0.534	DIP2A	
	rs2839376	46897344	0.082	WLTI	Memory	1.27E-03	0.041	PRMT2	
	rs2839182	46510580	0.131	WLTD	Memory	4.80E-04	0.013	MCM3AP	
	rs2839193	46522771	0.212	Recogn	Memory	8.91E-04	0.038	MCM3AP	
	rs2075906	46449972	0.129	Zexec	Executive	4.48E-03	0.082	LSS	

Results are shown for the traits presented in Table 3 and the significant traits within the region. Significant results are shown in bold. † p-value after 10.000 permutations. ‡ cognitive tests that were found in the linkage analysis as shown in Table 3. WTI: AVLT immediate recall; Learn: word learning test trial b2; WVLTD: AVLT delayed recall; Recogn: AVLT recognition; zmen: composite score for memory; Stroop-ratio; TMT: trial making test ratio; WF: verbal fluency; zexec: composite score for executive function; VSA: Block Design Test; zglob: composite score for global cognition; DAT: DutA f Ault Reading Test.

Table 4

Finally, at chromosome 20p13 fine-mapping showed significant association with immediate memory to a variant at the terminal end of the chromosome near the defensin beta 125 gene (DEFB125) (p after adjustment for multiple testing=0.02). However, the Rotterdam study did not confirm this, although nominal significance was found with delayed recall and immediate memory and a variant at 680 kb distance (nominal p=0.005 and 0.04).

Regional plot for associations in the linkage regions in the Rotterdam Study



Chromosomal positions are depicted on the x-axis; the minus logarithms of nominal p-values are depicted on the y-axis. The horizontal line represent the significance threshold (p<0.05). The red dots represent the p-values that were significant after Bonferonni correction. Open symbols represent memory related cognitive tests; filled symbols represent executive function related cognitive tests.

Figure 2

Discussion

Our linkage analysis of cognitive function in the ERF Study yielded a total of 6 genome-wide significant regions. Significant linkage of cognitive tests was found to chromosomes 1p13.1, 12q24.33, 19q13.43, 20p13, 21q22.13 and 21q22.3. The strongest evidence for a genetic determinant of cognitive function in our study is found for chromosome 21q22.13. The two regions on chromosome 21 are implicated in the so-called Down's syndrome critical region, which is an intensively studied region within the human genome and is recognized to be crucial for Alzheimer's disease as the amyloid-precursorprotein (APP) gene is in the region, however, APP is located 3 Mb from our linkage peak [42-44]. After fine-mapping of these regions, the region at chromosome 21g22.13 was replicated in an independent population-based sample from an outbred population. Earlier, Luciano et al. found evidence for linkage of verbal IQ to the 21g22 region [19]. The variant within the highest LOD score on chromosome 21g22.13 lies in an intron of the tetratricopeptide-repeat-domain-3 (TTC3) gene. This gene is involved in protein metabolism and may play a role in neuronal development [45], but associations with cognition or related phenotypes have not been described so far. Other candidates in the linkage peak are the KCNJ6 and dual-specificity-tyrosine-phosphorylation-regulated-kinase-1A (DYRK1A) genes. Combining the linkage result with that of association (fine-mapping and replication). the most convincing evidence concerns the KCNJ6 gene for which 1 intronic variant was associated to executive function and global cognition in ERF and to memory in the Rotterdam study. The KCNJ6 gene is expressed in the brain and may play a role in long term potentiation, which is thought to be one of the cellular mechanisms involved in learning and memory [46]. A recent genome-wide association study suggests that this gene is associated with bipolar schizoaffective disorder [47]. The DYRK1A gene was previously associated with the personality trait of conscientiousness in a genomewide association study [48]. Levels of DYRK1A mRNA were elevated in brains of AD patients [49] and the gene has been related to learning and memory deficits in Down syndrome mouse models [50,51].

The other regions which we found in ERF were not replicated in the Rotterdam Study, however, they could be of interest, because each contains promising candidate genes. One explanation why we could not replicate the findings in the population-based study, could be the different characteristics of a family-based study and population-based study. Family-based linkage studies are mainly designed to find rare variants associated with the disease, while association studies are more powerful in finding common variants [25].

Chromosome 21q22.3 contains the S-100-calcium-binding-protein-beta-chain gene (S100B), which was associated to cognitive ability previously [52,53]. The gene is located less than 0.4 Mb to the most significant variant in the fine-mapping analysis. The region on chromosome 1 was linked to Alzheimer's disease in a large linkage meta-analysis [54] and there are multiple genes of interest in this region. The most significant variant in our linkage analysis lies in an intron of the immunoglobulin-superfamily-member-2 gene (IGSF2), a gene that is involved in the inhibition of T-cell proliferation.

Although this is an interesting candidate, there may be other neighboring genes explaining our findings. The nerve-growth-factor-beta-peptide gene (NGF), is involved in the regulation of neuronal growth and differentiation and has been associated with AD and personality traits [55]. Our fine-mapping analysis shows some significance of executive function to variants near IGSF2 and NGF. The chromosome 12 region which showed significant linkage in our analysis is adjacent to the region which was found to be suggestively linked to prospective memory in a previous study [23]. In the fine-mapping of this region, an intronic SNP within the fibrosin-like-1 (FBRSL1) gene was associated to delayed recall. This gene is expressed in the brain, but its function is largely unknown, asking for further research. We could not confirm the region on chromosome 19q13.43 with association. Chromosome 20p13, which we found linked to visual spatial ability and memory contains an obvious candidate gene, the prion-protein gene (PRNP), which is associated with multiple cognitive traits in the general population [56-58]. The highest LOD-score in the current study was found at 4.5 Mb distance from PRNP.

We confirmed some interesting chromosomal regions previously found in related phenotypes and found possible new regions. Interestingly, most regions were related to multiple cognitive domains, which fits the 'generalist gene' hypothesis indicating that there are pleiotrophic pathways involved in cognitive function [5]. There are some issues that need to be discussed. First, 3 of the 6 linkage peaks were located at the terminal ends of the chromosomes, which may have lead to inflation of the estimates, because there are no neighbouring markers. However, there were multiple markers contributing to the peaks leading us to believe that the height of the peak was not due to a single marker estimate. Second, although we used a linkage panel, the markers were relatively dense compared to STR marker sets, which may have caused inflation of peaks due to LD between markers. This may partly explain our exceptionally high peaks. Therefore, our results should be interpreted in light of the simulation based thresholds for significant and suggestive findings as shown in this paper. It is of interest that the variants identified with our association analyses (fine-mapping and replication) were intronic variants of KCNJ6 that were significant after adjustment for multiple testing, which may imply that they are not directly involved in protein transcription.

The strengths of the study are the cohort design of the discovery as well as the replication sample, and the fact that study subjects were not selected on the trait of interest. Moreover, the population is derived from a young genetic isolate, which increases the study power because of the smaller gene pool. Yet, the findings can be extended to the general population, because the genetic makeup does not differ largely from the general population [25].

In summary, our study demonstrates evidence for significant linkage of chromosomes 1p13.1, 12q24.33, 19q13.43, 20p13, 21q22.13 and 21q22.3 to cognitive function. The region 21q22.13 was replicated in an independent sample and is most likely explained by the KCNL6 gene.

70

References

- 1. Green MF. Cognitive impairment and functional outcome in schizophrenia and bipolar disorder. J Clin Psychiatry 2006;67 Suppl 9:3-8; discussion 36-42.
- 2. Seidman LJ, Biederman J, Weber W, Hatch M, Faraone SV. Neuropsychological function in adults with attention-deficit hyperactivity disorder. Biol Psychiatry 1998;44:260-8.
- 3. Huntley JD, Howard RJ. Working memory in early Alzheimer's disease: a neuropsychological review. Int J Geriatr Psychiatry 2009.
- 4. Sleegers K, de Koning I, Aulchenko YS, van Rijn MJ, Houben MP, Croes EA, van Swieten JC, Oostra BA, van Duijn CM. Cerebrovascular risk factors do not contribute to genetic variance of cognitive function: the ERF study. Neurobiology of Aging 2007;28:735-41.
- 5. Butcher LM, Kennedy JK, Plomin R. Generalist genes and cognitive neuroscience. Curr Opin Neurobiol 2006;16:145-51.
- 6. van Duijn CM. Prospects of genetic research of mild cognitive impairment. J Intern Med 2004;256:235-9.
- 7. Pardo LM, van Duijn CM. In search of genes involved in neurodegenerative disorders. Mutat Res 2005;592:89-101.
- 8. Deary IJ, Johnson W, Houlihan LM. Genetic foundations of human intelligence. Hum Genet 2009;126:215-32.
- 9. Payton A. The Impact of Genetic Research on our Understanding of Normal Cognitive Ageing: 1995 to 2009. Neuropsychol Rev 2009.
- 10. Corder EH, Lannfelt L, Bogdanovic N, Fratiglioni L, Mori H. The role of APOE polymorphisms in late-onset dementias. Cell Mol Life Sci 1998;54:928-34.
- van Duijn CM, de Knijff P, Cruts M, Wehnert A, Havekes LM, Hofman A, Van Broeckhoven C. Apolipoprotein E4 allele in a population-based study of early-onset Alzheimer's disease. Nat Genet 1994;7:74-8.
- 12. Wisdom NM, Callahan JL, Hawkins KA. The effects of apolipoprotein E on non-impaired cognitive functioning: A meta-analysis. Neurobiol Aging 2009.
- 13. Seshadri S, DeStefano AL, Au R, Massaro JM, Beiser AS, Kelly-Hayes M, Kase CS, D'Agostino RB, Sr., Decarli C, Atwood LD, Wolf PA. Genetic correlates of brain aging on MRI and cognitive test measures: a genome-wide association and linkage analysis in the Framingham Study. BMC Med Genet 2007;8 Suppl 1:S15.
- 14. Papassotiropoulos A, Stephan DA, Huentelman MJ, Hoerndli FJ, Craig DW, Pearson JV, Huynh KD, Brunner F, Corneveaux J, Osborne D, Wollmer MA, Aerni A, Coluccia D, Hanggi J, Mondadori CR, Buchmann A, Reiman EM, Caselli RJ, Henke K, de Quervain DJ. Common Kibra alleles are associated with human memory performance. Science 2006;314:475-8.
- 15. Papassotiropoulos A, Henke K, Stefanova E, Aerni A, Muller A, Demougin P, Vogler C, Sigmund JC, Gschwind L, Huynh KD, Coluccia D, Mondadori CR, Hanggi J, Buchmann A, Kostic V, Novakovic I, van den Bussche H, Kaduszkiewicz H, Weyerer S, Bickel H, Riedel-Heller S, Pentzek M, Wiese B, Dichgans M, Wagner M, Jessen F, Maier W, de Quervain DJ. A genome-wide survey of human short-term memory. Mol Psychiatry 2009.
- Rommelse NN, Arias-Vasquez A, Altink ME, Buschgens CJ, Fliers E, Asherson P, Faraone SV, Buitelaar JK, Sergeant JA, Oosterlaan J, Franke B. Neuropsychological endophenotype approach to genome-wide linkage analysis identifies susceptibility loci for ADHD on 2q21.1 and 13q12.11. Am J Hum Genet 2008;83:99-105.
- 17. Buyske S, Bates ME, Gharani N, Matise TC, Tischfield JA, Manowitz P. Cognitive traits link to human chromosomal regions. Behav Genet 2006;36:65-76.
- Dick DM, Aliev F, Bierut L, Goate A, Rice J, Hinrichs A, Bertelsen S, Wang JC, Dunn G, Kuperman S, Schuckit M, Nurnberger J, Jr., Porjesz B, Beglieter H, Kramer J, Hesselbrock V. Linkage analyses of IQ in the collaborative study on the genetics of alcoholism (COGA) sample. Behav Genet 2006;36:77-86.
- 19. Luciano M, Wright MJ, Duffy DL, Wainwright MA, Zhu G, Evans DM, Geffen GM, Montgomery GW, Martin NG. Genome-wide scan of IQ finds significant linkage to a quantitative trait locus on 2q. Behav Genet 2006;36:45-55.
- Posthuma D, Luciano M, Geus EJ, Wright MJ, Slagboom PE, Montgomery GW, Boomsma DI, Martin NG. A genomewide scan for intelligence identifies quantitative trait loci on 2q and 6p. Am J Hum Genet 2005;77:318-26.
- Almasy L, Gur RC, Haack K, Cole SA, Calkins ME, Peralta JM, Hare E, Prasad K, Pogue-Geile MF, Nimgaonkar V, Gur RE. A genome screen for quantitative trait loci influencing schizophrenia and neurocognitive phenotypes. Am J Psychiatry 2008;165:1185-92.
- Paunio T, Ekelund J, Varilo T, Parker A, Hovatta I, Turunen JA, Rinard K, Foti A, Terwilliger JD, Juvonen H, Suvisaari J, Arajarvi R, Suokas J, Partonen T, Lonnqvist J, Meyer J, Peltonen L. Genome-wide scan in a nationwide study sample of schizophrenia families in Finland reveals susceptibility loci on chromosomes 2q and 5q. Hum Mol Genet 2001;10:3037-48.
- 23. Singer JJ, Falchi M, Macgregor AJ, Cherkas LF, Spector TD. Genome-wide scan for prospective memory suggests linkage to chromosome 12q22. Behav Genet 2006;36:18-28.
- 24. Schol-Gelok S, Janssens ACJW, Tiemeier HW, Liu F, Lopez-Leon S, Zorkoltseva IV, Axenovich TI, van Swieten JC, Aulchenko YS, Oostra BA, van Duijn CM. A genome-wide screen for depression in two independent Dutch populations. Biological Psychiatry Accepted.
- 25. Pardo LM, MacKay I, Oostra B, van Duijn CM, Aulchenko YS. The effect of genetic drift in a young genetically isolated population. Annals of Human Genetics 2005;69:288-95.
- 26. Aulchenko YS, Heutink P, Mackay I, Bertoli-Avella AM, Pullen J, Vaessen N, Rademaker TA, Sandkuijl LA, Cardon L, Oostra B, van Duijn CM. Linkage disequilibrium in young genetically isolated Dutch population. Eur J Hum Genet 2004;12:527-34.
- Sleegers K, Roks G, Theuns J, Aulchenko YS, Rademakers R, Cruts M, van Gool WA, Van Broeckhoven C, Heutink P, Oostra BA, van Swieten JC, van Duijn CM. Familial clustering and genetic risk for dementia in a genetically isolated Dutch population. Brain 2004;127:1641-9.
- 28. Liu F, Pardo LM, Schuur M, Sanchez-Juan P, Isaacs A, Sleegers K, de Koning I, Zorkoltseva IV, Axenovich TI, Witteman JC, Janssens AC, van Swieten JC, Aulchenko YS, Oostra BA, van Duijn CM. The apolipoprotein E gene and its age-specific effects on cognitive function. Neurobiology of Aging 2008.

- 29. Prins ND, van Dijk EJ, den Heijer T, Vermeer SE, Jolles J, Koudstaal PJ, Hofman A, Breteler MM. Cerebral small-vessel disease and decline in information processing speed, executive function and memory. Brain 2005;128:2034-41.
- 30. Liu F, Kirichenko A, Axenovich TI, van Duijn CM, Aulchenko YS. An approach for cutting large and complex pedigrees for linkage analysis. European Journal of Human Genetics 2008;16:854-60.
- 31. Abecasis GR, Cherny SS, Cookson WO, Cardon LR. Merlin--rapid analysis of dense genetic maps using sparse gene flow trees. Nat Genet 2002;30:97-101.
- 32. O'Connell JR, Weeks DE. PedCheck: a program for identification of genotype incompatibilities in linkage analysis. Am J Hum Genet 1998;63:259-66.
- 33. Mukhopadhyay N, Almasy L, Schroeder M, Mulvihill WP, Weeks DE. Mega2: data-handling for facilitating genetic linkage and association analyses. Bioinformatics 2005;21:2556-7.
- 34. Kong A, Cox NJ. Allele-sharing models: LOD scores and accurate linkage tests. Am J Hum Genet 1997;61:1179-88.
- 35. Kruglyak L, Daly MJ, Reeve-Daly MP, Lander ES. Parametric and nonparametric linkage analysis: a unified multipoint approach. Am J Hum Genet 1996;58:1347-63.
- *36. Whittemore AS, Halpern J. A class of tests for linkage using affected pedigree members. Biometrics* 1994;50:118-27.
- Amin N, Aulchenko YS, Dekker MC, Ferdinand RF, van Spreeken A, Temmink AH, Verhulst FC, Oostra BA, van Duijn CM. Suggestive linkage of ADHD to chromosome 18q22 in a young genetically isolated Dutch population. Eur J Hum Genet 2009;17:958-66.
- 38. Devlin B, Roeder K, Wasserman L. Genomic control for association studies: a semiparametric test to detect excess-haplotype sharing. Biostatistics 2000;1:369-87.
- 39. Aulchenko YS, Ripke S, Isaacs A, van Duijn CM. GenABEL: an R library for genome-wide association analysis. Bioinformatics 2007;23:1294-6.
- 40. Amin N, van Duijn CM, Aulchenko YS. A genomic background based method for association analysis in related individuals. PLoS ONE 2007;2:e1274.
- Hofman A, Breteler MM, van Duijn CM, Janssen HL, Krestin GP, Kuipers EJ, Stricker BH, Tiemeier H, Uitterlinden AG, Vingerling JR, Witteman JC. The Rotterdam Study: 2010 objectives and design update. Eur J Epidemiol 2009;24:553-72.
- 42. Van Broeckhoven C, Kumar-Singh S. Genetics and pathology of alpha-secretase site AbetaPP mutations in the understanding of Alzheimer's disease. J Alzheimers Dis 2006;9:389-98.
- 43. Harris SE, Fox H, Wright AF, Hayward C, Starr JM, Whalley LJ, Deary IJ. A genetic association analysis of cognitive ability and cognitive ageing using 325 markers for 109 genes associated with oxidative stress or cognition. BMC Genet 2007;8:43.
- 44. Chumakov I, Rigault P, Guillou S, Ougen P, Billaut A, Guasconi G, Gervy P, LeGall I, Soularue P, Grinas L, et al. Continuum of overlapping clones spanning the entire human chromosome 21q. Nature 1992;359:380-7.
- 45. Berto G, Camera P, Fusco C, Imarisio S, Ambrogio C, Chiarle R, Silengo L, Di Cunto F. The Down syndrome critical region protein TTC3 inhibits neuronal differentiation via RhoA and Citron kinase. J Cell Sci 2007;120:1859-67.

- 46. Chung HJ, Ge WP, Qian X, Wiser O, Jan YN, Jan LY. G protein-activated inwardly rectifying potassium channels mediate depotentiation of long-term potentiation. Proc Natl Acad Sci U S A 2009;106:635-40.
- 47. Hamshere ML, Green EK, Jones IR, Jones L, Moskvina V, Kirov G, Grozeva D, Nikolov I, Vukcevic D, Caesar S, Gordon-Smith K, Fraser C, Russell E, Breen G, St Clair D, Collier DA, Young AH, Ferrier IN, Farmer A, McGuffin P, Holmans PA, Owen MJ, O'Donovan MC, Craddock N. Genetic utility of broadly defined bipolar schizoaffective disorder as a diagnostic concept. Br J Psychiatry 2009;195:23-9.
- 48. Terracciano A, Sanna S, Uda M, Deiana B, Usala G, Busonero F, Maschio A, Scally M, Patriciu N, Chen WM, Distel MA, Slagboom EP, Boomsma DI, Villafuerte S, Sliwerska E, Burmeister M, Amin N, Janssens AC, van Duijn CM, Schlessinger D, Abecasis GR, Costa PT, Jr. Genome-wide association scan for five major dimensions of personality. Mol Psychiatry 2008.
- Kimura R, Kamino K, Yamamoto M, Nuripa A, Kida T, Kazui H, Hashimoto R, Tanaka T, Kudo T, Yamagata H, Tabara Y, Miki T, Akatsu H, Kosaka K, Funakoshi E, Nishitomi K, Sakaguchi G, Kato A, Hattori H, Uema T, Takeda M. The DYRK1A gene, encoded in chromosome 21 Down syndrome critical region, bridges between beta-amyloid production and tau phosphorylation in Alzheimer disease. Hum Mol Genet 2007;16:15-23.
- 50. Altafaj X, Dierssen M, Baamonde C, Marti E, Visa J, Guimera J, Oset M, Gonzalez JR, Florez J, Fillat C, Estivill X. Neurodevelopmental delay, motor abnormalities and cognitive deficits in transgenic mice overexpressing Dyrk1A (minibrain), a murine model of Down's syndrome. Hum Mol Genet 2001;10:1915-23.
- 51. Park J, Oh Y, Chung KC. Two key genes closely implicated with the neuropathological characteristics in Down syndrome: DYRK1A and RCAN1. BMB Rep 2009;42:6-15.
- 52. Lambert JC, Ferreira S, Gussekloo J, Christiansen L, Brysbaert G, Slagboom E, Cottel D, Petit T, Hauw JJ, DeKosky ST, Richard F, Berr C, Lendon C, Kamboh MI, Mann D, Christensen K, Westendorp R, Amouyel P. Evidence for the association of the S100beta gene with low cognitive performance and dementia in the elderly. Mol Psychiatry 2007;12:870-80.
- 53. Rothermundt M, Peters M, Prehn JH, Arolt V. S100B in brain damage and neurodegeneration. Microsc Res Tech 2003;60:614-32.
- 54. Butler AW, Ng MY, Hamshere ML, Forabosco P, Wroe R, Al-Chalabi A, Lewis CM, Powell JF. Meta-analysis of linkage studies for Alzheimer's disease-A web resource. Neurobiol Aging 2009.
- 55. Hashimoto R, Hirata Y, Asada T, Yamashita F, Nemoto K, Mori T, Moriguchi Y, Kunugi H, Arima K, Ohnishi T. Effect of the brain-derived neurotrophic factor and the apolipoprotein E polymorphisms on disease progression in preclinical Alzheimer's disease. Genes Brain Behav 2009;8:43-52.
- 56. Berr C, Richard F, Dufouil C, Amant C, Alperovitch A, Amouyel P. Polymorphism of the prion protein is associated with cognitive impairment in the elderly: the EVA study. Neurology 1998;51:734-7.

04 Genome-wide linkage screen of cognitive function identifies susceptible chromosomal regions

- 57. Houlihan LM, Harris SE, Luciano M, Gow AJ, Starr JM, Visscher PM, Deary IJ. Replication study of candidate genes for cognitive abilities: the Lothian Birth Cohort 1936. Genes Brain Behav 2009;8:238-47.
- 58. Rujescu D, Hartmann AM, Gonnermann C, Moller HJ, Giegling I. M129V variation in the prion protein may influence cognitive performance. Mol Psychiatry 2003;8:937-41.

05 Genome-wide association study of cognitive executive functions: Meta-analysis of the CHARGE consortium

Abstract

Introduction

Executive function is an important part of cognitive function. To explore common variants that contribute to the normal variation in cognition, we conducted genome-wide association analyses in the Cohorts for Heart and Aging Research in Genomic Epidemiology (CHARGE) consortium, which includes large prospective population-based cohorts.

Materials and Methods

Neuropsychological testing was available for 13 cohorts. For executive function and processing speed tasks, we included the Trail Making Test (TMT) parts A and B and the Stroop Color and Word Test in the analyses. All the individual studies used their genotyped data to impute to 2.5 million single nucleotide polymorphisms (SNPs). The analyses were performed in Caucasians older than 45 years who were free of dementia and clinical stroke at times of cognitive testing. Each study applied an additive genetic model using linear regression models adjusting for age and sex. Meta-analyses were performed for TMT-A, TMT-B, TMT-B minus TMT-A using fixed-effects models and on Stroop using sample-sized weighted models.

Results

The meta-analysis included 5,477 participants for the TMT-A, 6,212 for TMT-B and TMT-B minus TMT-A and 7,777 for the Stroop. The most significant association was found with TMT-B and a SNP on chromosome 18. This SNP was just above the genome-wide significant threshold with a p-value of 6.95*10-8. When comparing our findings of the GWAS on cognitive function to the GWAS of AD and schizophrenia published earlier, we identified two other genes of interest: STXBP6 and PCDH9. However, these SNPs did not reach genome wide significance in the present analysis nor in those of the disease outcomes.

Conclusions

In conclusion, we found compelling evidence for a region on chromosome 18 that is involved in TMT-B and preliminary evidence for STXBP6 and PCDH9. However, our findings await replication which is at present ongoing.

Introduction

Cognitive function is a broad concept referring to multiple cognitive domains, among which are memory, language, executive function and visuospatial ability. Normal cognitive ability is an important determinant of quality of life. Impairment of these functions is seen in patients with various diseases including dementia, bipolar disorder, schizophrenia and attention deficit hyperactivity disorder (ADHD) [1-3]. Executive function is one of the major processes of the frontal lobe, and includes a range of tasks, among which are response inhibition, attention, cognitive flexibility and planning [4]. Genetic factors account for over 20% of the variability in executive function traits [5-9]. Finding susceptibility genes for cognitive functioning may provide insight into the normal variation in executive function, but

may also increase the knowledge of diseases that are associated with cognitive impairment.

Although various genes have been identified, consistency is yet to be found, which is due to lack of replication and meta-analyses [10]. Of candidate genes, the apolipoprotein E (APOE) gene seems the most promising genetic risk factor for executive function and memory [11]. Linkage regions that were previously found for executive function tasks were located on chromosomes 2q, 5q, 11q, 13q and 14q [12-14]. To our knowledge there are currently 4 genome-wide association studies on cognitive traits in adults [15-18]. In these studies associations were found of memory with the calsynthenin-2 (CLSTN2) gene on chromosome 3, with the WW-and-C2-domain-containing-1 (KIBRA) gene on chromosome 5 and with the sodium-channel-voltage-gated-type I-alpha subunit (SCN1A) gene on chromosome 2. Abstract reasoning was associated with the sortilin-related-receptor (SORL1) gene located on chromosome 11. However, no genes were identified for executive function by GWAs. Furthermore, the studies of memory and abstract reasoning were small and liberal significance thresholds were used.

To explore common variants that contribute to the normal variation in executive function, we performed a large-scale meta-analysis combining GWAS from the Cohorts for Heart and Aging Research in Genomic Epidemiology (CHARGE) consortium, including 12 prospective population-based cohorts. We present results from four meta-analyses.

Materials and Methods

Consortium

The Cohorts for Heart and Aging Research in Genomic Epidemiology (CHARGE) consortium includes large prospective population-based cohorts [19]. Each cohort has extensive phenotypic information on various traits and genome-wide data available. Guidelines for collaboration were approved and phenotype-specific workgroups agreed on phenotype harmonization, the selection of covariates and analytic plans for each study and for the overall meta-analyses. Each study obtained approval from Institutional Review Boards. All participants gave written consent for study participation and use of DNA for genetic research.

Setting

Details of cohort selection, risk factor assessment and cognitive testing in the 5 cohorts included in the current study have been described previously and included in the supplementary data. Briefly, the Aging Gene-Environment Susceptibility – Reykjavik Study (AGES) included 30,795 persons born between 1907 and 1937 who lived in Reykjavik at the 1967 baseline examination. Re-examination of surviving members of the cohort was initiated in 2002 as part of the AGES-Reykjavik Study (N=2300) [20]. The Atherosclerosis Risk in Communities Study (ARIC) included 15792 persons of 45 to 64 years of age from 4 US communities between 1987 to1989. The study included blacks as well as whites (N=11478) [21]. The Austrian Stroke Prevention Study (ASPS) included 2007 persons living

in Graz, Austria. An extended diagnostic work-up was performed in a subset of age 45 to 85 years between 1991 and 1994 and between 1999 and 2003 [22,23]. The Cardiovascular Health Study (CHS) enrolled persons aged 65 years and older from 4 US communities. Baseline examination was either in 1989 to 1999 or 1992 to 1993 (N=5888, including 4925 whites) [24]. The Erasmus Rucphen Family study (ERF) enrolled 3000 persons from a genetically isolated population located in the South West of the Netherlands. Baseline examination was performed between 2002 and 2005 [25]. The Framingham Heart Study (FHS) included 3 generation of participants from the US. Cognitive testing data was collected between 1999 and 2004 for survivors from the Original cohort (N=5209) that has been followed since 1948 and the offspring cohort since 1971 (N=5124) [26,27]. The Rotterdam Study is a population-based cohort study among inhabitants of a district of Rotterdam (Ommoord), The Netherlands. In 1990-1993, 7,983 persons participated and were re-examined every 3 to 4 years [28].

Cognitive testing

The study program of each study included neuropsychological testing. We focused on the Trail Making Test parts A and B (TMT). The TMT is a time-demanding task in which participants are asked to connect a randomly placed series of letters and numbers as quickly as possible. In TMT-A only the numbers have to be connected, in TMT-B numbers and letters have to be connected alternately (from 1 to A, to 2 to B etc). The score was defined as the time in seconds to complete the task. Participants who passed the maximum test time for TMT-B were given the maximum time score of 300 seconds. There were no participants who timed out on TMT-A. For analysis, we used time in seconds on TMT-A and TMT-B. Additionally, we used the time difference between TMT-B and -A (TMT-BminusA). The times were transformed by taking the natural logarithm. The Stroop Color and Word Test is also a time demanding task consisting of three cards. In card I, participants have to name the right words as quickly as possible, in card II they have to name the right colors as quickly as possible. On this card the meaning of the word is different than the color the word is printed in (e.g. blue is written whereas the word is printed in red). ARIC recorded number of words; AGES, ASPS, ERF and RS recorded time in seconds. For analysis, we used the time difference between card III and II (Stroop interference).

Genotyping

The consortium was formed after the individual studies had finalized their GWAS platforms, and the studies included used different platforms: the Illumina HumanCNV370-Duo BeadChip® for AGES and CHS; the Affymetrix GeneChip SNP Array 6.0® for ARIC; the Illumina Human610-Quad BeadChip® for ASPS; the Affymetrix GeneChip Human Mapping 500K Array Set® and 50K Human Gene Focused Panel® for FHS; a combination of the Illumina HumanHap 320K array, Illumina HumanHap 6k Beadchip array, the Illumina Human 370K-Duo SNP array and the Affymetrix GeneChip® Human

80

Mapping 250K Nsp Array for ERF; and the Illumina HumanHap550-Duo BeadChip® for the Rotterdam Study.

All studies included used their genotyped data to impute to 2.5 million non-monomorphic, autosomal single nucleotide polymorphisms (SNPs) using release 22 HapMap CEU population as a reference. As detailed previously [29], cohort-specific quality controls included filters for call rate, heterozygosity, and number of Mendelian errors per individual. SNP-specific guality controls included filters for call rate, minor allele frequency, Hardy-Weinberg equilibrium, and differential missingness by outcome or genotype (mishap test in PLINK, http://pngu.mgh.harvard.edu/purcell/plink/). The set of genotyped input SNPs used for imputation in each study was selected based on their highest quality GWA data. We used a call rate >95% in ARIC, >97% in CHS and FHS, and >98% in AGES-Reykjavik, ASPS, ERF and Rotterdam; a minor allele frequency >0.01 in each study; a Hardy-Weinberg p>1x10-5 in ARIC and CHS and p>1x10-6 in AGES-Reykjavik, ASPS, ERF, FHS and Rotterdam; a test of differential missingness by the "mishap" test in PLINK p>1x10-9 in each study except ERF and ARIC. We used either the Markov Chain Haplotyping (MaCH) package (http://www.sph.umich.edu/csg/ abecasis/MACH, version 1.0.15 or 1.0.16 software) for AGES-Reykjavik, ARIC, ASPS, ERF, FHS and Rotterdam for imputation to the plus strand of NCBI build 36, HapMap release #22, or BIM-BAM15 programs for CHS for imputation to the plus strand of NCBI build 35. For each imputed SNP informativity of imputation was estimated as the ratio of the empirically observed dosage variance to the expected binomial dosage variance.

Study population

Participants were eligible for the discovery cohort if they did not have prevalent dementia, clinical stroke and were Caucasian. Participants were excluded when consent was declined or genotyping failed. Only participants aged 45 and older were included.

Statistical analysis within studies

An additive genetic model was used in each study, fitting a 1-degree-of-freedom test relating genotype dosage (0, 1 or 2 copies of the minor allele) to cognitive tests. Linear regression models were used adjusting for sex and age and secondary adjusting for age, sex and education. ARIC and CHS additionally adjusted for study site; FHS and ERF the additionally adjusted for family structure. We did not adjust for additional covariates to avoid adjusting for variables that might be involved in the causal pathway. Education was divided into 4 categories corresponding to the years and level of education. Category 0 corresponded to less than high school (0-11 years); category 1 corresponded to a high school degree (12 years); category 2 corresponded to more than 12 years of education, but not a college degree and category 3 corresponded to college degree and above.

Studies were screened for latent population substructure, including cryptic relatedness, using suitable programs: EIGENSTRAT in ARIC, FHS and AGES [30,31], an IBD matrix in ASPS and Rotterdam [32], and using principal component analysis in CHS. When appropriate, components related to the cognitive phenotype under study were included as covariates in the linear regression. In ERF, analyses were performed in GenABEL (R-library) [33], using the mmscore method, with a kinship matrix that was estimated from the genotype data to adjust for relatedness of the population [34].

We studied quantile-quantile (Q-Q) plots to ensure that the p-value distributions in each of the cohorts conformed to a null distribution at all but the extreme tail. We also calculated the genomic inflation factor lambda, which measures over-dispersion of test-statistics from association tests indicating population stratification and can be used to apply genomic control [33]. The lambda for TMT was less than 1.02 for all of the studies.



Quantile-quantile plots of TMT-A, TMT-B, TMT-BminusA and Stroop results

Q-Q plot of the inverse variance meta-analyses, it shows the distribution of the observed test statistic (negative log of p-values, on the y-axis) plotted against the distribution of test statistic expected under the null-hypothesis (on the x-axis).

Figure 1

Meta-analysis

After quality control and filtering within each study, AGES had either genotyped or imputed data for 2,532,729, ARIC for 2,543,887 SNPS, ASPS for 2,543,887 SNPs, CHS for 2,531,169 SNPs, FHS for 2,540,223 SNPs, the Rotterdam study for 2,543,887 SNPs and ERF for 2,543,887 SNPs. We restricted our meta-analysis to autosomal SNPs that were common to all studies and had a minor allele frequency >0.01 and an imputation quality >0.3.

Cohort		TMT-A	TMT-B	TMT-BminusA	Stroop interference
		seconds	seconds	seconds	seconds
AGES	N	-	-	-	2644
	Age	-	-	-	75.9 (5.33)
	Gender (% female)	-	-	-	58.9
	Test score	-	-	-	41.7 (21.5)
ARIC	N	438	436	436	430
	Age	72.6 (4.2)	72.6 (4.2)	72.6 (4.2)	72.5 (4.2)
	Gender (% female)	59.8	59.6	59.6	60.0
	Test score †	41.6 (17.6)	114.6 (49.6)	73.6 (43.1)	-29.9 (9.5) †
ASPS	N	-	830	-	261
	Age	-	65.2 (8.0)	-	65.1 (7.5)
	Gender (% female)	-	56.9	-	57.1
	Test score	-	128.0 (60.2)	-	47.2 (22.7)
CHS	N	1249	1249	1249	
	Age	79.4 (3.8)	79.4 (3.8)	79.4 (3.8)	-
	Gender (% female)	61.0%	61.0%	61.0%	-
	Test score	52.3 (22.0)	141.5 (62.5)	89.2 (53.3)	-
ERF	N	1267	1255	1255	1238
	Age	58.7 (8.9)	58.7 (8.8)	58.7 (8.8)	58.6 (8.8)
	Gender (% female)	54.8	54.7	54.7	55.5
	Test score	45.9 (20.8)	139.5 (78.0)	94.0 (67.2)	54.2 (33.0)
FHS	N	2475	2440	2440	-
	Age	64.7 (11.9)	64.6 (10.9)	64.6 (10.9)	-
	Gender (% female)	54.8	54.7	54.7	-
	Test score	35.7 (20.2)	92.0 (55.6)	60.0 (46.2)	-
RS	N	-	-	-	3204
	Age	-	-	-	64.4 (6.7)
	Gender (% female)	-	-	-	57.9
	Test score	-	-	-	32.1 (18.9)

chapter 05

Table 1

We used an inverse-variance meta-analysis as our primary method after applying genomic control within each individual study. Beta estimates were weighted by their inverse variance and a combined estimate was obtained by summing the weighted betas and dividing by the summed weights. Hence results for SNPs imputed with low certainty were down-weighted because the low informativity of imputation ensures a large variance. In contrast, studies with large sample sizes and with directly genotyped or well-imputed SNPs had a greater effect on the meta-analyses p-value because of small variances. In a secondary analysis we used an effective sample size weighted meta-analysis technique after applying genomic control within each individual study. For each SNP the z-statistic was weighted by the effective sample size (product of the sample size and the ratio of the empirically observed dosage variance to the expected binomial dosage variance for imputed SNPs). A combined estimate was obtained by summing the weighted z-statistics and dividing by the summed weights. Hence results for SNPs imputed with low certainty were down-weighted. In contrast, studies with

large sample sizes and with directly genotyped or well-imputed SNPs had a greater effect on the meta-analyses p-value. We undertook the meta-analysis using the METAL software by Abecasis and Willer in 2007.



Genome-wide plots of TMT-A, TMT-B, TMT-BminusA and Stroop results

Minus log p-values (x-axis) are plotted against their genomic position (y-axis). The plot is based the fixed-effects meta-analysis.

Figure 2

We estimated the genomic inflation factor lambda after meta-analysis. The estimate of lambda was 1.010 for TMT-A, 1.022 for TMT-B and 1.024 for TMT-BminusA indicating no significant inflation of p-values. The quantile-quantile (Q-Q) plots of our inverse variance meta-analyses results for the 3 tests show the distribution of the observed test statistic (negative log of p-values, on the y-axis) plotted against the distribution of test statistic expected under the null-hypothesis (on the x-axis) (Supplementary Figures 1).

A p-value \leq 5.0*10-8 was considered genome-wide significant [36] and a p-values between 5.0*10-8 and 1.0*10-5 were considered highly suggestive associations. P-values <1.0*10-3 were considered suggestive associations.

We additionally studied overlap between our suggestive associations and previously reported genome-wide data from studies in cognitive function and in diseases for which cognition may be an endophenotype. We used three published meta-analyses with publicly available data on various cognitive tests [18], Alzheimer's disease (AD) [37] and schizophrenia (SCZ) [38].

Results

Baseline characteristics of the study populations are depicted in Table 1. The mean age of the study population was 68.3 years and 57.6% were women. The mean test score on TMT-A was 43.9 seconds among the 5,477 participants that were included in the meta-analysis. The mean test score of the 6,212 participants included in the TMT-B analysis was 123.1 seconds and the mean time difference on TMT-BminusA was 79.2 seconds among 5,377 participants. For Stroop interference, the mean test score for the cohorts using time in seconds was 43.8 seconds, the mean test score for ARIC was -29.9 words.

Figure 2 shows the genome-wide plots for each trait, depicting the p-values by their genomic position. Highly suggestive loci with a p-value smaller than 10-5 are presented in Supplementary Tables 1 to 4. The most significant finding was seen for TMT-B for a SNP on chromosome 18. This SNP was just above the genome-wide significant threshold with a p-value of 6.95*10-8. A detailed plot for all SNPs within a region of 250kb is shown in Figure 3, which shows that there are multiple SNPs in the region in linkage disequilibrium with this SNP. All cohorts show similar effect sizes and direction of effect for this SNP (Figure 4). The p-value for this SNP is 0.09 for TMT-A and of 3.35*10-5 for TMT-BminusA. The SNP is intergenic, and located in between two interesting genes.

The lowest p-value for TMT-A was 3.35*10-7 for a SNP on chromosome 13. The second ranked SNP with a p-value of 7.69*10-7 was located in a region that was also associated with TMT-B (p-value 7.30*10-6). For TMT-BminusA, the lowest p-value was 1.11*10-6 for a SNP on chromosome 10. Finally, on chromosomes 18 the same SNP was associated with TMT-B and TMT-BminusA (p-value 1.39*10-6 and 8.19*10-6, respectively). Stroop interference was associated to two regions on chromosome 3. The first ranked SNP was associated with a p-value of 2.38*10-7, which is rare variant (MAF=0.03).

Regional plot for associations in the region around the top hit (+/- 250kb)



Chromosome 18 position (hg18) (kb)

All SNP are plotted with their meta-analysis probability values against their genomic position. The color of the triangles represents the linkage disequilibrium between SNP. Light blue line represents estimated recombination rates.

Figure 3



Forest plot of the results per cohort for the top hit – TMT-B

The values on the x-axis represent the betas.

Figure 4

Next, we performed a data mining study comparing our findings to the GWAS findings of the largest studies on AD and schizophrenia [35,36]. At the suggestive threshold of 10-3, 2731 SNPs were associated with TMT-A, 2790 SNPs with TMT-B, 2761 with TMT-BminusA and 2980 SNPs with Stroop. When comparing our suggestive associations (p<10-3) with the disease based genome-wide studies (Table 2), we did not see an overlap for the top SNPs identified for TMT-A, TMT-B, TMT-BminusA or Stroop. However, one putative AD gene, the sortilin-related-receptor 1 gene (SORL1) did show some overlap with Stroop. There are two additional interesting findings. First, we found one region with overlapping SNPs for both TMT-B and TMT-BminusA. This region on chromosome 14 overlapped with the findings in Alzheimer's disease [35] and contains the neuro-oncological ventral antigen 1 (NOVA1) and the syntaxin binding protein 6 (STXBP6) genes. Second, the region identified in the top ranking of the Stroop data, which concerned SNP within the protocadherin 9 gene (PCDH9) (p=8.49*10-6; supplementary table 4), was identified also in the AD GWAS, although a different SNP was reported.

Overlap of the	GWAS findi	ings for TMT	-А, ТМТ-В,	TMT-Bminu	sA and Stroop in	terference with C	WAS of related	outcomes		
SNP	Chrom	Effect	SE	MAF	Our p-value	Direction	Gene	Flanking genes	AD	SCZ
TMT-A					P					
SNPa	8	-0.0298	0.0088	0.1980	7.68E-04		C8orf79			1.76E-04
SNPb	15	0.0268	0.008	0.2521	8.68E-04	++++		BNC1, SH3GL3	1.79E-04	
SNPc	19	-0.0293	0.0085	0.2658	5.97E-04	+	ZNF610		5.28E-04	
ТМТ-В										
SNPd	1	-0.0448	0.0136	0.36	9.55E-04	+-	DAB1			8.20E-04
SNPe	4	-0.0358	0.0106	0.16	7.15E-04	-+	intergenic	BOD1L, AC095052.1		8.67E-04
SNPf *	6	0.0427	0.0113	0.17	1.56E-04	+++++	intergenic	HLA-region	9.32E-04	
SNPg *	14	0.029	0.0083	0.46	4.83E-04	+++++	intergenic	STXBP6, NOVA1	9.43E-04	
TMT- BminusA										
SNPh	3	-0.0519	0.0139	0.38	1.90E-04		CACNA2D3			2.35E-04
SNPi	10	-0.0497	0.0148	0.23	8.12E-04	-+	LIPA			3.71E-04
SNPj	14	-0.0462	0.0133	0.47	5.39E-04		intergenic	STXBP6, NOVA1	9.51E-05 8.22E-04	
SNPk *	18	-0.0523	0.0133	0.29	8.70E-05		intergenic	C18orf22, ADNP2	(apoe neg) 5.86E-04 (apoe neg)	
SNPI *	18	0.0412	0.0124	0.42	9.18E-04	++++	DCC			6.44E-05
Stroop †		Weight	Z-score							
SNPm	6	7435.37	3.464	0.3311	5.33E-04	+++++	JARID2			0.03
SNPn	8	7521.56	-3.305	0.2080	9.50E-04		C8orf79			1.76E-04
SNPo	11	7573.78	3.401	0.0252	6.71E-04	+++++	SORL1		9.70E-03	
SNPp	13	7732.09	-3.916	0.3469	9.02E-05	-+	PCDH9		5.74E-04	
SNPq	15	7720.07	-3.352	0.1405	8.01E-04		intergenic	CA12, USP3	(apoe neg) 6.80E-04 (apoe neg)	
SNPr	15	6967.1	3.512	0.2833	4.44E-04	++-++	intergenic	KLHL25, AGBL1	(apoe neg)	3.43E-04

P-values are on the fixed effects meta-analysis adjusted for age and sex. † P-values are based on the sample-size weighted meta-analysis of adjusted for age and sex Highlighted SNPs overlapped between our traits. * More than one SNP showed overlap.

Table 2

chapter 05

Discussion

The current study is a large meta-analysis of genome-wide associations studies performed in 7 cohorts including over 6000 subjects. Although not fulfilling the criteria for genome-wide significance, by far the most interesting region is the chromosome 18 region that shows association to TMT-B. We found one SNP approaching genome wide significance in the analysis of TMT-B. The SNP is located between two interesting genes. Unfortunately, none of the other tests showed p-values that were

close to significance. At this stage, our study is awaiting replication for the top ranked SNPs. When considering the highest ranked SNPs based on p-values, however, we find little overlap between the SNPs associated with the various cognitive tests. The lack of overlap for genes involved in TMT-A, in TMT-B and Stroop may be explained by the fact that TMT-A is an outcome related to processing speed while TMT-B and Stroop are outcomes related to executive function. There may be several explanations for the lack of overlap between the findings for TMT-B and Stroop. We may have lacked statistical power to identify true positive SNPs and the top SNPs may primarily represent noise. Alternatively, the overlap in brain function as measured by TMT-B and Stroop may be limited. While all the measures were timed and likely share a common processing speed factor, the executive factors underlying TMT-B and Stroop likely reflect diverse higher order factors. If this is the case, one may expect different genes to be involved in different tests.

Despite the low power, we still see an overlap of our GWAS with previous GWAS on AD and schizophrenia, for which cognitive functioning is an endophenotype. It is of interest that not the apolipoprotein E gene (APOE) but rather SORL1 pops up in both cognitive function and AD analyses. SORL1 as been identified as a gene that may be involved in AD [39,40]. When comparing our findings to those of the 4 earlier genome-wide association studies on cognitive traits in adults [15-18], we only see an overlap with the relation of SORL1 to abstract reasoning. The gene of most interest in the comparative analysis of our findings to those in AD and Schizophrenia is STXBP6 [37]. This gene forms non-fusogenic complexes with the synaptosomal-associated-protein-25kDa (SNAP25) and syntaxin-1A (STX1A) genes and may thereby modulate the formation of functional SNARE complexes and may be involved in exocytosis [41]. SNAP25 has been associated both to schizophrenia and Attention Deficit Hyperactivity Disorder (ADHD) [42,43]. Further, a region was identified in the top of the GWAs of Stroop data (PCDH9), which overlapped with the AD GWAS. PCDH9 belongs to the protocadherin gene family, a subfamily of the cadherin superfamily [44]. The gene encodes a cadherin-related neuronal receptor that localizes to synaptic junctions and is putatively involved in specific neuronal connections and signal transduction. Protocadherin 9 cadherin-related neuronal receptor, predominantly expressed in brain, localizes to synaptic junctions and is involved in specific neuronal connections and signal transduction. The gene is also expressed in other tissues in which it exerts a developmentally regulated expression pattern.

In conclusion, we found compelling evidence for a region on chromosome 18 that is involved in TMT-B. When comparing our findings of the GWAS on cognitive function to the GWAS of AD and schizophrenia we identified to other genes of interest: STXBP6 and PCDH9. For these two genes the evidence for association is very preliminary as these SNPs did not reach genome wide significance in the present analysis nor in those of the disease outcomes. Our findings await replication studies that are presently ongoing.

88

TMT-A: Listing of all SNPs with a p-value < 10⁻⁵

SNP	Chrom	Effect	SE	MAF	P-value	Direction Gene		Flanking genes	SNPs <10 ⁻⁵	
Adusted for age+sex										
SNP1 SNP2	13 4	0.0483 -0.0379	0.0094 0.0077	0.1676 0.71	3.01E-07 7.69E-07	++++	MYO16 ADH5		2 1	
SNP3	12	0.0463	0.0097	0.8324	1.70E-06	+-++	OR6C70		5	
SNP4	7	-0.0341	0.0073	0.3707	3.03E-06		AC006397.1			
SNP5	14	-0.0537	0.0117	0.818	4.10E-06		intergenic	SNX6, RPL23AP8, CFL2	1	
SNP6	8	0.0338	0.0073	0.4737	4.12E-06	++++	intergenic	LOC392180, RPL23AP54	3	
SNP7	1	0.033	0.0073	0.5565	6.12E-06	++++	KCND3		0	
SNP8	3	0.039	0.0086	0.2207	6.13E-06	++++	TGM4		0	
SNP9	1	-0.0346	0.0078	0.5545	8.60E-06		intergenic	KIAA1026, PRDM2		

P-values are based on the fixed effects meta-analysis of CHS, FHS, ERF, and ARIC. Number of SNPs refers to additional SNPs within 250kb of the top SNP.

Supplementary table 1

SNP	Chrom	Effect	SE	MAF	P-value	Direction	Gene	Flanking genes	Number
									of SNPs <10 ^{.5}
Adusted for age+sex									
SNP1	18	0.0725	0.0134	0.1137	6.95E-08	+++++	intergenic	PIK3C3, RIT2	6
SNP2	21	0.0378	0.0078	0.4460	1.23E-06	+++++	intergenic	CLIC6, RUNX1	0
SNP3	18	-0.0573	0.0119	0.1127	1.39E-06		ADNP2		8
SNP4	4	0.0459	0.0097	0.2094	2.25E-06	+++++	SEPT11		5
SNP5	7	0.0385	0.0081	0.3615	1.99E-06	+++++	intergenic	BET1, Col1A2	1
SNP6	4	0.0869	0.0194	0.0459	7.30E-06	+++++	ADH4		0
SNP7	6	0.0572	0.0122	0.2159	2.76E-06	+++++	intergenic	LOC728275, LOC728316	0
SNP8	16	0.0615	0.0134	0.0899	4.41E-06	+++++	intergenic	A2BP1, LOC283953	2
SNP9	16	0.0589	0.013	0.1163	5.52E-06	+++++	intergenic	CDH8	1
SNP10	15	0.035	0.0078	0.3980	7.13E-06	+++++	intergenic	SQRDL, SLC24A5	8
SNP11	19	-0.0363	0.0082	0.3403	9.18E-06		intergenic	AC022145.1, ZNF730	1

P-values are based on the fixed effects meta-analysis of ASPS, ARIC, ERF, CHS, and FHS. Number of SNPs refers to additional SNPs within 250kb of the top SNP.

Supplementary table 2

TMT-BminusA: Listing	TMT-BminusA: Listing of all SNPs with a p-value < 10 ⁻													
SNP	Chrom	Effect	SE	MAF	P-value	Direction	Gene	Flanking genes	Number of SNPs <10 ⁻⁵					
Adusted for age+sex														
SNP1	10	0.1587	0.0326	0.0532	1.11E-06	++++	intergenic	GPR158, MYO3A	2					
SNP2	6	-0.0761	0.0157	0.1946	1.33E-06		PRIM2	BAG2	24					
SNP3	8	-0.1512	0.0326	0.0437	3.52E-06		NRG1		0					
SNP4	12	-0.1224	0.027	0.0758	5.82E-06		intergenic	E2F7, NAV3	0					
SNP5	18	-0.0872	0.0196	0.1115	8.19E-06		ADNP2		0					
SNP6	4	0.0709	0.0156	0.1891	5.19E-06	++++	HERC3		0					
SNP7	5	0.0596	0.0131	0.3743	5.67E-06	++++	MAST4		0					
SNP8 SNP9	5 1	-0.0974	0.0215	0.1039	6.12E-06 8.71E-06		intergenic	LOC100289569, EFNA5	1					
00		0.0074	0.0107	0.1000	0E-00		1107112							

P-values are based on the fixed effects meta-analysis of ARIC, ERF, CHS and FHS adjusted for age and sex. Number of SNPs refers to additional SNPs within 250kb of the top SNP.

.

Supplementary table 3

Stroop interfere	Stroop interference: Listing of all SNPs with a p-value < 10^{-5}													
SNP	Chrom	Weight	Z-score	MAF	P-value	Direction	Gene	Nearby*	Number of SNPs <10 ⁻⁵					
SNP1	3	7184.95	5.106	0.0307	2.38E-07	++-+++	ATP13A5		0					
SNP2	15	7417.36	-4.698	0.4470	2.63E-06		TMOD2		2					
SNP3	6	5869.31	4.661	0.094	3.15E-06	+++++	intergenic	POPDC3, PREP	1					
SNP4	5	7440.17	4.614	0.1168	3.96E-06	+-+++	GPX3		0					
SNP5	6	7333.42	-4.613	0.2180	3.96E-06		intergenic	LOC100129554, RPS3AP24	9					
SNP6	6	6560.34	4.526	0.4701	6.03E-06	+++++	intergenic	NUFIP1P, RNU7- 66P	0					
SNP7	3	7745.52	4.497	0.2397	6.91E-06	++++++	CPNE4		13					
SNP8	9	7416.22	-4.476	0.040	7.59E-06		intergenic	LOC392285 (RPL4P5), C9orf123	1					
SNP9	13	7729.08	-4.452	0.3639	8.49E-06		PCDH9		11					

P-values are based on the sample-size weighted meta-analysis of AGES, ARIC, ASPS, ERF, RS (N=7,777) adjusted for age and sex. *: nearest gene to snp; number of SNPs refers to additional SNPs within 250kb of the top SNP. SNPs highlighted overlap between the two models.

Supplementary table 4

References

- 1. Green MF. Cognitive impairment and functional outcome in schizophrenia and bipolar disorder. J Clin Psychiatry 2006;67 Suppl 9:3-8; discussion 36-42.
- 2. Seidman LJ, Biederman J, Weber W, Hatch M, Faraone SV. Neuropsychological function in adults with attention-deficit hyperactivity disorder. Biol Psychiatry 1998;44:260-8.
- 3. Huntley JD, Howard RJ. Working memory in early Alzheimer's disease: a neuropsychological review. Int J Geriatr Psychiatry 2009.
- 4. Smith EE, Jonides J. Storage and executive processes in the frontal lobes. Science 1999;283:1657-61.
- Wright M, De Geus E, Ando J, Luciano M, Posthuma D, Ono Y, Hansell N, Van Baal C, Hiraishi K, Hasegawa T, Smith G, Geffen G, Geffen L, Kanba S, Miyake A, Martin N, Boomsma D. Genetics of cognition: outline of a collaborative twin study. Twin Res 2001;4:48-56.
- 6. Sleegers K, de Koning I, Aulchenko YS, van Rijn MJ, Houben MP, Croes EA, van Swieten JC, Oostra BA, van Duijn CM. Cerebrovascular risk factors do not contribute to genetic variance of cognitive function: the ERF study. Neurobiol Aging 2007;28:735-41.
- 7. Butcher LM, Kennedy JK, Plomin R. Generalist genes and cognitive neuroscience. Curr Opin Neurobiol 2006;16:145-51.
- McClearn GE, Johansson B, Berg S, Pedersen NL, Ahern F, Petrill SA, Plomin R. Substantial genetic influence on cognitive abilities in twins 80 or more years old. Science 1997;276:1560-3.
- 9. Swan GE, Carmelli D, Reed T, Harshfield GA, Fabsitz RR, Eslinger PJ. Heritability of cognitive performance in aging twins. The National Heart, Lung, and Blood Institute Twin Study. Arch Neurol 1990;47:259-62.
- 10. Payton A. The Impact of Genetic Research on our Understanding of Normal Cognitive Ageing: 1995 to 2009. Neuropsychol Rev 2009.
- 11. Wisdom NM, Callahan JL, Hawkins KA. The effects of apolipoprotein E on non-impaired cognitive functioning: A meta-analysis. Neurobiol Aging 2009.
- 12. Buyske S, Bates ME, Gharani N, Matise TC, Tischfield JA, Manowitz P. Cognitive traits link to human chromosomal regions. Behav Genet 2006;36:65-76.
- 13. Rommelse NN, Arias-Vasquez A, Altink ME, Buschgens CJ, Fliers E, Asherson P, Faraone SV, Buitelaar JK, Sergeant JA, Oosterlaan J, Franke B. Neuropsychological endophenotype approach to genome-wide linkage analysis identifies susceptibility loci for ADHD on 2q21.1 and 13q12.11. Am J Hum Genet 2008;83:99-105.
- Almasy L, Gur RC, Haack K, Cole SA, Calkins ME, Peralta JM, Hare E, Prasad K, Pogue-Geile MF, Nimgaonkar V, Gur RE. A genome screen for quantitative trait loci influencing schizophrenia and neurocognitive phenotypes. Am J Psychiatry 2008;165:1185-92.
- 15. Papassotiropoulos A, Stephan DA, Huentelman MJ, Hoerndli FJ, Craig DW, Pearson JV, Huynh KD, Brunner F, Corneveaux J, Osborne D, Wollmer MA, Aerni A, Coluccia D, Hanggi J, Mondadori CR, Buchmann A, Reiman EM, Caselli RJ, Henke K, de Quervain DJ. Common Kibra alleles are associated with human memory performance. Science 2006;314:475-8.

- 16. Seshadri S, DeStefano AL, Au R, Massaro JM, Beiser AS, Kelly-Hayes M, Kase CS, D'Agostino RB, Sr., Decarli C, Atwood LD, Wolf PA. Genetic correlates of brain aging on MRI and cognitive test measures: a genome-wide association and linkage analysis in the Framingham Study. BMC Med Genet 2007;8 Suppl 1:S15.
- 17. Papassotiropoulos A, Henke K, Stefanova E, Aerni A, Muller A, Demougin P, Vogler C, Sigmund JC, Gschwind L, Huynh KD, Coluccia D, Mondadori CR, Hanggi J, Buchmann A, Kostic V, Novakovic I, van den Bussche H, Kaduszkiewicz H, Weyerer S, Bickel H, Riedel-Heller S, Pentzek M, Wiese B, Dichgans M, Wagner M, Jessen F, Maier W, de Quervain DJ. A genome-wide survey of human short-term memory. Mol Psychiatry 2009.
- 18. Cirulli ET, Kasperaviciute D, Attix DK, Need AC, Ge D, Gibson G, Goldstein DB. Common genetic variation and performance on standardized cognitive tests. Eur J Hum Genet 2010.
- Psaty BM, O'Donnell CJ, Gudnason V, Lunetta KL, Folsom AR, Rotter JI, Uitterlinden AG, Harris TB, Witteman JC, Boerwinkle E. Cohorts for Heart and Aging Research in Genomic Epidemiology (CHARGE) Consortium: Design of prospective meta-analyses of genomewide association studies from 5 cohorts. Circ Cardiovasc Genet 2009;2:73-80.
- Harris TB, Launer LJ, Eiriksdottir G, Kjartansson O, Jonsson PV, Sigurdsson G, Thorgeirsson G, Aspelund T, Garcia ME, Cotch MF, Hoffman HJ, Gudnason V. Age, Gene/Environment Susceptibility-Reykjavik Study: multidisciplinary applied phenomics. Am J Epidemiol 2007;165:1076-87.
- 21. ARIC-investigators. The Atherosclerosis Risk in Communities (ARIC) Study: design and objectives. The ARIC investigators. Am J Epidemiol 1989;129:687-702.
- 22. Schmidt R, Lechner H, Fazekas F, Niederkorn K, Reinhart B, Grieshofer P, Horner S, Offenbacher H, Koch M, Eber B, et al. Assessment of cerebrovascular risk profiles in healthy persons: definition of research goals and the Austrian Stroke Prevention Study (ASPS). Neuroepidemiology 1994;13:308-13.
- Schmidt R, Schmidt H, Pichler M, Enzinger C, Petrovic K, Niederkorn K, Horner S, Ropele S, Watzinger N, Schumacher M, Berghold A, Kostner GM, Fazekas F. C-reactive protein, carotid atherosclerosis, and cerebral small-vessel disease: results of the Austrian Stroke Prevention Study. Stroke 2006;37:2910-6.
- Fried LP, Borhani NO, Enright P, Furberg CD, Gardin JM, Kronmal RA, Kuller LH, Manolio TA, Mittelmark MB, Newman A, et al. The Cardiovascular Health Study: design and rationale. Ann Epidemiol 1991;1:263-76.
- Aulchenko YS, Heutink P, Mackay I, Bertoli-Avella AM, Pullen J, Vaessen N, Rademaker TA, Sandkuijl LA, Cardon L, Oostra B, van Duijn CM. Linkage disequilibrium in young genetically isolated Dutch population. Eur J Hum Genet 2004;12:527-34.
- 26. Feinleib M, Kannel WB, Garrison RJ, McNamara PM, Castelli WP. The Framingham Offspring Study. Design and preliminary data. Prev Med 1975;4:518-25.
- 27. Dawber TR, Kannel WB. The Framingham study. An epidemiological approach to coronary heart disease. Circulation 1966;34:553-5.
- Hofman A, Breteler MM, van Duijn CM, Janssen HL, Krestin GP, Kuipers EJ, Stricker BH, Tiemeier H, Uitterlinden AG, Vingerling JR, Witteman JC. The Rotterdam Study: 2010 objectives and design update. Eur J Epidemiol 2009;24:553-72.

- 29. Ikram MA, Seshadri S, Bis JC, Fornage M, DeStefano AL, Aulchenko YS, Debette S, Lumley T, Folsom AR, van den Herik EG, Bos MJ, Beiser A, Cushman M, Launer LJ, Shahar E, Struchalin M, Du Y, Glazer NL, Rosamond WD, Rivadeneira F, Kelly-Hayes M, Lopez OL, Coresh J, Hofman A, DeCarli C, Heckbert SR, Koudstaal PJ, Yang Q, Smith NL, Kase CS, Rice K, Haritunians T, Roks G, de Kort PL, Taylor KD, de Lau LM, Oostra BA, Uitterlinden AG, Rotter JI, Boerwinkle E, Psaty BM, Mosley TH, van Duijn CM, Breteler MM, Longstreth WT, Jr., Wolf PA. Genomewide association studies of stroke. N Engl J Med 2009;360:1718-28.
- 30. Price AL, Patterson NJ, Plenge RM, Weinblatt ME, Shadick NA, Reich D. Principal components analysis corrects for stratification in genome-wide association studies. Nat Genet 2006;38:904-9.
- 31. Patterson N, Price AL, Reich D. Population structure and eigenanalysis. PLoS Genet 2006;2:e190.
- 32. Richards JB, Rivadeneira F, Inouye M, Pastinen TM, Soranzo N, Wilson SG, Andrew T, Falchi M, Gwilliam R, Ahmadi KR, Valdes AM, Arp P, Whittaker P, Verlaan DJ, Jhamai M, Kumanduri V, Moorhouse M, van Meurs JB, Hofman A, Pols HA, Hart D, Zhai G, Kato BS, Mullin BH, Zhang F, Deloukas P, Uitterlinden AG, Spector TD. Bone mineral density, osteoporosis, and osteoporotic fractures: a genome-wide association study. Lancet 2008;371:1505-12.
- 33. Aulchenko YS, Ripke S, Isaacs A, van Duijn CM. GenABEL: an R library for genome-wide association analysis. Bioinformatics 2007;23:1294-6.
- 34. Chen WM, Abecasis GR. Family-based association tests for genomewide association scans. Am J Hum Genet 2007;81:913-26.
- 35. Bacanu SA, Devlin B, Roeder K. The power of genomic control. Am J Hum Genet 2000;66:1933-44.
- *36.* McCarthy MI, Abecasis GR, Cardon LR, Goldstein DB, Little J, Ioannidis JP, Hirschhorn JN. Genome-wide association studies for complex traits: consensus, uncertainty and challenges. Nat Rev Genet 2008;9:356-69.
- 37. Harold D, Abraham R, Hollingworth P, Sims R, Gerrish A, Hamshere ML, Pahwa JS, Moskvina V, Dowzell K, Williams A, Jones N, Thomas C, Stretton A, Morgan AR, Lovestone S, Powell J, Proitsi P, Lupton MK, Brayne C, Rubinsztein DC, Gill M, Lawlor B, Lynch A, Morgan K, Brown KS, Passmore PA, Craig D, McGuinness B, Todd S, Holmes C, Mann D, Smith AD, Love S, Kehoe PG, Hardy J, Mead S, Fox N, Rossor M, Collinge J, Maier W, Jessen F, Schurmann B, van den Bussche H, Heuser I, Kornhuber J, Wiltfang J, Dichgans M, Frolich L, Hampel H, Hull M, Rujescu D, Goate AM, Kauwe JS, Cruchaga C, Nowotny P, Morris JC, Mayo K, Sleegers K, Bettens K, Engelborghs S, De Deyn PP, Van Broeckhoven C, Livingston G, Bass NJ, Gurling H, McQuillin A, Gwilliam R, Deloukas P, Al-Chalabi A, Shaw CE, Tsolaki M, Singleton AB, Guerreiro R, Muhleisen TW, Nothen MM, Moebus S, Jockel KH, Klopp N, Wichmann HE, Carrasquillo MM, Pankratz VS, Younkin SG, Holmans PA, O'Donovan M, Owen MJ, Williams J. Genome-wide association study identifies variants at CLU and PICALM associated with Alzheimer's disease. Nat Genet 2009;41:1088-93.

- 38. Shi J, Levinson DF, Duan J, Sanders AR, Zheng Y, Pe'er I, Dudbridge F, Holmans PA, Whittemore AS, Mowry BJ, Olincy A, Amin F, Cloninger CR, Silverman JM, Buccola NG, Byerley WF, Black DW, Crowe RR, Oksenberg JR, Mirel DB, Kendler KS, Freedman R, Gejman PV. Common variants on chromosome 6p22.1 are associated with schizophrenia. Nature 2009;460:753-7.
- 39. Rogaeva E, Meng Y, Lee JH, Gu Y, Kawarai T, Zou F, Katayama T, Baldwin CT, Cheng R, Hasegawa H, Chen F, Shibata N, Lunetta KL, Pardossi-Piquard R, Bohm C, Wakutani Y, Cupples LA, Cuenco KT, Green RC, Pinessi L, Rainero I, Sorbi S, Bruni A, Duara R, Friedland RP, Inzelberg R, Hampe W, Bujo H, Song YQ, Andersen OM, Willnow TE, Graff-Radford N, Petersen RC, Dickson D, Der SD, Fraser PE, Schmitt-Ulms G, Younkin S, Mayeux R, Farrer LA, St George-Hyslop P. The neuronal sortilin-related receptor SORL1 is genetically associated with Alzheimer disease. Nat Genet 2007;39:168-77.
- 40. Bertram L, McQueen MB, Mullin K, Blacker D, Tanzi RE. Systematic meta-analyses of Alzheimer disease genetic association studies: the AlzGene database. Nat Genet 2007;39:17-23.
- 41. Pobbati AV, Stein A, Fasshauer D. N- to C-terminal SNARE complex assembly promotes rapid membrane fusion. Science 2006;313:673-6.
- 42. Feng Y, Crosbie J, Wigg K, Pathare T, Ickowicz A, Schachar R, Tannock R, Roberts W, Malone M, Swanson J, Kennedy JL, Barr CL. The SNAP25 gene as a susceptibility gene contributing to attention-deficit hyperactivity disorder. Mol Psychiatry 2005;10:998-1005, 973.
- Carroll LS, Kendall K, O'Donovan MC, Owen MJ, Williams NM. Evidence that putative ADHD low risk alleles at SNAP25 may increase the risk of schizophrenia. Am J Med Genet B Neuropsychiatr Genet 2009;150B:893-9.
- 44. Strehl S, Glatt K, Liu QM, Glatt H, Lalande M. Characterization of two novel protocadherins (PCDH8 and PCDH9) localized on human chromosome 13 and mouse chromosome 14. Genomics 1998;53:81-9.

05 Genome-wide association study of cognitive executive functions: Meta-analysis of the CHARGE consortium

Part III

Genetic Determinants of Alzheimer's Disease

06 Cathepsin D gene and the risk of Alzheimer's disease: A population-based study and meta-analysis

Abstract

Cathepsin D (CTSD) is a gene involved in amyloid precursor protein processing and is considered a candidate for Alzheimer's disease (AD). The aim of the current study was to examine if variation in CTSD increases the risk of AD. We performed a candidate-gene analysis in a population-based cohort-study (N=7983), and estimated the effect of CTSD on the risk of AD. Additionally, a large meta-analysis was performed incorporating our data and previously published data. The T-allele of CTSD rs17571 was associated with an increased risk of AD (p-value 0.007) in the Rotterdam Study. This association was predominantly found in APOE ε 4 noncarriers. A meta-analysis of previously published data showed a significantly increased risk of AD in carriers of the T-allele of rs17571 (OR 1.22, 95% CI 1.03-1.44), irrespective of APOE ε 4 carrier status. This study adds to the evidence that CTSD increases the risk of AD, although the effect size is moderate.

Introduction

Neurodegenerative diseases like Alzheimer's disease (AD) are highly prevalent diseases in the elderly and a major burden on society [1,2]. Neuropathologically, AD is characterized by neuritic plaques (amyloid) and neurofibrillary tangles (hyperphosphorylated tau protein) [3,4]. Progress has been made in understanding the etiology of AD, but the exact pathogenesis has yet to be determined. Alzheimer's disease is most likely caused by various interacting environmental and genetic risk factors [5]. The best known and most consistently reported genetic risk factor is apolipoprotein E (APOE), which is involved in about 50% of late-onset AD cases [6,7]. Other established genes in AD, such as the beta amyloid precursor protein gene (APP), and the presenilin 1 and 2 genes (PSEN1 and PSEN2) [8.9], are mainly implicated in early-onset AD [10]. Although various other genes have been reported in AD, most of them lack sufficient replication. There is particular interest in genes coding for proteins involved in the cleavage of amyloid precursor protein or clearance of beta-amyloid [11-13]. One such protein is cathepsin D, a lysosomal enzyme found in neuritic plaques [14,15]. This protein has secretase activity, and presumably, a role in the processing of tau protein, APP, amyloid beta and apolipoprotein E [16-19]. A polymorphism in exon two of the gene results in a C- to T-transition, resulting in an alanine to valine substitution in the protein. This polymorphism is associated with AD, although reports are inconsistent and a meta-analysis failed to show an association [20]. An explanation for this inconsistency might be that most previous studies used prevalent cases of AD, usually in a clinic-based setting, which might result in biases due to selective case-ascertainment, low-response rates and, possibly, case-fatalities.

The aim of the current study was to examine if polymorphisms in CTSD increase the risk of incident AD in a population-based setting. Additionally, we studied a set of dense SNPs covering a 150kb region flanking CTSD to evaluate the presence of other genetic variants that might be associated with AD. Finally, we performed a meta-analysis pooling our results with previously published data.

Materials and Methods

Study population

The current study was performed in the Rotterdam Study, which is an ongoing population-based study of 7983 elderly Caucasians (55 years and older). The study aims to find determinants of chronic disease in the elderly [21]. At baseline (1990-1993) and three follow-up rounds (1993-1994, 1997-1999, 2002-2004) participants were invited for extensive examinations. The study was approved by the Medical Ethics Committee at ErasmusMC, and all participants gave written informed consent. Only people who were not demented at baseline were included in the current study (N=6886).

Genotyping

Genomic DNA was extracted from whole blood samples according to standard methods [22]. The samples were genotyped for polymorphisms in APOE and CTSD with Taqman allelic discrimination Assays-By-Design (Applied Biosystems, Foster City, CA). All measurements were performed in accordance with the manufacturer's protocols and primer and probe sequences for the SNPs are available from the manufacturer. Details on APOE have been published previously [23], and for CTSD, 3 SNPs were typed (rs2292963, rs2292962, rs17571). These SNPs as tagging SNPs covering the gene were selected based on the blocks that were seen in the CEU population available from the HapMap database (release 2005). We selected the functional variant and 2 SNPs with a minor allele frequency \geq 0.05 (http://www.hapmap.org) [24] [25]. All genotypes were in Hardy-Weinberg equilibrium.

To test whether additional SNPs in the region were associated with AD, we also used data from a genome-wide screen. This screen was part of a large project on the genetics of complex diseases: samples were typed on version 3 of the Illumina-Infinium-II HumanHap550SNP array [26]. We utilized SNPs in a 150kb region centered on the CTSD gene.

The mismatch between the two platforms was previously tested for 24 other SNPs and corresponded to 0.3% (range 0.2-0.6%). The cross platform agreement for rs17571 in the current study was 99.7%. Only samples of good quality DNA were genotyped, and after genotypic quality control 6112 individuals (493 with incident AD) were available for analysis.

Ascertainment of incident AD

Incident AD was diagnosed with a three-step protocol [27]. During the visits to the research centre, all individuals were screened with two cognitive tests (Mini Mental State Examination (MMSE) and Geriatric Mental State schedule (GMS)). Further testing with the Cambridge examination for mental disorders of the elderly (Camdex) was performed in individuals with scores on the MMSE < 26 or on the GMS > 0. When dementia was suspected and additional testing was required for the diagnosis, individuals were examined by a neuropsychologist. Additionally, imaging data was

used when available. A team consisting of a neurologist, a neuropsychologist and a research physician finally ascertained the diagnosis of AD according to internationally accepted criteria. In addition to the visits to the research center, the population was continuously monitored for incident AD through the medical records of general practitioners and the Regional Institute for Outpatient Mental Health Care. There were no cases with a family-history suggestive of autosomal dominant AD. Follow-up was complete through January 1 2005, during which 493 persons developed AD.

Statistical analyses

General descriptive statistics were performed with T-tests for normally distributed variables and Chi-square statistics for categorical or dichotomous variables (version 15.0 of SPSS). To test the associations between the polymorphisms and AD, we used Chi-square statistics and logistic regression models adjusted for age and sex. Additionally, we used Cox regression models to incorporate time-to-event. To further explore the association, we repeated the analyses stratified by APOE ϵ 4 carrier status. A p-value <0.05, adjusted for multiple testing, was considered significant.

A meta-analysis was done for CTSD rs17571 using rmeta (R package). To find previously published studies, we searched PubMed using the key-words: CTSD, Alzheimer's disease, gene and association. Additionally, we checked the reference lists of these papers and searched the AlzGene database (http://www.alzgene.org) [28]. Meta-analysis was conducted on data from previously published studies (from 1999 to January 1st 2009) both overall and stratified by APOE ɛ4 carrier status. For the meta-analyses, random effects models were applied.

	Controls	Cases	p-value
N	5619	493	
Sex (% women)	3242 (58%)	366 (74%)	<0.01
Age	68.2	76.3	<0.01
Mean follow-up	9.6	6.4	<0.01
APOF £4 carriers (%)	1423 (26%)	203 (43%)	<0.01

Table 1

Results

There were 493 incident AD cases in this population of 6112 individuals with a mean age of onset of 82.8 (±6.9) years. The baseline characteristics of the study population are shown in Table 1. Cases were older and were more often women and carriers of the APOE ε 4 allele than control subjects.

Table 2 shows the genotype frequencies for the three tested SNPs. Carriers of the T-allele of rs17571 had a higher risk of AD (OR 1.36, 95% CI 1.09-1.71). This effect was equally strong in the Cox analysis (HR 1.34, 95% CI 1.08-1.64). The two other SNPs were not associated with a

06 Cathepsin D gene and the risk of Alzheimer's disease: A population-based study and meta-analysis

higher risk of AD. Next, the population was stratified by APOE ε 4 carrier status (Table 3). The association of rs17517 with AD was predominantly found in APOE ε 4 noncarriers (adjusted HR 1.42, 95% CI 1.09-1.87). No association was found in APOE ε 4 carriers.

Additionally, we studied a set of dense SNPs in a 150kb region flanking CTSD to test whether other genetic variants explained the association of rs17571 with AD. The results are shown in Figure 1. Rs17571 and rs2292963 were the only SNPs within the gene, and rs2292962 was not present on the array. The 85 other SNPs in the flanking region were neither in strong LD with these two SNPs nor significantly associated with AD. When ranking the p-values of the whole genome-wide association study, rs17571 was ranked at the 5115th position, which is within the top 10% of the results.

Association o	Association of CTSD with AD													
SNP	Physical position	Contro	s				Cases					p-value	OR (95% CI)	
rs2292963	1732908	CC	TC	TT	C-allele	T-allele	CC	TC	TT	C-allele	T-allele			
		3343	1903	269	0.78	0.22	293	161	36	0.76	0.24	0.24	1.10 (0.94-1.28)	
												†0.35	1.08 (0.92-1.27)	
rs2292962	1734897	GG	AG	AA	G-allele	A-allele	GG	AG	AA	G-allele	A-allele			
		3998	1292	89	0.86	0.14	371	101	8	0.88	0.12	0.20	0.88 (0.72-1.07)	
												†0.16	0.86 (0.69-1.06)	
rs17571	1739170	CC	TC	TT	C-allele	T-allele	CC	TC	TT	C-allele	T-allele			
		4628	735	32	0.93	0.07	386	81	6	0.90	0.10	0.007	1.36 (1.09-1.71)	
												10.008	1.38 (1.09-1.75)	

†: adjusted for age and sex

Table 2

Association of CTSD with AD, stratified by APOE ɛ4 carrier status

	APOE	E4 carrie	ers						APOE	APOE 64 honcarriers							
	Controls			Cases			р	OR (95% CI)	Contro	Controls		Case	s		р	OR (95% CI)	
		то			то					то			то				
152252503		10			10				00	10			10				
	847	490	54	122	69	11	0.64	1.06 (0.83-1.36)	2394	1330	206	160	82	24	0.22	1.14 (0.93-1.39)	
							0.91†	1.02 (0.78-1.32)							0.24†	1.131 (0.92-1.40)	
rs2292962	GG	AG	AA	GG	AG	AA			GG	AG	AA	GG	AG	AA			
	991	347	19	147	45	5	0.90	0.98 (0.72-1.33)	2862	900	68	207	51	3	0.09	0.78 (0.59-1.04)	
							0.84†	0.97 (0.70-1.34)							0.06†	0.75 (0.55-1.01)	
rs17571	CC	тс	π	CC	тс	π		,	CC	TC	π	CC	TC	TT		,	
	1152	195	5	161	30	3	0.24	1.25 (0.86-1.81)	3314	509	27	207	48	3	0.009	1.48 (1.11-1.99)	
							0.29†	1.24 (0.84-1.83)							0.008†	1.52 (1.12-2.07)	

t: adjusted for age and sex

Table 3

Finally, we performed a large meta-analysis pooling our data with previously published data. For this analysis, 26 studies reporting on the association of CTSD with AD in Caucasians were identified [13,29-51]. Eight of these [13,31-37] were not analyzed, because they overlapped with other published studies or sufficient data was not provided in the published paper. The latter concerned three publications, of which one found an association between the T-allele of rs17571 and AD, one did not find an association and one did not report this outcome. Additionally, one study was dropped because the control genotypes were out of Hardy-Weinberg Equilibrium (HWE) [44]. An overview of eligible studies is given in Supplementary Table 1. The other 17 studies included a total of 3798 AD cases and 3865 controls. Thirteen included AD patients diagnosed according to the National Institute of Neurological and Communicative Disorders and Stroke-Alzheimer's Disease and Related Disorders Association (NINCDS-ADRDA) criteria, one study used the Diagnostic and Statistical Manual of Mental Disorders (DSM-IV) criteria to identify cases, and one study analyzed autopsy confirmed cases only. Most studies included sporadic late-onset cases, however, there were studies with combinations of early-onset, lateonset, sporadic and familial cases. There was one study, which included only non-familiar earlyonset cases [32]. Controls did not have a diagnosis of Alzheimer's disease, and were selected either from a population-based or clinical sample. The T-allele frequency ranged from 0.05 to 0.11 in controls and from 0.05 to 0.14 in cases. The summary odds ratio of the meta-analysis was 1.22 (95% CI 1.03-1.44). There was significant between-study heterogeneity (p=0.002). When the original study was excluded, the pooled odds ratio dropped to 1.17 (95% CI 0.99-1.37), which was borderline significant (p=0.06). When excluding both Emahazion et al. and the original report, the between-study heterogeneity disappeared (p=0.09) and the summary odds ratio became 1.21 (95% CI 1.05-1.40). To explore how evidence accumulated over time, we performed a cumulative meta-analysis, which is shown in Figure 2. The results show a stable





Figure 1

06 Cathepsin D gene and the risk of Alzheimer's disease: A population-based study and meta-analysis

Cumulative meta-analysis of CTSD rs17571 in AD



*: original study

Study reference

meta-analysis of CTSD rs17571 in AD stratified by APOE ϵ 4 carrier status







*: original study

Odds ratio

Figure 2

chapter 06

summary odds ratio of 1.2 since 2006. When we analyze the data without the present study, the summary odds ratio is 1.21 with a 95% CI ranging from 1.00 to 1.46 (p-value of 0.053). The meta-analysis shown on the Alzgene database (http://www.alzgene.org) presents an OR of 1.17 (95% CI 0.98-1.41) in Caucasians. The main explanation for the difference with our results is the exclusion of the study in which the control genotypes were not in HWE.

For the APOE ε 4 stratified analyzes, 6 studies were available [30,38,40,47-49] with a total of 1446 cases and 1699 controls. The results of this meta-analysis are shown in Figure 3. The pooled odds ratio was greater in APOE ε 4 carriers than noncarriers, but non-significant.

Discussion

In this population-based cohort study and meta-analysis, we found that rs17571 in CTSD was associated with incident AD. This association was more pronounced in APOE ε4 noncarriers, however, a stratified meta-analysis did not confirm this finding. When ranking the p-values of the GWAS, rs17571 ranked 5115th, which is within the top 10%.

Evidence for association of CTSD with AD, thus far, has been very inconsistent. Most studies failed to replicate the original study, with one even showing a significant opposite effect [32] and only a few replicating the original finding [43,49]. Also, meta-analyses thus far did not find a significant association (http://www.alzgene.org) [20,52]. Pooling our longitudinal data with previously published data strengthens the evidence for an association of CTSD with AD. The meta-analysis clearly shows an association in Caucasians, even in a heterogeneous group of early-onset, late-onset, sporadic and familial cases. The data reported in previous studies do not allow us to perform separate analyses in sporadic late-onset AD cases, which would be of interest for further research. Although differences in genotype distribution between early-onset and late-onset cases were not found in a previous study [41], excluding the study of Emahazion et al., which included only early-onset AD cases, reduced between-study heterogeneity. The between-study heterogeneity disappeared when additionally excluding the original study. With the accumulation of data over time we saw a stabilizing odds ratio and a narrowing of the 95% confidence level boundaries.

There were no genome-wide association studies (GWAS) with publicly accessible data that included the SNP of interest. These studies typed two SNPs in LD with rs17571, but these were not associated with AD (rs7938305: p=0.26, OR 0.86; rs17834326: p=0.54, OR 0.97) [53,54]. We decided not to include these in the meta-analysis, because we intended to analyze the functional SNP for the candidate gene analysis only.

Most studies do not find differences between the effect of CTSD in APOE ϵ 4 carriers and noncarriers or a significant interaction between these two genes [29,30,41,42]. Although the results in our study show the effect of the T-allele of rs17571 predominantly in APOE ϵ 4 noncarriers, the meta-analysis does not confirm this finding. In fact, the meta-analysis showed a more pronounced effect in ϵ 4 carriers, however, the difference was not significant.

106

This finding is in line with the suggestion that in AD, CTSD influences APOE independent pathways [55]. CTSD encodes a protein found in senile plaques and CSF [14,18,56]. The mRNA is upregulated in neurons in brains of AD patients, and the immunoreactivity is greater than in normal brains [14,15]. The CTSD protein is able to cleave APP [57] and tau protein [16], although the enzyme seems not to be essential for APP processing [58]. Animal studies show that mutations in CTSD causing dysfunction of the protein, result in loss of neurons in the brain and neurodegeneration [59]. In humans, the biological implication of the polymorphism rs17571 is not clear, although multiple biological associations have been found.

None of the published GWAS, including our own, showed genome-wide evidence for a role of CTSD in AD, however, rs17571 reached the top 10% of our GWA findings. Most GWAS are underpowered to find genes with small effects. The relatively high ranking is in accordance with simulations of GWAS by Zaykin et al.[60].

The strength of the current study is the inclusion of a large population-based cohort study, which has several advantages over clinic-based case-control studies. The main advantages are reduction in biases due to selective case-ascertainment, defining the control group, and low-response rates. Furthermore, a major advantage is the use of incidence instead of prevalence cases, reducing prevalence-incidence bias that may occur when genetic variants are determining mortality. A limitation of the study could be misclassification of cases, but this misclassification would have occurred randomly and would thus not have lead to an overestimation of the effect size. Finally, the population-based design limits the number of cases available.

In conclusion, although the effect of CTSD on the risk of AD is small (OR 1.2), it might be important as one of multiple genetic risk factors adding up to a higher risk of AD. Our findings add to the evidence of an association with AD, and show that the odds ratio stabilized over time.
Overview of	studies	elinihle	for the	meta-analysis	
Overview of	SILULIES	enuble	IUI IIIE	111010-011017515	

First author	Year	Inclu	uded in meta-analysis	Controls	AD	Control		Case		Reference
				N	Ν	T-allele	C-allele	C-allele	T-allele	
						frequency	frequency	frequency	frequency	
*Papassotiropoulos	1999	yes		351	102	0.07	0.93	0.86	0.14	(Papassotiropoulos et al., 1999)
McIlroy	1999	yes		187	183	0.05	0.95	0.92	0.08	(McIlroy et al., 1999)
Bhojak	2000	yes		316	531	0.09	0.91	0.90	0.10	(Bhojak et al., 2000)
Crawford	2000	yes		120	210	0.08	0.92	0.90	0.10	(Crawford et al., 2000)
Papassotiropoulos	2000 (b)	yes		184	127	0.05	0.95	0.88	0.12	(Papassotiropoulos et al., 2000b)
Bertram	2001	yes		182	200	0.09	0.91	0.91	0.09	(Bertram et al., 2001)
Emahazion	2001 (I)	yes		149	120	0.11	0.89	0.95	0.05	(Emahazion et al., 2001)
Matsui	2001	yes		50	69	0.08	0.92	0.93	0.07	(Matsui et al., 2001)
Menzer	2001	yes		302	324	0.06	0.94	0.92	0.08	(Menzer et al., 2001)
Bagnoli	2002	yes		126	197	0.11	0.89	0.88	0.12	(Bagnoli et al., 2002)
Ingegni	2003	yes		120	142	0.10	0.90	0.87	0.13	(Ingegni et al., 2003)
Styczynska	2003	yes		100	100	0.05	0.96	0.94	0.07	(Styczynska et al., 2003)
Beyer	2005	yes		181	206	0.08	0.92	0.92	0.08	(Beyer et al., 2005)
Blomqvist	2006	yes		173	385	0.06	0.94	0.94	0.06	(Blomqvist et al., 2006)
Davidson	2006	yes		767	560	0.08	0.92	0.92	0.08	(Davidson et al., 2006)
Mariani	2006	yes		136	100	0.07	0.93	0.86	0.14	(Mariani et al., 2006)
Capurso	2008	yes		421	242	0.08	0.92	0.94	0.06	(Capurso et al., 2008)
Blomqvist	2006	no	overlap Emahazion et al. 2001	-	-	-	-	-	-	(Blomqvist et al., 2006)
Capurso	2005	no	overlap Capurso et al. 2008	218	168	-	-	-	-	(Capurso et al., 2005)
Corder	2006	no	no data provided	120	180	-	-	-	-	(Corder et al., 2006)
Emahazion	2001 (II)	no	no data provided	176	116	-	-	-	-	(Emahazion et al., 2001)
Kolsch	2004	no	overlap Papassotiropoulos et al. 1999	-	-	-	-	-	-	(Kolsch et al., 2004)
Papassotiropoulos	2000 (a)	no	overlap Papassotiropoulos et al. 1999&2000	-	-	-	-	-	-	(Papassotiropoulos et al., 2000a)
Papassotiropoulos	2002	no	no data provided	24	41	-	-	-	-	(Papassotiropoulos et al., 2002)
Prince	2001	no	overlap Blomqvist et al. 2005	-	-	-		-		(Prince et al., 2001)
Mateo	2002	no	deviation from HWE in control group	346	311	0.10	0.90	0.91	0.09	(Mateo et al., 2002)

*Original report; N: number; allele frequencies are reported for rs17571

Supplementary Table

References

- 1. Cummings JL. Alzheimer's disease. N Engl J Med 2004;351:56-67.
- Caracciolo B, Palmer K, Monastero R, Winblad B, Backman L, Fratiglioni L. Occurrence of cognitive impairment and dementia in the community. A 9-year-long prospective study. Neurology 2008.
- NIA-Working-Group. Consensus recommendations for the postmortem diagnosis of Alzheimer's disease. The National Institute on Aging, and Reagan Institute Working Group on Diagnostic Criteria for the Neuropathological Assessment of Alzheimer's Disease. Neurobiol Aging 1997;18:S1-2.
- Mirra SS, Heyman A, McKeel D, Sumi SM, Crain BJ, Brownlee LM, Vogel FS, Hughes JP, van Belle G, Berg L. The Consortium to Establish a Registry for Alzheimer's Disease (CERAD). Part II. Standardization of the neuropathologic assessment of Alzheimer's disease. Neurology 1991;41:479-86.
- Selkoe DJ. Alzheimer's disease: genes, proteins, and therapy. Physiol Rev 2001;81:741-66.
- Strittmatter WJ, Saunders AM, Schmechel D, Pericak-Vance M, Enghild J, Salvesen GS, Roses AD. Apolipoprotein E: high-avidity binding to beta-amyloid and increased frequency of type 4 allele in late-onset familial Alzheimer disease. Proc Natl Acad Sci U S A 1993;90:1977-81.
- Coon KD, Myers AJ, Craig DW, Webster JA, Pearson JV, Lince DH, Zismann VL, Beach TG, Leung D, Bryden L, Halperin RF, Marlowe L, Kaleem M, Walker DG, Ravid R, Heward CB, Rogers J, Papassotiropoulos A, Reiman EM, Hardy J, Stephan DA. A high-density wholegenome association study reveals that APOE is the major susceptibility gene for sporadic late-onset Alzheimer's disease. J Clin Psychiatry 2007;68:613-8.
- Campion D, Dumanchin C, Hannequin D, Dubois B, Belliard S, Puel M, Thomas-Anterion C, Michon A, Martin C, Charbonnier F, Raux G, Camuzat A, Penet C, Mesnage V, Martinez M, Clerget-Darpoux F, Brice A, Frebourg T. Early-onset autosomal dominant Alzheimer disease: prevalence, genetic heterogeneity, and mutation spectrum. Am J Hum Genet 1999;65:664-70.
- 9. Patterson C, Feightner JW, Garcia A, Hsiung GY, MacKnight C, Sadovnick AD. Diagnosis and treatment of dementia: 1. Risk assessment and primary prevention of Alzheimer disease. Cmaj 2008;178:548-56.
- 10. Waring SC, Rosenberg RN. Genome-wide association studies in Alzheimer disease. Arch Neurol 2008;65:329-34.
- 11. Hardy J, Selkoe DJ. The amyloid hypothesis of Alzheimer's disease: progress and problems on the road to therapeutics. Science 2002;297:353-6.
- 12. Rogaeva E, Meng Y, Lee JH, Gu Y, Kawarai T, Zou F, Katayama T, Baldwin CT, Cheng R, Hasegawa H, Chen F, Shibata N, Lunetta KL, Pardossi-Piquard R, Bohm C, Wakutani Y, Cupples LA, Cuenco KT, Green RC, Pinessi L, Rainero I, Sorbi S, Bruni A, Duara R, Friedland RP, Inzelberg R, Hampe W, Bujo H, Song YQ, Andersen OM, Willnow TE, Graff-Radford N, Petersen RC, Dickson D, Der SD, Fraser PE, Schmitt-Ulms G, Younkin S,

Mayeux R, Farrer LA, St George-Hyslop P. The neuronal sortilin-related receptor SORL1 is genetically associated with Alzheimer disease. Nat Genet 2007;39:168-77.

- 13. Blomqvist ME, Reynolds C, Katzov H, Feuk L, Andreasen N, Bogdanovic N, Blennow K, Brookes AJ, Prince JA. Towards compendia of negative genetic association studies: an example for Alzheimer disease. Hum Genet 2006;119:29-37.
- 14. Cataldo AM, Barnett JL, Berman SA, Li J, Quarless S, Bursztajn S, Lippa C, Nixon RA. Gene expression and cellular content of cathepsin D in Alzheimer's disease brain: evidence for early up-regulation of the endosomal-lysosomal system. Neuron 1995;14:671-80.
- 15. Ginsberg SD, Hemby SE, Lee VM, Eberwine JH, Trojanowski JQ. Expression profile of transcripts in Alzheimer's disease tangle-bearing CA1 neurons. Ann Neurol 2000;48:77-87.
- 16. Kenessey A, Nacharaju P, Ko LW, Yen SH. Degradation of tau by lysosomal enzyme cathepsin D: implication for Alzheimer neurofibrillary degeneration. J Neurochem 1997;69:2026-38.
- 17. Chevallier N, Vizzavona J, Marambaud P, Baur CP, Spillantini M, Fulcrand P, Martinez J, Goedert M, Vincent JP, Checler F. Cathepsin D displays in vitro beta-secretase-like specificity. Brain Res 1997;750:11-9.
- 18. Zhou W, Scott SA, Shelton SB, Crutcher KA. Cathepsin D-mediated proteolysis of apolipoprotein E: possible role in Alzheimer's disease. Neuroscience 2006;143:689-701.
- 19. Ladror US, Snyder SW, Wang GT, Holzman TF, Krafft GA. Cleavage at the amino and carboxyl termini of Alzheimer's amyloid-beta by cathepsin D. J Biol Chem 1994;269:18422-8.
- Ntais C, Polycarpou A, Ioannidis JP. Meta-analysis of the association of the cathepsin D Ala224Val gene polymorphism with the risk of Alzheimer's disease: a HuGE gene-disease association review. Am J Epidemiol 2004;159:527-36.
- 21. Hofman A, Breteler MM, van Duijn CM, Krestin GP, Pols HA, Stricker BH, Tiemeier H, Uitterlinden AG, Vingerling JR, Witteman JC. The Rotterdam Study: objectives and design update. Eur J Epidemiol 2007;22:819-29.
- 22. Miller SA, Dykes DD, Polesky HF. A simple salting out procedure for extracting DNA from human nucleated cells. Nucleic Acids Res 1988;16:1215.
- Slooter AJ, Cruts M, Kalmijn S, Hofman A, Breteler MM, Van Broeckhoven C, van Duijn CM. Risk estimates of dementia by apolipoprotein E genotypes from a population-based incidence study: the Rotterdam Study. Arch Neurol 1998;55:964-8.
- 24. The, International, HapMap, Consortium. The International HapMap Project. Nature 2003;426:789-96.
- 25. Thorisson GA, Smith AV, Krishnan L, Stein LD. The International HapMap Project Web site. Genome Res 2005;15:1592-3.
- 26. Richards JB, Rivadeneira F, Inouye M, Pastinen TM, Soranzo N, Wilson SG, Andrew T, Falchi M, Gwilliam R, Ahmadi KR, Valdes AM, Arp P, Whittaker P, Verlaan DJ, Jhamai M, Kumanduri V, Moorhouse M, van Meurs JB, Hofman A, Pols HA, Hart D, Zhai G, Kato BS, Mullin BH, Zhang F, Deloukas P, Uitterlinden AG, Spector TD. Bone mineral density, osteoporosis, and osteoporotic fractures: a genome-wide association study. Lancet 2008;371:1505-12.

- 27. Ott A, Breteler MM, van Harskamp F, Stijnen T, Hofman A. Incidence and risk of dementia. The Rotterdam Study. American Journal of Epidemiology 1998;147:574-80.
- 28. Bertram L, McQueen M, Mullin K, Blacker D, Tanzi R. The AlzGene Database. Alzheimer Research Forum. Vol. 2009.
- 29. Crawford FC, Freeman MJ, Schinka J, Abdullah LI, Richards D, Sevush S, Duara R, Mullan MJ. The genetic association between Cathepsin D and Alzheimer's disease. Neurosci Lett 2000;289:61-5.
- 30. Bhojak TJ, DeKosky ST, Ganguli M, Kamboh MI. Genetic polymorphisms in the cathespin D and interleukin-6 genes and the risk of Alzheimer's disease. Neurosci Lett 2000;288:21-4.
- Capurso C, Solfrizzi V, D'Introno A, Colacicco AM, Capurso SA, Mastroianni F, Liaci M, Vendemiale G, Capurso A, Panza F. The cathepsin D gene exon 2 (C224T) polymorphism and sporadic Alzheimer's disease in European populations. J Gerontol A Biol Sci Med Sci 2005;60:991-6.
- 32. Emahazion T, Feuk L, Jobs M, Sawyer SL, Fredman D, St Clair D, Prince JA, Brookes AJ. SNP association studies in Alzheimer's disease highlight problems for complex disease analysis. Trends Genet 2001;17:407-13.
- 33. Kolsch H, Ptok U, Majores M, Schmitz S, Rao ML, Maier W, Heun R. Putative association of polymorphism in the mannose 6-phosphate receptor gene with major depression and Alzheimer's disease. Psychiatr Genet 2004;14:97-100.
- 34. Papassotiropoulos A, Bagli M, Jessen F, Maier W, Forstl H, Kurz A, Heun R. Interaction of two genes possibly involved in the regulation of the amyloid precursor protein (APP) processing. Mol Psychiatry 2000;5:240-1.
- Papassotiropoulos A, Lewis HD, Bagli M, Jessen F, Ptok U, Schulte A, Shearman MS, Heun R. Cerebrospinal fluid levels of beta-amyloid(42) in patients with Alzheimer's disease are related to the exon 2 polymorphism of the cathepsin D gene. Neuroreport 2002;13:1291-4.
- 36. Prince JA, Feuk L, Sawyer SL, Gottfries J, Ricksten A, Nagga K, Bogdanovic N, Blennow K, Brookes AJ. Lack of replication of association findings in complex disease: an analysis of 15 polymorphisms in prior candidate genes for sporadic Alzheimer's disease. Eur J Hum Genet 2001;9:437-44.
- Corder EH, Huang R, Cathcart HM, Lanham IS, Parker GR, Cheng D, Smith S, Poduslo SE. Membership in genetic groups predicts Alzheimer disease. Rejuvenation Res 2006;9:89-93.
- Bagnoli S, Nacmias B, Tedde A, Guarnieri BM, Cellini E, Ciantelli M, Petruzzi C, Bartoli A, Ortenzi L, Serio A, Sorbi S. Cathepsin D polymorphism in Italian sporadic and familial Alzheimer's disease. Neurosci Lett 2002;328:273-6.
- 39. Beyer K, Lao JI, Latorre P, Ariza A. Age at onset: an essential variable for the definition of genetic risk factors for sporadic Alzheimer's disease. Ann N Y Acad Sci 2005;1057:260-78.
- 40. Capurso C, Solfrizzi V, D'Introno A, Colacicco AM, Capurso SA, Bifaro L, Menga R, Santamato A, Seripa D, Pilotto A, Capurso A, Panza F. Short arm of chromosome 11 and sporadic Alzheimer's disease: catalase and cathepsin D gene polymorphisms. Neurosci Lett 2008;432:237-42.

- 41. Davidson Y, Gibbons L, Pritchard A, Hardicre J, Wren J, Tian J, Shi J, Stopford C, Julien C, Thompson J, Payton A, Thaker U, Hayes AJ, Iwatsubo T, Pickering-Brown SM, Pendleton N, Horan MA, Burns A, Purandare N, Lendon CL, Neary D, Snowden JS, Mann DM. Genetic associations between cathepsin D exon 2 C-->T polymorphism and Alzheimer's disease, and pathological correlations with genotype. J Neurol Neurosurg Psychiatry 2006;77:515-7.
- 42. Ingegni T, Nocentini G, Mariani E, Spazzafumo L, Polidori MC, Cherubini A, Catani M, Cadini D, Senin U, Mecocci P. Cathepsin D polymorphism in Italian elderly subjects with sporadic late-onset Alzheimer's disease. Dement Geriatr Cogn Disord 2003;16:151-5.
- 43. Mariani E, Seripa D, Ingegni T, Nocentini G, Mangialasche F, Ercolani S, Cherubini A, Metastasio A, Pilotto A, Senin U, Mecocci P. Interaction of CTSD and A2M polymorphisms in the risk for Alzheimer's disease. J Neurol Sci 2006;247:187-91.
- 44. Mateo I, Sanchez-Guerra M, Combarros O, Llorca J, Infante J, Gonzalez-Garcia J, del Molino JP, Berciano J. Lack of association between cathepsin D genetic polymorphism and Alzheimer disease in a Spanish sample. Am J Med Genet 2002;114:31-3.
- 45. Matsui T, Morikawa Y, Tojo M, Okamura N, Maruyama M, Hirai H, Chiba H, Matsushita S, Higuchi S, Arai H, Sasaki H. Cathepsin D polymorphism not associated with Alzheimer's disease in Japanese. Ann Neurol 2001;49:544-5.
- 46. McIlroy SP, Dynan KB, McGleenon BM, Lawson JT, Passmore AP. Cathepsin D gene exon 2 polymorphism and sporadic Alzheimer's disease. Neurosci Lett 1999;273:140-1.
- Menzer G, Muller-Thomsen T, Meins W, Alberici A, Binetti G, Hock C, Nitsch RM, Stoppe G, Reiss J, Finckh U. Non-replication of association between cathepsin D genotype and late onset Alzheimer disease. Am J Med Genet 2001;105:179-82.
- 48. Papassotiropoulos A, Bagli M, Feder O, Jessen F, Maier W, Rao ML, Ludwig M, Schwab SG, Heun R. Genetic polymorphism of cathepsin D is strongly associated with the risk for developing sporadic Alzheimer's disease. Neurosci Lett 1999;262:171-4.
- Papassotiropoulos A, Bagli M, Kurz A, Kornhuber J, Forstl H, Maier W, Pauls J, Lautenschlager N, Heun R. A genetic variation of cathepsin D is a major risk factor for Alzheimer's disease. Ann Neurol 2000;47:399-403.
- Styczynska M, Religa D, Pfeffer A, Luczywek E, Wasiak B, Styczynski G, Peplonska B, Gabryelewicz T, Golebiowski M, Kobrys M, Barcikowska M. Simultaneous analysis of five genetic risk factors in Polish patients with Alzheimer's disease. Neurosci Lett 2003;344:99-102.
- 51. Bertram L, Guenette S, Jones J, Keeney D, Mullin K, Crystal A, Basu S, Yhu S, Deng A, Rebeck GW, Hyman BT, Go R, McInnis M, Blacker D, Tanzi R. No evidence for genetic association or linkage of the cathepsin D (CTSD) exon 2 polymorphism and Alzheimer disease. Ann Neurol 2001;49:114-6.
- 52. Bertram L, McQueen MB, Mullin K, Blacker D, Tanzi RE. Systematic meta-analyses of Alzheimer disease genetic association studies: the AlzGene database. Nat Genet 2007;39:17-23.

- 53. Li H, Wetten S, Li L, St Jean PL, Upmanyu R, Surh L, Hosford D, Barnes MR, Briley JD, Borrie M, Coletta N, Delisle R, Dhalla D, Ehm MG, Feldman HH, Fornazzari L, Gauthier S, Goodgame N, Guzman D, Hammond S, Hollingworth P, Hsiung GY, Johnson J, Kelly DD, Keren R, Kertesz A, King KS, Lovestone S, Loy-English I, Matthews PM, Owen MJ, Plumpton M, Pryse-Phillips W, Prinjha RK, Richardson JC, Saunders A, Slater AJ, St George-Hyslop PH, Stinnett SW, Swartz JE, Taylor RL, Wherrett J, Williams J, Yarnall DP, Gibson RA, Irizarry MC, Middleton LT, Roses AD. Candidate single-nucleotide polymorphisms from a genomewide association study of Alzheimer disease. Arch Neurol 2008;65:45-53.
- 54. Reiman EM, Webster JA, Myers AJ, Hardy J, Dunckley T, Zismann VL, Joshipura KD, Pearson JV, Hu-Lince D, Huentelman MJ, Craig DW, Coon KD, Liang WS, Herbert RH, Beach T, Rohrer KC, Zhao AS, Leung D, Bryden L, Marlowe L, Kaleem M, Mastroeni D, Grover A, Heward CB, Ravid R, Rogers J, Hutton ML, Melquist S, Petersen RC, Alexander GE, Caselli RJ, Kukull W, Papassotiropoulos A, Stephan DA. GAB2 alleles modify Alzheimer's risk in APOE epsilon4 carriers. Neuron 2007;54:713-20.
- 55. Sjogren M, Minthon L, Davidsson P, Granerus AK, Clarberg A, Vanderstichele H, Vanmechelen E, Wallin A, Blennow K. CSF levels of tau, beta-amyloid(1-42) and GAP-43 in frontotemporal dementia, other types of dementia and normal aging. J Neural Transm 2000;107:563-79.
- 56. Schwagerl AL, Mohan PS, Cataldo AM, Vonsattel JP, Kowall NW, Nixon RA. Elevated levels of the endosomal-lysosomal proteinase cathepsin D in cerebrospinal fluid in Alzheimer disease. J Neurochem 1995;64:443-6.
- Sadik G, Kaji H, Takeda K, Yamagata F, Kameoka Y, Hashimoto K, Miyanaga K, Shinoda T. In vitro processing of amyloid precursor protein by cathepsin D. Int J Biochem Cell Biol 1999;31:1327-37.
- Saftig P, Peters C, von Figura K, Craessaerts K, Van Leuven F, De Strooper B. Amyloidogenic processing of human amyloid precursor protein in hippocampal neurons devoid of cathepsin D. J Biol Chem 1996;271:27241-4.
- 59. Shacka JJ, Klocke BJ, Young C, Shibata M, Olney JW, Uchiyama Y, Saftig P, Roth KA. Cathepsin D deficiency induces persistent neurodegeneration in the absence of Baxdependent apoptosis. J Neurosci 2007;27:2081-90.
- 60. Zaykin DV, Zhivotovsky LA. Ranks of genuine associations in whole-genome scans. Genetics 2005;171:813-23.

07 Transferrin and HFE genes interact in Alzheimer's disease risk: the Epistasis Projects

Abstract

Iron overload may contribute to the risk of Alzheimer's disease (AD). In the Epistasis Project, with 1757 cases of AD and 6295 controls, we studied four variants in two genes of iron metabolism: haemochromatosis (HFE) C282Y and H63D, and transferrin (TF) C2 and

-2G/A. We replicated the reported interaction between HFE 282Y and TF C2 in the risk of AD: synergy factor = 1.75 (95% confidence interval: 1.1-2.8, p = 0.02) in Northern Europeans. The synergy factor was 3.1 (1.4-6.9, 0.007) in subjects with the APOEs4 allele. We found another interaction, between HFE 63HH and TF -2AA, markedly modified by age. Both interactions were found mainly or only in Northern Europeans. The interaction between HFE 282Y and TF C2 has now been replicated twice, in altogether 2313 cases of AD and 7065 controls, and has also been associated with increased iron load. We therefore suggest that iron overload may be a causative factor in the development of AD. Treatment for iron overload might thus be protective in some cases.

Introduction

Brain iron in ageing and in Alzheimer's disease

Since the 1950s, iron has been reported to increase with age in the brain [1] and has been proposed a role in Alzheimer's disease (AD) [2]. Although a more rapid rise in brain levels of non-haem iron is seen in a person's first 30 years [1], levels of iron and of iron-related proteins continue to rise with normal ageing in many, but not all brain regions [3-9]. Disturbances of iron metabolism have been widely reported in AD brain (below) and also in autopsy brain tissue and CSF and in mild cognitive impairment (MCI) [10,11]. Raised iron levels have been commonly found in AD, e.g. in the basal ganglia [in vivo: [12,13]; autopsy: [14]], in the hippocampus [in vivo: [13,15]; autopsy: [16,17]] and in the neocortex [autopsy: [14,17,18]]. Iron has been located in postmortem tissue in and around senile plaques and neurofibrillary tangles [19-22]. There have been varying reports concerning a possible cellular imbalance of the iron:ferritin ratio [4,7,18,23] and also on the levels and activity of the important ferroxidase, ceruloplasmin [24-26] in AD brain. Altered patterns of transferrin expression have been reported in AD brain [4,14]. Increased expression of lactoferrin has been found both in neurones and glia in numerous regions of AD brain [27]. This may suggest increased use of non-transferrin routes of cellular iron uptake, e.g. via lactoferrin.

Excess iron and oxidative stress

Higher brain iron levels have been correlated with greater cognitive impairment in AD [13,28]. It has been proposed that excess iron in AD leads to oxidative damage [29-32]. Such damage to lipids [33-36], to proteins [37-39] to DNA [40] and to RNA [41] has been found in AD, including early AD, and in MCI. RNA oxidation can lead to reduced rates of protein synthesis, which have been reported in MCI and AD brain tissue [42]. Markers of lipid peroxidation have also been found in the CSF of MCI patients, and in MCI patients who subsequently converted to AD [43,44]. In short, oxidative stress

is seen early in AD, indeed in the pre-clinical stage, and provokes an antioxidant response [45]. Moreover, markers of lipid peroxidation in ventricular fluid have been correlated with cortical atrophy, reduced brain weight and severity of AD [46]. Excess iron may also contribute to neurodegeneration via, for instance, the metabolism and toxicity of β -amyloid [29,47-51], the loss of calcium homeostasis [52] and the degradation of a subset of microglia [53].

Transferrin and HFE in AD

Two genes of iron metabolism have been much studied in AD: transferrin (TF) and its C2 variant (rs1049296), and the haemochromatosis gene (HFE) and its C282Y (rs1800562) and H63D (rs1799945) variants. These HFE variants affect blood iron status, with the rare 282Y homozygotes (frequency < 1%) having the highest transferrin saturation [54]. Since 1993 when an association of TF C2 with AD was first proposed [55], there have been altogether 14 independent studies of that association. Although most studies have been negative, the AlzGene meta-analysis of the allele [56] (http://www.alzforum.org/res/com/gen/alzgene/) currently shows a significant, though low, odds ratio of AD: 1.2 (95% confidence interval: 1.04-1.3) (10 December 2009), with a similar pattern in Caucasians and East Asians. Also, a large family study [57] supported the association. There have been 13 independent, association studies of HFE, with mixed results. The AlzGene meta-analysis of the 282Y allele currently gives an odds ratio of AD of 1.04 (0.9-1.2). However, three studies [58-60] have proposed that the 63D allele may be associated with lower onset age of AD, possibly in interaction with apolipoprotein £4 (APOE £4). Also, the Oxford Project to Investigate Memory and ageing (OPTIMA) reported an interaction between TF C2 and HFE 282Y in the risk of AD [61]. This interaction was recently replicated [62] in a large collaborative study. Further, our preliminary examination of data from the Rotterdam Study revealed another potential interaction in the risk of AD. That was between HFE H63D and a single nucleotide polymorphism (SNP) just two base-pairs upstream of the start codon of TF (rs1130459, -2G/A) (data not shown). We therefore examined four SNPs, HFE C282Y, HFE H63D, TF C2 and TF -2G/A (Table 1), in the Epistasis Project. This project is a collaboration of seven AD research groups, contributing DNA samples from 1757 cases of AD and 6295 controls [63].

Methods

Study population

The Epistasis Project aims primarily to replicate genetic interactions that have been reported to affect the risk of AD. Sample-sets were drawn from narrow geographical regions with relatively homogeneous, Caucasian populations, by seven AD research groups: Bonn, Bristol, Nottingham, OPTIMA, Oviedo, Rotterdam and Santander. All AD cases were diagnosed "definite" or "probable" by CERAD or NINCDS-ADRDA criteria. Full details are given in our previous paper [63].

Genotyping

Genotyping for the six centres other than Rotterdam (below) was performed at the Wellcome Trust Sanger Institute, using the iPLEX Gold assay (Sequenom Inc.). Whole genome amplified DNA was used for 82% of samples; genomic DNA was used for the 18% of samples that were not suitable for whole genome amplification. A Sequenom iPLEX, designed for quality control purposes, was used to assess genotype concordance between genomic and whole genome amplified DNA for 168 individuals. Assays for all SNPs were designed using the eXTEND suite and MassARRAY Assay Design software version 3.1 (Sequenom Inc.). Samples were amplified in multiplexed PCR reactions before allele specific extension. Allelic discrimination was obtained by analysis with a MassARRAY Analyzer Compact mass spectrometer. Genotypes were automatically assigned and manually confirmed using MassArray TyperAnalyzer software version 4.0 (Sequenom Inc.). Gender markers were included in all iPLEX assays as a quality control metric for confirmation of plate/sample identity. Genotyping of rs1130459 was carried out using KASPar technology by KBioscience (http://www.kbioscience.co.uk). Repeat genotyping of rs1049296 was performed using TaqMan technology according to standard conditions (C_7505275_10; Applied Biosystems, Foster City, CA, U.S.A.).

Genotyping in the Rotterdam cohort was done on Version 3 Illumina-Infinium-II HumanHap550 SNP array (Illumina, San Diego, USA) and additionally, SNPs were imputed using MACH software (http:// www.sph.umich.edu/csg/abecasis/MACH/) with HapMap CEU Release 22 as a reference [64]. The reliability of imputation was estimated for each imputed SNP with the ratio of expected and observed dosage variance (O/E ratio). Only samples with high-quality extracted DNA were genotyped; 5974 were available with good quality genotyping data; 5502 of these had reliable phenotypes. For this study, rs1800562 and rs1049296 were genotyped, and rs1799945 and rs1130459 were imputed.

Statistical analysis

We assessed associations with logistic regression models and synergy factor analysis [65], controlling for age, gender, study centre and the ε 4 allele of apolipoprotein (APOE ε 4), using R Version 2.6.1 (R Foundation for Statistical Computing, Vienna, Austria). Heterogeneity between centers was controlled by fitting a fixed effect corresponding to contrasts between the baseline centre and the six other centers (having compared models with fixed- and random-effect terms in centre, goodness of fit was measured using Akaike's Information Criterion, which favored using fixed effects only). Overdispersion was controlled by fitting generalized linear models with a quasi-binomial family with logit link. Where the overall synergy factor was significant at p < 0.05, the seven individual centers and the two geographical regions, North Europe and North Spain, were also examined. Power calculations were based on the synergy factor values. Comparisons of allelic frequencies between North Spain and North Europe were by Fisher's exact test. Linear regression models were used to compare onset ages. Linkage disequilibrium data were estimated using the R genetics library (http://cran.r-project. org/web/packages/genetics/ index.html). All tests of significance and power calculations were twosided.

Results

Hardy-Weinberg analysis, allelic frequencies and linkage disequilibrium

Hardy-Weinberg analysis was performed for the four SNPs of Table 1 in both cases and controls of the Rotterdam samples, genotyped by Rotterdam, and of the samples from the other six centres, genotyped by the Sanger Institute. In one of these 16 analyses, i.e. of rs1049296 in controls from the six centres, the genotypes were not in Hardy-Weinberg equilibrium. These samples were therefore retyped by a different method. The two methods, Sequenom and TaqMan, were in 99.6% agreement in the 2317 samples and neither were in Hardy-Weinberg equilibrium in controls (p = 0.01). There are three reported, overlapping copy number variations, each between 1.3 and 1.7 kb long, all just over 6 kb downstream of TF and just over 9kb from rs1049296 (TF C2) (Database of Genomic Variants: http://projects.tcag.ca/variation/).

Studied St	SNPs							
Gene	SNP	Minor allele fi	requency in contro	ols	Linkage	disequilib	rium in cor	trols
		North	North	Difference:	North Eu	urope	North Sp	pain
		Europe	Spain	p	D'	r²	D'	r
HFE	rs1800562 C282Y	6.3% (Y)	3.45% (Y)	0.0003				
					0.992	0.011	0.992	0.011
	rs1799945 H63D	14.5% (D)	24.0% (D)	4.5x10 ⁻¹⁶				
TF	rs1049296 C2 = P589S	16.4% (S)	17.65% (S)	0.34				
					0.602	0.080	0.338	0.026
	rs1130459 -2G/A	47.2% (A)	48.0% (A)	0.64				

SNP = single nucleotide polymorphism; HFE = the haemochromatosis gene; TF = the transferrin gene; D' = ratio of observed linkage disequilibrium; r = correlation coefficient.

Table 1

Table 1 gives the allelic frequencies and structure of linkage disequilibrium (LD) of the two pairs of SNPs in controls. The allelic frequencies of HFE were strikingly different between North Europe and North Spain (Supplementary Table 1), but the frequencies were rather similar for TF. In contrast, the LD structure of HFE was nearly identical in the two regions, but differed somewhat for TF: D' = 0.6 versus 0.3 (Table 1). Supplementary Table 2 shows the allele frequencies by country, which were largely consistent in North Europe.

Gene	Genotype	Numbers				Adjusted ^a odds ratios	s (95% Cl, p)	
		North Euro	pe	North Spa	ain	All	North Europe	North Spain
		Controls	AD cases	Controls	AD cases			
HFE	282YY+CY	667	117	30	38	1.0 (0.8-1.2, 0.8)	0.9 (0.7-1.2, 0.55)	1.3 (0.7-2.15, 0.4)
	282CC	4830	945	428	428		,	
	63DD+HD	1478	281	193	191	1.0 (0.8-1.1, 0.6)	0.9 (0.8-1.1. 0.4)	1.1 (0.8-1.4, 0.7)
	63HH	4016	774	267	269			
TF	589SS+PS	1647	333	139	163	1.1 (0.96-1.3, 0.2)	1.0 (0.9-1.2, 0.7)	1.2 (0.9-1.7, 0.2)
	589PP	3847	724	317	296			
	-2AA+GA	1213	211	100	141	1.1 (0.97-1.3, 0.1)	1.0 (0.8-1.2, 0.9)	1.6 (1.1-2.2, 0.01)
	-2GG	4258	716	348	297			



SNP = single nucleotide polymorphism; HFE = the haemochromatosis gene; TF = the transferrin gene; AD = Alzheimer's disease; CI = confidence interval. ^aAll analyses controlled for centre, age, gender and genotype of apolipoprotein ɛ4.

Table 2

Main effects: AD risk and onset age

We found no significant main effects on AD risk of three of the four SNPs (Table 2). But the genotype, TF -2AA+GA, was associated with risk of AD only in North Spain: odds ratio = 1.6 (95% confidence interval: 1.1-2.2, p = 0.01). Although we found an apparent association of the HFE 63D allele with lower onset age of AD (p = 0.0008) in unadjusted analysis, we did not replicate the association on controlling for centre, age, gender and APOE ε 4 (p = 0.36). In the five centres with data on onset ages (Bonn, Bristol, OPTIMA, Santander, Rotterdam: n = 1249), the median onset age was 74.0 years (interquartile range: 68.0-80.0, n = 389) in patients with HFE 63D, while it was 76.0 years (69.0-82.6, 860) in HFE 63H homozygotes. There was no interaction between HFE H63D and APOEε4 in onset age.

Interaction	Dataset	Numbers		Power ^a	Adjusted ^b SF
		Controls	AD		(95% CI, <i>p</i>)
HFE 282Y+	All	6227	1672	43%	1.4 (0.9-2.2, 0.15)
× TF C2+	North Europe	5723	1152	34%	1.75 (1.1-2.8, 0.02)
	North Spain	504	520	17%	0.5 (0.2-1.8, 0.3)
HFE 63HH	All	6206	1496	76%	1.5 (1.1-2.1, 0.02)
× TF -2AA	North Europe	5706	1008	58%	2.0 (1.3-3.05, 0.002)
	North Spain	500	488	40%	1.3 (0.7-2.55, 0.4)

Interactions between HFE and TF SNPs in the risk of Alzheimer's disease

SNP = single nucleotide polymorphism; HFE = the haemochromatosis gene; TF = the transferrin gene; SF = synergy factor; CI = confidence interval; Y4 and C2+ group the genotypes; YY4CY and C2/C2+C2/, respectively. ¹⁷Power to detect an SF of 1 4 (first interaction) or of 1.5 (second interaction) at p = 0.05. ⁵All analyses controlled for centre, age, gender and genotype of apolipoprotein c4.

Interactions

We also replicated both interactions in the risk of AD, i.e. the interaction between HFE 282Y and TF C2 and that between HFE 63HH and TF -2AA, but only in Northern Europeans (Table 3): synergy factors = 1.75 (95% confidence interval: 1.1-2.8, p = 0.02) for HFE 282Y × TF C2 and 2.0 (1.3-3.05, 0.002) for HFE 63HH × TF -2AA. Neither result was due to any distortion in control frequencies (data not shown). In view of reported sex differences in iron status and related genetics [66], we also examined these interactions by sex. We found no significant differences between men and women in either interaction (data not shown). In the OPTIMA report [61], we had suggested that there might be a further interaction between HFE 282Y, TF C2 and APOE ϵ 4. Here we found that the interaction between these two iron-related SNPs only occurred in subjects with APOE ϵ 4, where the synergy factor in North Europeans was 3.1 (1.4-6.9, 0.007), but not in APOE ϵ 4 negatives, where the synergy factor was 1.05 (0.55-2.0, 0.87). However, there were no significant interactions between APOE ϵ 4 and either SNP or both together.

Effect on the i	nteraction betwe	en HFE 63HH and TF-2AA of stratification I	by age ± 80 years

Age	TF-2G/A	Adjusted* OR: HFE 63HH vs 63D-positive (95% Cl, p)		Adjusted ^a SF: HFE 63HH × TF -2AA (95% CI, p)		
	genotype	North Europe North Spain		North Europe	North Spain	
< 80 years	-2AA	1.55 (0.85-2.9, 0.2)	0.9 (0.4-1.8, 0.7)	2 2 /1 1 4 4 0 02)	10(040010)	
	-2G-positive	0.8 (0.55-1.1, 0.2)	0.9 (0.5-1.4, 0.5)	2.2 (1.1-4.4, 0.03)	1.0 (0.4*2.3, 1.0)	
>80 years	-2AA	2.2 (1.2-4.0, 0.01)	3.1 (0.9-11.2, 0.08)	0.0 (1.1.4.0.0.00)		
	-2G-positive	1.0 (0.8-1.3, 0.95)	1.1 (0.6-2.0, 0.85)	2.2 (1.1-4.2, 0.02)	2.9 (0.0-10.6, 0.1)	

HFE = the haemochromatosis gene; TF = the transferrin gene; OR = odds ratio; CI = confidence interval; SF = synergy factor; age is at death or last examination (an age of 80 years is equivalent to an onset age of 75 years in our dataset). Numbers of cases and controls are given in Supplementary Table 3. ^aAll analyses controlled for centre, age, gender and genotype of apolioportelin e4.

Table 4

chapter 07

Studied S	NPs, by genotype					
Gene	SNP	Subjects	Numbers, by		Totals	
HFE	rs1800562 C282Y	Controls	AA: 23	GA: 716	GG: 5532	6271
		AD	7	166	1556	1729
	rs1799945 H63D	Controls	GG: 161	CG: 1589	CC: 4512	6262
		AD	61	464	1178	1703
TF	rs1049296 C2 = P589S	Controls	TT: 196	CT: 1671	CC: 4374	6241
		AD	64	482	1136	1682
	rs1130459 -2G/A	Controls	GG: 1711	GA: 3137	AA: 1373	6221
		AD	397	735	386	1518

SNP = single nucleotide polymorphism; HFE = the haemochromatosis gene; TF = the transferrin gene

Supplementary Table 1

Note that the second interaction above suggests that HFE 63HH is a risk factor for AD, depending on the TF -2G/A genotype, whereas in apparent contradiction, age data suggest that HFE 63HH might be associated with higher onset age. We therefore examined this genetic interaction by age. There was a clear interaction in our overall dataset between HFE 63HH and age at death or last examination as a continuous variable (p < 0.00001) and between that genotype and age ± 80 years [synergy factor = 1.9 (1.45-2.6, 0.00001)], indicating increased risk associated with that genotype in the older subset. However, Table 4 shows that the interaction between HFE 63HH and TF -2AA applied equally to both age subsets in Northern Europeans. The Northern Spanish dataset lacked power in this subset analysis. However, it was consistent with North Europe in the older subset, but showed no effect in the younger subset (Table 4). Stratification by the median age of controls, 76.88 years instead of 80 vears, produced similar results, except that the synergy factor for Northern Europeans in the younger subset dropped below significance: 2.25 (0.97-5.2, 0.06).

Control a	Control allele frequencies by country							
Gene	SNP	Minor allele freq	uency in controls					
		Britain	The	Germany	Spain			
			Netherlands					
HFE	rs1800562 C282Y	51/816	652/10222	23/462	36/1042			
		= 6.25% (Y)	= 6.4% (Y)	= 5.0% (Y)	= 3.45% (Y)			
	rs1799945 H63D	134/802	1472/10222	56/462	249/1038			
		= 16.7% (D)	= 14.4% (D)	= 12.1% (D)	= 24.0% (D)			
TF	rs1049296 C2 = P589S	125/800	1688/10220	70/442	180/1020			
		= 15.6% (S)	= 16.5% (S)	= 15.8% (S)	= 17.65% (S)			
	rs1130459 -2G/A	358/746	4829/10220	209/462	487/1014			
		= 48.0% (A)	= 47.25% (A)	= 45.2% (A)	= 48.0% (A)			

SNP = single nucleotide polymorphism; HFE = the haemochromatosis gene; TF = the transferrin gene

Supplementary Table 2

Age TF-2G/A		Numbers				Adjusted* OR: HFE	53HH vs 63D-	Adjusted* SF: HFE 63HH × TF -2AA	
	genotype	North Europe		North Spain		positive (95% CI, p)		(95% CI, <i>p</i>)	
		Controls	AD	Controls	AD	North Europe	North Spain	North Europe	North Spain
< 80	-2AA	804	112	59	109	1.55 (0.85-2.9, 0.2)	0.9 (0.4-1.8, 0.7)	2.2 (1.1-4.4.	2.2 (1.1-4.4 , 10 (0.4-2.3, 1.0)
years	-2G- positive	2843	342	200	227	0.8 (0.55-1.1, 0.2)	0.9 (0.5-1.4, 0.5)	0.03)	
> 80 years	-2AA	457	115	53	50	2.2 (1.2-4.0, 0.01)	3.1 (0.9-11.2, 0.08)		
	-2G- positive	1610	447	194	116	1.0 (0.8-1.3, 0.95)	1.1 (0.6-2.0, 0.85)	2.2 (1.1-4.2, 0.02)	2.9 (0.8-10.8, 0.1)

Supplementary Table 3 Effect on the interaction between HEE 63HH and TE-200 of stratification by age + 80 years

HFE = the haemochromatosis gene: TF = the transferrin gene: OR = odds ratio: CI = confidence interval: SF = synergy factor; age is at death or ast examination (an age of 80 years is equivalent to an onset age of 75 years in our dataset). *All analyses controlled for centre, age, gender and genotype of apolipoprotein e4.

Supplementary Table 3

Supplementary Table 4 shows how the association with AD of each risk factor (genotype) is changed by the presence of the interacting factor. For instance, the association with AD of HFE 63HH is changed from an odds ratio of 0.9 (0.75-1.1, 0.4) to an odds ratio of 1.8 (1.2-2.6, 0.004) by the presence of TF -2AA.

Odds ratio of	In the subset:-	Numbers		Adjusted* odds ratios of AD
AD for:-		Controls	AD cases	(95%Cl, p)
HFE 282Y+	TF C2-	Y+ 500	Y+ 71	0.8 (0.6-1.04, 0.08)
		Y- 3514	Y- 725	
	TFC2+	Y+ 199	Y+ 49	1.3 (0.9-2.0, 0.2)
		Y- 1510	Y- 307	
TFC2+	HFE 282Y-	C2+ 1510	C2+ 307	1.0 (0.8-1.2, 0.8)
		C2- 3514	C2- 725	
	HFE 282Y+	C2+ 199	C2+ 49	1.8 (1.1-2.9, 0.015)
		C2- 500	C2- 71	
HFE 63HH	TF-2G+	HH 3279	HH 570	0.9 (0.75-1.1, 0.4)
		D+ 1167	D+ 213	
	TF-2AA	HH 906	HH 173	1.8 (1.2-2.6, 0.004)
		D+ 354	D+ 52	
TF-2AA	HFE 63D+	AA 354	AA 52	0.6 (0.4-0.9, 0.01)
		G+ 1167	G+ 213	
	HFE 63HH	AA 906	AA 173	1.25 (1.02-1.5, 0.035)
		G+ 3279	G+ 570	

Supplementary Table 4. Odds ratios of Alzheimer's disease for interacting genotypes, stratified by each other, in Northern Europeans

HFE = the haemochromatosis gene; TF = the transferrin gene; CI = confidence interval; Y+, C2+, G+ and D+ group the genotypes, YY+CY, C2/C2+C2/, GG + GA and DD + DH, respectively. *All analyses controlled for centre, age, gender and genotype of apolipoprotein ϵ 4.

Supplementary Table 4

Discussion

The interactions

Our results reveal a complex three-way interaction between HFE 63HH, TF -2AA and age (Table 4). This explains the paradox that HFE 63HH is a risk factor for AD, contingent on TF -2G/A genotype, yet may also be associated with a higher onset age of AD. The explanation is that the risk effect of HFE 63HH only applies to older people, possibly through additional interactions with survival-promoting factors. Nevertheless, we consider that this interaction should remain tentative until replicated in another large study.

In contrast, the interaction between HFE 282Y and TF C2 has now been replicated in two large, independent studies: Kauwe et al [62] and the present study. Kauwe et al used 1161 cases and 1342 controls, and they reported a combined synergy factor of 2.4 (1.4-4.2, 0.002), controlling for centre, age, gender and APOEɛ4 (as we did). The discordant results of association studies of these two SNPs when examined individually [see the AlzGene meta-analyses [56] may be partly due to this interaction. For instance, in our Northern European dataset, TF C2 was only associated with AD risk in the presence of HFE 282Y [odds ratio = 1.8 (1.1-2.9, 0.015)], whereas it had no effect in the latter's absence [odds ratio = 1.0 (0.8-1.2, 0.8)] (Supplementary Table 4). It is noticeable that nothing was obtained from the examination of the main effects of HFE 282Y and TF C2 (Table 2). But an important result has been derived from the study of their interaction. This interaction is so far the only example of epistasis in AD to have been consistently replicated in such numbers: altogether 2313 cases and 7065 controls.

Both interactions were found only or mainly in Northern Europeans (Tables 3 and 4). The samples used by Kauwe et al were also mainly Northern European. We found differences between North Europe and North Spain both in allelic frequencies and in LD structure (Table 1). There are several examples of differences between North and South Europe in genetic studies [67-69]. In our metaanalysis of the indel in the angiotensin 1-converting enzyme in AD [69], we were only able to remove the marked heterogeneity in our analyses by geographic stratification. This revealed clear differences between North and South Europe. Also in iron metabolism, a higher proportion of haemochromatosis patients are HFE 282Y homozygotes in North than in South Europe [67]. Other, as yet unknown, genes may modify the associations with AD of our studied SNPs in the Northern Spanish. However, we cannot rule out either interaction in that population, in view of the relatively low power in that subset (Table 3).

Although there is some LD between the two SNPs in HFE and also between the two in TF (Table 1), neither interaction is due to the effects of such LD. We cannot rule out that either interaction is due to LD with other polymorphisms. However, we suggest that each interaction may independently contribute to AD risk, by separate mechanisms, both of which lead to iron overload. There are a number of potential mechanisms, which we outline in the next section.

The four SNPs: potential mechanisms of the associations with AD

The HFE protein regulates iron metabolism in at least three ways. First, the protein has been shown to bind transferrin receptor 1 (TfR1) [70], thereby reducing the affinity of TfR1 for transferrin and decreasing cellular iron uptake. Second, HFE has been reported to lower cellular iron levels without binding TfR1 [71]. Third, HFE inhibits iron export from various cells, including macrophages [72-74]. The most likely mechanism of the association of the 282Y variant with AD may be the loss of the first function of HFE above. That is because the variant fails to reach the cell surface and thus to bind TfR1 [70,75]. Hence it leaves transferrin free to bind TfR1 with high affinity, leading to increased cellular uptake of iron. HFE 282Y may also be defective in the third function above [72]. However, that function may be the mechanism of the association of the H63D polymorphism with AD, assuming that HFE inhibits iron export from microglia, as it does from macrophages [72,73]. A subset of microglia store excess brain iron in ferritin [53]. This role may lead to their dystrophy [53,76] where brain iron accumulation is excessive, as in ageing and in AD (Introduction). The proportion of ferritin-positive microglia increases with ageing and further increases in AD [53]. However, the 63D variant [73] and possibly also 282Y [72] have lost the ability to block iron export, which may result in greater release of iron from microglia. H63D may also affect the first function of HFE (above): although the variant binds TfR1, it is reported to have little influence on the affinity of TfR1 for transferrin [70]. Other potential mechanisms associated with H63D include the promotion of glutamate toxicity [77] and of tau phosphorylation [78]. Oxidative stress-related mechanisms of these two variants are consistent with three findings of a Rotterdam study [79]. First, that both variants were associated with higher serum levels of the antioxidant, bilirubin; second, that serum bilirubin correlated with iron load; and third, that high serum bilirubin was associated with reduced mortality in 282Y heterozygotes and 63D homozygotes. The interaction between HFE 282Y and TF C2 may be modified by APOEɛ4 (3.3 above), which itself has been associated with oxidative stress [80-84] and with greater vulnerability to Fe2+ ions [85].

The C2 variant of transferrin has also been proposed to increase the risk of oxidative stress [86], but the mechanisms remain unresolved. In vitro studies [87,88] have shown no differences in the ironbinding properties of the variant. Its changed glycosylation pattern [89] could be relevant, however. Glycosylation patterns are altered in CSF in AD [90], notably of transferring [91], which also has a higher oxidation index in AD plasma [92]. Although such partial changes in the glycosylation of transferrin do not affect its receptor binding [93,94], they may speed its degradation [89]. Changes in the glycosylation patterns of transferrin have been reported in various diseases, e.g. rheumatoid arthritis [95]. Total iron-binding capacity in blood may be marginally lower in C2 homozygotes [96-98], although the difference was only significant in one study [96]. In an OPTIMA study [99] with a subset of subjects from this study, transferrin saturation was higher in non-demented elderly with both HFE 282Y and TF C2, although neither variant alone had any effect. This increased transferrin saturation was due rather to raised serum iron in bi-carriers of these variants, than to lower ironbinding capacity. Iron load did not differ by genetic combination in AD [99]. The C2 variant may also have a role in β-amyloid metabolism [100,101]. The function of the -2G/A variant of TF has not yet been studied. However, given its position between the start site and the promoter of TF, very close to the former, it may well affect the expression of the gene.

Conclusions

We suggest that the combination, HFE 282Y with TF C2, and possibly also HFE 63HH with TF -2AA, may contribute to iron overload and thus to oxidative stress in the pre-clinical phase of AD. The effect of the former combination is influenced by APOEɛ4 and that of the latter combination depends on age. These interactions may partly explain the discordant results of previous studies. There are various potential mechanisms to obtain these effects, but these mechanisms remain unproven. Further study should include the adequately-powered examination of brain iron levels in subjects with these genetic combinations, in AD and particularly in MCI and elderly controls. Excess brain iron has also been found in other neurodegenerative conditions, such as Parkinson's disease [102,103]. It may be of interest to examine these genetic interactions in those conditions as well, and also in relation to cognitive performance in elderly controls, provided sufficient power is available. To have even 50% power to replicate the interaction between HFE 282Y and TF C2 at p = 0.05 in a Northern European sample, i.e. with control allelic frequencies similar to those in Table 1, would require 2400 cases and 2400 controls. It would require an even larger dataset in other populations, which have

still lower frequencies of HFE 282Y. The interaction between HFE 63HH and TF -2AA would require 1025 cases and 1025 controls to have 50% power. However, the former interaction has now been replicated twice independently, in samples totalling 2313 cases and 7065 controls, i.e. in Kauwe et al [62] and in the Northern Europeans of this study. Risk factors for AD may act many years before disease onset, as with high blood pressure [104,105] and with high serum cholesterol [106]. In the OPTIMA study cited above [99], the 282Y/C2 combination was also associated with higher iron load in non-demented elderly. We may therefore conclude that iron overload can be one of the causative factors in the development of AD. In our controls, 3.5% of Northern Europeans have the 282Y/C2 combination and 15.9% have the 63HH/-2AA combination (but the latter interaction needs further replication). Altogether, 18.1% have one or other combination. We suggest that treatment for iron overload, e.g. venesection [107] or iron chelation [108,109] might benefit elderly people with these genetic combinations and needs to be explored further.

References

- 1. Hallgren B, Sourander P. The effect of age on the non-haemin iron in the human brain. J Neurochem 1958;3:41-51.
- 2. Goodman L. Alzheimer's disease; a clinico-pathologic analysis of twenty-three cases with a theory on pathogenesis. J Nerv Ment Dis 1953;118:97-130.
- 3. Milton WJ, Atlas SW, Lexa FJ, Mozley PD, Gur RE. Deep gray matter hypointensity patterns with aging in healthy adults: MR imaging at 1.5 T. Radiology 1991;181:715-9.
- 4. Connor JR, Snyder BS, Beard JL, Fine RE, Mufson EJ. Regional distribution of iron and iron-regulatory proteins in the brain in aging and Alzheimer's disease. J Neurosci Res 1992;31:327-35.
- 5. Bartzokis G, Sultzer D, Mintz J, Holt LE, Marx P, Phelan CK, Marder SR. In vivo evaluation of brain iron in Alzheimer's disease and normal subjects using MRI. Biol Psychiatry 1994;35:480-7.
- 6. Bartzokis G, Tishler TA, Lu PH, Villablanca P, Altshuler LL, Carter M, Huang D, Edwards N, Mintz J. Brain ferritin iron may influence age- and gender-related risks of neurodegeneration. Neurobiol Aging 2007;28:414-23.
- 7. Connor JR, Snyder BS, Arosio P, Loeffler DA, LeWitt P. A quantitative analysis of isoferritins in select regions of aged, parkinsonian, and Alzheimer's diseased brains. J Neurochem 1995;65:717-24.
- 8. Martin WR, Ye FQ, Allen PS. Increasing striatal iron content associated with normal aging. Mov Disord 1998;13:281-6.
- 9. Hirose W, Ikematsu K, Tsuda R. Age-associated increases in heme oxygenase-1 and ferritin immunoreactivity in the autopsied brain. Leg Med (Tokyo) 2003;5 Suppl 1:S360-6.
- 10. Lavados M, Guillon M, Mujica MC, Rojo LE, Fuentes P, Maccioni RB. Mild cognitive impairment and Alzheimer patients display different levels of redox-active CSF iron. J Alzheimers Dis 2008;13:225-32.
- 11. Smith MA, Zhu X, Tabaton M, Liu G, McKeel Jr DW, Cohen ML, Wang X, Siedlak SL, Hayashi T, Nakamura M, Nunomura A, Perry G. Increased Iron and Free Radical Generation in Preclinical Alzheimer Disease and Mild Cognitive Impairment. J Alzheimers Dis 2009.
- 12. Bartzokis G, Sultzer D, Cummings J, Holt LE, Hance DB, Henderson VW, Mintz J. In vivo evaluation of brain iron in Alzheimer disease using magnetic resonance imaging. Arch Gen Psychiatry 2000;57:47-53.
- 13. Ding B, Chen KM, Ling HW, Sun F, Li X, Wan T, Chai WM, Zhang H, Zhan Y, Guan YJ. Correlation of iron in the hippocampus with MMSE in patients with Alzheimer's disease. J Magn Reson Imaging 2009;29:793-8.
- 14. Loeffler DA, Connor JR, Juneau PL, Snyder BS, Kanaley L, DeMaggio AJ, Nguyen H, Brickman CM, LeWitt PA. Transferrin and iron in normal, Alzheimer's disease, and Parkinson's disease brain regions. J Neurochem 1995;65:710-24.
- 15. Schenck JF, Zimmerman EA, Li Z, Adak S, Saha A, Tandon R, Fish KM, Belden C, Gillen RW, Barba A, Henderson DL, Neil W, O'Keefe T. High-field magnetic resonance imaging of brain iron in Alzheimer disease. Top Magn Reson Imaging 2006;17:41-50.
- 16. Deibel MA, Ehmann WD, Markesbery WR. Copper, iron, and zinc imbalances in severely degenerated brain regions in Alzheimer's disease: possible relation to oxidative stress. J Neurol Sci 1996;143:137-42.
- 17. Cornett CR, Markesbery WR, Ehmann WD. Imbalances of trace elements related to oxidative damage in Alzheimer's disease brain. Neurotoxicology 1998;19:339-45.
- 18. Dedman DJ, Treffry A, Candy JM, Taylor GA, Morris CM, Bloxham CA, Perry RH, Edwardson JA, Harrison PM. Iron and aluminium in relation to brain ferritin in normal individuals and Alzheimer's-disease and chronic renal-dialysis patients. Biochem J 1992;287 (Pt 2):509-14.

- 19. Jellinger K, Paulus W, Grundke-Iqbal I, Riederer P, Youdim MB. Brain iron and ferritin in Parkinson's and Alzheimer's diseases. J Neural Transm Park Dis Dement Sect 1990;2:327-40.
- 20. Connor JR, Menzies SL, St Martin SM, Mufson EJ. A histochemical study of iron, transferrin, and ferritin in Alzheimer's diseased brains. J Neurosci Res 1992;31:75-83.
- 21. Good PF, Perl DP, Bierer LM, Schmeidler J. Selective accumulation of aluminum and iron in the neurofibrillary tangles of Alzheimer's disease: a laser microprobe (LAMMA) study. Ann Neurol 1992;31:286-92.
- 22. Morris CM, Kerwin JM, Edwardson JA. Non-haem iron histochemistry of the normal and Alzheimer's disease hippocampus. Neurodegeneration 1994;3:267-75.
- 23. Pinero DJ, Hu J, Connor JR. Alterations in the interaction between iron regulatory proteins and their iron responsive element in normal and Alzheimer's diseased brains. Cell Mol Biol (Noisy-le-grand) 2000;46:761-76.
- 24. Connor JR, Tucker P, Johnson M, Snyder B. Ceruloplasmin levels in the human superior temporal gyrus in aging and Alzheimer's disease. Neurosci Lett 1993;159:88-90.
- 25. Loeffler DA, LeWitt PA, Juneau PL, Sima AA, Nguyen HU, DeMaggio AJ, Brickman CM, Brewer GJ, Dick RD, Troyer MD, Kanaley L. Increased regional brain concentrations of ceruloplasmin in neurodegenerative disorders. Brain Res 1996;738:265-74.
- 26. Castellani RJ, Smith MA, Nunomura A, Harris PL, Perry G. Is increased redox-active iron in Alzheimer disease a failure of the copper-binding protein ceruloplasmin? Free Radic Biol Med 1999;26:1508-12.
- 27. Kawamata T, Tooyama I, Yamada T, Walker DG, McGeer PL. Lactotransferrin immunocytochemistry in Alzheimer and normal human brain. Am J Pathol 1993;142:1574-85.
- 28. Zhu WZ, Zhong WD, Wang W, Zhan CJ, Wang CY, Qi JP, Wang JZ, Lei T. Quantitative MR phase-corrected imaging to investigate increased brain iron deposition of patients with Alzheimer disease. Radiology 2009;253:497-504.
- 29. Schubert D, Chevion M. The role of iron in beta amyloid toxicity. Biochem Biophys Res Commun 1995;216:702-7.
- 30. Sayre LM, Perry G, Harris PL, Liu Y, Schubert KA, Smith MA. In situ oxidative catalysis by neurofibrillary tangles and senile plaques in Alzheimer's disease: a central role for bound transition metals. J Neurochem 2000;74:270-9.
- Honda K, Smith MA, Zhu X, Baus D, Merrick WC, Tartakoff AM, Hattier T, Harris PL, Siedlak SL, Fujioka H, Liu Q, Moreira Pl, Miller FP, Nunomura A, Shimohama S, Perry G. Ribosomal RNA in Alzheimer disease is oxidized by bound redox-active iron. J Biol Chem 2005;280:20978-86.
- 32. Jiang D, Li X, Williams R, Patel S, Men L, Wang Y, Zhou F. Ternary complexes of iron, amyloid-beta, and nitrilotriacetic acid: binding affinities, redox properties, and relevance to iron-induced oxidative stress in Alzheimer's disease. Biochemistry 2009;48:7939-47.
- Pratico D, V MYL, Trojanowski JQ, Rokach J, Fitzgerald GA. Increased F2-isoprostanes in Alzheimer's disease: evidence for enhanced lipid peroxidation in vivo. Faseb J 1998;12:1777-83.
- 34. Keller JN, Schmitt FA, Scheff SW, Ding Q, Chen Q, Butterfield DA, Markesbery WR. Evidence of increased oxidative damage in subjects with mild cognitive impairment. Neurology 2005;64:1152-6.
- 35. Markesbery WR, Kryscio RJ, Lovell MA, Morrow JD. Lipid peroxidation is an early event in the brain in amnestic mild cognitive impairment. Ann Neurol 2005;58:730-5.
- 36. Williams TI, Lynn BC, Markesbery WR, Lovell MA. Increased levels of 4-hydroxynonenal and acrolein, neurotoxic markers of lipid peroxidation, in the brain in Mild Cognitive Impairment and early Alzheimer's disease. Neurobiol Aging 2006;27:1094-9.

- 37. Butterfield DA, Poon HF, St Clair D, Keller JN, Pierce WM, Klein JB, Markesbery WR. Redox proteomics identification of oxidatively modified hippocampal proteins in mild cognitive impairment: insights into the development of Alzheimer's disease. Neurobiol Dis 2006;22:223-32.
- 38. Cecarini V, Ding Q, Keller JN. Oxidative inactivation of the proteasome in Alzheimer's disease. Free Radic Res 2007;41:673-80.
- 39. Terni B, Boada J, Portero-Otin M, Pamplona R, Ferrer I. Mitochondrial ATP-Synthase in the Entorhinal Cortex Is a Target of Oxidative Stress at Stages I/II of Alzheimer's Disease Pathology. Brain Pathol 2009.
- 40. Wang J, Markesbery WR, Lovell MA. Increased oxidative damage in nuclear and mitochondrial DNA in mild cognitive impairment. J Neurochem 2006;96:825-32.
- 41. Nunomura A, Perry G, Aliev G, Hirai K, Takeda A, Balraj EK, Jones PK, Ghanbari H, Wataya T, Shimohama S, Chiba S, Atwood CS, Petersen RB, Smith MA. Oxidative damage is the earliest event in Alzheimer disease. J Neuropathol Exp Neurol 2001;60:759-67.
- 42. Ding Q, Markesbery WR, Chen Q, Li F, Keller JN. Ribosome dysfunction is an early event in Alzheimer's disease. J Neurosci 2005;25:9171-5.
- 43. Pratico D, Clark CM, Liun F, Rokach J, Lee VY, Trojanowski JQ. Increase of brain oxidative stress in mild cognitive impairment: a possible predictor of Alzheimer disease. Arch Neurol 2002;59:972-6.
- 44. Brys M, Pirraglia E, Rich K, Rolstad S, Mosconi L, Switalski R, Glodzik-Sobanska L, De Santi S, Zinkowski R, Mehta P, Pratico D, Saint Louis LA, Wallin A, Blennow K, de Leon MJ. Prediction and longitudinal study of CSF biomarkers in mild cognitive impairment. Neurobiol Aging 2009;30:682-90.
- 45. Aksenov MY, Tucker HM, Nair P, Aksenova MV, Butterfield DA, Estus S, Markesbery WR. The expression of key oxidative stress-handling genes in different brain regions in Alzheimer's disease. J Mol Neurosci 1998;11:151-64.
- 46. Montine TJ, Markesbery WR, Zackert W, Sanchez SC, Roberts LJ, 2nd, Morrow JD. The magnitude of brain lipid peroxidation correlates with the extent of degeneration but not with density of neuritic plaques or neurofibrillary tangles or with APOE genotype in Alzheimer's disease patients. Am J Pathol 1999;155:863-8.
- 47. Mantyh PW, Ghilardi JR, Rogers S, DeMaster E, Allen CJ, Stimson ER, Maggio JE. Aluminum, iron, and zinc ions promote aggregation of physiological concentrations of betaamyloid peptide. J Neurochem 1993;61:1171-4.
- 48. Bodovitz S, Falduto MT, Frail DE, Klein WL. Iron levels modulate alpha-secretase cleavage of amyloid precursor protein. J Neurochem 1995;64:307-15.
- 49. Rogers JT, Randall JD, Cahill CM, Eder PS, Huang X, Gunshin H, Leiter L, McPhee J, Sarang SS, Utsuki T, Greig NH, Lahiri DK, Tanzi RE, Bush AI, Giordano T, Gullans SR. An iron-responsive element type II in the 5'-untranslated region of the Alzheimer's amyloid precursor protein transcript. J Biol Chem 2002;277:45518-28.
- 50. Rottkamp CA, Raina AK, Zhu X, Gaier E, Bush AI, Atwood CS, Chevion M, Perry G, Smith MA. Redox-active iron mediates amyloid-beta toxicity. Free Radic Biol Med 2001;30:447-50.
- 51. Kuperstein F, Yavin E. Pro-apoptotic signaling in neuronal cells following iron and amyloid beta peptide neurotoxicity. J Neurochem 2003;86:114-25.
- 52. Hidalgo C, Nunez MT. Calcium, iron and neuronal function. IUBMB Life 2007;59:280-5.
- 53. Lopes KO, Sparks DL, Streit WJ. Microglial dystrophy in the aged and Alzheimer's disease brain is associated with ferritin immunoreactivity. Glia 2008;56:1048-60.
- 54. Jackson HA, Carter K, Darke C, Guttridge MG, Ravine D, Hutton RD, Napier JA, Worwood M. HFE mutations, iron deficiency and overload in 10,500 blood donors. Br J Haematol 2001;114:474-84.

- 55. van Rensburg SJ, Carstens ME, Potocnik FC, Aucamp AK, Taljaard JJ. Increased frequency of the transferrin C2 subtype in Alzheimer's disease. Neuroreport 1993;4:1269-71.
- 56. Bertram L, McQueen MB, Mullin K, Blacker D, Tanzi RE. Systematic meta-analyses of Alzheimer disease genetic association studies: the AlzGene database. Nat Genet 2007;39:17-23.
- 57. Schjeide BM, McQueen MB, Mullin K, DiVito J, Hogan MF, Parkinson M, Hooli B, Lange C, Blacker D, Tanzi RE, Bertram L. Assessment of Alzheimer's disease case-control associations using family-based methods. Neurogenetics 2009;10:19-25.
- 58. Sampietro M, Caputo L, Casatta A, Meregalli M, Pellagatti A, Tagliabue J, Annoni G, Vergani C. The hemochromatosis gene affects the age of onset of sporadic Alzheimer's disease. Neurobiol Aging 2001;22:563-8.
- 59. Combarros O, Garcia-Roman M, Fontalba A, Fernandez-Luna JL, Llorca J, Infante J, Berciano J. Interaction of the H63D mutation in the hemochromatosis gene with the apolipoprotein E epsilon 4 allele modulates age at onset of Alzheimer's disease. Dement Geriatr Cogn Disord 2003;15:151-4.
- 60. Alizadeh BZ, Njajou OT, Millan MR, Hofman A, Breteler MM, van Duijn CM. HFE variants, APOE and Alzheimer's disease: findings from the population-based Rotterdam study. Neurobiol Aging 2009;30:330-2.
- 61. Robson KJ, Lehmann DJ, Wimhurst VL, Livesey KJ, Combrinck M, Merryweather-Clarke AT, Warden DR, Smith AD. Synergy between the C2 allele of transferrin and the C282Y allele of the haemochromatosis gene (HFE) as risk factors for developing Alzheimer's disease. J Med Genet 2004;41:261-5.
- 62. Kauwe JS, Bertelsen S, Mayo K, Cruchaga C, Abraham R, Hollingworth P, Harold D, Owen MJ, Williams J, Lovestone S, Morris JC, Goate AM. Suggestive synergy between genetic variants in TF and HFE as risk factors for Alzheimer's disease. Am J Med Genet B Neuropsychiatr Genet 2009.
- 63. Combarros O, van Duijn CM, Hammond N, Belbin O, Arias-Vasquez A, Cortina-Borja M, Lehmann MG, Aulchenko YS, Schuur M, Kolsch H, Heun R, Wilcock GK, Brown K, Kehoe PG, Harrison R, Coto E, Alvarez V, Deloukas P, Mateo I, Gwilliam R, Morgan K, Warden DR, Smith AD, Lehmann DJ. Replication by the Epistasis Project of the interaction between the genes for IL-6 and IL-10 in the risk of Alzheimer's disease. J Neuroinflammation 2009;6:22.
- 64. Ikram MA, Seshadri S, Bis JC, Fornage M, DeStefano AL, Aulchenko YS, Debette S, Lumley T, Folsom AR, van den Herik EG, Bos MJ, Beiser A, Cushman M, Launer LJ, Shahar E, Struchalin M, Du Y, Glazer NL, Rosamond WD, Rivadeneira F, Kelly-Hayes M, Lopez OL, Coresh J, Hofman A, DeCarli C, Heckbert SR, Koudstaal PJ, Yang Q, Smith NL, Kase CS, Rice K, Haritunians T, Roks G, de Kort PL, Taylor KD, de Lau LM, Oostra BA, Uitterlinden AG, Rotter JI, Boerwinkle E, Psaty BM, Mosley TH, van Duijn CM, Breteler MM, Longstreth WT, Jr., Wolf PA. Genomewide association studies of stroke. N Engl J Med 2009;360:1718-28.
- 65. Cortina-Borja M, Smith AD, Combarros O, Lehmann DJ. The synergy factor: a statistic to measure interactions in complex diseases. BMC Res Notes 2009;2:105.
- 66. Deugnier Y, Jouanolle AM, Chaperon J, Moirand R, Pithois C, Meyer JF, Pouchard M, Lafraise B, Brigand A, Caserio-Schoenemann C, Mosser J, Adams P, Le Gall JY, David V. Genderspecific phenotypic expression and screening strategies in C282Y-linked haemochromatosis: a study of 9396 French people. Br J Haematol 2002;118:1170-8.
- 67. Merryweather-Clarke AT, Pointon JJ, Jouanolle AM, Rochette J, Robson KJ. Geography of HFE C282Y and H63D mutations. Genet Test 2000;4:183-98.
- 68. Capurso C, Solfrizzi V, D'Introno A, Colacicco AM, Capurso SA, Mastroianni F, Liaci M, Vendemiale G, Capurso A, Panza F. The cathepsin D gene exon 2 (C224T) polymorphism and sporadic Alzheimer's disease in European populations. J Gerontol A Biol Sci Med Sci 2005;60:991-6.

- 69. Lehmann DJ, Cortina-Borja M, Warden DR, Smith AD, Sleegers K, Prince JA, van Duijn CM, Kehoe PG. Large meta-analysis establishes the ACE insertion-deletion polymorphism as a marker of Alzheimer's disease. Am J Epidemiol 2005;162:305-17.
- 70. Feder JN, Penny DM, Irrinki A, Lee VK, Lebron JA, Watson N, Tsuchihashi Z, Sigal E, Bjorkman PJ, Schatzman RC. The hemochromatosis gene product complexes with the transferrin receptor and lowers its affinity for ligand binding. Proc Natl Acad Sci U S A 1998;95:1472-7.
- Zhang AS, Davies PS, Carlson HL, Enns CA. Mechanisms of HFE-induced regulation of iron homeostasis: Insights from the W81A HFE mutation. Proc Natl Acad Sci U S A 2003;100:9500-5.
- 72. Montosi G, Paglia P, Garuti C, Guzman CA, Bastin JM, Colombo MP, Pietrangelo A. Wildtype HFE protein normalizes transferrin iron accumulation in macrophages from subjects with hereditary hemochromatosis. Blood 2000;96:1125-9.
- 73. Drakesmith H, Sweetland E, Schimanski L, Edwards J, Cowley D, Ashraf M, Bastin J, Townsend AR. The hemochromatosis protein HFE inhibits iron export from macrophages. Proc Natl Acad Sci U S A 2002;99:15602-7.
- 74. Davies PS, Enns CA. Expression of the hereditary hemochromatosis protein HFE increases ferritin levels by inhibiting iron export in HT29 cells. J Biol Chem 2004;279:25085-92.
- 75. Feder JN, Tsuchihashi Z, Irrinki A, Lee VK, Mapa FA, Morikang E, Prass CE, Starnes SM, Wolff RK, Parkkila S, Sly WS, Schatzman RC. The hemochromatosis founder mutation in HLA-H disrupts beta2-microglobulin interaction and cell surface expression. J Biol Chem 1997;272:14025-8.
- 76. Streit WJ, Miller KR, Lopes KO, Njie E. Microglial degeneration in the aging brain--bad news for neurons? Front Biosci 2008;13:3423-38.
- 77. Mitchell RM, Lee SY, Simmons Z, Connor JR. HFE polymorphisms affect cellular glutamate regulation. Neurobiol Aging 2009.
- 78. Hall EC, 2nd, Lee SY, Mairuae N, Simmons Z, Connor JR. Expression of the HFE allelic variant H63D in SH-SY5Y cells affects tau phosphorylation at serine residues. Neurobiol Aging 2009.
- 79. Alizadeh BZ, Njajou OT, Houwing-Duistermaat JJ, de Jong G, Vergeer JM, Hofman A, Pols HA, van Duijn CM. Does bilirubin protect against hemochromatosis gene (HFE) related mortality? Am J Med Genet A 2004;129A:39-43.
- 80. Miyata M, Smith JD. Apolipoprotein E allele-specific antioxidant activity and effects on cytotoxicity by oxidative insults and beta-amyloid peptides. Nat Genet 1996;14:55-61.
- Ramassamy C, Averill D, Beffert U, Bastianetto S, Theroux L, Lussier-Cacan S, Cohn JS, Christen Y, Davignon J, Quirion R, Poirier J. Oxidative damage and protection by antioxidants in the frontal cortex of Alzheimer's disease is related to the apolipoprotein E genotype. Free Radic Biol Med 1999;27:544-53.
- Ramassamy C, Averill D, Beffert U, Theroux L, Lussier-Cacan S, Cohn JS, Christen Y, Schoofs A, Davignon J, Poirier J. Oxidative insults are associated with apolipoprotein E genotype in Alzheimer's disease brain. Neurobiol Dis 2000;7:23-37.
- Tamaoka A, Miyatake F, Matsuno S, Ishii K, Nagase S, Sahara N, Ono S, Mori H, Wakabayashi K, Tsuji S, Takahashi H, Shoji S. Apolipoprotein E allele-dependent antioxidant activity in brains with Alzheimer's disease. Neurology 2000;54:2319-21.
- 84. Kharrazi H, Vaisi-Raygani A, Rahimi Z, Tavilani H, Aminian M, Pourmotabbed T. Association between enzymatic and non-enzymatic antioxidant defense mechanism with apolipoprotein E genotypes in Alzheimer disease. Clin Biochem 2008;41:932-6.
- 85. Mazur-Kolecka B, Dickson D, Frackowiak J. Induction of vascular amyloidosis-beta by oxidative stress depends on APOE genotype. Neurobiol Aging 2006;27:804-14.
- 86. Beckman L, Beckman G. Transferrin C2 as an enhancer of cyto- and genotoxic damage. Prog Clin Biol Res 1986;209B:221-4.

- 87. Van Landeghem GF, Sikstrom C, Beckman LE, Adolfsson R, Beckman L. Transferrin C2, metal binding and Alzheimer's disease. Neuroreport 1998;9:177-9.
- 88. Zatta P, Messori L, Mauri P, van Rensburg SJ, van Zyl J, Gabrielli S, Gabbiani C. The C2 variant of human serum transferrin retains the iron binding properties of the native protein. Biochim Biophys Acta 2005;1741:264-70.
- 89. van Rensburg SJ, Berman P, Potocnik F, MacGregor P, Hon D, de Villiers N. 5- and 6-glycosylation of transferrin in patients with Alzheimer's disease. Metab Brain Dis 2004;19:89-96.
- Fodero LR, Saez-Valero J, Barquero MS, Marcos A, McLean CA, Small DH. Wheat germ agglutinin-binding glycoproteins are decreased in Alzheimer's disease cerebrospinal fluid. J Neurochem 2001;79:1022-6.
- 91. Taniguchi M, Okayama Y, Hashimoto Y, Kitaura M, Jimbo D, Wakutani Y, Wada-Isoe K, Nakashima K, Akatsu H, Furukawa K, Arai H, Urakami K. Sugar chains of cerebrospinal fluid transferrin as a new biological marker of Alzheimer's disease. Dement Geriatr Cogn Disord 2008;26:117-22.
- 92. Yu HL, Chertkow HM, Bergman H, Schipper HM. Aberrant profiles of native and oxidized glycoproteins in Alzheimer plasma. Proteomics 2003;3:2240-8.
- 93. Mason AB, Miller MK, Funk WD, Banfield DK, Savage KJ, Oliver RW, Green BN, MacGillivray RT, Woodworth RC. Expression of glycosylated and nonglycosylated human transferrin in mammalian cells. Characterization of the recombinant proteins with comparison to three commercially available transferrins. Biochemistry 1993;32:5472-9.
- 94. Hoefkens P, Huijskes-Heins MI, de Jeu-Jaspars CM, van Noort WL, van Eijk HG. Influence of transferrin glycans on receptor binding and iron-donation. Glycoconj J 1997;14:289-95.
- 95. Feelders RA, Vreugdenhil G, de Jong G, Swaak AJ, van Eijk HG. Transferrin microheterogeneity in rheumatoid arthritis. Relation with disease activity and anemia of chronic disease. Rheumatol Int 1992;12:195-9.
- 96. Wong CT, Saha N. Effects of transferrin genetic phenotypes on total iron-binding capacity. Acta Haematol 1986;75:215-8.
- 97. Cleve H, Schwendner E, Rodewald A, Bidlingmaier F. Genetic transferrin types and ironbinding: a comparative study of a European and an African population sample. Hum Genet 1988;78:16-20.
- 98. Sikstrom C, Beckman L, Hallmans G, Asplund K. Transferrin types, iron-binding capacity and body iron stores. Hum Hered 1993;43:337-41.
- 99. Lehmann DJ, Worwood M, Ellis R, Wimhurst VL, Merryweather-Clarke AT, Warden DR, Smith AD, Robson KJ. Iron genes, iron load and risk of Alzheimer's disease. J Med Genet 2006;43:e52.
- 100. Giunta S, Galeazzi R, Valli MB, Corder EH, Galeazzi L. Transferrin neutralization of amyloid beta 25-35 cytotoxicity. Clin Chim Acta 2004;350:129-36.
- 101. Kauwe JS, Wang J, Mayo K, Morris JC, Fagan AM, Holtzman DM, Goate AM. Alzheimer's disease risk variants show association with cerebrospinal fluid amyloid beta. Neurogenetics 2009;10:13-7.
- 102.Berg D, Hochstrasser H. Iron metabolism in Parkinsonian syndromes. Mov Disord 2006;21:1299-310.
- 103. Altamura S, Muckenthaler MU. Iron toxicity in diseases of aging: Alzheimer's disease, Parkinson's disease and atherosclerosis. J Alzheimers Dis 2009;16:879-95.
- 104.Skoog I, Lernfelt B, Landahl S, Palmertz B, Andreasson LA, Nilsson L, Persson G, Oden A, Svanborg A. 15-year longitudinal study of blood pressure and dementia. Lancet 1996;347:1141-5.
- 105. Kivipelto M, Helkala EL, Laakso MP, Hanninen T, Hallikainen M, Alhainen K, Soininen H, Tuomilehto J, Nissinen A. Midlife vascular risk factors and Alzheimer's disease in later life: longitudinal, population based study. Bmj 2001;322:1447-51.

- 106. Solomon A, Kivipelto M, Wolozin B, Zhou J, Whitmer RA. Midlife serum cholesterol and increased risk of Alzheimer's and vascular dementia three decades later. Dement Geriatr Cogn Disord 2009;28:75-80.
- 107. Dwyer BE, Zacharski LR, Balestra DJ, Lerner AJ, Perry G, Zhu X, Smith MA. Getting the iron out: phlebotomy for Alzheimer's disease? Med Hypotheses 2009;72:504-9.
- 108. Crapper McLachlan DR, Dalton AJ, Kruck TP, Bell MY, Smith WL, Kalow W, Andrews DF. Intramuscular desferrioxamine in patients with Alzheimer's disease. Lancet 1991;337:1304-8.
- 109. Amit T, Avramovich-Tirosh Y, Youdim MB, Mandel S. Targeting multiple Alzheimer's disease etiologies with multimodal neuroprotective and neurorestorative iron chelators. Faseb J 2008;22:1296-305.

Part IV

Genetic determinants of age-related brain pathology in hypertensive patients

08 Genetic risk factors for cerebral small vessel disease in hypertensive patients from in a genetically isolated population

Abstract

Background

Asymptomatic cerebral lesions on MRI such as white matter lesions (WML), lacunes and microbleeds are commonly seen in the elderly. We examined the role of a series of candidate genes involved in blood pressure regulation and amyloid metabolism.

Materials & Methods

The study was embedded in a family-based cohort sampled from a Dutch genetically isolated population. We selected individuals between 55 and 75 years of age with hypertension (N=129). Volumes of WML and presence of lacunes and microbleeds were assessed with MRI. We studied 3 genes involved in blood pressure regulation (Angiotensin, Angiotensin II type 1 Receptor, alpha-Adducin), and 2 genes involved in the amyloid pathway (Apolipoprotein E (APOE) and sortilin-related receptor gene (SORL1)).

Results

All participants had WML (median volume, 3.1 mL, interquartile range, 1.5 – 6.5 mL), lacunar infarcts were present in 15.5% and microbleeds in 23.3%. Homozygosity for the APOE ɛ4 allele was associated with lacunes (OR, 4.8; 95% CI, 1.2-19.3). Individuals carrying two copies of the variant allele of 4 SNPs located at the 3'-end of SORL1 (rs1699102, rs3824968, rs2282649, rs1010159), had significantly more often microbleeds (highest odds ratio, 6.87; 95% CI, 1.78-26.44).

Conclusion

The association of SORL1 with microbleeds suggests that the amyloid cascade is involved in the etiology of microbleeds in populations with hypertension.

Introduction

Asymptomatic cerebral lesions, such as white matter lesions (WML), lacunes, and microbleeds are common in the elderly, and associated with stroke and cognitive impairment [1-4]. The mechanisms underlying the pathogenesis of these lesions are largely unknown. Several studies, however, suggest that changes in blood vessels in the brain lead to ischemic damage, causing WML and lacunes, and that leakage of red blood cells might lead to hemosiderin depositions or microbleeds [5,6]. Damage to vessels is most likely a consequence of hypertension and atherosclerosis, but there is also data suggesting that lesions are related to amyloid angiopathy [7].

Since hypertension is a risk factor for cerebral small vessel disease (CSVD), various genetic studies targeted genes that are involved in blood pressure regulation. Most widely studied are the alfaadducin gene (ADD1), which increases renal sodium absorption, and the angiotensin (AGT) and angiotensin II type 1 receptor (AGTR1) genes, which play a role in the renin-angiotensin system [8]. The ADD1-Gly460Trp, AGT-M235T and AGTR1-C573T polymorphisms are associated with increased risk of atherosclerosis and CSVD [9-11]. Also genes involved in the amyloidogenic pathway have been implicated in CSVD. Two identified genes are the apolipoprotein E (APOE) and sortilin-related receptor (SORL1) genes. Although the exact function of APOE in the brain is not fully unraveled yet, this gene is thought to aid beta-amyloid clearance [12]. Carriers of the APOE ε4 allele have an increased risk of WML and microbleeds [13,14]. SORL1 codes for a neuronal apolipoprotein receptor, and is thought to regulate processing of the amyloid precursor protein in the brain [15]. Multiple SNPs within SORL1 are associated with Alzheimer's disease (AD) and cognitive impairment, and recently also with cerebral atrophy and cerebrovascular disease [16,17]. We selected the key SNPs from the original report for the current study [16].

The aim of the study was to examine the role of genes involved in blood pressure regulation and amyloid processing in the development of WML, lacunes, microbleeds and cognitive impairment.

Materials and Methods

The study was embedded in a population-based study in a genetically isolated population (Erasmus Rucphen Family (ERF)-Study). Considering hypertension as major risk factor for CSVD, we selected participants aged 55 to 75 years with hypertension to ensure a high prevalence of pathology. Hypertension was defined as systolic blood pressure \geq 160 and/or diastolic blood pressure \geq 100 and/or use of antihypertensive medication. Persons with a history of stroke or dementia or with MRI-contraindications were excluded. A random subset of 261 was invited out of 330 individuals who were eligible for the study; 135 agreed to participate. These had higher levels of education than nonparticipants (p=0.01). All participants gave informed consent and the study was approved by the Medical Ethics Committee.

Brain imaging was done on a 1.5-T MRI scanner (Signa Excite II, General Electric Healthcare, Milwaukee, WI, USA) with use of previously described protocols.[18] Volumes of WML were obtained by an automatic brain-tissue segmentation method.[19] To take into account differences in headsize, these volumes were analyzed as percentage of intracranial volume. Lacunes and microbleeds were rated by two trained reviewers; when there was no consensus a neuroradiologist decided. Previous agreement rates in our group with the same reviewers and neuroradiologist were good (κ =0.87 (intraobserver) and κ =0.85 (interobserver)).[14] MRI-scans could not be acquired in 4 participants because of physical constraints and 2 persons were excluded from analyses (both had a large incidentally discovered brain tumor). In total, complete information was available for 129 individuals. Cognitive function was assessed with a word learning test, the Trail Making Test (TMT), the Stroop Color and Word (CW-) test, verbal fluency tests, and the block-design subtest of WAISIII.[20] From the word learning test, we derived four scores: working memory, learning, delayed recall, and recognition. From the TMT, a ratio score was computed: time on part B divided by time on part A. This was also done for the Stroop CW-test (time on card III divided by time on card II).

Blood pressure was measured by a mercury-based sphygmomanometer. Blood was taken for estimation of levels of total cholesterol, high-density lipoprotein cholesterol (HDL-c) and low-density lipoprotein cholesterol (LDL-c).[20] Samples were genotyped for ADD1-Gly460Trp, AGT-M235T, AGTR1-C573T, APOE (rs429358, rs7412), and SORL1 (rs668387, rs689021, rs641120, rs1699102, rs3824968, rs2282649, rs1010159) with Taqman allelic discrimination Assays-By-Design (Applied Biosystems, Foster City, CA).[9,11,20]

To analyze the association of genotypes with WML volume, lacunes and microbleeds, regression models were used. White matter lesion volume was transformed by taking the natural logarithm. Haplotypes were derived with the software package SimWalk2 [21]; these haplotypes were considered as a fixed factor in a general linear model. Student's T-test and Chi-square statistics were used to perform general descriptive statistics (version 15.0 of SPSS). To compare baseline characteristics, we defined CSVD as a WML volume greater than the 75th percentile, or presence of lacunes or microbleeds. Analyses were adjusted for age and sex, and p-values were adjusted for family relationship. We corrected for multiple testing by performing 10.000 permutations.

Results

The mean age of the participants was 64.5 (\pm 4.6) years and 52.7% were female. In this hypertensive population, WML were seen in all participants (median, 3.11 mL; interquartile range, 1.51-6.50 mL). Lacunes were present in 20 individuals (15.5%), and microbleeds in 30 individuals (23.3%) with lobar location in 63%. Those with WML greater than the 75th percentile more often had lacunes (p<0.001) and microbleeds (p=0.003). Age was associated with larger WML volumes (p<0.001) and presence of microbleeds (p=0.02). Other associations between CSVD and cardiovascular risk factors were not observed (Table 1).

Individuals with CVSD versus individuals without CVSD							
	CSVD -	CSVD +	p-value				
Number	77	52					
Female %	51.9	53.8	0.49				
Age	63.3 (4.5)	66.3 (4.1)	<0.001				
Systolic blood pressure (mmHg)	143.8 (18.6)	149.5 (17.3)	0.08				
Diastolic blood pressure (mmHg)	83.2 (10.0)	84.5 (9.2)	0.45				
BMI	29.5 (4.7)	28.6 (3.6)	0.29				
Cholesterol (mmol/L)	5.2 (1.1)	5.2 (1.2)	0.89				
HDL-C (mmol/L)	1.5 (0.3)	1.3 (0.3)	0.32				
LDL-C (mmol/L)	3.4 (1.0)	3.3 (1.0)	0.65				
Creatinine (mmol/L)	73.3 (20.3)	72.5 (17.9)	0.88				
Diabetes (%)	15.6	15.4	0.98				
Current smokers (%)	71.4	73.1	0.50				
Use of alcohol (%)	62.3	67.3	0.58				
Depressive symptoms† (%)	16.9	19.2	0.82				
Education (% with primary education)	32.5	36.5	0.71				

Values are presented as mean (±sd). CVSD: Cerebral small vessel disease, defined as wml ≥ P75 or lacune or cerebral microbleeds. BMI: Body Mass Index † defined as a score on CES-D ≥ 16 and/or a score of ≥ 11 on HADS-D and/or use of antidepressants

Table 1

The results of the association analyses between the polymorphisms and WML volume, lacunes and microbleeds are shown in Table 2. Genotype frequencies of ADD1, AGT and AGTR1 were not associated with WML volume, lacunes or microbleeds. Individuals carrying two copies of the APOE ϵ 4 allele had more WML, lacunes and microbleeds than those without the ϵ 4 allele; however, the effects were small. This difference was significant for lacunes (p=0.04).

The sortilin-related receptor gene was consistently associated with the presence of microbleeds. Individuals carrying two copies of the variant allele of the 4 SNPs located at the 3'-end of the gene (rs1699102, rs3824968, rs2282649, rs1010159) more often had microbleeds (highest odds ratio, 6.87; 95% Cl, 1.78-26.44). In this hypertensive population, these SNPs were also associated with cognitive function. The SNPs rs1699102 (p=0.001), rs3824968 (p=0.004), rs2282649 (p=0.001), and rs1010159 (p=0.002) were all associated with TMT-ratio. The haplotype analysis of the SORL1 SNPs also showed association with microbleeds (p=0.05), most significantly when the haplotypes were based on rs1699102, rs3824968, rs2282649 and rs1010159 only (p=0.03). None of the SORL1 SNPs showed evidence for association with WML or lacunes. Genotype frequencies for the related SNPs were similar for participants and nonparticipants.

Association of AGT, AGTR1, ADD1, APOE and SORL1 with WML, microbleeds and lacunar infarcts

Gene	Genotype	WML‡			Lacunar infarcts				Microbleeds			
		N	Mean difference (95% CI)	Р	Absent	Present	OR (95% CI)	Р	Absent	Present	OR (95% CI)	P
AGT	MM	33	ref		28	5	ref		26	7	ref	
	MT	64	-0.32 (-0.73 - 0.08)	0.12	52	12	1.32 (0.40-4.37)	0.65	48	16	1.37 (0.47-3.97)	0.57
	TT	23	-0.25 (-0.76 - 0.26)	0.33	21	2	0.57 (0.10-3.34)	0.53	18	5	1.19 (0.31-4.57)	0.80
AGTR1	TT	24	ref		20	4	ref		20	4	ref	
	CT	53	-0.10 (-0.57 - 0.36)	0.66	46	7	0.71 (0.18-2.78)	0.62	38	15	1.94 (0.55-6.86)	0.30
	CC	42	0.05 (-0.44 - 0.54)	0.84	35	7	1.19 (0.29-4.82)	0.81	33	9	1.60 (0.41-6.20)	0.50
ADD1	GG	77	ref		63	14	ref		54	23	ref	
	GT/TT	46	-0.22 (-0.56 - 0.12)	0.21	41	5	0.55 (0.18-1.66)	0.29	39	7	0.41 (0.16-1.07)	0.07
APOE	0 copies ɛ4	78	ref		66	12	ref		61	17	ref	
	1 copy ε4	39	-0.01 (-0.37 - 0.36)	0.97	36	3	0.39 (0.10-1.52)	0.18	30	9	0.96 (0.37-2.48)	0.94
	2 copy ε4	12	0.31 (-0.27 - 0.88)	0.29	7	5	4.77 (1.18-19.32)	0.04	8	4	1.99 (0.50-7.94)	0.33
SORL1 rs668387	CC	30	ref		26	4	ref		22	8	ref	
	CT	62	-0.04 (-0.45 - 0.38)	0.86	52	10	1.15 (0.32-4.11)	0.83	52	10	0.46 (0.15-1.36)	0.16
	TT	31	-0.11 (-0.58 - 0.37)	0.66	25	6	1.32 (0.32-5.42)	0.70	20	11	1.28 (0.41.4.00)	0.67
SORL1 rs689021	GG	26	ref		22	4	ref		19	7	ref	
	AG	63	-0.07 (-0.51 - 0.36)	0.74	54	9	0.85 (0.23-3.10)	0.80	52	11	0.50 (0.16-1.54)	0.23
	AA	31	-0.18 (-0.68 - 0.32)	0.47	25	6	1.12 (0.27-4.64)	0.88	21	10	1.06 (0.32-3.51)	0.92
SORL1 rs641120	CC	29	ref		25	4	ref		21	8	ref	
	CT	61	-0.07 (-0.49 - 0.35)	0.74	52	9	0.99 (0.27-3.59)	0.99	51	10	0.44 (0.15-1.32)	0.14
	TT	31	-0.18 (-0.66 - 0.30)	0.47	25	6	1.26 (0.31-5.18)	0.73	21	10	1.01 (0.32-3.22)	0.99
SORL1 rs1699102	TT	49	ref		40	9	ref		42	7	ref	
	CT	55	0.15 (-0.22 - 0.51)	0.43	48	7	0.65 (0.22-1.93)	0.34	43	12	1.76 (0.61-5.07)	0.30
	CC	17	0.15 (-0.37 - 0.68)	0.56	15	2	0.60 (0.11-3.20)	0.55	9	8	6.81 (1.79-25.97)	0.005
SORL1 rs3824968	TT	51	ref		41	10	ref		42	9	ref	
	AT	53	0.06 (-0.30 - 0.41)	0.75	47	6	0.50 (0.16-1.52)	0.22	42	11	1.19 (0.43-3.30)	0.74
	AA	15	0.28 (-0.25 - 0.81)	0.30	13	2	0.74 (0.14-3.92)	0.72	8	7	5.90 (1.54-22.70)	0.01
SORL1 rs2282649	CC	56	ref		47	9	ref		47	9	ref	
	CT	49	0.13 (-0.24 - 0.49)	0.49	42	7	0.88 (0.30-2.63)	0.82	38	11	1.60 (0.58-4.46)	0.37
	TT	15	0.32 (-0.22 - 0.87)	0.24	13	2	0.96 (0.18-5.15)	0.96	8	7	6.87 (1.78-26.44)	0.005
SORL1 rs1010159	TT	52	ref		43	9	ref		44	8	ref	
	CT	57	0.30 (-0.05 - 0.65)	0.10	48	9	0.89 (0.32-2.49)	0.83	42	15	2.05 (0.76-5.49)	0.16
	CC	20	0.21 (-0.28 - 0.69)	0.40	18	2	0.65 (0.12-3.38)	0.61	13	7	4.17 (1.18-14.70)	0.03

WML: white matter lesion volume, P: p-value adjusted for age, sex and family-relationship. ref: reference genotype. ‡ natural log transformed variable

Table 2

Discussion

This study shows that two genes involved in the amyloidogenic pathway and previously described in AD, APOE and SORL1, were associated with CSVD. The presence of two APOE-ɛ4 alleles was associated with lacunes. Given the role of apolipoprotein in the brain, this finding suggests involvement of beta-amyloid clearance in the pathogenesis of lacunes.[22] Alternatively, the association in this hypertensive population might be explained by the effect of APOE on cardiovascular factors.[20]

Our observation that SORL1 was associated with microbleeds has not been previously reported. This gene regulates APP processing and SORL1 deficiency leads to increased levels of betaamyloid and enhances amyloid pathology in the brain.[16] In our study, microbleeds were predominantly located in lobar brain regions, implying that most of the lesions resulted from amyloid angiopathy.[14] In amyloid angiopathy-related AD, the majority of plaques is centered on vessel walls or in the immediate perivascular regions.[23] The relationship between SORL1 and microbleeds is of interest, because of increasing evidence that beta-amyloid plays a role in neurodegeneration through perivascular interstitial fluid drainage.[24] Through its role in microbleeds, SORL1, may link to dementia.[1] In line with this hypothesis, we found that the SNPs associated with microbleeds were marginally associated with cognitive function. This indirect role of SORL1 in plaque formation may also explain why the effects of the gene are modest in AD (http://www.Alzgene.org).

The genes involved in blood pressure regulation were not associated with CVSD. Earlier findings have been inconsistent.[9,11,25] The lack of an association in our study might be due to the fact that all participants were hypertensive, and that the blood pressure ranges were small. Indeed, blood pressure itself was not significantly associated with lesions.

The strength of our study is the population-based design, performed in a homogenous group of middle-aged individuals with hypertension derived from a genetically isolated population, which increases statistical power of genetic association studies. The size of the study is limited, however, and allows mainly detecting genetic variants with large effects. The association of SORL1 with microbleeds therefore remains to be confirmed. A question that remains to be answered in larger populations is whether hypertension interacts with SORL1 in the association with microbleeds, which might possibly be an alternative interpretation of our results.

The observed association of SORL1 with microbleeds in our study suggests that the amyloid cascade is involved in the etiology of microbleeds in populations with hypertension.

References

- Werring DJ, Frazer DW, Coward LJ, Losseff NA, Watt H, Cipolotti L, Brown MM, Jager HR. Cognitive dysfunction in patients with cerebral microbleeds on T2*-weighted gradient-echo MRI. Brain 2004;127:2265-75.
- Kuller LH, Shemanski L, Manolio T, Haan M, Fried L, Bryan N, Burke GL, Tracy R, Bhadelia R. Relationship between ApoE, MRI findings, and cognitive function in the Cardiovascular Health Study. Stroke 1998;29:388-98.
- 3. Vermeer SE, Longstreth WT, Jr., Koudstaal PJ. Silent brain infarcts: a systematic review. Lancet Neurol 2007;6:611-9.
- Kato H, Izumiyama M, Izumiyama K, Takahashi A, Itoyama Y. Silent cerebral microbleeds on T2*-weighted MRI: correlation with stroke subtype, stroke recurrence, and leukoaraiosis. Stroke 2002;33:1536-40.
- Jagust WJ, Zheng L, Harvey DJ, Mack WJ, Vinters HV, Weiner MW, Ellis WG, Zarow C, Mungas D, Reed BR, Kramer JH, Schuff N, DeCarli C, Chui HC. Neuropathological basis of magnetic resonance images in aging and dementia. Ann Neurol 2008;63:72-80.
- Fazekas F, Kleinert R, Roob G, Kleinert G, Kapeller P, Schmidt R, Hartung HP. Histopathologic analysis of foci of signal loss on gradient-echo T2*-weighted MR images in patients with spontaneous intracerebral hemorrhage: evidence of microangiopathy-related microbleeds. AJNR Am J Neuroradiol 1999;20:637-42.
- 7. Chen YW, Gurol ME, Rosand J, Viswanathan A, Rakich SM, Groover TR, Greenberg SM, Smith EE. Progression of white matter lesions and hemorrhages in cerebral amyloid angiopathy. Neurology 2006;67:83-7.
- 8. Manunta P, Burnier M, D'Amico M, Buzzi L, Maillard M, Barlassina C, Lanella G, Cusi D, Bianchi G. Adducin polymorphism affects renal proximal tubule reabsorption in hypertension. Hypertension 1999;33:694-7.
- 9. van Rijn MJ, Bos MJ, Isaacs A, Yazdanpanah M, Arias-Vasquez A, Stricker BH, Klungel OH, Oostra BA, Koudstaal PJ, Witteman JC, Hofman A, Breteler MM, van Duijn CM. Polymorphisms of the renin-angiotensin system are associated with blood pressure, atherosclerosis and cerebral white matter pathology. J Neurol Neurosurg Psychiatry 2007;78:1083-7.
- Takami S, Imai Y, Katsuya T, Ohkubo T, Tsuji I, Nagai K, Satoh H, Hisamichi S, Higaki J, Ogihara T. Gene polymorphism of the renin-angiotensin system associates with risk for lacunar infarction. The Ohasama study. Am J Hypertens 2000;13:121-7.
- 11. van Rijn MJ, Bos MJ, Yazdanpanah M, Isaacs A, Arias-Vasquez A, Koudstaal PJ, Hofman A, Witteman JC, van Duijn CM, Breteler MM. Alpha-adducin polymorphism, atherosclerosis, and cardiovascular and cerebrovascular risk. Stroke 2006;37:2930-4.
- Lind J, Persson J, Ingvar M, Larsson A, Cruts M, Van Broeckhoven C, Adolfsson R, Backman L, Nilsson LG, Petersson KM, Nyberg L. Reduced functional brain activity response in cognitively intact apolipoprotein E epsilon4 carriers. Brain 2006;129:1240-8.
- 13. Hirono N, Yasuda M, Tanimukai S, Kitagaki H, Mori E. Effect of the apolipoprotein E epsilon4 allele on white matter hyperintensities in dementia. Stroke 2000;31:1263-8.
- 14. Vernooij MW, van der Lugt A, Ikram MA, Wielopolski PA, Niessen WJ, Hofman A, Krestin GP, Breteler MM. Prevalence and risk factors of cerebral microbleeds: the Rotterdam Scan Study. Neurology 2008;70:1208-14.
- 15. Andersen OM, Reiche J, Schmidt V, Gotthardt M, Spoelgen R, Behlke J, von Arnim CA, Breiderhoff T, Jansen P, Wu X, Bales KR, Cappai R, Masters CL, Gliemann J, Mufson EJ, Hyman BT, Paul SM, Nykjaer A, Willnow TE. Neuronal sorting protein-related receptor sorLA/LR11 regulates processing of the amyloid precursor protein. Proc Natl Acad Sci U S A 2005;102:13461-6.
- 16. Rogaeva E, Meng Y, Lee JH, Gu Y, Kawarai T, Zou F, Katayama T, Baldwin CT, Cheng R, Hasegawa H, Chen F, Shibata N, Lunetta KL, Pardossi-Piquard R, Bohm C, Wakutani Y, Cupples LA, Cuenco KT, Green RC, Pinessi L, Rainero I, Sorbi S, Bruni A, Duara R, Friedland RP, Inzelberg R, Hampe W, Bujo H, Song YQ, Andersen OM, Willnow TE, Graff-Radford N, Petersen RC, Dickson D, Der SD, Fraser PE, Schmitt-Ulms G, Younkin S, Mayeux R, Farrer LA, St George-Hyslop P. The neuronal sortilin-related receptor SORL1 is genetically associated with Alzheimer disease. Nat Genet 2007;39:168-77.
- Cuenco KT, Lunetta KL, Baldwin CT, McKee AC, Guo J, Cupples LA, Green RC, St George-Hyslop PH, Chui H, Decarli C, Farrer LA, for the MSG. Association of Distinct Variants in SORL1 With Cerebrovascular and Neurodegenerative Changes Related to Alzheimer Disease. Arch Neurol 2008;65:1640-1648.
- Vernooij MW, Ikram MA, Tanghe HL, Vincent AJ, Hofman A, Krestin GP, Niessen WJ, Breteler MM, van der Lugt A. Incidental findings on brain MRI in the general population. N Engl J Med 2007;357:1821-8.
- 19. de Boer R, Vrooman HA, van der Lijn F, Vernooij MW, Ikram MA, van der Lugt A, Breteler MM, Niessen WJ. White matter lesion extension to automatic brain tissue segmentation on MRI. Neuroimage 2009;45:1151-61.
- 20. Liu F, Pardo LM, Schuur M, Sanchez-Juan P, Isaacs A, Sleegers K, de Koning I, Zorkoltseva IV, Axenovich TI, Witteman JC, Janssens AC, van Swieten JC, Aulchenko YS, Oostra BA, van Duijn CM. The apolipoprotein E gene and its age-specific effects on cognitive function. Neurobiology of Aging 2008.
- 21. Sobel E, Lange K. Descent graphs in pedigree analysis: applications to haplotyping, location scores, and marker-sharing statistics. Am J Hum Genet 1996;58:1323-37.
- 22. van Dijk EJ, Prins ND, Vermeer SE, Hofman A, van Duijn CM, Koudstaal PJ, Breteler MM. Plasma amyloid beta, apolipoprotein E, lacunar infarcts, and white matter lesions. Ann Neurol 2004;55:570-5.
- Kumar-Singh S, Julliams A, Nuydens R, Ceuterick C, Labeur C, Serneels S, Vennekens K, Van Osta P, Geerts H, De Strooper B, Van Broeckhoven C. In vitro studies of Flemish, Dutch, and wild-type beta-amyloid provide evidence for two-staged neurotoxicity. Neurobiol Dis 2002;11:330-40.
- 24. Weller RO, Subash M, Preston SD, Mazanti I, Carare RO. Perivascular drainage of amyloidbeta peptides from the brain and its failure in cerebral amyloid angiopathy and Alzheimer's disease. Brain Pathol 2008;18:253-66.

08 Genetic risk factors for cerebral small vessel disease in hypertensive patients from in a genetically isolated population

25. Gormley K, Bevan S, Markus HS. Polymorphisms in genes of the renin-angiotensin system and cerebral small vessel disease. Cerebrovasc Dis 2007;23:148-55.

09 The relation of the sortilin-related receptor gene to hippocampal volume and plasma amyloid beta levels in a family-based study of hypertensive patients

Abstract

Introduction

The sortilin-related receptor gene (SORL1) is an interesting candidate involved in Alzheimer's disease (AD). SORL1 consists of two functional domains, one functions in the cholesterol pathway and the other in the APP processing pathway. The gene has been associated to AD and cerebrovascular disease. We studied the relationship of SORL1 with hippocampal volume and plasma Aβ levels. *Materials & Methods*

The study was embedded in a family-based cohort sampled from a Dutch genetically isolated population. We selected individuals between 55 and 75 years of age with hypertension (N=128). Volumes of the hippocampi were assessed with MRI and non-fasting blood sample were taken to assess plasma A β levels. We studied the effect of 7 variants within SORL1 that were previously reported in AD, by performing the variable screening analysis under the polygenic model and haplotype analyses.

Results

Three variants located near the 3'-end of SORL1 were significantly associated to hippocampal volume. The 3-SNP haplotypes for rs1699102, rs3824968 and rs2282649 (CAT) and for rs3824968, rs2282649 and rs1010159 (ATC) were associated to higher hippocampal volumes on MRI. We did not find significant associations of single variants with plasma Aβ levels.

Conclusion

Variants near the 3'end of SORL1 are associated to hippocampal volume. Our findings need replication in larger cohorts.

Introduction

Neurodegenerative diseases like Alzheimer's disease (AD) are highly prevalent diseases in the elderly and a major burden on society [1,2]. Neuropathologically, AD is characterized by neuritic plaques and neurofibrillary tangles [3,4]. Neuritic plaques contain amyloid beta (A β) proteins, which are formed after proteolytic processing of the amyloid precursor protein (APP). A β proteins are present in fulllength species (β 40 and β 42) and in shorter amino-terminal truncated species (β n40 and β n42). The latter accounts for 60% of all A β species in pre-clinical AD stages [5]. A β pathology may also be present in brains of cognitively healthy elderly at postmortem and can visualized with PIB-PET during life [6,7]. The deposition and clearance of A β in the brain is related to A β levels in plasma, but the mechanism is not fully understood [8]. Both increased and decreased risks as well as no effects have been reported for A β 40 and A β 42 [9-14].

Another early biomarker of AD is hippocampal atrophy [15,16]. Although the hippocampus has a low A β load in non-AD brains [17], the correlation of hippocampal atrophy on MRI with A β CSF levels and with 11C-PIB uptake in PET studies of AD patients and healthy controls is high [18-20].

The heritability of plasma $A\beta$ levels and medial temporal lobe atrophy have both been estimated

around 60%, since no major genes have been identified yet, it is of interest to find genetic risk factors that influence the variability of these traits [21-25]. Previous studies have suggested overlap between genes affecting brain volumes and those affecting blood pressure [26], making it interesting to study genetic susceptibility in hypertensive patients.

In the current study, we focused on the sortilin-related receptor gene (SORL1), which is involved in the APP pathway. The SORL1 protein may act as a sorting receptor for APP [27,28] and variants in the SORL1 gene are associated to AD although the effect size is small and evidence is unstable [29-31]. We have recently found association of variants in SORL1 with cerebral microbleeds [32]. In addition to a role in APP processing, the SORL1 protein is related to low-density lipoprotein receptors and may be involved in atherosclerotic processes [33], which was also suggested by associations of variants in SORL1 with cerebrovascular disease and cerebral atrophy reported by a recent study [34]. The aim of the current study was to examine whether SORL1 was related to plasma $A\beta$ levels and hippocampal atrophy on MRI. We studied this in a hypertensive subset of the Erasmus Rucphen Family (ERF) study, which is a family-based study in a genetically isolated population.

Materials and Methods

Study population

The study was embedded in a population-based study in a genetically isolated population in the Netherlands: the Erasmus Rucphen Family (ERF)-Study. Participants are all descendents of a limited number of founders living in the 19th century. Extensive genealogical data is available to the year 1600 [35,36]. The current study was designed to find genetic risk factors for cerebral small vessel disease (CSVD). Considering hypertension as major risk factor for CSVD, we selected participants aged 55 to 75 years with hypertension to ensure a high prevalence of pathology. Hypertension was defined as systolic blood pressure \geq 160 and/or diastolic blood pressure \geq 100 and/or use of antihypertensive medication. Persons with a history of stroke or dementia or with MRI-contraindications were excluded. A random subset of 261 was invited out of 330 individuals who were eligible for the study; 135 agreed to participants. There was a small difference in completed years of education, which was higher for participants who had a mean level of 7.5 years compared to 9 years in non-participants (p=0.02). All participants gave informed consent and the study was approved by the Medical Ethics Committee at Erasmus MC University Medical Center.

Brain imaging

Brain imaging was done on a 1.5-T MRI scanner (Signa Excite II, General Electric Healthcare, Milwaukee, WI, USA) with use of a previously described protocol [37]. Volumes of the hippocampi, normal white matter (WM), white matter lesions (WML), grey matter (GM) and cerebrospinal fluid (CSF) were obtained by an automatic brain-tissue segmentation method which which was validated

previously [38,39]. The hippocampal formation included CA1 to CA4, the gyrus dentatus, and the subiculum. Volumes of the left and right hippocampus were measured with an automatic segmentation method based on intensity and regularity energy models. All samples were visually checked and when necessary manually segmented with specific software (http://www.bic.mni.mcgill.ca/software/) on coronal slides with continuous reference to all orientations. For analyses, the volumes of the left and right hippocampus were summed. To take into account differences in headsize, hippocampal volumes were analyzed as percentage of intracranial volume, which was defined as the total volume of WML, WM, GM and CSF. MRI-scans could not be acquired in 4 participants because of physical constraints and 2 persons were excluded from analyses (both had a large incidentally discovered brain tumor).

Aβ measurements

Non-fasting blood samples were obtained, which were immediately cooled on ice. From the samples plasma was extracted and stored at -80 °C. Plasma Aβ concentrations were measured using a fluorimetric bead-based immunoassay using xMAP® technology (Innogenetics®). We obtained Aβ40, Aβ42, and the truncated forms Aβn40 and Aβn42. For analyses, we also used Aβ42/Aβ40 ratio. In total, complete information was available for 128 individuals.

Covariates

Body weight and height were measured in centimeters and the body-mass index (BMI) was calculated from these measurements. Blood was taken for estimation of levels of total cholesterol, high-density lipoprotein cholesterol (HDL-c), low-density lipoprotein cholesterol (LDL-c) and creatinine in serum (Roche®). Glycosylated hemoglobin (HbA1c) was measured from full blood obtained in vacucontainers containing sodium citrate (Bio-Rad®) and analyzed using High-Performance Liquid Chromatography.

Genotyping

Genomic DNA was extracted from whole blood samples using the salting out method [40]. Samples were genotyped for APOE (rs429358, rs7412), and SORL1 (rs668387, rs689021, rs641120, rs1699102, rs3824968, rs2282649, rs1010159) with Taqman allelic discrimination Assays-By-Design (Applied Biosystems, Foster City, CA) [41].

The SNPs within SORL1 were chosen, because we considered them as the key SNPs from the original report [31].

Statistical analyses

General descriptive statistics were estimated with one-way ANOVA and Chi-square statistics as implemented in the software SPSS (version 15.0). The observed frequencies of the genotypes were tested for deviations from Hardy-Weinberg equilibrium (HWE) by exact test. All genotypes were in HWE (p>0.05).

We first estimated whether plasma A β levels were associated to hippocampal volume on MRI with linear regression models. Second, to analyze the association of the genotypes with hippocampal volume and plasma A β levels, we performed the variable screening analysis under the polygenic model using the SOLAR software version 4.1.0 [42]. This software allows adjusting for family-relationships taking into account the pedigree structure. To reduce computational time, the large pedigree was cut into smaller subpedigrees of 18 bitsize before analysis [43]. One person could not be linked to these subpedigrees. The analyses in SOLAR were adjusted for age, sex and inbreeding coefficient, which was estimated from the genealogical data of the complete pedigree of the population using the software PEDIG [44]. The genotypes were entered in the model as a covariate with the reference genotype group coded as 0, the heterozygote genotypes coded as 1 and the other genotype group coded as 2. For APOE, 0 was used for the $\epsilon 2/\epsilon 2$, $\epsilon 2/\epsilon 3$ and $\epsilon 3/\epsilon 3$ genotypes, 1 for the $\epsilon 2/\epsilon 4$ and $\epsilon 3/\epsilon 4$ genotypes and 2 for the $\epsilon 4/\epsilon 4$ genotype.

Haplotypes for SORL1 were derived with the software package SimWalk2 [45]. Haplotypes with a frequency >5% were implemented as binary factors in the models and analyzed in SOLAR. To adjust for multiple testing we used Bonferroni correction adjusting for 7 independent tests.

Results

Baseline characteristics of the study population are shown in Table 1. The mean age was 64.6 years and there were slightly more women than men. Hippocampal volume was not significantly associated to plasma levels of $A\beta$.

Descriptive characteristics of the study population										
N	128									
Sex (% women)	52.3%									
Age	64.6 (4.5)									
BMI	29.2 (4.3)									
HDL-c (mmol/L)	1.3 (0.3)									
Creatinine (µmol/L	73.0 (19.4)									
HbA1c (%)	5.8 (0.6)									
White matter lesion volume*	0.49 (0.56)									
APOE E4 carriers (%)	39.1%									

BMI: body-mass index, APOE: Apolipoprotein E gene. *percentage of intracranial volume

Table 1

The SORL1 gene was significantly associated with hippocampal volume (Table 2). Single SNP analyses showed that subjects with the minor genotype of 3 SNPs located near the 3'-end of SORL1 had significantly higher hippocampal volumes than subjects with the major genotype (Figure). Additional adjustment for the APOE genotype resulted in greater effect size and more significant p-values for these associations.



Regional plots depicting the associations of SORL1 with hippocampal volume (first) and Aß levels (second)

The x-axis depicts the physical position on the chromosome (kb); the y-axis depicts the minus log p-values.

Figure 1

Haplotype analyses revealed significant associations of the 3-SNP haplotypes for rs1699102, rs3824968 and rs2282649 (CAT) and for rs3824968, rs2282649 and rs1010159 (ATC) with higher hippocampal volumes (Table 3). These haplotypes were the second most frequent in our population (18.5% and 19.3% respectively). The p-value of the CAT-haplotype remained significant after adjustment for multiple testing. Additional adjustment for APOE genotype did not change these results. The haplotypes consisting of the 3 SNPs located at the 5'-end of SORL1 (rs668387, rs689021 and rs641120) were not significantly associated.

The SORL1 SNPs were not significantly associated with plasma A β levels (Table 4 and Figure). The results were similar when adjusting these analyses additionally for creatinine, high-density lipoprotein, HbA1c, BMI and white matter lesion volume (data not shown).

09 The relation of the sortilin-related receptor gene to hippocampal volume and plasma amyloid beta levels in a family-based study of hypertensive patients

Gene variant	Genotype	Hippocampal volume‡									
		N		Mean (SD)	P1	P2					
SORL1 rs668387	CC	36	CC	0.52 (0.03)	0.55	0.58					
	CT	60	CT	0.53 (0.06)							
	TT	31	TT	0.51 (0.06)							
SORL1 rs689021	GG	34	GG	0.51 (0.05)	0.80	0.83					
	AG	62	AG	0.53 (0.05)							
	AA	31	AA	0.51 (0.06)							
SORL1 rs641120	CC	36	CC	0.51 (0.05)	0.73	0.76					
	CT	60	CT	0.53 (0.05)							
	Π	31	TT	0.51 (0.06)							
SORL1 rs1699102	TT	55	TT	0.51 (0.05)	0.01	0.01					
	CT	55	CT	0.52 (0.05)							
	CC	17	CC	0.54 (0.06)							
SORL1 rs3824968	TT	59	TT	0.51 (0.05)	0.04	0.03					
	AT	53	AT	0.52 (0.06)							
	AA	15	AA	0.55 (0.03)							
SORL1 rs2282649	CC	63	CC	0.51 (0.05)	0.03	0.02					
	CT	49	CT	0.52 (0.06)							
	TT	15	TT	0.55 (0.03)							
SORL1 rs1010159	TT	51	TT	0.51 (0.04)	0.10	0.07					
	CT	56	CT	0.52 (0.07)							
	CC	20	CC	0.54 (0.04)							

P1: p-value adjusted for age, sex and family-relationship. P2: P1 additionally adjusted for APOE genotype. ‡ percentage of intracranial volume

Table 2

chapter 09

SNPs	Haplotype	Frequency % (N)	P*	Effect	P**
rs668387- rs689021- rs641120	CGC	37.8 (96)	0.61		0.63
	CAT	33.5 (85)	0.90	+	
	TAT	7.9 (20)	0.53	+	
	TGC	7.5 (15)	0.73	+	
rs1699102-rs3824968- rs2282649	TTC	53.5 (136)	0.21		0.16
	CAT	18.5 (47)	0.005	+	
	CAC	6.3 (16)	0.43	-	
	CTT	5.1 (13)	0.60	+	
	CTC	5.1 (13)	0.58	-	
rs3824968- rs2282649- rs1010159	тст	50.4 (128)	0.10		0.25
	ATC	19.3 (49)	0.05	+	
	TCC	8.3 (21)	0.53	+	
	TTC	5.9 (15)	0.60	+	
	ACT	5.9 (15)	0.74	+	

P: p-value adjusted for age, sex and family-relationship. ‡ percentage of intracranial volume. "p-value for haplotype; "p-value for region. Effect: direction of effect. The + sign refers to increased hippocampal volumes and the – sign to decreased hippocampal volumes.

Table 3

Gene variant	Genotype	Aβ40 (pg/mL)			Aβ42 (pg/mL)			Aβn40 (pg/mL)				Aβn42 (pg/mL)			Aβ42/Ab40	
		N	Mean (SD)	Р	Ν	Mean (SD)	Р	N	Mean (SD)	Р	N	Mean (SD)	Р	N	Mean (SD)	Р
SORL1 rs668387	CC	34	172.24 (43.60)	0.94	35	39.15 (14.75)	0.77	35	168.74 (39.21)	0.64	33	27.33 (6.14)	0.71	34	0.24 (0.08)	0.46
	CT	61	182.59 (43.07)		61	41.83 (11.97)		61	178.48 (40.71)		56	28.78 (7.80)		61	0.24 (0.07)	
	TT	31	175.15 (27.93)		31	43.11 (14.74)		31	174.31 (29.25)		30	26.76 (4.82)		31	0.25 (0.07)	
SORL1 rs689021	GG	32	175.12 (39.07)	0.85	33	41.59 (16.27)	0.65	33	175.63 (30.76)	0.59	33	28.13 (6.03)	0.43	32	0.25 (0.07)	0.99
	AG	63	182.50 (43.91)		63	41.33 (12.12)		63	176.95 (43.34)		57	28.43 (7.81)		63	0.24 (0.08)	
	AA	31	171.71 (31.88)		31	41.36 (13.17)		31	169.46 (32.43)		29	26.46 (4.86)		31	0.24 (0.06)	
SORL1 rs641120	CC	34	173.53 (40.92)	0.94	35	41.11 (16.33)	0.75	35	172.47 (36.82)	0.69	34	27.90 (6.09)	0.53	34	0.25 (0.09)	0.97
	CT	61	183.62 (42.92)		61	41.60 (11.91)		61	178.81 (40.79)		56	28.58 (7.80)		61	0.23 (0.07)	
	TT	31	171.71 (31.88)		31	41.36 (13.17)		31	169.46 (32.43)		29	26.46 (4.86)		31	0.24 (0.06)	
SORL1 rs1699102	TT	55	174.53 (40.52)	0.44	56	41.69 (14.38)	0.22	56	169.50 (38.34)	0.76	53	28.64 (6.72)	0.45	55	0.25 (0.07)	0.21
	CT	54	184.19 (37.86)		54	41.83 (13.40)		54	183.52 (37.21)		50	27.16 (6.45)		54	0.23 (0.09)	
	CC	17	169.34 (44.34)		17	39.13 (10.59)		17	164.42 (33.31)		16	27.51 (7.63)		17	0.24 (0.04)	
SORL1 rs3824968	TT	59	173.28 (40.10)	0.21	60	41.86 (14.24)	0.30	60	169.84 (38.12)	0.55	57	28.50 (6.47)	0.44	59	0.25 (0.06)	0.12
	AT	53	184.54 (38.27)		53	41.33 (13.27)		53	182.39 (37.24)		49	27.22 (6.69)		53	0.23 (0.09)	
	AA	14	172.81 (45.37)		14	39.76 (11.17)		14	167.16 (35.40)		13	27.55 (8.07)		14	0.24 (0.04)	
SORL1 rs2282649	CC	63	175.97 (42.13)	0.35	64	41.63 (14.10)	0.36	64	173.09 (38.96)	0.89	60	28.39 (6.38)	0.50	63	0.25 (0.07)	0.21
	CT	49	182.01 (35.91)		49	41.58 (13.38)		49	179.16 (36.98)		46	27.27 (6.84)		49	0.23 (0.09)	
	TT	14	172.81 (45.37)		14	39.76 (11.17)		14	167.16 (35.40)		13	27.55 (8.07)		15	0.24 (0.04)	
SORL1 rs1010159	TT	52	175.92 (43.42)	0.55	52	41.75 (12.71)	0.27	52	170.69 (39.69)	0.95	48	27.88 (6.70)	0.90	52	0.24 (0.07)	0.31
	CT	56	181.93 (36.84)		56	42.55 (14.05)		56	181.60 (37.46)		53	27.94 (6.70)		56	0.24 (0.09)	
	CC	18	171.57 (40.20)		19	37.09 (13.46)		19	165.87 (30.75)		18	27.61 (7.17)		18	0.23 (0.04)	

P: p-value adjusted for age, sex and family-relationship. SD: standard deviation

Table 4

Discussion

The present study shows that variants in SORL1 are associated to hippocampal volume with a protective effect of the AT-haplotype consisting of rs3824968 and rs2282649. This is in line with the results of a previous study reporting associations of SORL1 with cerebrovascular disease and brain atrophy [34]. Although not significant in our study, Cuenco et al. also found that lower hippocampal volumes were associated with the TC-haplotypes of these SNPs. Since neuronal loss in AD can be seen as hippocampal atrophy on MRI, these SORL1 variants may exert their effect on the processing of APP [46,47], however, the cross-sectional design of the current study limits the interpretation of underlying pathophysiological mechanisms. The haplotypes constructed from the SNPs at the 5'-end of SORL1 were not significantly associated to hippocampal volume, but the direction of effect was similar to the direction seen in AD, in which the CGC-haplotype was associated to an increased risk and the TAT-haplotype to a decreased risk of AD [31]. The opposite effects of these haplotypes on the risk of cerebrovascular disease that were found in a previous study [34] may suggest that one of the functional domains in SORL1 could be involved in the cholesterol pathway and the other in the APP processing pathway [48]. This may explain the heterogeneous findings in AD and MRI-traits (http:// www.Alzgene.org) [30,34].

Our study did not find an association of SORL1 with plasma A β . One explanation could be that plasma A β , although linked to brain A β may not accurately reflect brain A β pathology. Studies measuring A β levels in CSF or postmortem studies may provide more accurate measurements of brain A β pathology. Also, the effect of SORL1 on AD pathology may be contributable to other mechanisms in the APP-pathway than A β generation. Plasma A β may be an indicator of vascular damage, rather than neurodegeneration, since it has been associated to white matter lesions and cerebral small vessel disease [49,50]. Increased levels of plasma A β could be a cause of microvascular damage, but could also be a consequence if damage to small vessels results in leakage of A β from the brain to the circulation. The effect of SORL1 on CSF A β levels warrants further study and may elucidate the role of SORL1 and A β in cerebrovascular and neurodegenerative disease processes.

Our observation of a strong association of SORL1 with hippocampal volume in hypertensive patients may reflect a possible interaction of SORL1 with hypertension, which is also supported by previous findings of a stronger effect of SORL1 with AD in patients with cerebrovascular disease [34]. The pathophysiology of AD is likely multifactorial with vascular as well as neurodegenerative factors influencing the disease process [51]. Both cerebrovascular disease and hippocampal atrophy are associated with an increased risk of dementia and cognitive dysfunction [52-55]. This may be due to interaction between vascular and neurodegenerative processes in their effect on dementia [56], as has been demonstrated by increased white matter lesion volumes in patients with greater hippocampal atrophy [57].

The strength of our study is that it was performed in a homogenous group of middle-aged individuals with hypertension derived from a genetically isolated population, which increases statistical power of genetic association studies. The size of the study is limited, however, and allows mainly detecting genetic variants with large effects. A question that remains to be answered in larger populations is whether hypertension interacts with SORL1.

To summarize, variants near the 3'end of SORL1 are associated to hippocampal volume on MRI. Our findings need replication in larger cohorts.

References

- 1. Cummings JL. Alzheimer's disease. N Engl J Med 2004;351:56-67.
- Caracciolo B, Palmer K, Monastero R, Winblad B, Backman L, Fratiglioni L. Occurrence of cognitive impairment and dementia in the community. A 9-year-long prospective study. Neurology 2008.
- 3. NIA-Working-Group. Consensus recommendations for the postmortem diagnosis of Alzheimer's disease. The National Institute on Aging, and Reagan Institute Working Group on Diagnostic Criteria for the Neuropathological Assessment of Alzheimer's Disease. Neurobiol Aging 1997;18:S1-2.
- Mirra SS, Heyman A, McKeel D, Sumi SM, Crain BJ, Brownlee LM, Vogel FS, Hughes JP, van Belle G, Berg L. The Consortium to Establish a Registry for Alzheimer's Disease (CERAD). Part II. Standardization of the neuropathologic assessment of Alzheimer's disease. Neurology 1991;41:479-86.
- Sergeant N, Bombois S, Ghestem A, Drobecq H, Kostanjevecki V, Missiaen C, Wattez A, David JP, Vanmechelen E, Sergheraert C, Delacourte A. Truncated beta-amyloid peptide species in pre-clinical Alzheimer's disease as new targets for the vaccination approach. J Neurochem 2003;85:1581-91.
- 6. Fotenos AF, Mintun MA, Snyder AZ, Morris JC, Buckner RL. Brain volume decline in aging: evidence for a relation between socioeconomic status, preclinical Alzheimer disease, and reserve. Arch Neurol 2008;65:113-20.
- Bennett DA, Schneider JA, Arvanitakis Z, Kelly JF, Aggarwal NT, Shah RC, Wilson RS. Neuropathology of older persons without cognitive impairment from two community-based studies. Neurology 2006;66:1837-44.
- DeMattos RB, Bales KR, Cummins DJ, Paul SM, Holtzman DM. Brain to plasma amyloidbeta efflux: a measure of brain amyloid burden in a mouse model of Alzheimer's disease. Science 2002;295:2264-7.
- 9. van Oijen M, Hofman A, Soares HD, Koudstaal PJ, Breteler MM. Plasma Abeta(1-40) and Abeta(1-42) and the risk of dementia: a prospective case-cohort study. Lancet Neurol 2006;5:655-60.
- 10. Mayeux R, Honig LS, Tang MX, Manly J, Stern Y, Schupf N, Mehta PD. Plasma A[beta]40 and A[beta]42 and Alzheimer's disease: relation to age, mortality, and risk. Neurology 2003;61:1185-90.
- 11. Graff-Radford NR, Crook JE, Lucas J, Boeve BF, Knopman DS, Ivnik RJ, Smith GE, Younkin LH, Petersen RC, Younkin SG. Association of low plasma Abeta42/Abeta40 ratios with increased imminent risk for mild cognitive impairment and Alzheimer disease. Arch Neurol 2007;64:354-62.
- 12. Pomara N, Willoughby LM, Sidtis JJ, Mehta PD. Selective reductions in plasma Abeta 1-42 in healthy elderly subjects during longitudinal follow-up: a preliminary report. Am J Geriatr Psychiatry 2005;13:914-7.

- 13. Fukumoto H, Tennis M, Locascio JJ, Hyman BT, Growdon JH, Irizarry MC. Age but not diagnosis is the main predictor of plasma amyloid beta-protein levels. Arch Neurol 2003;60:958-64.
- 14. Lambert JC, Schraen-Maschke S, Richard F, Fievet N, Rouaud O, Berr C, Dartigues JF, Tzourio C, Alperovitch A, Buee L, Amouyel P. Association of plasma amyloid beta with risk of dementia: the prospective Three-City Study. Neurology 2009;73:847-53.
- 15. Jack CR, Jr., Petersen RC, Xu YC, Waring SC, O'Brien PC, Tangalos EG, Smith GE, Ivnik RJ, Kokmen E. Medial temporal atrophy on MRI in normal aging and very mild Alzheimer's disease. Neurology 1997;49:786-94.
- Cuenco KT, Green RC, Zhang J, Lunetta K, Erlich PM, Cupples LA, Farrer LA, DeCarli C. Magnetic resonance imaging traits in siblings discordant for Alzheimer disease. J Neuroimaging 2008;18:268-75.
- 17. Price JL, Morris JC. Tangles and plaques in nondemented aging and "preclinical" Alzheimer's disease. Ann Neurol 1999;45:358-68.
- 18. Fagan AM, Head D, Shah AR, Marcus D, Mintun M, Morris JC, Holtzman DM. Decreased cerebrospinal fluid Abeta(42) correlates with brain atrophy in cognitively normal elderly. Ann Neurol 2009;65:176-83.
- 19. Frisoni GB, Lorenzi M, Caroli A, Kemppainen N, Nagren K, Rinne JO. In vivo mapping of amyloid toxicity in Alzheimer disease. Neurology 2009;72:1504-11.
- 20. Bourgeat P, Chetelat G, Villemagne VL, Fripp J, Raniga P, Pike K, Acosta O, Szoeke C, Ourselin S, Ames D, Ellis KA, Martins RN, Masters CL, Rowe CC, Salvado O. Beta-amyloid burden in the temporal neocortex is related to hippocampal atrophy in elderly subjects without dementia. Neurology 2010;74:121-7.
- Ertekin-Taner N, Graff-Radford N, Younkin LH, Eckman C, Adamson J, Schaid DJ, Blangero J, Hutton M, Younkin SG. Heritability of plasma amyloid beta in typical late-onset Alzheimer's disease pedigrees. Genet Epidemiol 2001;21:19-30.
- 22. Turner ST, Fornage M, Jack CR, Jr., Mosley TH, Knopman DS, Kardia SL, Boerwinkle E, de Andrade M. Genomic susceptibility Loci for brain atrophy, ventricular volume, and leukoaraiosis in hypertensive sibships. Arch Neurol 2009;66:847-57.
- 23. Lunetta KL, Erlich PM, Cuenco KT, Cupples LA, Green RC, Farrer LA, Decarli C. Heritability of magnetic resonance imaging (MRI) traits in Alzheimer disease cases and their siblings in the MIRAGE study. Alzheimer Dis Assoc Disord 2007;21:85-91.
- 24. Kaymaz N, van Os J. Heritability of structural brain traits an endophenotype approach to deconstruct schizophrenia. Int Rev Neurobiol 2009;89:85-130.
- 25. Peper JS, Brouwer RM, Boomsma DI, Kahn RS, Hulshoff Pol HE. Genetic influences on human brain structure: a review of brain imaging studies in twins. Hum Brain Mapp 2007;28:464-73.
- 26. Turner ST, Fornage M, Jack CR, Jr., Mosley TH, Kardia SL, Boerwinkle E, de Andrade M. Genomic susceptibility loci for brain atrophy in hypertensive sibships from the GENOA study. Hypertension 2005;45:793-8.

- 27. Andersen OM, Reiche J, Schmidt V, Gotthardt M, Spoelgen R, Behlke J, von Arnim CA, Breiderhoff T, Jansen P, Wu X, Bales KR, Cappai R, Masters CL, Gliemann J, Mufson EJ, Hyman BT, Paul SM, Nykjaer A, Willnow TE. Neuronal sorting protein-related receptor sorLA/LR11 regulates processing of the amyloid precursor protein. Proc Natl Acad Sci U S A 2005;102:13461-6.
- 28. Dodson SE, Andersen OM, Karmali V, Fritz JJ, Cheng D, Peng J, Levey AI, Willnow TE, Lah JJ. Loss of LR11/SORLA enhances early pathology in a mouse model of amyloidosis: evidence for a proximal role in Alzheimer's disease. J Neurosci 2008;28:12877-86.
- 29. Liu F, Ikram MA, Janssens AC, Schuur M, de Koning I, Isaacs A, Struchalin M, Uitterlinden AG, den Dunnen JT, Sleegers K, Bettens K, Van Broeckhoven C, van Swieten J, Hofman A, Oostra BA, Aulchenko YS, Breteler MM, van Duijn CM. A study of the SORL1 gene in Alzheimer's disease and cognitive function. J Alzheimers Dis 2009;18:51-64.
- 30. Bertram L, McQueen MB, Mullin K, Blacker D, Tanzi RE. Systematic meta-analyses of Alzheimer disease genetic association studies: the AlzGene database. Nat Genet 2007;39:17-23.
- 31. Rogaeva E, Meng Y, Lee JH, Gu Y, Kawarai T, Zou F, Katayama T, Baldwin CT, Cheng R, Hasegawa H, Chen F, Shibata N, Lunetta KL, Pardossi-Piquard R, Bohm C, Wakutani Y, Cupples LA, Cuenco KT, Green RC, Pinessi L, Rainero I, Sorbi S, Bruni A, Duara R, Friedland RP, Inzelberg R, Hampe W, Bujo H, Song YQ, Andersen OM, Willnow TE, Graff-Radford N, Petersen RC, Dickson D, Der SD, Fraser PE, Schmitt-Ulms G, Younkin S, Mayeux R, Farrer LA, St George-Hyslop P. The neuronal sortilin-related receptor SORL1 is genetically associated with Alzheimer disease. Nat Genet 2007;39:168-77.
- 32. Schuur M, van Swieten JC, Schol-Gelok S, Ikram MA, Vernooij MW, Liu F, Isaacs A, de Boer R, de Koning I, Niessen WJ, Vrooman H, Oostra BA, van der Lugt A, Breteler MMB, van Duijn CM. Genetic risk factors for cerebral small vessel disease in hypertensive patients from in a genetically isolated population. J Neurol Neurosurg Psychiatry In press.
- Ohwaki K, Bujo H, Jiang M, Yamazaki H, Schneider WJ, Saito Y. A secreted soluble form of LR11, specifically expressed in intimal smooth muscle cells, accelerates formation of lipidladen macrophages. Arterioscler Thromb Vasc Biol 2007;27:1050-6.
- Cuenco KT, Lunetta KL, Baldwin CT, McKee AC, Guo J, Cupples LA, Green RC, St George-Hyslop PH, Chui H, Decarli C, Farrer LA, for the MSG. Association of Distinct Variants in SORL1 With Cerebrovascular and Neurodegenerative Changes Related to Alzheimer Disease. Arch Neurol 2008;65:1640-1648.
- 35. Pardo LM, MacKay I, Oostra B, van Duijn CM, Aulchenko YS. The effect of genetic drift in a young genetically isolated population. Annals of Human Genetics 2005;69:288-95.
- Aulchenko YS, Heutink P, Mackay I, Bertoli-Avella AM, Pullen J, Vaessen N, Rademaker TA, Sandkuijl LA, Cardon L, Oostra B, van Duijn CM. Linkage disequilibrium in young genetically isolated Dutch population. Eur J Hum Genet 2004;12:527-34.
- 37. Vernooij MW, Ikram MA, Tanghe HL, Vincent AJ, Hofman A, Krestin GP, Niessen WJ, Breteler MM, van der Lugt A. Incidental findings on brain MRI in the general population. N Engl J Med 2007;357:1821-8.

- de Boer R, Vrooman HA, van der Lijn F, Vernooij MW, Ikram MA, van der Lugt A, Breteler MM, Niessen WJ. White matter lesion extension to automatic brain tissue segmentation on MRI. Neuroimage 2009;45:1151-61.
- 39. van der Lijn F, den Heijer T, Breteler MM, Niessen WJ. Hippocampus segmentation in MR images using atlas registration, voxel classification, and graph cuts. Neuroimage 2008;43:708-20.
- 40. Miller SA, Dykes DD, Polesky HF. A simple salting out procedure for extracting DNA from human nucleated cells. Nucleic Acids Res 1988;16:1215.
- 41. Liu F, Pardo LM, Schuur M, Sanchez-Juan P, Isaacs A, Sleegers K, de Koning I, Zorkoltseva IV, Axenovich TI, Witteman JC, Janssens AC, van Swieten JC, Aulchenko YS, Oostra BA, van Duijn CM. The apolipoprotein E gene and its age-specific effects on cognitive function. Neurobiology of Aging 2008.
- 42. Almasy L, Blangero J. Multipoint quantitative-trait linkage analysis in general pedigrees. Am J Hum Genet 1998;62:1198-211.
- 43. Liu F, Kirichenko A, Axenovich TI, van Duijn CM, Aulchenko YS. An approach for cutting large and complex pedigrees for linkage analysis. Eur J Hum Genet 2008;16:854-60.
- 44. Boichard D. PEDIG: a FORTRAN package for pedigree analysis studied for large populations. Proceeding of the 7th Word Congress of Genet Appl Livest Prod 2002:28.
- 45. Sobel E, Lange K. Descent graphs in pedigree analysis: applications to haplotyping, location scores, and marker-sharing statistics. Am J Hum Genet 1996;58:1323-37.
- 46. Bobinski M, de Leon MJ, Wegiel J, Desanti S, Convit A, Saint Louis LA, Rusinek H, Wisniewski HM. The histological validation of post mortem magnetic resonance imagingdetermined hippocampal volume in Alzheimer's disease. Neuroscience 2000;95:721-5.
- 47. Mortimer JA, Gosche KM, Riley KP, Markesbery WR, Snowdon DA. Delayed recall, hippocampal volume and Alzheimer neuropathology: findings from the Nun Study. Neurology 2004;62:428-32.
- 48. Hermey G. The Vps10p-domain receptor family. Cell Mol Life Sci 2009;66:2677-89.
- Gurol ME, Irizarry MC, Smith EE, Raju S, Diaz-Arrastia R, Bottiglieri T, Rosand J, Growdon JH, Greenberg SM. Plasma beta-amyloid and white matter lesions in AD, MCI, and cerebral amyloid angiopathy. Neurology 2006;66:23-9.
- 50. van Dijk EJ, Prins ND, Vermeer SE, Hofman A, van Duijn CM, Koudstaal PJ, Breteler MM. Plasma amyloid beta, apolipoprotein E, lacunar infarcts, and white matter lesions. Ann Neurol 2004;55:570-5.
- 51. Schneider JA, Boyle PA, Arvanitakis Z, Bienias JL, Bennett DA. Subcortical infarcts, Alzheimer's disease pathology, and memory function in older persons. Ann Neurol 2007;62:59-66.
- 52. Vermeer SE, Prins ND, den Heijer T, Hofman A, Koudstaal PJ, Breteler MM. Silent brain infarcts and the risk of dementia and cognitive decline. N Engl J Med 2003;348:1215-22.
- 53. Ikram MA, Vrooman HA, Vernooij MW, den Heijer T, Hofman A, Niessen WJ, van der Lugt A, Koudstaal PJ, Breteler MM. Brain tissue volumes in relation to cognitive function and risk of dementia. Neurobiol Aging 2010;31:378-86.

- 54. den Heijer T, Geerlings MI, Hoebeek FE, Hofman A, Koudstaal PJ, Breteler MM. Use of hippocampal and amygdalar volumes on magnetic resonance imaging to predict dementia in cognitively intact elderly people. Arch Gen Psychiatry 2006;63:57-62.
- 55. Kuller LH, Shemanski L, Manolio T, Haan M, Fried L, Bryan N, Burke GL, Tracy R, Bhadelia R. Relationship between ApoE, MRI findings, and cognitive function in the Cardiovascular Health Study. Stroke 1998;29:388-98.
- Wu CC, Mungas D, Petkov CI, Eberling JL, Zrelak PA, Buonocore MH, Brunberg JA, Haan MN, Jagust WJ. Brain structure and cognition in a community sample of elderly Latinos. Neurology 2002;59:383-91.
- 57. den Heijer T, Launer LJ, Prins ND, van Dijk EJ, Vermeer SE, Hofman A, Koudstaal PJ, Breteler MM. Association between blood pressure, white matter lesions, and atrophy of the medial temporal lobe. Neurology 2005;64:263-7.

09 The relation of the sortilin-related receptor gene to hippocampal volume and plasma amyloid beta levels in a family-based study of hypertensive patients

10 Polymorphisms of the renin-angiotensin system and alfa-adducin in associate to circulating amyloid beta levels

Abstract

Background

The renin-angiotensin system (RAS) has been implicated in Alzheimer's disease (AD) and degradation of amyloid beta (A β). Genetic studies focused on the angiotensin converting enzyme gene (ACE). Less is known for the other genes implicated in the RAS system. We studied the association of three commonly studied genetic variants with circulating A β .

Materials & Methods

The study was embedded in a family-based cohort sampled from a Dutch genetically isolated population. We selected individuals between 55 and 75 years of age with hypertension (N=128). Non-fasting blood sample were taken to assess plasma A β levels using xMAP® technology. We studied the effect of 3 variants within the angiotensin, angiotensin II type 1 receptor and adducin genes (AGT-M235T, AGTR1-C573T and ADD1-Gly460Trp) that were previously reported in cerebrovascular and cardiovascular disease.

Results

The AGT-M235T TT-genotype was significantly associated with higher levels of plasma A β 42 (p=0.008) and truncated A β n42 (p=0.02). The association to A β 42 remained significant after adjusting for potential confounders and multiple testing. Persons with the variant genotype of ADD1-Gly460Trp tended to have lower levels of A β n40 than persons with the wild type genotype (p=0.02), but this association did not remain significant after adjustment for confounders and multiple testing. No associations were found between plasma A β and AGTR1-C573T.

Conclusion

Our study shows that the AGT-M235T polymorphism is associated to plasma A β levels in hypertensive patients.

Introduction

Alzheimer's disease (AD) is characterized by neuritic plaques of which amyloid beta (A β) is the main component [1-3]. A β is present in full-length species (A β 40 and A β 42) and in shorter amino-terminal truncated species (A β n40 and A β n42). The latter concerns proteins which may predominantly be spliced to pathogenic forms and account for 60% of all A β species in pre-clinical AD stages [4]. A β pathology is also found in brains of cognitively healthy elderly at postmortem examination and visualized during life using PET studies [5,6]. The deposition and clearance of A β in the brain is related to A β levels in plasma, but the mechanism is not fully understood [7].

The renin-angiotensin system (RAS) is mainly involved in the regulation of blood pressure and salt homeostasis, but has also been implicated AD [8]. In AD brains, RAS is activated [9] and receptors for angiotensin II are present in brain tissue [10]. The gene encoding the angiotensin converting enzyme (ACE), is one of the most studied genes involved in the RAS system and has consistently been associated with AD [11,12]. There is experimental evidence for a role of ACE in the degradation of

amyloid beta (A β) in the brain [13,14] and inhibition of ACE activity results in increased A β deposition in the brain [15].

Far less is known for the other genes implicated in the RAS system in relation to AD pathology. The main other genes that have been studied are the angiotensin (AGT) and angiotensin II type 1 receptor (AGTR1) genes. These genes play a role in RAS by affecting the protein translation [16,17] and specific polymorphisms (AGT-M235T and AGTR1-C573T) have been associated to increased risk of cardiovascular and cerebrovascular disease [18-23]. Another gene that could be of interest given its strong link to RAS is the alfa-adducin gene (ADD1). ADD1 has been associated to increased renal sodium uptake [24]. Alfa-adducin is highly expressed in brain regions, including the hippocampus [25]. Different expression of the adducin subunits (α , β and γ) has been related to impaired learning, impaired motor function and synaptic plasticity in the hippocampus of mice [26]. We have previously shown a relationship of the ADD1-Gly460Trp variant and ischemic brain changes, mainly in hypertensive patients [27]. The reason why RAS genes interact with hypertension is far from clear. Hypertension early in life has been associated to AD at old age [28,29]. This may suggest that hypertensive patients are at increased risk of early AD pathology, despite the fact that they do not show symptoms of AD or cognitive decline[30]. Early amyloid pathology, however, may induce increased levels of A β in the blood.

To follow-up our work and that of others on the role of RAS genes other than ACE in A β pathology, we conducted a study relating these genes to plasma A β levels in individuals with hypertension. In the current study, we were interested in the role of A β to genes that we earlier implicated in MRI pathology: AGT-M235T, AGTR1-C573T and ADD1-Gly460Trp. We studied the association of these variants with plasma A β levels in individuals with hypertension from the Erasmus Rucphen Family (ERF) study, which is a family-based study in a genetically isolated population.

Materials and Methods

Study population

The study was conducted in a family-based study in a genetically isolated population in the Netherlands: the Erasmus Rucphen Family (ERF) study. Participants are all descendents of a limited number of founders and extensive genealogical data is available [31,32]. The current study was designed to find genetic risk factors for cerebral small vessel disease and included participants aged 55 to 75 years with hypertension. Hypertension was defined as systolic blood pressure ≥ 160 and/or diastolic blood pressure ≥ 100 and/or use of antihypertensive medication. Persons with a history of stroke or dementia or with MRI-contraindications were excluded. Two-hundred sixty one persons were invited; 135 agreed to participants. There was a small difference in completed years of education, which was higher for participants who had a mean level of 7.5 years compared to 9 years in non-participants (p=0.02). All participants gave informed consent and the study was approved by

the Medical Ethics Committee at Erasmus MC University Medical Centre.

Aβ measurements

Non-fasting blood samples were obtained during the visit at the research center, following standardized protocols. Samples were immediately cooled on ice. Plasma was extracted within the same day and stored at -80 °C. Plasma Aβ concentrations were measured using a fluorimetric bead-based immunoassay using xMAP® technology (Innogenetics®) using the manufacturers protocols. We obtained Aβ40, Aβ42, and the truncated forms Aβn40 and Aβn42. For analyses, we also used Aβ42/Aβ40 ratio.

Covariates

We evaluated the association of Aβ levels to covariates that were reported in earlier studies [33,34]. Body weight (kilograms) and height (centimeters) were measured and the body-mass index (BMI) was calculated from these measurements. Blood pressure was measured twice in sitting position by one physician using a sphygmomanometer. Blood was taken for the assessment of levels of high-density lipoprotein cholesterol (HDL-c) and creatinine in serum (Roche®) [35]. Glycosylated hemoglobin (HbA1c) was measured from full blood obtained in vacucontainers containing sodium citrate (Bio-Rad®) and analyzed using High-Performance Liquid Chromatography.

Genotyping

Genomic DNA was extracted from whole blood samples using the salting out method [36]. Samples were genotyped for ADD1-Gly460Trp, AGT-M235T, AGTR1-C573T, with Taqman allelic discrimination Assays-By-Design (Applied Biosystems, Foster City, CA) [19,27]. Samples were also genotyped for Apolipoprotein E (APOE) (rs429358, rs7412) to assess ɛ4 carrier status. In total, complete information was available for 128 individuals.

The observed frequencies of the genotypes were tested for deviations from Hardy-Weinberg equilibrium using the exact test for multiple alleles [37]. For all genetic variants, the allele and genotype distribution were in Hardy-Weinberg equilibrium (p>0.05).

Statistical analyses

General descriptive statistics were estimated with one-way ANOVA and Chi-square statistics as implemented in the software SPSS (version 15.0). Correlation coefficients between A β levels and covariates were assessed using SPSS. To analyze the association of the genotypes with and plasma A β levels, we performed the variable screening analysis under the polygenic model using the SOLAR software version 4.1.0 [38]. This software allows adjusting for family-relationships taking into account the pedigree structure. To reduce computational time, the large pedigree was cut into smaller subpedigrees of 18 bitsize before analysis [39]. One person could not be linked to these

subpedigrees, therefore we could analyze 128 persons. The analyses in SOLAR were adjusted for age, sex and inbreeding coefficient, which was estimated from the genealogical data of the complete pedigree of the population using the software PEDIG [40]. In a second model, we adjusted additionally for creatinine, HDL-c, HbA1c and BMI. The genotypes were entered in the model as a covariate with the reference genotype group coded as 0, the heterozygote genotypes coded as 1 and the rare homogeneous genotype group coded as 2.

N	128						
Sex (% women)	52.3%						
Age	64.6 (4.5)						
BMI	29.2 (4.3)						
HbA1c (%)	5.8 (0.6)						
Creatinine (µmol/L	73.0 (19.4)						
HDL-c (mmol/L)	1.3 (0.3)						
APOE 24	39.1%						
BMI: body-mass index. SBP: s diastolic blood pressure; HbA1	ystolic blood pressure; DBP: c: Glycosylated hemoglobin; c. cholesterol:						

Values represent percentages or mean (standard deviation).

Table 1

chapter 10

Correlation between plasma A													
	Αβ40	Aβn40	Αβ42	Aβn42	Αβ42/Αβ40	Sex	Age	BMI	HbA1C	Creat	HDL-C (mmol/L)	APOE	
Αβ40	1	0.77***	0.40***	0.23**	-0.38***	0.01	-0.02	0.24*	0.05	0.08	-0.18*	0.13	
Aβn40		1	0.32***	0.38***	-0.31***	0.05	-0.01	0.20	0.16	0.15	-0.18*	0.10	
Αβ42			1	0.35***	0.66***	-0.14	-0.01	0.10	0.12	0.04	0.04	-0.13	
Aβn42				1	0.22**	-0.09	-0.02	0.04	0.05	0.10	0.03	-0.10	
Δβ42/Αβ40					1	-0.22*	0.04	-0.11	0.07	0.02	0.19*	-0.21*	

Significant correlations are depicted in bold. Significant correlations are depicted in bold. * p<0.05; ** p<0.01; *** p<0.001. BMI: body-mass index; HbA1c: Glycosylated hemoglobin; Creat: Creatinne; HDL-c: high-density ligoprotein cholesterol; APCE: Apolipoprotein E gene defined as having 0, 1 or 2 copies of the e4 allele.

Table 2

Results

Table 1 shows the baseline characteristics of the study population. The mean age was 64.6 years and 52.3% were women. Correlations between the different Aβ measurements and covariates are given in Table 2. Of the covariates, sex and HDL-c were significantly associated to the Aβ42/Aβ40 ratio. HDL-c and BMI were related to Aβ40 and Aβn40. APOE was correlated to the Aβ42/Aβ40 ratio (R=-0.21; p=0.03), but not to single A β measurements. As expected, the different A β proteins were highly correlated (all p-values <10-5) and there was a strong correlation between A β 42 and the A β 42/ Aβ40 ratio (R=0.66; p=3.03*10-17).

Table 3 shows mean AB levels per genotype group for each gene. AGT-235T was significantly associated with higher levels of AB42 (p=0.008) and ABn42 (p=0.02). Individuals with the MMgenotype had mean AB42 plasma levels of 39.3 pg/mL compared to 48.0 pg/mL in individuals with the TT-genotype. For Aβn42 levels, a mean of 25.9 pg/mL was seen in individuals with the MM

Associatio	4ssociation of AGT, AGTR1, and ADD1 plasma abeta levels															
Gene	ne Genotype Aβ40		Αβ42			2	Αβη40				Αβn42			Αβ42/Αβ40		
		N	Mean (SD)	Р	N	Mean (SD)	Р	N	Mean (SD)	Р	N	Mean (SD)	Р	N	Mean (SD)	Р
AGT	MM	40	179.0 (36.7)	0.57	41	39.3 (14.2)	0.008	41	176.6 (31.8)	0.98	39	25.9 (5.9)	0.02	40	0.23 (0.08)	0.15
	MT	63	171.6 (35.6)		63	40.4 (11.7)		63	169.9 (37.5)		58	28.3 (6.1)		63	0.24 (0.07)	
	Π	23	193.6 (52.6)		23	48.1 (14.9)		23	185.0 (46.7)		22	30.2 (8.7)		23	0.25 (0.07)	
ACTES	TT	20	100.0 (25.2)	0.17	22	40.0 (15.1)	0.92	22	174 9 (00 9)	0.00	20	20.2 (6.1)	0.15	20	0.02 (0.09)	0.19
AGINI	07	32	102.9 (33.3)	0.17	33	40.2 (13.1)	0.62	33	174.0 (29.0)	0.99	30	29.3 (0.1)	0.15	32	0.23 (0.08)	0.10
	01	52	184.4 (42.2)		52	41.9 (13.4)		52	178.7 (41.3)		50	27.0 (0.8)		52	0.23 (0.06)	
	CC	42	166.3 (39.0)		42	41.7 (12.4)		42	169.9 (39.4)		39	27.2 (7.1)		42	0.26 (0.08)	
ADD1	GG	81	182.9 (41.7)	0.06	82	42.1 (14.2)	0.64	82	179.8 (38.3)	0.06	78	28.4 (6.8)	0.31	81	0.24 (0.08)	0.49
			,			· /			(,							
	GT/TT	45	169.1 (35.4)		45	40.1 (11.9)		45	165.6 (35.3)		41	26.9 (6.5)		45	0.24 (0.06)	
			,			,										

P: p-value adjusted for age, sex and family-relationship. Values are in pg/mL.

Table 3

Associations of AGT, AGTR1, and ADD1 with plasma Aβ levels, adjusted for covariates



Genotypes for AGT-M235T (A), AGTR1-C573T (B) and ADD1-Gly460Trp (C) are depicted on the x-axis. Values on the y-axis represent the differences in the mean plasma A β levels compared to the reference genotype with 95% confidence intervals. The differences were adjusted for age, sex, family-relationship, creatinine, high-density lipoprotein cholesterol, glycosylated hemoglobin and body mass index.

genotype and 30.2 pg/mL in those with the TT-genotype. ADD1 was associated with A β n40 (p=0.02). Individuals with the GT/TT genotype had significantly lower levels of A β n40 (165.6 pg/mL) than persons with the GG genotype (179.8 pg/mL). No significant association was seen to the A β 42/A β 40 ratio for any of the genes.

The Figure shows the differences in the mean A β levels compared to the reference genotype, adjusting for additional covariates. The findings remained significant for the relation of AGT with A β 42 and A β n42. Regarding the A β 40 levels and the A β 42/A β 40 ratio there was no significant trend (A). After adjusting, no association was seen between AGTR1 and plasma A β (B) and the association of ADD1 with A β n40 became borderline significant (p-value 0.06) (C).

Discussion

The present study shows that individuals with the AGT-M235T TT-genotype have significantly higher levels of plasma A β 42 and A β n42. There was also some evidence, although not significant after adjusting, that individuals with GT/TT genotypes of the ADD1-Gly460Trp variant have lower levels of A β n40, but there was no association between AGTR1 and A β . APOE was correlated to a lower A β 42/A β 40 ratio.

Before interpreting the findings, a few methodological issues need to be addressed. First, we did not do a formal Bonferroni correction of the threshold p-value, because the A β outcomes were all related (Table 2). Correcting the p-value for the number of association analyses (Table 3; N=15) would therefore have been a too conservative approach. Alternatively, when using for example the 3 genes and 2 A β outcomes (A β 40 and A β 42), the p-value observed for AGT would be below the 0.0083 threshold that is then obtained.

Second, the sample size of the current population was relatively small. In this respect our finding that APOE was correlated to a lower A β 42/A β 40 ratio is important, serving as a proof of principle. However, also the allele distributions are important when considering the power of the study. Because some genotypes were less frequent than APOE ϵ 4, the small sample size may explain that the major differences in mean A β levels between the genotype groups of AGTR1 and ADD1 that were observed were not significant. For example, large differences in the mean A β 40 levels were observed for AGT and ADD1 (Figure 1) and for ADD1 large differences were observed for A β 40, but none of the differences reached significance. Because these relationships do have biological plausibility, replication of our findings in larger datasets is warranted.

It is also important to realize that the associations of plasma $A\beta$ levels with dementia have been inconsistent. Increased risk has been reported for high plasma levels of both $A\beta40$ and $A\beta42$, but also decreased risk for high levels of $A\beta42$ have been found [33,34,41-45]. An increased risk with higher levels of plasma $A\beta42$ has also been reported for cognitive decline [43]. One of the explanations for these discrepancies may be that plasma $A\beta$ levels increase with aging, but not necessarily with disease progression [44]. It is thought that with disease progression $A\beta42$ is increasingly deposited in the brain and levels are consequently lower in plasma. In a young population with an increased

risk of AD pathology like ours, higher levels of A β 42 could then be expected in persons carrying risk genotypes. The age of the study population and timing of A β measurements may be important contributors to the inconsistencies in literature.

From a biological perspective it is of interest that we found higher levels of plasma A β 42 in persons with the TT-genotype of AGT-M235T. This genotype was previously described to increase the risk of cerebrovascular small vessel disease [23]. Previous studies have also reported a role in large vessel disease, but a recent meta-analysis did not confirm these findings [46]. This may imply that AGT exerts its effects via an independent brain RAS system [47] and not through increased levels of AGT II affecting atherogenic processes [48]. Not many groups have studied this gene in relation to dementia, and the findings are not convincing [49,50] with possibly an increased risk in patients with the AGT-M235T TT-genotype [51]. The RAS proteins have been implicated in AD pathology, because receptors for angiotensin II are present in brain tissue and RAS is activated in AD brains [9,10]. Along with cerebrovascular effects, RAS could thus have a direct effect on brain pathology and possibly on AB. Indirect evidence is given by the association of ACE, another gene involved in this system, with AD [11,12] and the role of ACE in the degradation of Aβ in the brain [13,14]. It has been shown that inhibition of ACE activity results in increased AB deposition in the brain [15]. Like RAS, alfa-adducin also plays a role in the brain [25,26]. Our results are suggestive for an association of the variant allele of ADD1-Gly460Trp to lower levels of A β 40, but these findings should be interpreted with caution as they were not significant when adjusting for multiple testing using Bonferroni correction. Whereas it is an interesting candidate for AD pathology [26], the evidence was found in animal studies and no human studies have been conducted to our knowledge. The role of ADD1 in AD amyloid pathology is not known and as mentioned, inconsistent associations have been reported between plasma Aβ40 and dementia. Our findings therefore warrant further study.

Alternative explanations of our findings are possible. A β may have a direct effect on brain vasculature and vascular damage may have contributed to our findings, since the studied variants are all implicated in atherogenic effects [52]. Associations have been reported for A β with cerebral small vessel disease [53,54] and vascular dysfunction has been suggested in AD pathology [55]. On the other hand, damage to cerebral small vessels due to other processes may result in leakage of A β causing higher levels of circulating A β . We found moderate correlations between plasma A β and cardiovascular factors, which have been shown to be highly correlated to cerebral small vessel disease, and adjusting for these factors did not affect our main finding. The studied variants and A β are also both correlated to renal function, however, adjusting for creatinine as a measure for kidney function did not alter our findings.

The strength of our study is the family-based design, performed in a homogenous group of middleaged individuals with hypertension derived from a genetically isolated population, which increases statistical power of genetic association studies. The size of the study is limited, however, which increases the chance of false-positive and -negative findings. In sum, we found a consistent association of the AGT-M235T TT-genotype to plasma A β 42 and A β n42. A question that remains to be answered in longitudinal data is the role of AGT and ADD1 in midlife plasma A β levels and the effect of changing A β levels on AD pathology.

References

- 1. NIA-Working-Group. Consensus recommendations for the postmortem diagnosis of Alzheimer's disease. The National Institute on Aging, and Reagan Institute Working Group on Diagnostic Criteria for the Neuropathological Assessment of Alzheimer's Disease. Neurobiol Aging 1997;18:S1-2.
- Mirra SS, Heyman A, McKeel D, Sumi SM, Crain BJ, Brownlee LM, Vogel FS, Hughes JP, van Belle G, Berg L. The Consortium to Establish a Registry for Alzheimer's Disease (CERAD). Part II. Standardization of the neuropathologic assessment of Alzheimer's disease. Neurology 1991;41:479-86.
- 3. Braak H, Braak E. Neuropathological stageing of Alzheimer-related changes. Acta Neuropathol 1991;82:239-59.
- Sergeant N, Bombois S, Ghestem A, Drobecq H, Kostanjevecki V, Missiaen C, Wattez A, David JP, Vanmechelen E, Sergheraert C, Delacourte A. Truncated beta-amyloid peptide species in pre-clinical Alzheimer's disease as new targets for the vaccination approach. J Neurochem 2003;85:1581-91.
- 5. Fotenos AF, Mintun MA, Snyder AZ, Morris JC, Buckner RL. Brain volume decline in aging: evidence for a relation between socioeconomic status, preclinical Alzheimer disease, and reserve. Arch Neurol 2008;65:113-20.
- Bennett DA, Schneider JA, Arvanitakis Z, Kelly JF, Aggarwal NT, Shah RC, Wilson RS. Neuropathology of older persons without cognitive impairment from two community-based studies. Neurology 2006;66:1837-44.
- 7. DeMattos RB, Bales KR, Cummins DJ, Paul SM, Holtzman DM. Brain to plasma amyloidbeta efflux: a measure of brain amyloid burden in a mouse model of Alzheimer's disease. Science 2002;295:2264-7.
- 8. Kehoe PG, Wilcock GK. Is inhibition of the renin-angiotensin system a new treatment option for Alzheimer's disease? Lancet Neurol 2007;6:373-8.
- 9. Savaskan E, Hock C, Olivieri G, Bruttel S, Rosenberg C, Hulette C, Muller-Spahn F. Cortical alterations of angiotensin converting enzyme, angiotensin II and AT1 receptor in Alzheimer's dementia. Neurobiol Aging 2001;22:541-6.
- 10. Wright JW, Yamamoto BJ, Harding JW. Angiotensin receptor subtype mediated physiologies and behaviors: new discoveries and clinical targets. Prog Neurobiol 2008;84:157-81.
- 11. Lehmann DJ, Cortina-Borja M, Warden DR, Smith AD, Sleegers K, Prince JA, van Duijn CM, Kehoe PG. Large meta-analysis establishes the ACE insertion-deletion polymorphism as a marker of Alzheimer's disease. Am J Epidemiol 2005;162:305-17.
- 12. Bertram L, McQueen MB, Mullin K, Blacker D, Tanzi RE. Systematic meta-analyses of Alzheimer disease genetic association studies: the AlzGene database. Nat Genet 2007;39:17-23.
- 13. Hemming ML, Selkoe DJ. Amyloid beta-protein is degraded by cellular angiotensin-converting enzyme (ACE) and elevated by an ACE inhibitor. J Biol Chem 2005;280:37644-50.

- 14. Hu J, Igarashi A, Kamata M, Nakagawa H. Angiotensin-converting enzyme degrades Alzheimer amyloid beta-peptide (A beta); retards A beta aggregation, deposition, fibril formation; and inhibits cytotoxicity. J Biol Chem 2001;276:47863-8.
- Zou K, Yamaguchi H, Akatsu H, Sakamoto T, Ko M, Mizoguchi K, Gong JS, Yu W, Yamamoto T, Kosaka K, Yanagisawa K, Michikawa M. Angiotensin-converting enzyme converts amyloid beta-protein 1-42 (Abeta(1-42)) to Abeta(1-40), and its inhibition enhances brain Abeta deposition. J Neurosci 2007;27:8628-35.
- Kim HS, Krege JH, Kluckman KD, Hagaman JR, Hodgin JB, Best CF, Jennette JC, Coffman TM, Maeda N, Smithies O. Genetic control of blood pressure and the angiotensinogen locus. Proc Natl Acad Sci U S A 1995;92:2735-9.
- 17. Sethi AA, Nordestgaard BG, Tybjaerg-Hansen A. Angiotensinogen gene polymorphism, plasma angiotensinogen, and risk of hypertension and ischemic heart disease: a metaanalysis. Arterioscler Thromb Vasc Biol 2003;23:1269-75.
- 18. Takami S, Imai Y, Katsuya T, Ohkubo T, Tsuji I, Nagai K, Satoh H, Hisamichi S, Higaki J, Ogihara T. Gene polymorphism of the renin-angiotensin system associates with risk for lacunar infarction. The Ohasama study. Am J Hypertens 2000;13:121-7.
- 19. van Rijn MJ, Bos MJ, Isaacs A, Yazdanpanah M, Arias-Vasquez A, Stricker BH, Klungel OH, Oostra BA, Koudstaal PJ, Witteman JC, Hofman A, Breteler MM, van Duijn CM. Polymorphisms of the renin-angiotensin system are associated with blood pressure, atherosclerosis and cerebral white matter pathology. J Neurol Neurosurg Psychiatry 2007;78:1083-7.
- 20. Rodriguez-Perez JC, Rodriguez-Esparragon F, Hernandez-Perera O, Anabitarte A, Losada A, Medina A, Hernandez E, Fiuza D, Avalos O, Yunis C, Ferrario CM. Association of angiotensinogen M235T and A(-6)G gene polymorphisms with coronary heart disease with independence of essential hypertension: the PROCAGENE study. Prospective Cardiac Gene. J Am Coll Cardiol 2001;37:1536-42.
- Plat AW, Stoffers HE, Klungel OH, van Schayck CP, de Leeuw PW, Soomers FL, Schiffers PM, Kester AD, Kroon AA. The contribution of six polymorphisms to cardiovascular risk in a Dutch high-risk primary care population: the HIPPOCRATES project. J Hum Hypertens 2009;23:659-67.
- 22. Brenner D, Labreuche J, Poirier O, Cambien F, Amarenco P. Renin-angiotensin-aldosterone system in brain infarction and vascular death. Ann Neurol 2005;58:131-8.
- 23. Schmidt R, Schmidt H, Fazekas F, Launer LJ, Niederkorn K, Kapeller P, Lechner A, Kostner GM. Angiotensinogen polymorphism M235T, carotid atherosclerosis, and small-vessel disease-related cerebral abnormalities. Hypertension 2001;38:110-5.
- 24. Manunta P, Burnier M, D'Amico M, Buzzi L, Maillard M, Barlassina C, Lanella G, Cusi D, Bianchi G. Adducin polymorphism affects renal proximal tubule reabsorption in hypertension. Hypertension 1999;33:694-7.
- 25. Seidel B, Zuschratter W, Wex H, Garner CC, Gundelfinger ED. Spatial and sub-cellular localization of the membrane cytoskeleton-associated protein alpha-adducin in the rat brain. Brain Res 1995;700:13-24.

- 26. Porro F, Rosato-Siri M, Leone E, Costessi L, Iaconcig A, Tongiorgi E, Muro AF. betaadducin (Add2) KO mice show synaptic plasticity, motor coordination and behavioral deficits accompanied by changes in the expression and phosphorylation levels of the alpha- and gamma-adducin subunits. Genes Brain Behav 2009.
- 27. van Rijn MJ, Bos MJ, Yazdanpanah M, Isaacs A, Arias-Vasquez A, Koudstaal PJ, Hofman A, Witteman JC, van Duijn CM, Breteler MM. Alpha-adducin polymorphism, atherosclerosis, and cardiovascular and cerebrovascular risk. Stroke 2006;37:2930-4.
- 28. Kivipelto M, Ngandu T, Fratiglioni L, Viitanen M, Kareholt I, Winblad B, Helkala EL, Tuomilehto J, Soininen H, Nissinen A. Obesity and vascular risk factors at midlife and the risk of dementia and Alzheimer disease. Arch Neurol 2005;62:1556-60.
- 29. Launer LJ, Ross GW, Petrovitch H, Masaki K, Foley D, White LR, Havlik RJ. Midlife blood pressure and dementia: the Honolulu-Asia aging study. Neurobiol Aging 2000;21:49-55.
- 30. Skoog I, Gustafson D. Hypertension, hypertension-clustering factors and Alzheimer's disease. Neurol Res 2003;25:675-80.
- 31. Pardo LM, MacKay I, Oostra B, van Duijn CM, Aulchenko YS. The effect of genetic drift in a young genetically isolated population. Annals of Human Genetics 2005;69:288-95.
- 32. Aulchenko YS, Heutink P, Mackay I, Bertoli-Avella AM, Pullen J, Vaessen N, Rademaker TA, Sandkuijl LA, Cardon L, Oostra B, van Duijn CM. Linkage disequilibrium in young genetically isolated Dutch population. Eur J Hum Genet 2004;12:527-34.
- 33. Lambert JC, Schraen-Maschke S, Richard F, Fievet N, Rouaud O, Berr C, Dartigues JF, Tzourio C, Alperovitch A, Buee L, Amouyel P. Association of plasma amyloid beta with risk of dementia: the prospective Three-City Study. Neurology 2009;73:847-53.
- 34. van Oijen M, Hofman A, Soares HD, Koudstaal PJ, Breteler MM. Plasma Abeta(1-40) and Abeta(1-42) and the risk of dementia: a prospective case-cohort study. Lancet Neurol 2006;5:655-60.
- 35. Liu F, Pardo LM, Schuur M, Sanchez-Juan P, Isaacs A, Sleegers K, de Koning I, Zorkoltseva IV, Axenovich TI, Witteman JC, Janssens AC, van Swieten JC, Aulchenko YS, Oostra BA, van Duijn CM. The apolipoprotein E gene and its age-specific effects on cognitive function. Neurobiology of Aging 2008.
- 36. Miller SA, Dykes DD, Polesky HF. A simple salting out procedure for extracting DNA from human nucleated cells. Nucleic Acids Res 1988;16:1215.
- 37. Guo SW, Thompson EA. Performing the exact test of Hardy-Weinberg proportion for multiple alleles. Biometrics 1992;48:361-72.
- 38. Almasy L, Blangero J. Multipoint quantitative-trait linkage analysis in general pedigrees. Am J Hum Genet 1998;62:1198-211.
- 39. Liu F, Kirichenko A, Axenovich TI, van Duijn CM, Aulchenko YS. An approach for cutting large and complex pedigrees for linkage analysis. Eur J Hum Genet 2008;16:854-60.
- 40. Boichard d. PEDIG: a FORTRAN package for pedigree analysis studied for large populations. Proceeding of the 7th Word Congress of Genet Appl Livest Prod 2002:28.
- 41. Mayeux R, Honig LS, Tang MX, Manly J, Stern Y, Schupf N, Mehta PD. Plasma A[beta]40 and A[beta]42 and Alzheimer's disease: relation to age, mortality, and risk. Neurology 2003;61:1185-90.

- 42. Graff-Radford NR, Crook JE, Lucas J, Boeve BF, Knopman DS, Ivnik RJ, Smith GE, Younkin LH, Petersen RC, Younkin SG. Association of low plasma Abeta42/Abeta40 ratios with increased imminent risk for mild cognitive impairment and Alzheimer disease. Arch Neurol 2007;64:354-62.
- 43. Pomara N, Willoughby LM, Sidtis JJ, Mehta PD. Selective reductions in plasma Abeta 1-42 in healthy elderly subjects during longitudinal follow-up: a preliminary report. Am J Geriatr Psychiatry 2005;13:914-7.
- 44. Fukumoto H, Tennis M, Locascio JJ, Hyman BT, Growdon JH, Irizarry MC. Age but not diagnosis is the main predictor of plasma amyloid beta-protein levels. Arch Neurol 2003;60:958-64.
- 45. Schupf N, Tang MX, Fukuyama H, Manly J, Andrews H, Mehta P, Ravetch J, Mayeux R. Peripheral Abeta subspecies as risk biomarkers of Alzheimer's disease. Proc Natl Acad Sci U S A 2008;105:14052-7.
- 46. Wang X, Cheng S, Brophy VH, Erlich HA, Mannhalter C, Berger K, Lalouschek W, Browner WS, Shi Y, Ringelstein EB, Kessler C, Luedemann J, Lindpaintner K, Liu L, Ridker PM, Zee RY, Cook NR. A meta-analysis of candidate gene polymorphisms and ischemic stroke in 6 study populations: association of lymphotoxin-alpha in nonhypertensive patients. Stroke 2009;40:683-95.
- 47. Bunnemann B, Fuxe K, Ganten D. The brain renin-angiotensin system: localization and general significance. J Cardiovasc Pharmacol 1992;19 Suppl 6:S51-62.
- 48. Burnier M, Brunner HR. Angiotensin II receptor antagonists. Lancet 2000;355:637-45.
- 49. Giedraitis V, Kilander L, Degerman-Gunnarsson M, Sundelof J, Axelsson T, Syvanen AC, Lannfelt L, Glaser A. Genetic analysis of Alzheimer's disease in the Uppsala Longitudinal Study of Adult Men. Dement Geriatr Cogn Disord 2009;27:59-68.
- 50. Edwards TL, Pericak-Vance M, Gilbert JR, Haines JL, Martin ER, Ritchie MD. An association analysis of Alzheimer disease candidate genes detects an ancestral risk haplotype clade in ACE and putative multilocus association between ACE, A2M, and LRRTM3. Am J Med Genet B Neuropsychiatr Genet 2009;150B:721-35.
- 51. Taylor A, Ezquerra M, Bagri G, Yip A, Goumidi L, Cottel D, Easton D, Evans JG, Xuereb J, Cairns NJ, Amouyel P, Chartier-Harlin MC, Brayne C, Rubinsztein DC. Alzheimer disease is not associated with polymorphisms in the angiotensinogen and renin genes. Am J Med Genet 2001;105:761-4.
- 52. Thomas T, Thomas G, McLendon C, Sutton T, Mullan M. beta-Amyloid-mediated vasoactivity and vascular endothelial damage. Nature 1996;380:168-71.
- 53. van Dijk EJ, Prins ND, Vermeer SE, Hofman A, van Duijn CM, Koudstaal PJ, Breteler MM. Plasma amyloid beta, apolipoprotein E, lacunar infarcts, and white matter lesions. Ann Neurol 2004;55:570-5.
- 54. Gurol ME, Irizarry MC, Smith EE, Raju S, Diaz-Arrastia R, Bottiglieri T, Rosand J, Growdon JH, Greenberg SM. Plasma beta-amyloid and white matter lesions in AD, MCI, and cerebral amyloid angiopathy. Neurology 2006;66:23-9.
- 55. Schneider JA, Arvanitakis Z, Bang W, Bennett DA. Mixed brain pathologies account for most dementia cases in community-dwelling older persons. Neurology 2007;69:2197-204.

Part V

General discussion and summary

11 Findings of this thesis
In this thesis I describe my research on genetic determinants of cognitive function and age-related brain changes. I have used outcomes that are highly heritable as endophenotypes for my studies of Alzheimer's disease, including cognitive function, Aβ plasma levels and age-related brain changes as visible on magnetic resonance imaging (MRI). Different study-designs were chosen to investigate our research questions including candidate gene studies, genome wide linkage analysis and genome wide association studies. In the following chapter, I will discuss the main findings of this thesis.

One of the most extensively studied candidate gene in Alzheimer's disease is the apolipoprotein E gene (APOE) [1-3]. The ɛ4 allele of this gene is a well-established determinant of AD with a large effect on disease risk. Based on the hypothesis that cognitive function may be a relevant endophenotype for AD, we studied the relation between APOE and cognitive function in **chapter 3**. We found that the APOE*ɛ4 allele was significantly associated with lower test scores on the Adult Verbal Learning Test in individuals older than 50 years of age. This effect of APOE*ɛ4 was independent of the effect of APOE*ɛ4 on vascular risk factors and most pronounced on learning ability. Similar to the findings of others [4], we found that the APOE*ɛ4 allele has an effect on cognitive function, but that in contrast to AD the effect is relatively small. We focused our gene discovery studies on cognitive function, since this outcome showed the most consistent association to APOE [4] and may therefore be the most promising endophenotype.

To explore new susceptibility regions for cognitive functioning without prior assumptions of pathways involved, we conducted a hypothesis-free genome-wide search on a range of cognitive tests. In **chapter 4** we present the findings of a non-parametric linkage analyses in the Erasmus Rucphen Family (ERF) Study, which is a family-based study in a genetically isolated population. Since we were targeting genes with a major effect, we selected individuals from the lower extremes of the trait distribution for the linkage analysis. Thresholds for significant and suggestive linkage were estimated by a simulation study. Significant linkage (LOD > 3.78) to cognitive functioning was found on chromosomes 1p13.1, 12q24.33, 19q13.43, 20p13, 21q22.13 and 21q22.3. For the fine-mapping of the region, we used dense genotyping in the regions under the linkage peak in ERF and replicated these findings in a large outbred, population-based cohort, the Rotterdam Study (RS) [5]. Fine-mapping showed significant associations to chromosome 1 (p-value=0.03) and 21 (p-value=0.01) after correction for multiple testing, and association with the latter region on 21q22.13 was replicated in the Rotterdam Study (nominal p-value 0.003). Both fine-mapping and replication pointed to variants within the potassium inwardly-rectifying channel, subfamily J, member 6 gene (KCNJ6).

Whereas linkage analysis in the extremes of the distribution specifically targets variants with larger effects, we conducted a genome-wide association study of cognitive function as a continuous outcome in search of common variants with small effects. In **chapter 5** we describe a meta-analysis of different

genome-wide association studies performed in the Cohorts for Heart and Aging Research in Genomic Epidemiology (CHARGE) consortium. This consortium includes large prospective population-based cohorts. Neuropsychological testing was available for 13 cohorts. In this thesis, we focussed on executive function and processing speed tasks including the Trail Making Test (TMT) parts A and B and the Stroop Color and Word Test in the analyses. All individual studies used their genotyped data to impute to 2.5 million single nucleotide polymorphisms (SNPs). The analyses were performed in Caucasians older than 45 years who were free of dementia and clinical stroke at times of cognitive testing. The most significant finding was found with TMT-B and a SNP on chromosome 18. This SNP was just above the genome-wide significant threshold with a p-value of 6.95*10-8 and located between two plausible candidate genes. We further conducted an exploratory analysis in which we searched for overlap between our findings and the genome-wide association analyses published for AD and schizophrenia. Overlap with previous genome-wide association studies was found for multiple other SNPs with a p-value smaller than 1.0*10-3, of which the sortilin-related-receptor-1 (SORL1), the syntaxin-binding-protein-6 (STXBP6) and the protocadherin-9 (PCDH9) genes are the most interesting genes. The genes in the regions that we identified in this study may provide further insights into the pathways involved in the normal variation of cognition. Our findings, however, await replication, which is currently ongoing.

A preliminary comparison between the findings of the genome-wide linkage and association analysis suggests no overlap in genes, which may be expected in light of the mechanisms underlying the methods. Linkage is designed to target rare variants with large effects and association on the other hand is designed to find common variants with moderate effects.

Of interest is also that we did not find evidence for a role of APOE, or the recently discovered AD genes, PICALM, CRI and CLU [11,12] in cognitive function in our genome-wide association analyses. This finding reveals, again as expected, that findings on endophenotypes cannot be translated 1:1 to the disease of interest. Another issue to realize is that tests assess different aspects of cognitive function. Indeed we found that APOE was associated to the Adult Verbal Learning Test in chapter 3 but non-significantly to TMT-A, B or Stroop.

Having studied cognitive function as endophenotype, we further studied age-related brain changes as a second group of endophenotypes. We considered plasma Aβ levels as biomarkers for the presence of senile plaques and amyloid angiopathy, and asymptomatic brain lesions on MRI as age-related brain changes. We have focused on lacunar infarcts, white matter lesions (WML), microbleeds and hippocampal atrophy. All are associated with hypertension, stroke, dementia and cognitive impairment [13-18], and are also found in healthy elderly.

We examined the role of candidate genes involved in blood pressure regulation and in amyloid metabolism. We studied APOE, the renin-angiotensin system (RAS) related genes (Angiotensin, Angiotensin II type 1 Receptor, alpha-Adducin) and the sortilin-related receptor (SORL1) gene. RAS genes are involved in the regulation of blood pressure and salt homeostasis and the RAS proteins have also been implicated in Alzheimer's disease [19]. Receptors for angiotensin II are present in brain tissue [20] and an increased activation of RAS is seen in AD brains [21]. As already mentioned, APOE has consistently been associated with AD and there is increasing evidence that also SORL1 is associated with AD [4,22,23]. SORL1 consists of two functional regions, one functioning in the cholesterol pathway and the other in the APP processing pathway [24,25]. Interestingly, the gene has also been associated to cerebrovascular disease in a previous study [26] and also emerged in our comparative analysis in the genome-wide association study (**chapter 5**).

First, we studied all five variants in relation to the MRI endophenotypes: volumes of WML and presence of lacunes and microbleeds in a subgroup of the ERF study aged 55 and 75 years with hypertension (**chapter 8**). WML was present in variable severity in all participants, whereas lacunar infarcts were present in 15.5% and microbleeds in 23.3%. Homozygosity for the APOE ɛ4 allele was associated with lacunes (OR, 4.8; 95% CI, 1.2-19.3). Individuals carrying two copies of the variant allele of 4 SNPs located at the 3'-end of SORL1 (rs1699102, rs3824968, rs2282649, rs1010159), had an increased risk of microbleeds (highest odds ratio, 6.87; 95% CI, 1.78-26.44), which is suggestive for the hypothesis that the amyloid cascade is involved in the etiology of microbleeds in populations with hypertension.

Second, in **chapter 9** we studied SORL1 in relation to hippocampal volume and plasma A β levels in the same subgroup of the ERF study. Hippocampal volumes were quantitatively measured on MRI and plasma A β levels were determined in non-fasting blood samples. We studied the effect of 7 variants within SORL1 that were previously reported in AD. Three variants located near the 3'-end of SORL1 were significantly associated to hippocampal volume. The 3-SNP haplotypes for rs1699102, rs3824968 and rs2282649 (CAT) and for rs3824968, rs2282649 and rs1010159 (ATC) were associated to higher hippocampal volumes when adjusting for multiple testing. We did not find significant associations of single variants with plasma A β levels.

Third, we studied the association of the three variants within the angiotensin, angiotensin II type 1 receptor and adducin genes (AGT-M235T, AGTR1-C573T and ADD1-Gly460Trp) in the same middleaged hypertensive subset of ERF. Variants in these genes were previously reported in cerebroand cardiovascular disease in relation to circulating levels of plasma A β (**chapter 10**). The AGT-M235T TT-genotype was significantly associated with higher levels of plasma A β 42 (p=0.008) and truncated A β n42 (p=0.02). The association to A β 42 remained significant after adjusting for potential confounders and multiple testing. No significant associations were found between AGTR1-C573T or ADD1-Gly460Trp and plasma A β .

Taken together, the most interesting finding of our studies may be the associations that were found for SORL1 in various study designs. Our candidate gene analyses showed association of SORL1 with cognition as well as microbleeds and hippocampal volume. SORL1 also emerged in our genomewide association meta-analyses of cognitive function. A word of caution is, however, needed: our candidate gene studies were performed in a small sample size and were restricted to hypertensive individuals. These findings therefore need replication in larger cohorts in the general population.

Finally, we conducted two candidate gene studies in Alzheimer's disease to elucidate the role of two interesting pathways. Iron overload may contribute to the risk of Alzheimer's disease. We earlier have studied the genes implicated in hemochromatosis in relation to AD [27]. We found an effect of the hemochromatosis gene (HFE) on the age of onset of AD. The HFE-63D mutation was related to an earlier onset in APOE*ɛ4 carriers, but not to the disease risk. Other groups reported evidence in other variants in hemochromatosis genes HFE-C282Y and -H63D, and transferrin (TF) [22]. In the Epistasis Project, with 1757 AD cases and 6295 controls, we studied four variants in two genes of iron metabolism: HFE-C282Y and -H63D, and TF-C2 and -2G/A (**chapter 7**). We replicated the interactive effect between HFE-282Y and TF-C2 on the risk of AD in Northern Europeans. We also found an interaction between HFE-63HH and TF-2AA, which was markedly modified by age. The interaction between HFE-282Y and TF-C2 has now been replicated twice, in a total of 2313 cases of AD and 7065 control. There are a number of limitations of this study that hamper firm conclusions.

First, both interactions were found mainly or only in Northern Europeans. In fact, there was an absence of a relation between HFE and AD in a Northern Spanish population. From a statistical perspective, the exclusion of the Spanish data is problematic. Although the allele frequencies in Northern Spain differed from those in the Northern Europeans, this does not imply that the relation to AD should be different. A second problem is that although we pooled the data, the numbers are small and as a consequence the study power is low, making the analysis susceptible to false positive findings.

We also studied the Cathepsin D gene (CTSD) in relation to AD (**chapter 6**). CTSD is involved in amyloid precursor protein processing and is therefore considered a candidate for AD. We performed a candidate-gene analysis in the Rotterdam Study, which is a population-based cohort-study (N=7983) and estimated the effect of CTSD variants on the risk of AD. Additionally, we performed a large meta-analysis incorporating our data and previously published data. The T-allele of CTSD rs17571 was associated with an increased risk of AD (p-value 0.007) in the Rotterdam Study. This association was predominantly found in APOE ε4 noncarriers. A meta-analysis of previously published data showed

a significantly increased risk of AD in carriers of the T-allele of rs17571 (OR 1.22, 95% Cl 1.03-1.44), irrespective of APOE ɛ4 carrier status.

Besides these genetic studies, in chapter 2 we also performed a classical epidemiological study in which we studied a combination of cardiovascular risk factors as composed in the metabolic syndrome (MetS) in relation to cognition. While type 2 diabetes is known to be associated with poorer cognitive performance [28,29], fewer studies have reported on the association of MetS and contributing factors, such as insulin-resistance (HOMA-IR), low adiponectin-, and high C-reactive protein (CRP)- levels [30.31]. We studied whether these factors are related to cognitive function and which of the MetS components are independently associated. Also this study was performed in the ERF study where extensive data on physical examination, biomedical measurements and neuropsychological testing were available. Linear regression models were used to determine the association between MetS. HOMA-IR, adiponectin levels, CRP, and cognitive test scores. We found that predominantly women with MetS and high HOMA-IR had lower scores on executive function tests (p=0.03 and p=0.009). The most consistent individual component of MetS, contributing to the association with executive test scores was systolic blood pressure. We interpret these results with caution, however, since the design was cross-sectional and with very strict multiple testing adjustment using Bonferroni would result in only borderline significant p-values. Longitudinal studies will be needed to gain insight in the causality of our reported findings and may result in more conclusive findings.

A huge challenge in genetic and epidemiological research, especially in the candidate gene studies and an exploratory study such as we conducted for MetS, is how to improve the quality and validation of candidate genes in AD. This will be discussed in the next chapter.

References

- 1. Poirier J. Apolipoprotein E and Alzheimer's disease. A role in amyloid catabolism. Ann N Y Acad Sci 2000;924:81-90.
- van Duijn CM, de Knijff P, Cruts M, Wehnert A, Havekes LM, Hofman A, Van Broeckhoven C. Apolipoprotein E4 allele in a population-based study of early-onset Alzheimer's disease. Nat Genet 1994;7:74-8.
- 3. Farrer LA, Cupples LA, Haines JL, Hyman B, Kukull WA, Mayeux R, Myers RH, Pericak-Vance MA, Risch N, van Duijn CM. Effects of age, sex, and ethnicity on the association between apolipoprotein E genotype and Alzheimer disease. A meta-analysis. APOE and Alzheimer Disease Meta Analysis Consortium. Jama 1997;278:1349-56.
- 4. Wisdom NM, Callahan JL, Hawkins KA. The effects of apolipoprotein E on non-impaired cognitive functioning: A meta-analysis. Neurobiol Aging 2009.
- Hofman A, Breteler MM, van Duijn CM, Janssen HL, Krestin GP, Kuipers EJ, Stricker BH, Tiemeier H, Uitterlinden AG, Vingerling JR, Witteman JC. The Rotterdam Study: 2010 objectives and design update. Eur J Epidemiol 2009;24:553-72.
- 6. Harold D, Abraham R, Hollingworth P, Sims R, Gerrish A, Hamshere ML, Pahwa JS, Moskvina V, Dowzell K, Williams A, Jones N, Thomas C, Stretton A, Morgan AR, Lovestone S, Powell J, Proitsi P, Lupton MK, Brayne C, Rubinsztein DC, Gill M, Lawlor B, Lynch A, Morgan K, Brown KS, Passmore PA, Craig D, McGuinness B, Todd S, Holmes C, Mann D, Smith AD, Love S, Kehoe PG, Hardy J, Mead S, Fox N, Rossor M, Collinge J, Maier W, Jessen F, Schurmann B, van den Bussche H, Heuser I, Kornhuber J, Wiltfang J, Dichgans M, Frolich L, Hampel H, Hull M, Rujescu D, Goate AM, Kauwe JS, Cruchaga C, Nowotny P, Morris JC, Mayo K, Sleegers K, Bettens K, Engelborghs S, De Deyn PP, Van Broeckhoven C, Livingston G, Bass NJ, Gurling H, McQuillin A, Gwilliam R, Deloukas P, Al-Chalabi A, Shaw CE, Tsolaki M, Singleton AB, Guerreiro R, Muhleisen TW, Nothen MM, Moebus S, Jockel KH, Klopp N, Wichmann HE, Carrasquillo MM, Pankratz VS, Younkin SG, Holmans PA, O'Donovan M, Owen MJ, Williams J. Genome-wide association study identifies variants at CLU and PICALM associated with Alzheimer's disease. Nat Genet 2009;41:1088-93.
- Lambert JC, Heath S, Even G, Campion D, Sleegers K, Hiltunen M, Combarros O, Zelenika D, Bullido MJ, Tavernier B, Letenneur L, Bettens K, Berr C, Pasquier F, Fievet N, Barberger-Gateau P, Engelborghs S, De Deyn P, Mateo I, Franck A, Helisalmi S, Porcellini E, Hanon O, de Pancorbo MM, Lendon C, Dufouil C, Jaillard C, Leveillard T, Alvarez V, Bosco P, Mancuso M, Panza F, Nacmias B, Bossu P, Piccardi P, Annoni G, Seripa D, Galimberti D, Hannequin D, Licastro F, Soininen H, Ritchie K, Blanche H, Dartigues JF, Tzourio C, Gut I, Van Broeckhoven C, Alperovitch A, Lathrop M, Amouyel P. Genome-wide association study identifies variants at CLU and CR1 associated with Alzheimer's disease. Nat Genet 2009;41:1094-9.
- Werring DJ, Frazer DW, Coward LJ, Losseff NA, Watt H, Cipolotti L, Brown MM, Jager HR. Cognitive dysfunction in patients with cerebral microbleeds on T2*-weighted gradient-echo MRI. Brain 2004;127:2265-75.

- Kato H, Izumiyama M, Izumiyama K, Takahashi A, Itoyama Y. Silent cerebral microbleeds on T2*-weighted MRI: correlation with stroke subtype, stroke recurrence, and leukoaraiosis. Stroke 2002;33:1536-40.
- Kuller LH, Shemanski L, Manolio T, Haan M, Fried L, Bryan N, Burke GL, Tracy R, Bhadelia R. Relationship between ApoE, MRI findings, and cognitive function in the Cardiovascular Health Study. Stroke 1998;29:388-98.
- 11. Vermeer SE, Longstreth WT, Jr., Koudstaal PJ. Silent brain infarcts: a systematic review. Lancet Neurol 2007;6:611-9.
- 12. van Dijk EJ, Prins ND, Vrooman HA, Hofman A, Koudstaal PJ, Breteler MM. Progression of cerebral small vessel disease in relation to risk factors and cognitive consequences: Rotterdam Scan study. Stroke 2008;39:2712-9.
- 13. Goos JD, Kester MI, Barkhof F, Klein M, Blankenstein MA, Scheltens P, van der Flier WM. Patients with Alzheimer disease with multiple microbleeds: relation with cerebrospinal fluid biomarkers and cognition. Stroke 2009;40:3455-60.
- 14. Kehoe PG, Wilcock GK. Is inhibition of the renin-angiotensin system a new treatment option for Alzheimer's disease? Lancet Neurol 2007;6:373-8.
- 15. Wright JW, Yamamoto BJ, Harding JW. Angiotensin receptor subtype mediated physiologies and behaviors: new discoveries and clinical targets. Prog Neurobiol 2008;84:157-81.
- 16. Savaskan E, Hock C, Olivieri G, Bruttel S, Rosenberg C, Hulette C, Muller-Spahn F. Cortical alterations of angiotensin converting enzyme, angiotensin II and AT1 receptor in Alzheimer's dementia. Neurobiol Aging 2001;22:541-6.
- 17. Bertram L, McQueen MB, Mullin K, Blacker D, Tanzi RE. Systematic meta-analyses of Alzheimer disease genetic association studies: the AlzGene database. Nat Genet 2007;39:17-23.
- 18. Rogaeva E, Meng Y, Lee JH, Gu Y, Kawarai T, Zou F, Katayama T, Baldwin CT, Cheng R, Hasegawa H, Chen F, Shibata N, Lunetta KL, Pardossi-Piquard R, Bohm C, Wakutani Y, Cupples LA, Cuenco KT, Green RC, Pinessi L, Rainero I, Sorbi S, Bruni A, Duara R, Friedland RP, Inzelberg R, Hampe W, Bujo H, Song YQ, Andersen OM, Willnow TE, Graff-Radford N, Petersen RC, Dickson D, Der SD, Fraser PE, Schmitt-Ulms G, Younkin S, Mayeux R, Farrer LA, St George-Hyslop P. The neuronal sortilin-related receptor SORL1 is genetically associated with Alzheimer disease. Nat Genet 2007;39:168-77.
- Andersen OM, Reiche J, Schmidt V, Gotthardt M, Spoelgen R, Behlke J, von Arnim CA, Breiderhoff T, Jansen P, Wu X, Bales KR, Cappai R, Masters CL, Gliemann J, Mufson EJ, Hyman BT, Paul SM, Nykjaer A, Willnow TE. Neuronal sorting protein-related receptor sorLA/LR11 regulates processing of the amyloid precursor protein. Proc Natl Acad Sci U S A 2005;102:13461-6.
- 20. Dodson SE, Andersen OM, Karmali V, Fritz JJ, Cheng D, Peng J, Levey AI, Willnow TE, Lah JJ. Loss of LR11/SORLA enhances early pathology in a mouse model of amyloidosis: evidence for a proximal role in Alzheimer's disease. J Neurosci 2008;28:12877-86.

- Cuenco KT, Lunetta KL, Baldwin CT, McKee AC, Guo J, Cupples LA, Green RC, St George-Hyslop PH, Chui H, Decarli C, Farrer LA, for the MSG. Association of Distinct Variants in SORL1 With Cerebrovascular and Neurodegenerative Changes Related to Alzheimer Disease. Arch Neurol 2008;65:1640-1648.
- 22. Alizadeh BZ, Njajou OT, Millan MR, Hofman A, Breteler MM, van Duijn CM. HFE variants, APOE and Alzheimer's disease: findings from the population-based Rotterdam study. Neurobiol Aging 2009;30:330-2.
- 23. Gregg EW, Yaffe K, Cauley JA, Rolka DB, Blackwell TL, Narayan KM, Cummings SR. Is diabetes associated with cognitive impairment and cognitive decline among older women? Study of Osteoporotic Fractures Research Group. Arch Intern Med 2000;160:174-80.
- 24. Kumari M, Marmot M. Diabetes and cognitive function in a middle-aged cohort: findings from the Whitehall II study. Neurology 2005;65:1597-603.
- 25. Yaffe K, Kanaya A, Lindquist K, Simonsick EM, Harris T, Shorr RI, Tylavsky FA, Newman AB. The metabolic syndrome, inflammation, and risk of cognitive decline. Jama 2004;292:2237-42.
- 26. van den Berg E, Kloppenborg RP, Kessels RP, Kappelle LJ, Biessels GJ. Type 2 diabetes mellitus, hypertension, dyslipidemia and obesity: A systematic comparison of their impact on cognition. Biochim Biophys Acta 2009;1792:470-81.

12 Grading the credibility of genetic associations in Alzheimer's disease using the Venice criteria: practiccal considerations following from the Alzgene database

Abstract

Background

The epidemiological credibility of genetic associations in Alzheimer's disease (AD) in the 'top list' of the Alzgene database is graded using the so-called Venice criteria. We aimed to evaluate the robustness of these criteria

Materials and Methods

Hypothetical results from simulated studies were added to the meta-analyses of these associations that were graded with strong credibility or with moderate or weak credibility due to inconsistency of replication (high between-study heterogeneity (I2)) or low summary odds ratio (0.87 < OR < 1.15). Robustness was quantified as the sample size needed to change the grading.

Results

For 5 out of 8 associations graded with strong credibility, the grade changed to weak credibility because of small summary OR after the addition of studies with effects similar to the lowest/highest published OR and sample sizes ranging from 80 to 2000. For 4 out of 8, new studies could introduce large I2 when their sample sizes were 400 to 1600. These associations ended up with a small summary OR and one became non-significant. Two out of four associations graded with moderate and weak evidence because of I2 >25%, could not become strong evidence because of one outlier in each analysis. Finally, associations with weak credibility due to small OR only became non-significant when new studies with no effect had sample sizes ranging over 3400 to 6600.

Conclusion

The Venice criteria are very helpful criteria to grade the credibility of genetic associations, but its practical usefulness may be limited due to outliers and small effects. Further guidance is needed on how to deal with these situations.

Introduction

Unraveling the genetic basis of Alzheimer's Disease (AD) has proven a major challenge. After the discovery of the three genes that cause Mendelian forms of the disease (APP, PSEN1 and PSEN2) and the major susceptibility gene Apolipoprotein E (APOE) there have been no further major breakthroughs. Also, the first large-scale genome-wide association studies (GWAS) did not identify loci other than APOE [1-4], but recent large consortia meta-analyses discovered several new loci [5,6], including CR1 on chromosome 1, CLU on chromosome 8 and PICALM on chromosome 11. One of the reasons for the absence of gene discoveries is that many genetic association studies in AD were too small to detect low-risk susceptibility variants. Meta-analyses combining the results of multiple small studies are therefore more powerful and necessary approaches for the identification

of AD susceptibility genes. To facilitate these meta-analyses, the online Alzgene database was constructed (www.alzgene.org). This database currently includes 1,236 studies covering almost 600 genes and 2,400 polymorphisms (accessed 1th December 2009) [7].

It is likely that not all genetic associations reported are true associations, because some metaanalyses were rather small in sample size and some associations were only borderline significant. Distinguishing true from false positive findings is one of the greatest challenges in genetic epidemiology [8,9]. Recently, new guidelines were developed by a consensus workshop of the Human Genome Epidemiology Network for grading the epidemiological strength of cumulative evidence on genetic associations [10]. These so-called Venice criteria have recently been implemented in the Alzgene database to grade the credibility of epidemiological evidence for the genetic associations. Alzgene reports a 'top list' of 35 significant genetic associations (Table 1).

The Venice criteria value genetic associations based on 1. the amount of evidence; 2. consistency of replication and 3. protection from bias. Significant meta-analyses in large populations with little heterogeneity between the results of the individual studies and without evidence for bias are considered strong evidence for genetic association, whereas any violations to these three criteria result in moderate or weak evidence.

The Venice criteria are very useful to highlight guality differences between meta-analyses, but they may face limitations when applied too strictly. First, the criteria suggest that genetic associations with moderate or weak evidence can only be valued higher and that genetic associations with strong evidence cannot change to moderate or weak evidence after updating the analyses with newer studies. Second, according to the criteria genetic associations with small effect sizes (0.87< OR <1.15) are by definition graded as weak evidence because bias cannot be ruled out, except when they are investigated in consortia, which are considered protected from bias [11]. It is clear that a single study with a small OR has limited credibility, but when multiple large studies show the same small OR there may be less reason to consider these hampered by bias. Third, the criteria provide no guidance on how to handle outlying studies and subgroup analyses. The latter is particularly relevant with regard to genetic heterogeneity which may show genetic associations in some ethnic groups but not in an overall analysis. The aim of our study was to investigate the practical implications of these limitations. We evaluated the robustness of the Venice grades of genetic associations in AD by adding hypothetical results of simulated studies to the meta-analyses. The research questions that were addressed are: 1. what sample size is needed to change a grade of strong evidence to moderate evidence due to increased heterogeneity or to weak evidence due to small OR?; 2. what sample size is needed to change associations with weak evidence into non-significant associations?: 3. to what extent are grades of moderate or weak evidence due to high heterogeneity determined by the effect of a single study?

Materials and Methods

Venice criteria

The Venice criteria grade the strength of the epidemiological evidence for genetic associations taking account of the amount of evidence, consistency of replication and protection from bias [10]. For amount of evidence, associations receive an A when the combined number of cases and controls in the minor genetic group exceeds 1000, a B when the number is between 100 and 1000, and a C when it is below 100. For consistency of replication, the degree of between study heterogeneity in study results (I2) is considered. Point estimates of I2 below 25% receive an A, I2 between 25% and 50% receive a B and I2 exceeding 50% receive a C. A C is also given to non-significant associations. For protection from bias, the guidelines propose to consider potential sources of bias at the level of individual studies including errors in phenotypes, genotypes and confounding, and at the level of meta-analysis including publication and other selective reporting biases [10]. Associations receive an A when bias is not likely to affect the presence of the association, a B when there is no demonstrable bias, but important information is missing for its appraisal and a C when there is demonstrable clear or potential bias that had invalidated the association. Meta-analyses also receive an A when the OR deviates more than 1.15-fold from the null (>1.15 or <0.87). Strong epidemiological evidence for significant association was given to the meta-analyses that received three A's, moderate evidence to those that received any B, but not any C, and weak to those that received a C in any of the three criteria.

Alzgene database

The Alzgene database is a publicly available database of published genetic association studies in AD [7]. The studies are identified through systematic literature searches, which are continuously updated. The database includes studies that are published in English in peer-reviewed journals. Meta-analyses are conducted for associations that have been investigated in at least four independent samples. All genetic associations with a significant summary OR for at least one polymorphism are presented in a 'top list'. The epidemiological credibility of all loci in the top list is graded using the Venice criteria. The criteria are implemented as described above with the amount of evidence defined as the number of minor alleles.

Analyses

We performed meta-analyses on loci presented in the Alzgene's 'top list' (Table 1), that were graded with strong and moderate epidemiological credibility and on associations with weak epidemiological credibility due to high I2 or a small OR. We used data available from the Alzgene database [7], but excluded studies in which control genotypes were out of Hardy-Weinberg Equilibrium (HWE). To test the robustness of these grades, we updated the meta-analyses with results of simulated studies. Robustness was quantified as the sample needed to change the reported grades, which could be a

result of an increased I2, a smaller OR below the threshold of 1.15 or a non-significant summary OR. When the required sample size is large, substantial evidence is needed to change the credibility of the associations.

The meta-analyses were performed in R using the Rmeta library (www.r-project.org) using random effect models. The degree of heterogeneity between studies was assessed using the I2 statistic [12]. Fixed effects meta-analyses were used in the analysis of APOE with simulated data and in the meta-analysis with simulated data introducing high I2 in order to prevent adjustment of induced heterogeneity.

For the simulated data, we assumed that the allele frequencies of the controls were equal to the overall allele frequency in the meta-analysis from the Alzgene database. The allele frequencies in cases were calculated from the frequencies in controls and the OR, assuming HWE and a 50:50 case/control ratio. The betas were calculated by taking the natural logarithm of the OR and the standard errors were calculated by dividing the beta by the square root of the chisquare, which in turn was estimated from a contingency table of the observed and expected allele frequencies in cases and controls. Beta and standard error were then added to the meta-analysis. This was repeated with increasing sample sizes until an effect was seen on the summary OR or I2.

First, we addressed the robustness of associations graded with strong evidence by investigating the sample size needed to change this grade to moderate due to increased I2 or to weak due to small OR. To study the sample size needed to increase I2 >25%, we examined two scenarios in which we added a study with an effect similar to the lowest and highest OR observed in published studies. To study the sample size needed to change the summary OR towards the null, defined by the Venice criteria as 0.87 < OR < 1.15 [10] or to non-significance, we added the same hypothetical studies as mentioned above. Additionally, we investigated a scenario where we added a hypothetical study that showed no effect (OR=1).

Next, we addressed the robustness of associations with moderate and weak evidence by investigating the sample size needed to reduce I2 or to get a non-significant summary OR.

For I2, we studied the addition of simulated data with an effect similar to the current published summary OR. We investigated two scenarios, adding a single study and adding multiple studies with a sample size of 2000. For an effect on the summary OR, we examined whether the small effect size remained significant using two scenarios in which we added a study with no effect and a study with an effect similar to the lowest/highest OR observed in published studies.

Finally, we studied whether high I2 in associations with moderate and weak evidence were due to the effect of one single study by removing one study at a time.

Top fi	Top findings of genetic associations with Alzheimer's disease listed at Alzgene.org								
#	Gene	Ethnicity	Polymorphism	N minor	12	Bias Reason (Grade)	Overall	OR	Number of
		-		(Grade)	(Grade)		Grade		studies
1	APOE (22/3/4)	Caucasian	apoe ɛ2/3/4	3525 (A)	0 (A)	(A)	Strong	3.81 (3.38-4.29)	28
2	CLU	All	rs11136000	20271 (A)	0 (A)	(A)	Strong	0.85 (0.82-0.89)	8
3	PICALM	All	rs541458	13335 (A)	0 (A)	(A)	Strong	0.87 (0.83-0.91)	6
4	TNK1	All	rs1554948	5343 (A)	9 (A)	(A)	Strong	0.86 (0.80-0.93)	6
5	ACE	Caucasian	rs1800764	1371 (A)	0 (A)	(A)	Strong	0.79 (0.68-0.92)	4
6	TFAM	All	rs2306604	1604 (A)	0 (A)	(A)	Strong	0.82 (0.72-0.94)	5
7	CST3	Caucasian	rs1064039	1203 (A)	4 (A)	(A)	Strong	1.16 (1.00-1.13)	8
8	IL1B	Caucasian	rs1143634	1206 (A)	0 (A)	(A)	Strong	1.18 (1.04-1.39)	5
9	CR1	All	rs6656401	6653 (A)	44 (B)	(A)	Moderate	1.19 (1.09-1.28)	7
10	hCG2039140	All	rs1903908	768 (B)	0 (A)	(A)	Moderate	1.23 (1.06-1.44)	4
11	SORL1	Caucasian	rs12285364	680 (B)	7 (A)	(A)	Moderate	1.26 (1.06-1.49)	9
12	CHRNB2	All	rs4845378	227 (B)	0 (A)	(A)	Moderate	0.67 (0.50-0.90)	4
13	SORCS1	All	rs600879	567 (B)	0 (A)	(A)	Moderate	1.24 (1.04-1.48)	4
14	DAPK1	All	rs4878104	4219 (A)	0 (A)	Low OR, Regr (C)	Weak	0.88 (0.82-0.95)	7
15	PRNP	Caucasian	rs1799990	3521 (A)	7 (A)	Low OR, HWE (C)	Weak	0.91 (0.83-0.99)	10
16	MTHFR	All	rs1801133	8120 (A)	22 (A)	Low OR (C)	Weak	1.13 (1.04-1.24)	25
17	GAB2	Caucasian	rs10793294	1711 (A)	78 (C)	(A)	Weak	0.69 (0.54-0.88)	5
18	LOC651924	All	rs6907175	5072 (A)	3 (A)	Low OR (C)	Weak	0.89 (0.82-0.96)	6
19	GWA_14q32.13	All	rs11622883	5031 (A)	35 (B)	Low OR (C)	Weak	0.88 (0.80-0.97)	6
20	BDNF	Caucasian	rs6265	4145 (A)	0 (A)	Low OR (C)	Weak	1.09 (1.02-1.17)	16
21	NEDD9	All	rs760678	5336 (A)	39 (B)	Low OR, Regr (C)	Weak	0.89 (0.81-0.97)	8
22	CH25H	All	rs13500	729 (B)	65 (C)	Regr (C)	Weak	1.44 (1.08-1.93)	7
23	IL1A	Caucasian	rs1800587	4749 (A)	32 (B)	Low OR (C)	Weak	1.09 (1.00-1.19)	18
24	TF	All	rs1049296	2824 (A)	30 (B)	Regr (C)	Weak	1.18 (1.04-1.33)	14
25	LOC439999	All	rs498055	5288 (A)	49 (B)	F, HWE (C)	Weak	1.15 (1.03-1.29)	7
26	CALHM1	Caucasian	rs2986017	4523 (A)	68 (C)	F (C)	Weak	1.18 (1.03-1.35)	10
27	TNF	All	rs4647198	1515 (A)	0 (A)	F, HWE (C)	Weak	1.35 (1.04-1.76)	4
28	PGBD1	All	rs3800324	550 (B)	0 (A)	Regr, F (C)	Weak	1.21 (1.02-1.24)	7
29	THRA	All	rs939348	3248 (A)	0 (A)	Low OR, Regr (C)	Weak	1.10 (1.01-1.19)	6
30	ENTPD7	All	rs911541	2241 (A)	1 (A)	Low OR, F (C)	Weak	1.10 (1.01-1.21)	4
31	IL33	All	rs11792633	5896 (A)	63 (C)	F (C)	Weak	0.84 (0.72-0.99)	4
32	GAPDHS	All	rs4806173	2623 (A)	51 (C)	F (C)	Weak	0.87 (0.75-1.00)	4
33	отс	All	rs5963409	375 (B)	18 (A)	F (C)	Weak	X-chromosomal	10
34	GALP	All	rs3745833	4132 (A)	62 (C)	Low OR, Regr (C)	Weak	1.13 (1.00-1.29)	6
35	PSEN1	All	rs165932	13867 (A)	54 (C)	Low OB, HWE (C)	Weak	0.92 (0.86-1.00)	43
00			10100002	1.0007 (A)	0.(0)	(0)		0.02 (0.00-1.00)	-0

Derived from www.alzgene.org; accessed 1st Dec 2009. In bold: the associations that were studied in the current study. Grade: based on Venice criteria [10]. OR: odds ratio, Regr: modified regression test to assess positive publication bias, HWE: deviations from Hardy-Weinberg Equilibrium, F: exclusion of first study diminishes the association

Table1

Results

Descriptives

Of the 35 significant associations presented in Table 1, eight were graded with strong evidence for association, five with moderate evidence and 22 with weak evidence for association. The 8 associations with strong evidence included 3 loci identified in GWAS and 5 loci identified in metaanalyses of candidate gene studies. Figure 1A and 1B show that the effect sizes of the polymorphisms that were graded with strong evidence remained unchanged when the number of minor alleles was higher than 1,000. With the exception of APOE, most ORs were close to the 1.15 threshold specified in the Venice criteria.

There were only two reasons why genes were graded with moderate evidence. Four out of five genetic associations had insufficient amount of data and the fifth, CR1, had high I2. Of the 22 associations that were graded with weak evidence, three did not have a sufficient amount of data, and 19 had high I2, small OR or presence of biases. The associations graded with weak evidence due to HWE deviations, TNF, LOC439999, PRNP and PSEN1 were no longer significant after the studies in which controls were out of HWE were removed from the meta-analysis. Five associations were graded with

weak evidence only because of high I2 or small OR. Figure 2 shows that the cumulative OR of CR1 did not change in time, but the cumulative OR of GAB2 did. The cumulative ORs for LOC651924, GWA_14g32.13, BDNF and II1A fluctuated around a stable value (0.89 and 1.09) after the number of minor alleles reached 2500.

Sample sizes needed to obtain an I ² higher than 25% for loci that are grad	ded
with strong evidence of association	

	Range of published OR	When OR=lowest*	When OR=highest*			
Gene		Sample size	Sample size			
CLU	0.76-0.93	4400	10 000			
PICALM	0.83-0.95	>1 000 000	6000			
TNK1	0.76-1.05	1,600	520			
TFAM	0.60-1.00	400	1600			
ACE	0.74-0.84	>1 000 000	>1 000 000			
CST3	0.88-1.60	620	500			
IL1B	1.09-1.64	>1 000 000	560			
OR range: minimum and maximum odds ratio available from published studies included						

Table 2

chapter 12

Robustness of strong evidence

Table 2 shows that for ACE it was impossible to change its strong evidence to moderate evidence because of increased I2, because I2 could not increase over 25%. For PICALM and IL1B it was only possible when the study had the same effect as the published outlier and for CLU it was only possible with relatively large sample sizes. These findings did not change when we added multiple studies with moderate sample sizes (N=2000 each) instead of one single large study (data not shown).For TNK1, TFAM and CST3 even small sample sizes could increase I2 and change the grading from strong to moderate. In addition, the new summary OR became small (0.87 < OR <1.15) and for CST3 non-significant.

Samples size needed to obtain a summary OR lower than 1.15 or non-significant result for loci that are graded with strong

evidence of association							
					Small OR	Non-significance	
	Published			When OR=1	When OR=lowest/highest*	When OR=lowest/highest*	
Gene	OR	95% CI	OR range	Sample size	Sample size	Sample size	
APOE4	3.83	3.30-4.45	2.17-9.93	39000	Not Possible	Not Possible	
CLU	0.86	0.82-0.89	0.76-0.93	4000	9000	Not Possible	
PICALM	0.87	0.83-0.91	0.83-0.95	600	2000	Not Possible	
TNK1	0.86	0.80-0.93	0.76-1.05	800	600	6000	
TFAM	0.82	0.72-0.94	0.60-1.00	1000	1000	2500	
ACE	0.79	0.68-0.92	0.74-0.84	1200	Not Possible	Not Possible	
CST3	1.15	1.01-1.33	0.88-1.60	160	80	150	
IL1B	1.19	1.04-1.36	1.09-1.64	600	1500	Not Possible	

OR: summary odds ratio of meta-analysis; 95% CI: 95% confidence interval; OR range: minimum and maximum odds ratio from published studies included in the meta-analyses listed in Table 1. Sample size: number of cases and controls (50:50); * OR of the adde hypothetical studies was the lowesthighest of the range of published OR.

Table 3 shows that a small study could lower the summary OR below the threshold of 1.15, except for APOE, CLU and ACE. For APOE, only a very large study with no effect could lower the summary OR below the threshold, while for CLU and ACE a smaller study could. Yet for all three genes, an OR of 1 is outside the range of published ORs. For more realistic OR values, it was impossible to lower the summary OR of APOE and ACE and required a large study for CLU. For the other 5 genes, studies with 80 to 2000 participants were sufficient to lower the summary OR and change the grading from strong to weak. Three associations, TNK1, TFAM and CST3, became non-significant.

Robustness of moderate and weak evidence

Of the studies that were graded with moderate or weak evidence because of high I2, for 2 associations (CR1 and GAB2), it was impossible to reduce I2 below 25% with the addition of one study, but I2 could be reduced by the addition of 8 studies each with a sample size of 2000 (Table 4). For the other 2, GWA14q32.13 and IL1A, small sample sizes of only 200 participants were sufficient to reduce I2. Table 5 shows that for LOC651924, GWA_14q32.13, BDNF, IL1A and MTHFR, a sample size between 3400 and 6600 was needed to obtain a non-significant meta-analysis result after adding a study in which OR=1. A non-significant meta-analysis could be obtained at smaller sample sizes with the addition of studies with an effect size similar to the published ORs, however, for BDNF and IL1A these ORs were major outliers.

Sample sizes needed to obtain an I ² lower than 25% for loci that are graded with moderate or weak evidence because of high I ²								
Published				When adding one study*	When adding multiple studies (N=2000)*			
Gene	OR	95% CI	OR Range	Sample size	Studies added			
CR1	1.18	1.07-1.29	0.99-1.29	>1 000 000	8			
GAB2	0.64	0.47-0.86	0.51-0.91	>1 000 000	8			
GWA14q32.13	0.89	0.80-0.97	0.76-1.00	200	Not relevant			
IL1A	1.09	1.02-1.18	0.71-2.33	200	Not relevant			

OR: summary odds ratio of meta-analysis; Sample size: number of cases and controls (50:50);

OR of the added hypothetical studies was the summary published OR

Table 4

Sample size needed to obtain a non-significant OR for loci that are graded with weak evidence

	Published			When OR=1	When OR=lowest/highest*
Gene	OR	95% CI	OR Range	Sample size	Sample size
LOC651924	0.89	0.82-0.96	0.78-0.99	6000	6200
GWA_14q32.13	0.89	0.80-0.97	0.76-1.00	5600	5600
BDNF	1.09	1.02-1.17	0.66-1.90	6200	400
IL1A	1.09	1.02-1.18	0.71-2.33	3400	200
MTHFR	1.12	1.03-1.22	0.88-1.97	6600	1200

OR: summary odds ratio of meta-analysis; 95% CI: 95% confidence interval; OR range: minimum and maximum odds ratio available from published data on Alzgene Sample size: number of cases and controls (50:50). * OR of the adde hypothetical studies was the lowes/bhighest of the range of published OR.

Table 5

Finally, we investigated whether high I of associations with moderate or weak grades could be attributed to one single study. For CR1 and GAB2, the high I2 was indeed due to one contributing study, which showed an opposite direction of effect. Excluding the respective study from the meta-analysis reduced the heterogeneity from 48% to 0% for CR1 and from 78% to 0% for GAB2. The

summary OR also became stronger with less wide 95% confidence intervals. For the two other associations that were graded with weak evidence, the grading did not change when only one study was removed at the time.



Cumulative association of loci with strong evidence for association with AD from candidate gene studies

OR: cumulative odds ratio; N minor: number of minor alleles

Figure 1a

Cumulative association of loci with strong evidence for association with AD identified in genome wide association studies



OR: cumulative odds ratio; N minor: number of minor alleles

Figure 1b

chapter 12

Cumulative association of loci with moderate or weak evidence for association with AD



OR: cumulative odds ratio; N minor: number of minor alleles

Figure 2

Discussion

The Venice criteria are very useful in valuing genetic associations with AD. By assessing the sample size, heterogeneity and the presence of bias, the criteria identify potential short-comings in the data synthesis that have to be solved before one can rely on the associations found. However, our data show that there are some concerns and limitations. First, genetic associations that are graded with strong evidence can change to weak evidence when the analyses are updated with hypothetical, but realistic new data when the summary OR becomes smaller than the threshold of 1.15. Second, a number of genetic associations with small significant effect size seem robust, but would still be graded with weak evidence because of low OR. Third, for some genetic associations, the weak or moderate grade is determined by the effect of a single study with outlying effect.

The Venice criteria grade the strength of the epidemiological evidence for genetic associations taking account of the amount of evidence, consistency of replication and protection from bias [10]. The definition of the amount of evidence as defined in the criteria as the number in the minor genetic group could be specified more clearly. The Alzgene database is currently using the number in the minor allele group, which is obviously twice the size of the minor genotype group, which may as well be considered as the minor genetic group. The sample size is, however, a measurable estimate of the amount of evidence and the only debate could be the sample size that is used as a cut off, since it appears from our study and recent genome-wide association studies that larger sample sizes than 1000 are needed to find a robust estimate.

The consistency of replication as depicted in the I2, is also a measurable estimate of evidence. Our data show, however, that the associations with strong evidence can change to moderate evidence because the addition of future studies can result in an increased I2. Especially for associations that do not include large sample sizes (TNK1, TFAM and CST3) realistic sample sizes of only hundreds can cause an increased I2. Moreover, just one study can influence I2 and when removed associations graded with moderate and weak evidence can change to strong evidence, as is seen for CR1 and GAB2 in our analysis. Also, the grades can change with the addition of future studies causing I2 to decrease below 25%, which is what one could expect. Studies with sample sizes of only 200 could result in strong evidence for GWA14q32.13 and IL1A. Although our data show that a threshold of 25% is reasonable in real data, the application of a between-study heterogeneity threshold may be misleading. It should of course be taken into account when meta-analyzing genetic associations [13,14], but it may also reveal inconsistencies in study-designs rather than inconsistencies of genetic associations. We argue that when heterogeneity is present, finding genuine heterogeneity is very important, but using heterogeneity as a sole argument in grading genetic associations may not be sufficient [15]. Moreover, even in the presence of heterogeneity, if the direction of effect is the same among studies and only the effect sizes are different, the results may point to a true association. At least the latter will not affect the overall conclusion. Another argument is that most tests for heterogeneity lack power when sample sizes are small, resulting in relatively high estimates in large studies and are therefore hard to interpret reliably [16,17]. It has been suggested by others to include confidence intervals of the heterogeneity measurement to better evaluate meta-analyses and including such boundaries in the criteria would be of additional value [12].

In addition to considering potential sources of bias on the level of individual studies and meta-analyses, the criteria also propose the use of an OR threshold. The underlying idea being that studies with small effects are more likely biased. This criterion results in weak grading for all genetic associations with an OR smaller than 1.15 (or greater than 0.87). Our results show (below), however, that a number of these associations are robust. Strong evidence is currently given to 8 genetic associations with AD. While the published OR-ranges of the majority of strong graded associations are close to 1, future studies can affect the grading. In the current study, 5 associations are graded with weak evidence after the addition of new data and 3 associations even become non-significant. Only for APOE, ACE, and CLU large (unrealistic) samples sizes are required. Weak evidence because of a small OR is given to 5 associations with AD, but our data show that the associations will only become non-significant with the addition of either very large sample sizes or studies with an unrealistic OR. Therefore these associations may be true findings, underlining the importance to consider genetic associations with small effects. Moreover, genes with small effects are expected in light of the results from recent GWAS [18].

A valuable addition to published meta-analyses would be performing cumulative meta-analyses and plotting the results can be a useful application [17]. One could expect that with time evidence will accumulate leading to a stabilizing either significant or non-significant association. There is no easy solution for tackling the robustness of genetic associations, but initiating large consortia or meta-analyze published data as we have shown, seem to be the most promising methods [11].

We limited ourselves to the Alzgene database and genes that were published in the top list at the 1st of December 2009 and meanwhile new associations have emerged. During the preparation of this manuscript there was an update of the Alzgene database. The new top list, however, reassures our conclusions rather than compromising them. In the top list there are 4 new associations graded with strong evidence. Three are new (SORL1, IL8, LDLR) and the other is GWA_14q32.13, which moved up the list due to exclusion of one single study resulting in a greater summary OR (0.84 instead of 0.89). The 3 new genes graded with strong evidence are small studies or show only a large OR in a specific ethnic group (SORL1), and these will likely change with the addition of future data. Two associations dropped from strong to weak evidence, just like we predicted. IL1B dropped because of the addition of one non-significant study (N=465, OR=0.98), resulting in a non-significant OR when excluding the studies out of HWE. TFAM dropped because of the addition of one non-significant study of HWE. TFAM dropped because of the addition of one non-significant as the study with an OR beyond the published OR range so far (N=485, OR=1.15), causing increased I2 and first study bias.

To summarize, the Venice criteria are useful criteria for the appreciation of meta-analyses of genetic associations. However, clarification is needed on how to handle the grading in the presence of outliers

and when genetic variants have small effects. Our study evaluates the Venice criteria specifically in AD, and although many findings could be extrapolated to other diseases, the criteria should be evaluated in many field synopses. With the emerging of new data, associations should be re- and meta-analyzed and guidelines should be re-evaluated [17].

In conclusion, associations with strong credibility could change with the addition of new future data and some associations with weak credibility due to small OR may have sufficient amount of evidence. The practical usefulness of the interim Venice criteria may therefore be limited and further guidance is needed on how to deal with these situations.

References

- Coon KD, Myers AJ, Craig DW, Webster JA, Pearson JV, Lince DH, Zismann VL, Beach TG, Leung D, Bryden L, Halperin RF, Marlowe L, Kaleem M, Walker DG, Ravid R, Heward CB, Rogers J, Papassotiropoulos A, Reiman EM, Hardy J, Stephan DA. A high-density wholegenome association study reveals that APOE is the major susceptibility gene for sporadic late-onset Alzheimer's disease. J Clin Psychiatry 2007;68:613-8.
- Grupe A, Abraham R, Li Y, Rowland C, Hollingworth P, Morgan A, Jehu L, Segurado R, Stone D, Schadt E, Karnoub M, Nowotny P, Tacey K, Catanese J, Sninsky J, Brayne C, Rubinsztein D, Gill M, Lawlor B, Lovestone S, Holmans P, O'Donovan M, Morris JC, Thal L, Goate A, Owen MJ, Williams J. Evidence for novel susceptibility genes for late-onset Alzheimer's disease from a genome-wide association study of putative functional variants. Hum Mol Genet 2007;16:865-73.
- Li H, Wetten S, Li L, St Jean PL, Upmanyu R, Surh L, Hosford D, Barnes MR, Briley JD, Borrie M, Coletta N, Delisle R, Dhalla D, Ehm MG, Feldman HH, Fornazzari L, Gauthier S, Goodgame N, Guzman D, Hammond S, Hollingworth P, Hsiung GY, Johnson J, Kelly DD, Keren R, Kertesz A, King KS, Lovestone S, Loy-English I, Matthews PM, Owen MJ, Plumpton M, Pryse-Phillips W, Prinjha RK, Richardson JC, Saunders A, Slater AJ, St George-Hyslop PH, Stinnett SW, Swartz JE, Taylor RL, Wherrett J, Williams J, Yarnall DP, Gibson RA, Irizarry MC, Middleton LT, Roses AD. Candidate single-nucleotide polymorphisms from a genomewide association study of Alzheimer disease. Arch Neurol 2008;65:45-53.
- 4. Bertram L, Lange C, Mullin K, Parkinson M, Hsiao M, Hogan MF, Schjeide BM, Hooli B, Divito J, Ionita I, Jiang H, Laird N, Moscarillo T, Ohlsen KL, Elliott K, Wang X, Hu-Lince D, Ryder M, Murphy A, Wagner SL, Blacker D, Becker KD, Tanzi RE. Genome-wide association analysis reveals putative Alzheimer's disease susceptibility loci in addition to APOE. Am J Hum Genet 2008;83:623-32.
- Harold D, Abraham R, Hollingworth P, Sims R, Gerrish A, Hamshere ML, Pahwa JS, Moskvina V, Dowzell K, Williams A, Jones N, Thomas C, Stretton A, Morgan AR, Lovestone S, Powell J, Proitsi P, Lupton MK, Brayne C, Rubinsztein DC, Gill M, Lawlor B, Lynch A, Morgan K, Brown KS, Passmore PA, Craig D, McGuinness B, Todd S, Holmes C, Mann D, Smith AD, Love S, Kehoe PG, Hardy J, Mead S, Fox N, Rossor M, Collinge J, Maier W, Jessen F, Schurmann B, van den Bussche H, Heuser I, Kornhuber J, Wiltfang J, Dichgans M, Frolich L, Hampel H, Hull M, Rujescu D, Goate AM, Kauwe JS, Cruchaga C, Nowotny P, Morris JC, Mayo K, Sleegers K, Bettens K, Engelborghs S, De Deyn PP, Van Broeckhoven C, Livingston G, Bass NJ, Gurling H, McQuillin A, Gwilliam R, Deloukas P, Al-Chalabi A, Shaw CE, Tsolaki M, Singleton AB, Guerreiro R, Muhleisen TW, Nothen MM, Moebus S, Jockel KH, Klopp N, Wichmann HE, Carrasquillo MM, Pankratz VS, Younkin SG, Holmans PA, O'Donovan M, Owen MJ, Williams J. Genome-wide association study identifies variants at CLU and PICALM associated with Alzheimer's disease. Nat Genet 2009;41:1088-93.
- Lambert JC, Heath S, Even G, Campion D, Sleegers K, Hiltunen M, Combarros O, Zelenika D, Bullido MJ, Tavernier B, Letenneur L, Bettens K, Berr C, Pasquier F, Fievet N, Barberger-Gateau P, Engelborghs S, De Deyn P, Mateo I, Franck A, Helisalmi S, Porcellini E, Hanon

O, de Pancorbo MM, Lendon C, Dufouil C, Jaillard C, Leveillard T, Alvarez V, Bosco P, Mancuso M, Panza F, Nacmias B, Bossu P, Piccardi P, Annoni G, Seripa D, Galimberti D, Hannequin D, Licastro F, Soininen H, Ritchie K, Blanche H, Dartigues JF, Tzourio C, Gut I, Van Broeckhoven C, Alperovitch A, Lathrop M, Amouyel P. Genome-wide association study identifies variants at CLU and CR1 associated with Alzheimer's disease. Nat Genet 2009;41:1094-9.

- Bertram L, McQueen MB, Mullin K, Blacker D, Tanzi RE. Systematic meta-analyses of Alzheimer disease genetic association studies: the AlzGene database. Nat Genet 2007;39:17-23.
- 8. Hirschhorn JN, Lohmueller K, Byrne E, Hirschhorn K. A comprehensive review of genetic association studies. Genet Med 2002;4:45-61.
- 9. Abou-Sleiman PM, Hanna MG, Wood NW. Genetic association studies of complex neurological diseases. J Neurol Neurosurg Psychiatry 2006;77:1302-4.
- Ioannidis JP, Boffetta P, Little J, O'Brien TR, Uitterlinden AG, Vineis P, Balding DJ, Chokkalingam A, Dolan SM, Flanders WD, Higgins JP, McCarthy MI, McDermott DH, Page GP, Rebbeck TR, Seminara D, Khoury MJ. Assessment of cumulative evidence on genetic associations: interim guidelines. Int J Epidemiol 2008;37:120-32.
- Janssens AC, Gonzalez-Zuloeta Ladd AM, Lopez-Leon S, Ioannidis JP, Oostra BA, Khoury MJ, van Duijn CM. An empirical comparison of meta-analyses of published gene-disease associations versus consortium analyses. Genet Med 2009;11:153-62.
- 12. Higgins JP, Thompson SG, Deeks JJ, Altman DG. Measuring inconsistency in meta-analyses. Bmj 2003;327:557-60.
- 13. Ioannidis JP. Calibration of credibility of agnostic genome-wide associations. Am J Med Genet B Neuropsychiatr Genet 2008;147B:964-72.
- 14. Moonesinghe R, Khoury MJ, Liu T, Ioannidis JP. Required sample size and nonreplicability thresholds for heterogeneous genetic associations. Proc Natl Acad Sci U S A 2008;105:617-22.
- 15. Lohmueller KE, Pearce CL, Pike M, Lander ES, Hirschhorn JN. Meta-analysis of genetic association studies supports a contribution of common variants to susceptibility to common disease. Nat Genet 2003;33:177-82.
- 16. Hardy RJ, Thompson SG. Detecting and describing heterogeneity in meta-analysis. Stat Med 1998;17:841-56.
- 17. Kavvoura FK, loannidis JP. Methods for meta-analysis in genetic association studies: a review of their potential and pitfalls. Hum Genet 2008;123:1-14.
- 18. Khoury MJ, Bertram L, Boffetta P, Butterworth AS, Chanock SJ, Dolan SM, Fortier I, Garcia-Closas M, Gwinn M, Higgins JP, Janssens AC, Ostell J, Owen RP, Pagon RA, Rebbeck TR, Rothman N, Bernstein JL, Burton PR, Campbell H, Chockalingam A, Furberg H, Little J, O'Brien TR, Seminara D, Vineis P, Winn DM, Yu W, Ioannidis JP. Genome-wide association studies, field synopses, and the development of the knowledge base on genetic variation and human diseases. Am J Epidemiol 2009;170:269-79.

12 Grading the credibility of genetic associations in Alzheimer's disease using the Venice criteria: practiccal considerations following from the Alzgene database

13 Future research

Genetic epidemiologic research in neuroscience has many challenges and there are also many opportunities for research in the genomics era. A first step of research is genome-wide association study, which are present ongoing for many neurological diseases including stroke [1,2], Alzheimer's disease (AD) [3,4], frontal lobe dementia [5] and even for rare outcomes such as Creutzfeldt-Jakob disease [6]. Although these studies of stroke and AD have been successful and have revealed three new genes, the heritability of the disease is far from explained completely. The recent identified genes involved in AD, PICALM, CLU and CR1 contribute little to the discrimination of persons who will and will not develop disease (unpublished data). Age, sex and the apolipoprotein E (APOE) gene remain the main predictors for developing AD.

How to take this research further? There are ample opportunities for finding new genes by enlarging the data sets studied in terms of the number of patients included as well as the number of markers studied. The experience in other outcomes such as lipid levels and blood pressure is that enlarging the study sample still allows the identification of new genes with small effects. Also, preliminary studies show some improvement in findings using the data of the 1000 genomes as a base population for imputations instead of HAPMAP. The variation captured with HAPMAP is limited, because it is based on 60 individuals. Considering the successes of genome-wide association studies of quantitative outcomes (e.g. blood pressure), such studies have the potential to become more successful than those of binary outcomes (e.g. hypertension). Similarly, there are opportunities for genome-wide association of endophenotypes as cognitive functioning and brain imaging data such as hippocampal volumes, generalized brain atrophy and microbleeds.

In the near future, an important trend in neurogenetic research will be the formation of large consortia with samples sizes of thousands of participants. An important question that remains to be addressed is the genetic architecture underlying diseases such as AD and related disorders. Rare mutations have been implicated in the amyloid precursor protein (APP), and presenilin 1 (PSEN1) and 2 (PSEN2) genes in familial forms of AD. Further in sporadic AD, APOE is a common variant with a large effect while PICALM, CLU and CRI are common variants with small effects. The GWA studies conducted to date do not show evidence for common variants with effects comparable to that of APOE [3,4], whereas these studies had sufficient statistical power to reveal those. However, there are two types of variants for which the present studies were underpowered: 1. variants with small effect, similar or smaller than PICALM, CLU and CR1, and 2. rare variants with large effects. A first step that should be taken in the near future is to examine the evidence for a polygenic form of inheritance in which there are a large number of variants with very small effect covering the full genome. Recently, a new method to evaluate polygenic effects has been developed and applied successfully in schizophrenia and bipolar disorder [7].

High throughput sequencing offers opportunities to evaluate the presence of rare variants with large effects. The cost of high throughput sequencing is and will go down allowing the sequencing of larger groups of patients. However, data analysis is still a challenge. Based on our findings, the first regions of interest will be the sortilin-related receptor 1 gene (SORL1). Evidence is increasing for a role in Alzheimer's disease and cerebrovascular disease [8-10]. However, there are other regions for which there is consistent evidence for linkage in families and subsequent association that remain to be elucidated by deep sequencing including chromosome 10, chromosome 1 and chromosome 3 [11-14]. The latter region is identified in our own study. One may even argue that the chromosome 19 region, around APOE, requires further sequencing, as we and others have identified signals in genome wide association studies independent of APOE (unpublished data). Sequencing of the whole genome would also be of interest to target rare variants. The highest chances of success are probably to be expected in younger cohorts and family data. Studying rare variants would add to the field and may unravel new pathways involved in the disease. It goes without saying that it is of great importance to follow-up on the genome-wide results. Not only by performing replication studies and deep sequencing, but also by performing functional studies. Animal studies, for example can add much knowledge to the expression and function of newly discovered variants.

In future neuroscience research, genetic epidemiology has to explore whether alternative genetic mechanisms explain part of the missing heritability. Are there common or rare structural variants that explain part of the missing heritability? These have been implicated in various psychiatric outcomes and APP duplication has been seen in AD [15]. Another important question is whether there are epigenetic effects or post-translational modifications that are relevant for these disorders. Last but not least, the evidence for gene-interactions is to be explored. So far, no convincing evidence for genegene interaction has been found. Although technically it is possible to study interactions across the whole genome, there are still major computational problems to be resolved.

Gene-environment interactions have been Holy Grail in the field of genetic epidemiology. Large scale gene-environment interactions are within the scope of genetic epidemiological research. Although susceptible to false-positives, statistical approaches using robust standard errors have been developed and have enabled gene-interaction studies. However, in recent years there has been little progress in finding environmental risk factors that are consistently implicated in AD and related disorders. A point of consideration is how to develop the field of epidemiology to find consistent environmental risk factors. In the last few years, genetic epidemiology has been involved in genome-wide association studies. Large consortia were formed to improve the power to pinpoint SNPs that were replicated across different studies. To move the field of epidemiological research further, such large consortia should also be formed to target classical environmental risk factors. These consortia are thus not only needed to find new genetic risk factors, they are also needed to find consistent

207

evidence for environmental risk factors. Needless to say that there is an urge to find risk factors that can be used for early prevention of a devastating disease like Alzheimer's disease. The large number of non-replicable findings in epidemiology, however, has raised questions whether epidemiology is facing its limits. The lesson to be learned from genomic research: 1) target lower p-values as they have positive predictive values; 2) replicate findings before publication; 3) form large consortia to be able to meet 1 and 2.

References

- Ikram MA, Seshadri S, Bis JC, Fornage M, DeStefano AL, Aulchenko YS, Debette S, Lumley T, Folsom AR, van den Herik EG, Bos MJ, Beiser A, Cushman M, Launer LJ, Shahar E, Struchalin M, Du Y, Glazer NL, Rosamond WD, Rivadeneira F, Kelly-Hayes M, Lopez OL, Coresh J, Hofman A, DeCarli C, Heckbert SR, Koudstaal PJ, Yang Q, Smith NL, Kase CS, Rice K, Haritunians T, Roks G, de Kort PL, Taylor KD, de Lau LM, Oostra BA, Uitterlinden AG, Rotter JI, Boerwinkle E, Psaty BM, Mosley TH, van Duijn CM, Breteler MM, Longstreth WT, Jr., Wolf PA. Genomewide association studies of stroke. N Engl J Med 2009;360:1718-28.
- 2. Debette S, Bis JC, Fornage M, Schmidt H, Ikram MA, Sigurdsson S, Heiss G, Struchalin M, Smith AV, van der Lugt A, DeCarli C, Lumley T, Knopman DS, Enzinger C, Eiriksdottir G, Koudstaal PJ, DeStefano AL, Psaty BM, Dufouil C, Catellier DJ, Fazekas F, Aspelund T, Aulchenko YS, Beiser A, Rotter JI, Tzourio C, Shibata DK, Tscherner M, Harris TB, Rivadeneira F, Atwood LD, Rice K, Gottesman RF, van Buchem MA, Uitterlinden AG, Kelly-Hayes M, Cushman M, Zhu Y, Boerwinkle E, Gudnason V, Hofman A, Romero JR, Lopez O, van Duijn CM, Au R, Heckbert SR, Wolf PA, Mosley TH, Seshadri S, Breteler MM, Schmidt R, Launer LJ, Longstreth WT, Jr. Genome-wide association studies of MRI-defined brain infarcts: meta-analysis from the CHARGE Consortium. Stroke;41:210-7.
- Harold D, Abraham R, Hollingworth P, Sims R, Gerrish A, Hamshere ML, Pahwa JS, Moskvina V, Dowzell K, Williams A, Jones N, Thomas C, Stretton A, Morgan AR, Lovestone S, Powell J, Proitsi P, Lupton MK, Brayne C, Rubinsztein DC, Gill M, Lawlor B, Lynch A, Morgan K, Brown KS, Passmore PA, Craig D, McGuinness B, Todd S, Holmes C, Mann D, Smith AD, Love S, Kehoe PG, Hardy J, Mead S, Fox N, Rossor M, Collinge J, Maier W, Jessen F, Schurmann B, van den Bussche H, Heuser I, Kornhuber J, Wiltfang J, Dichgans M, Frolich L, Hampel H, Hull M, Rujescu D, Goate AM, Kauwe JS, Cruchaga C, Nowotny P, Morris JC, Mayo K, Sleegers K, Bettens K, Engelborghs S, De Deyn PP, Van Broeckhoven C, Livingston G, Bass NJ, Gurling H, McQuillin A, Gwilliam R, Deloukas P, Al-Chalabi A, Shaw CE, Tsolaki M, Singleton AB, Guerreiro R, Muhleisen TW, Nothen MM, Moebus S, Jockel KH, Klopp N, Wichmann HE, Carrasquillo MM, Pankratz VS, Younkin SG, Holmans PA, O'Donovan M, Owen MJ, Williams J. Genome-wide association study identifies variants at CLU and PICALM associated with Alzheimer's disease. Nat Genet 2009;41:1088-93.
- 4. Lambert JC, Heath S, Even G, Campion D, Sleegers K, Hiltunen M, Combarros O, Zelenika D, Bullido MJ, Tavernier B, Letenneur L, Bettens K, Berr C, Pasquier F, Fievet N, Barberger-Gateau P, Engelborghs S, De Deyn P, Mateo I, Franck A, Helisalmi S, Porcellini E, Hanon O, de Pancorbo MM, Lendon C, Dufouil C, Jaillard C, Leveillard T, Alvarez V, Bosco P, Mancuso M, Panza F, Nacmias B, Bossu P, Piccardi P, Annoni G, Seripa D, Galimberti D, Hannequin D, Licastro F, Soininen H, Ritchie K, Blanche H, Dartigues JF, Tzourio C, Gut I, Van Broeckhoven C, Alperovitch A, Lathrop M, Amouyel P. Genome-wide association study identifies variants at CLU and CR1 associated with Alzheimer's disease. Nat Genet 2009;41:1094-9.

- Van Deerlin VM, Sleiman PM, Martinez-Lage M, Chen-Plotkin A, Wang LS, Graff-Radford 5. NR, Dickson DW, Rademakers R, Boeve BF, Grossman M, Arnold SE, Mann DM, Pickering-Brown SM, Seelaar H, Heutink P, van Swieten JC, Murrell JR, Ghetti B, Spina S, Grafman J, Hodges J, Spillantini MG, Gilman S, Lieberman AP, Kaye JA, Woltjer RL, Bigio EH, Mesulam M, Al-Sarraj S, Troakes C, Rosenberg RN, White CL, 3rd, Ferrer I, Llado A, Neumann M, Kretzschmar HA, Hulette CM, Welsh-Bohmer KA, Miller BL, Alzualde A, de Munain AL, McKee AC, Gearing M, Levey AI, Lah JJ, Hardy J, Rohrer JD, Lashley T, Mackenzie IR, Feldman HH, Hamilton RL, Dekosky ST, van der Zee J, Kumar-Singh S, Van Broeckhoven C, Mayeux R, Vonsattel JP, Troncoso JC, Kril JJ, Kwok JB, Halliday GM, Bird TD, Ince PG, Shaw PJ, Cairns NJ, Morris JC, McLean CA, Decarli C, Ellis WG, Freeman SH, Frosch MP, Growdon JH, Perl DP, Sano M, Bennett DA, Schneider JA, Beach TG, Reiman EM, Woodruff BK, Cummings J, Vinters HV, Miller CA, Chui HC, Alafuzoff I, Hartikainen P, Seilhean D, Galasko D, Masliah E, Cotman CW, Tunon MT, Martinez MC, Munoz DG, Carroll SL, Marson D, Riederer PF, Bogdanovic N, Schellenberg GD, Hakonarson H, Trojanowski JQ, Lee VM. Common variants at 7p21 are associated with frontotemporal lobar degeneration with TDP-43 inclusions. Nat Genet 2010;42:234-9.
- Mead S, Poulter M, Uphill J, Beck J, Whitfield J, Webb TE, Campbell T, Adamson G, Deriziotis P, Tabrizi SJ, Hummerich H, Verzilli C, Alpers MP, Whittaker JC, Collinge J. Genetic risk factors for variant Creutzfeldt-Jakob disease: a genome-wide association study. Lancet Neurol 2009;8:57-66.
- Purcell SM, Wray NR, Stone JL, Visscher PM, O'Donovan MC, Sullivan PF, Sklar P. Common polygenic variation contributes to risk of schizophrenia and bipolar disorder. Nature 2009;460:748-52.
- Cuenco KT, Lunetta KL, Baldwin CT, McKee AC, Guo J, Cupples LA, Green RC, St George-Hyslop PH, Chui H, Decarli C, Farrer LA, for the MSG. Association of Distinct Variants in SORL1 With Cerebrovascular and Neurodegenerative Changes Related to Alzheimer Disease. Arch Neurol 2008;65:1640-1648.
- Rogaeva E, Meng Y, Lee JH, Gu Y, Kawarai T, Zou F, Katayama T, Baldwin CT, Cheng R, Hasegawa H, Chen F, Shibata N, Lunetta KL, Pardossi-Piquard R, Bohm C, Wakutani Y, Cupples LA, Cuenco KT, Green RC, Pinessi L, Rainero I, Sorbi S, Bruni A, Duara R, Friedland RP, Inzelberg R, Hampe W, Bujo H, Song YQ, Andersen OM, Willnow TE, Graff-Radford N, Petersen RC, Dickson D, Der SD, Fraser PE, Schmitt-Ulms G, Younkin S, Mayeux R, Farrer LA, St George-Hyslop P. The neuronal sortilin-related receptor SORL1 is genetically associated with Alzheimer disease. Nat Genet 2007;39:168-77.
- 10. Bertram L, McQueen MB, Mullin K, Blacker D, Tanzi RE. Systematic meta-analyses of Alzheimer disease genetic association studies: the AlzGene database. Nat Genet 2007;39:17-23.
- Myers A, Holmans P, Marshall H, Kwon J, Meyer D, Ramic D, Shears S, Booth J, DeVrieze FW, Crook R, Hamshere M, Abraham R, Tunstall N, Rice F, Carty S, Lillystone S, Kehoe P, Rudrasingham V, Jones L, Lovestone S, Perez-Tur J, Williams J, Owen MJ, Hardy J, Goate AM. Susceptibility locus for Alzheimer's disease on chromosome 10. Science 2000;290:2304-5.

- 12. Liu F, Arias-Vasquez A, Sleegers K, Aulchenko YS, Kayser M, Sanchez-Juan P, Feng BJ, Bertoli-Avella AM, van Swieten J, Axenovich TI, Heutink P, van Broeckhoven C, Oostra BA, van Duijn CM. A genomewide screen for late-onset Alzheimer disease in a genetically isolated Dutch population. The American Journal of Human Genetics 2007;81:17-31.
- 13. Butler AW, Ng MY, Hamshere ML, Forabosco P, Wroe R, Al-Chalabi A, Lewis CM, Powell JF. Meta-analysis of linkage studies for Alzheimer's disease-A web resource. Neurobiol Aging 2009.
- Hamshere ML, Holmans PA, Avramopoulos D, Bassett SS, Blacker D, Bertram L, Wiener H, Rochberg N, Tanzi RE, Myers A, Wavrant-De Vrieze F, Go R, Fallin D, Lovestone S, Hardy J, Goate A, O'Donovan M, Williams J, Owen MJ. Genome-wide linkage analysis of 723 affected relative pairs with late-onset Alzheimer's disease. Hum Mol Genet 2007;16:2703-12.
- 15. Sleegers K, Brouwers N, Gijselinck I, Theuns J, Goossens D, Wauters J, Del-Favero J, Cruts M, van Duijn CM, Van Broeckhoven C. APP duplication is sufficient to cause early onset Alzheimer's dementia with cerebral amyloid angiopathy. Brain 2006;129:2977-83.

14 Nederlandse samenvatting en conclusies

Dit proefschrift beschrijft ons onderzoek naar genetische determinanten van cognitieve functie en aan veroudering gerelateerde hersenenveranderingen. Voor dit onderzoek hebben we gebruikt gemaakt van uitkomsten die in hoge mate erfelijk zijn en deze gebruikt als endofenotypes voor de studie van de ziekte van Alzheimer, met inbegrip van cognitieve functies, Aβ plasma waarden en aan veroudering gerelateerde hersenenveranderingen die zichtbaar zijn op magnetische resonantie beeldvorming (MRI) van de hersenen. Verschillende studie ontwerpen werden gekozen om onze onderzoeksvragen te benaderen, zoals kandidaat-gen, genoom-wijd linkage en genoom-wijd associatie onderzoek.

Eén van de meest bestudeerde kandidaat-genen in de ziekte van Alzheimer (AD) is het Apolipoproteïne E gen (APOE) [1-3]. Het ε 4 allel van dit gen is een bekend en consistent gerapporteerde risicofactor voor AD. Gebaseerd op de hypothese dat cognitieve functies relevante endofenotypes zijn voor AD, bestudeerden we de relatie tussen APOE en cognitieve functies in **hoofdstuk 3**. We vonden dat het APOE* ε 4 allel significant geassocieerd was met lagere testscores op de Adult Verbal Learning Test bij personen ouder dan 50 jaar. Dit effect van APOE* ε 4 was onafhankelijk van het effect van APOE* ε 4 op vasculaire risicofactoren en het meest uitgesproken op het leervermogen. Vergelijkbaar met de bevindingen van anderen [4], vonden we dat het APOE* ε 4 allel een effect heeft op het cognitief functioneren, maar dat dit effect in tegenstelling tot AD relatief klein is. We hebben ons in de studies naar het ontdekken van nieuwe genen gericht op cognitieve functies, omdat deze uitkomsten de meest consistente associaties toonden met APOE [4], wat zou kunnen betekenen dat cognitief functioneren het meest belovende endofenotype is.

Om nieuwe genetische regio's voor cognitieve functioneren te onderzoeken zonder voorafgaande aannames van de onderliggende pathofysiologie hebben we een hypothese-vrije genoom-wijde studie verricht op verschillende cognitieve testen. In **hoofdstuk 4** presenteren we de bevindingen van een genoom-wijde linkage analyse in de Erasmus Rucphen Family (ERF) Studie, dat een familie-studie is in een genetisch geïsoleerde populatie. Aangezien we geïnteresseerd waren in genen met een groot effect, hebben we alleen personen met lage cognitieve scores geincludeerd deze linkage analyse. Afkapwaarden voor statistische significantie en suggestie werden bepaald met simulatie studies. Een significant resultaat (LOD> 3,78) werd gevonden met regio's op chromosoom 1p13.1, 12q24.33, 19q13.43, 20p13, 21q22.13 en 21q22.3. Voor het verder definiëren van genen in de regio's, analyseerden we de regio's onder de piek met dichtere genotypering en probeerden we deze bevindingen in een groot populatie cohort, de Rotterdam Studie, te repliceren [5]. Deze analyses resulteerden in significante associaties op chromosoom 1 (p-waarde = 0,03) en 21 (p-waarde = 0,01) in ERF en de laatstgenoemde regio op 21q22.13 werd gerepliceerd in de Rotterdam Studie (nominale p -waarde 0,003). Deze regio bevat het kalium kanaal familie J 6 gen (KCNJ6).

Linkage analyse in de laagste cognitieve scores is vooral gericht op het vinden van zeldzame genetische varianten met een groot effect. Om vaker voorkomende genetische varianten met een klein effect te vinden, verrichten we een genoom-wijde associatie studie op iedereen met cognitieve functie scores en daarbij gebruikten we deze scores als continue variabelen. In hoofdstuk 5 beschrijven we een meta-analyse van verschillende genoom-wijde associatie studies uitgevoerd in de cohorten voor het Heart and Aging Research in Genomic Epidemiology (CHARGE) consortium. Dit consortium bevat grote prospectieve bevolkings-cohorten. Neuropsychologische testen waren beschikbaar voor 13 cohorten. In dit proefschrift, hebben we ons gericht op executieve functies en snelheidstaken: de Trail Making Test (TMT) deel A en B en de Stroop Kleur-en Woord-Test. De analyses werden uitgevoerd in blanke deelnemers ouder dan 45 jaar, die geen dementie hadden en geen tekenen van herseninfarcten of -bloedingen ten tijde van de cognitieve testen. De belangrijkste bevinding werd gevonden met TMT-B en een SNP op chromosoom 18. Deze SNP had een p-waarde net boven de genoom-wijde significante afkapwaarde met een p-waarde van 6,95*10-8. Deze SNP is gelegen tussen twee plausibele kandidaat-genen. Wij hebben daarna een verkennende analyse uitgevoerd waarin we gezocht hebben naar overlap tussen onze bevindingen en andere genoomwijde associatie analyses gepubliceerd voor AD en schizofrenie. Overlap met vorige genoomwijde studies werd gevonden voor meerdere andere SNPs met een p-waarde kleiner dan 1.0*10-3, waarvan de sortilin-related-receptor-1 (SORL1), de syntaxin-bindend-eiwit-6 (STXBP6) en de protocadherine-9 (PCDH9) genen het meest interessant zijn. De genen in de regio's die in onze studies werden gevonden, kunnen nieuwe inzichten geven in de mechanismen onderliggend aan de normale variatie in het cognitief functioneren. Onze bevindingen moeten echter gerepliceerd worden, wat momenteel gaande is.

Een eerste vergelijking tussen de bevindingen van de genoom-wijde linkage en associatie studies toont geen overlap in genen. Overlap zou men ook niet mogen verwachten, omdat de mechanismen die ten grondslag liggen aan de beide methoden verschillen. Linkage is vooral bedoeld om zeldzame varianten met groot effect te vinden en associatie is vooral geschikt om veel voorkomende varianten te vinden met klein of matig effect.

Van belang is ook dat we in onze genoom-wijde associatie studie geen bewijs vinden voor een rol van APOE, of voor de recent ontdekte AD genen, PICALM, CR1 en CLU [11,12] in cognitieve functie. Deze bevinding toont, dat de resultaten van analyses op endofenotypes niet één op één kan worden vertaald tot de ziekte. Ook is het van belang te vermelden dat cognitieve testen verschillende aspecten van het cognitief functioneren beoordelen. Zo vonden we inderdaad dat APOE geassocieerd was met de Adult Verbal Learning Test in hoofdstuk 3, maar niet-significant geassocieerd was met TMT-A, TMT-B of Stroop.
Na het onderzoeken van cognitieve functie als endophenotype, bestudeerden we de aan veroudering gerelateerde hersenveranderingen als een tweede groep van endofenotypen. We hebben plasma Aβ waarden gebruikt als biomarkers voor de aanwezigheid van seniele plaques en amyloïd angiopathie, en asymptomatische laesies die zichtbaar zijn op MRI van de hersenen als leeftijd-gerelateerde hersenenveranderingen. We hebben ons in dit gedeelte van het onderzoek gericht op witte stof laesies (WML), lacunair infarcten, microbloedingen en hippocampus atrofie. Al deze parameters zijn geassocieerd met een hoge bloeddruk, beroerte, dementie en cognitieve stoornissen [13-18], maar worden ook gevonden in gezonde ouderen.

We onderzochten de rol van enkele kandidaat-genen die betrokken zijn bij de regulatie van bloeddruk en amyloid metabolisme. We bestudeerden het APOE gen, genen gerelateerd aan het renineangiotensine systeem (RAS) (Angiotensine, angiotensine II receptor type 1, alfa-Adducin) en het sortilin-gerelateerde receptor gen (SORL1).

Het is bekend dat RAS-genen betrokken zijn bij de regulatie van de bloeddruk en zout-homeostase en dat RAS-eiwitten betrokken kunnen zijn bij AD [19]. Receptoren voor angiotensine II zijn aanwezig in de hersenen [20] en een verhoogde activatie van RAS wordt gezien in de hersenen van AD patiënten [21]. Zoals boven genoemd, is APOE consequent in verband gebracht met AD en er zijn steeds meer aanwijzingen dat ook SORL1 een rol speelt in AD [4,22,23]. SORL1 bestaat uit twee functionele regio's, één functionerend in het cholesterol metabolisme en de andere in de verwerking van het amyloid voorloper eiwit (APP) [24,25]. Dit gen is recent tevens in verband gebracht met cerebrovasculaire ziekte [26] en opvallend genoeg vinden we SORL1 ook in onze verkennende vergelijkende analyses van de genoom-wijde associatie studies.

We bestudeerden eerst alle vijf de varianten in relatie tot de volgende MRI endofenotypes: WML, lacunaire infarcten en microbloedingen. We onderzochten dit in de ERF-studie in een subgroep van 55 tot 75 jaar met hypertensie (**hoofdstuk 8**). Alle deelnemers hadden enige mate van WML, terwijl lacunair infarcten aanwezig waren in 15,5% en microbloedingen in 23,3% van de deelnemers. Homozygositeit voor het APOE ε4 allel werd geassocieerd met lacunaire infarcten (OR, 4.8; 95% CI, 1.2-19.3). Individuen met twee kopieën van het variant allel van 4 SNPs gelegen aan het 3'-einde van SORL1 (rs1699102, rs3824968, rs2282649, rs1010159), hadden een verhoogd risico op microbloedingen (hoogste odds ratio, 6.87; 95% CI, 1.78-26.44). Dit zou de hypothese ondersteunen dat de amyloid cascade betrokken is bij de etiologie van microbloedingen in populaties met hypertensie.

Ten tweede bestudeerden we de relatie tussen het SORL1 gen en hippocampus volume en plasma Aβ waarden in **hoofdstuk 9**. Hiervoor gebruikten we dezelfde subgroep van de ERF-studie. Hippocampus volumes werden kwantitatief gemeten met MRI en plasma Aβ waarden werden bepaald in niet-nuchtere bloedmonsters. We bestudeerden het effect van dezelfde 7 varianten binnen SORL1, waarvan eerder een relatie met AD was aangetoond. Drie varianten gelegen nabij het 3'-einde van SORL1 waren significant geassocieerd met hippocampus volume. De 3-SNP haplotypen voor rs1699102, rs3824968 en rs2282649 (CAT) en voor rs3824968, rs2282649 en rs1010159 (ATC) waren geassocieerd met grotere hippocampus volumes. We vonden geen significante associaties van deze varianten met plasma Aβ waarden.

Ten derde bestudeerden we de associatie van drie varianten binnen de angiotensine, angiotensine II type 1 receptor en adducin genen (AGT-M235T, AGTR1-C573T en ADD 1-Gly460Trp) in relatie tot A β plasma waarden in dezelfde hypertensieve subgroep van de ERF-studie (**hoofdstuk 10**). Deze varianten werden in eerdere onderzoeken in verband gebracht met vasculaire ziekten van hart en hersenen. Het AGT-M235T TT-genotype was significant geassocieerd met hogere waarden van plasma A β 42 (p = 0,008) en van getrunceerd A β n42 (p = 0,02). De associatie met A β 42 bleef significant na correctie voor potentiële confounders en het toepassen van meerdere testen. We vonden geen significante associaties tussen AGTR1-C573T of ADD 1-Gly460Trp en plasma A β .

Samenvattend is de meest interessante bevinding van de bovengenoemde studies de associaties met SORL1 die werden gevonden in diverse studie-ontwerpen. Onze kandidaat-gen analyses toonden associatie van SORL1 met cognitieve functies evenals met microbloedingen en hippocampus volume. SORL1 kwam ook naar voren in onze genoom-wijde associatie meta-analyses van de cognitieve functies. Enige terughoudendheid in de interpretatie van onze resultaten is echter nodig: onze kandidaat-gen studies werden uitgevoerd in een kleine steekproef en waren beperkt tot hypertensieve personen. Onze bevindingen moeten dus gerepliceerd worden en dienen onderzocht te worden in grotere cohorten in de algemene bevolking.

In het derde gedeelte van dit proefschrift bespreken we twee kandidaat-gen studies in AD. Met deze studies wilden we de rol van twee interessante pathofysiologische mechanismen nader onderzoeken. Eén van de mechanismen is het ijzer-metabolisme, waarvan bekend is dat ijzerstapeling kan bijdragen aan het risico op AD. Onze onderzoeksafdeling heeft eerder al de relatie tussen hemochromatose-genen en AD onderzocht. In deze studies werd een effect van het hemochromatose gen (HFE) op de leeftijd waarop AD zich openbaart gevonden [27]. De HFE-63D mutatie was geassocieerd met het op jongere leeftijd ontstaan van AD in mensen die het APOE*ɛ4 allel bij zich dragen, maar deze mutatie was niet geassocieerd met een hoger risico op AD. Andere studie-groepen vonden associaties tussen AD en andere varianten in de hemochromatose genen HFE-C282Y en -H63D, en in het transferrine gen (TF) [22]. In het Epistasis Project, waarin 1757 AD patienten en 6295 controles zijn geïncludeerd, bestudeerden we vier varianten in twee genen die een rol spelen in het ijzer-metabolisme: HFE-C282Y, HFE -H63D, TF-C2 en TF-2G/A (hoofdstuk 7). Wij repliceerden

de interactieve werking tussen HFE-282Y en TF-C2 op het risico op AD in Noord-Europeanen. We vonden ook een interactie tussen HFE-63HH en TF-2AA, die aanzienlijk werd beïnvloed door de leeftijd. De interactie tussen HFE-282Y en TF-C2 is nu twee keer gerepliceerd in een totaal aantal van 2313 AD patiënten en 7065 gezonde controles.

Er zijn desondanks beperkingen in de interpretatie van de resultaten van deze studie, die het trekken van harde conclusies belemmeren. Ten eerste werden beide interacties voornamelijk in Noord-Europeanen gevonden en werd er geen relatie tussen HFE en AD gevonden in een Noord-Spaanse populatie. Vanuit statistisch oogpunt is de uitsluiting van de Spaanse gegevens problematisch. Hoewel de allel-frequenties in Noord-Spanje verschilden van die in de Noord-Europeanen, betekent dit niet dat de relatie met AD anders zou zijn. Een tweede probleem is dat ondanks het samenvoegen van de gegevens, de aantallen klein zijn en als gevolg daarvan de statistische power van deze studie laag is, waardoor de analyse gevoelig was voor vals-positieve bevindingen.

De tweede kandidaat-gen studie, die we verrichtten met betrekking tot AD, betrof het Cathepsine D gen (CTSD) (**hoofdstuk 6**). CTSD is betrokken bij het verwerken van het amyloid voorloper eiwit (APP) en is daarom een interessant kandidaat-gen voor AD. We onderzochten CTSD in relatie tot AD in de Rotterdam Studie, dat een populatie-gebaseerde cohort-studie is (N = 7983). Daarnaast deden we een grote meta-analyse waarin onze resultaten werden samengevoegd met resultaten van eerder gepubliceerde gegevens. In de Rotterdam Studie vonden we een verhoogd risico op AD in deelnemers die het T-allel van CTSD rs17571 droegen (p-waarde 0,007). Deze associatie werd voornamelijk gevonden in mensen die niet het APOE*ε4 allel hadden. De meta-analyse toonde eveneens een significant verhoogd risico op AD bij dragers van het T-allel van rs17571 (OR 1,22, 95% CI 1.03-1.44), maar dit risico bleek onafhankelijk van de APOE*ε4 status.

Naast bovengenoemde genetische studies, hebben we ook een klassieke epidemiologische studie verricht in **hoofdstuk 2** waarin we een combinatie van cardiovasculaire risicofactoren, samengesteld in het metabool syndroom (MetS) onderzochten met betrekking tot cognitieve functies. Diabetes type 2 is een bekende risicofactor voor slechtere cognitieve prestaties [28,29], maar er zijn minder studies verschenen over de associatie van MetS en bijdragende factoren, zoals insuline-resistentie (HOMA-IR), laag adiponectin, en hoge C-reactief proteïne (CRP) waarden in het bloed [30,31]. We onderzochten of deze factoren gerelateerd zijn aan cognitieve functies en ook welke MetS componenten onafhankelijk geassocieerd zijn. Voor deze studie gebruikten we de ERF-populatie waarin uitgebreide gegevens ten aanzien van lichamelijk onderzoek, biomedische bepalingen en neuropsychologische testen beschikbaar zijn. We vonden dat overwegend vrouwen met MetS en hoge HOMA-IR lagere scores op executieve cognitieve testen hadden (p = 0,03 en p = 0,009). De meest consistente individuele component van het MetS, was de systolische bloeddruk. We interpreteerden

218

deze resultaten met voorzichtigheid, aangezien het een cross-sectionele studie betrof en slechts borderline significante p-waarden werden gevonden na correctie voor het gebruik van meerdere testen. Longitudinale studies zullen nodig zijn om beter inzicht te krijgen in de causaliteit van onze bevindingen en zullen moeten bijdragen aan meer definitieve conclusies.

Een grote uitdaging in genetisch epidemiologisch onderzoek is de validering van kandidaat-genen in AD te verbeteren. Er worden jaarlijks vele nieuwe genetische associaties gerapporteerd, maar vrijwel geen kunnen worden gerepliceerd. In **hoofdstuk 12** hebben we gesproken over de toepassing van de zogenaamde *Venice*-criteria in de validering van genetische associaties in AD. The *Venice*-criteria evalueren de kwaliteit van de sterkte van het epidemiologische bewijs voor genetische associaties aan de hand van 3 punten: de hoeveelheid data, de consistentie van replicatie en mate van bias. Hoewel de eerste twee punten robuust bleken te zijn, was de laatste moeilijker te objectiveren, vooral wanneer milde uitschieters een rol spelen en wanneer de associaties de huidige indeling van de meest sterk bewezen genetische associaties beïnvloeden en slechts enkele associaties robuust zullen blijken. The *Venice*-criteria vereisen verdere verfijningen voor de praktische toepassing in genetisch epidemiologisch onderzoek.

Dankwoord

Het is een unieke ervaring om betrokken te zijn bij onderzoeksprojecten binnen het Erasmus Medisch Centrum. De toegang tot gegevens van twee unieke studies, de Erasmus Rucphen Familie (ERF) Studie en het Erasmus Rotterdam Gezondheid Onderzoek, is een voorrecht.

In het bijzonder wil ik enkele personen bedanken zonder wie de tot stand koming van dit proefschrift niet mogelijk was geweest.

Allereerst wil ik de deelnemers aan beide onderzoeken bedanken. Zonder de deelname van de bewoners uit omgeving Rucphen en Ommoord zou dit onderzoek niet mogelijk zijn geweest.

Dan natuurlijk mijn promotor en initiatiefnemer van de ERF-studie, prof. dr. C.M. van Duijn; beste Cock, om deel uit te maken van jouw onderzoeksteam was een bijzondere ervaring. Ik heb je leren kennen als een zeer gedreven onderzoeker en ik ben dankbaar voor de talloze mogelijkheden die je me geboden hebt om me te ontwikkelen als een kritisch onderzoeker. Je ongelooflijke geheugen en kennis zijn zeer uitdagend en motiverend en bovenal leerzaam.

Ik hoop van harte dat er nieuwe data in ERF gegenereerd zullen worden.

Daarnaast mijn andere promotor en mede-initiatiefnemer van de ERF-studie, prof. dr. B.A. Oostra; beste Ben, jij was de man van de grote lijnen en soms bijna vaderlijke bezorgdheid. Het advies om toch vooral op vakantie te gaan in de afrondende fase heb ik met dank ter harte genomen. Ik heb het erg gewaardeerd dat je de stabiele factor was binnen mijn promotietraject.

Natuurlijk wil ik ook mijn co-promotor, dr. J.C. van Swieten hier bedanken; beste John, zonder jouw inmenging was ik nooit betrokken geraakt bij dit promotieonderzoek. Jij zorgde voor de link tussen de kliniek en genetische epidemiologie. Jouw input in de stukken was onmisbaar en ik heb het erg gewaardeerd dat ik altijd bij je terecht kon.

Zonder de leden van de kleine en grote commissie had deze dag niet plaats kunnen vinden. Dr. Bonifati, Dr. Biessels en Prof. Dr. Boomsma, ik wil jullie heel hartelijk danken voor het op zo'n korte termijn lezen en beoordelen van het manuscript.

Ook de leden van de grote commissie wil ik danken voor hun tijd om te opponeren. Prof. dr. Koudstaal, dank, we zullen elkaar nog veel spreken de komende jaren, ik verheug me erop. Dr. de Leeuw, als één van mijn vroegere voorgangers, vind ik het een bijzondere eer dat je tijd hebt genomen in de commissie zitting te nemen.

Het vervolg onderzoek in ERF was nooit gelukt zonder de hulp van Prof. Dr. M.M.B. Breteler en Dr. A. van der Lugt.

Beste Monique, de eerste aanzet van het onderzoek is dankzij jou tot stand gekomen. We hebben

een aantal interessante papers geschreven mede door de toegang die ik had tot de ERGO data. Ik bewonder je gedrevenheid en betrokkenheid bij mijn project.

Beste Aad, dankzij jou inmenging konden we gelukkig doorgaan met het inplannen van deelnemers en liep de inclusie geen vertraging op. Je hebt me veel geleerd door het samen beoordelen van de scans. Ik heb je feedback op de manuscripten en enthousiasme voor het onderzoek erg gewaardeerd.

Verder wil ik Suzanne Gelok en Petra Veraart bedanken. Suzanne, je werd meteen in het diepe gegooid en hebt erg veel gedaan om het onderzoek in goede banen te leiden. Samen vroeg in de ochtend naar het onderzoekscentrum om de deelnemers te ontvangen en samen laat naar huis na het afdraaien en invriezen van het bloed. Ik hoop dat je een hoop hebt geleerd. Ik was onder de indruk van je professionaliteit en betrokkenheid.

Petra, jij hebt heel wat deelnemers heen en weer gereden tussen Brabant en Rotterdam. 's Ochtends vroeg al op pad en na het file-leed ging je met frisse moed met de deelnemers aan de slag. Ik was blij dat jij het platte Brabants goed onder de knie hebt, want wat mijn Groningse achtergrond kon ik het vaak niet verstaan.

Dr. M.W. Vernooij en dr. M.A. Ikram hebben geholpen alle scans te beoordelen. Beste Meike, dank voor de wijze lessen voor het opstarten van het onderzoek en het leren lezen van de scans op microbleeds en lacunaire infarcten.

Beste Arfan, we hebben heel wat analyses besproken en veel discussies gehad. Ik heb deze discussies altijd erg leuk gevonden en was blij dat ik vaak snel bij je terecht kon. Zeker aan het eind, omdat je er voor gezorgd hebt dat de replicatie data er op tijd waren. Ik zie je vast nog veel in de kliniek.

De andere co-auteurs van de manuscripten wil ik hier ook bedanken.

Beste Carola, Tom, Wiro, Henri, Renske, Fedde, Cecile en Peter: dank voor jullie input in de manuscripten!

Peter, je prachtige tekeningen staan nog steeds op het bord. Dank voor de gesprekjes tijdens je werk in Rotterdam. Fedde, veel succes met het afronden van je proefschrift!

Cecile, jij was natuurlijk meer dan een co-auteur. Ik heb ontzettend veel van je geleerd, dank voor je geduld en uitleg!

Met veel plezier heb ik de afgelopen jaren gewerkt op de 22e. Dankzij de hulp van mijn collega's werden zelfs de ingewikkeldste analyses een uitdaging en werden de dalen minder diep.

Yurii thanks for explaining the methodological parts and your input in the analyses.

Annelous, er is een hoop veranderd sinds we samen een kamer deelden. Ik heb veel van je geleerd

en ik ben blij dat ik in het begin met jou een kamer deelde. Je hebt me leren incasseren en relativeren. Dat je een olifant niet in één keer op kunt eten, maar beetje bij beetje, stukje voor stukje, herhaal ik nog vaak voor mezelf.

Leonieke, ik vond het leuk met je samen te werken. Hoe bevalt je nieuwe baan? Veel succes met het afronden van je proefschrift.

Fan, thanks for all the patience in explaining the software and your nice companionship. How are the Dutch lessons going?

Aaron, thanks for all the help and the nice chats, especially the ones on Friday!

Maksim, it was nice working with you. I like your sense of humor.

Linda, het manuscript komt tot een eind! Veel succes met de rest van je werk.

Ayse, I enjoyed the dinners with you and Najaf, finally some girl talk.

Sophie, goed dat je de CJD registratie onder je hoede hebt genomen. Veel succes met het afronden van je master.

Zonder de volgende mensen zouden er geen genotypes bepaald zijn: Jeannette,

Andy, Bernadette, Sue Ellen en Andrea dank voor jullie hulp in het lab, ook als ik pas laat met samples terug kwam. Jeannette, je weet gelukkig altijd alles terug te vinden. Ik vond het een erg prettige samenwerking!

Dames van het secretariaat: Marjolein, Brigitte, Marion dank voor jullie hulp.

Jeannette jouw hulp met het verzamelen van alle hoofdstukken en samenbinden van de laatste versies was onmisbaar, heel veel dank hiervoor.

Pascual, thanks for being so enthusiastic about your work, it is inspiring. Thanks for your advice on the CJD patients. We will keep in touch.

Liesbeth en Danielle, ik vond het erg leuk dat jullie mijn kamergenootjes zijn geworden. Onze trajecten verschillen niet zo veel en ik heb veel gehad aan jullie luisterend oor. Liesbeth, voor jou zit het er bijna op, veel succes volgende maand. Danielle, veel plezier op je bruiloft volgende maand. Martje, heel veel succes.

I also would like to thank the international collaborators.

Tatiana, Irina, Anatoly: thanks for all the work and help. It was nice meeting you guys in person in Rotterdam.

Donald: thank you for the fruitfull collaboration.

All the consortia in CHARGE: thank you for your trust in me, so I could do the meta-analyses. Dear Tom, Stephanie and Jan, it was nice working with you and hopefully we will continue collaborating in the near future.

Dankwoord

Co-dames: hopelijk volgen er nog vele dinertjes! Jullie geven me een hoop energie.

Sun-D: tijden veranderen, wat een verschil met twaalf jaar geleden! Helemaal geweldig dat jullie vrij konden nemen om er vandaag bij te zijn. Straks is er meer tijd om weer af te spreken, ik heb er zin in.

Riek en Marijke, jullie blijven in mijn gedachten, ik wou dat ik op jullie promoties had kunnen zijn. Berit, Christine, Groningen-dames en natuurlijk SHOT! dank voor jullie interesse en geduld, hopelijk kan ik me nu weer vaker bij jullie aansluiten, want ik heb een veel verhalen en gezelligheid gemist.

Dear paranimphs, many thanks for being here today!

Najaf, I enjoyed sharing the office with you. I am happy that we met and we get along so well. I miss the coffees in the morning and the chats. Hopefully, we will find more time to drink coffee and have dinner together!

Renate, super dat mijn paranimf bent vandaag. Ik ben erg blij dat je altijd in de buurt bent en gelukkig wil je meestal wel naar me luisteren, al is het de zoveelste keer hetzelfde probleem. Ik waardeer je professionele nuchtere advies. Zoals je weet kan ik me aan het begin van de week al verheugen op het vrijdagse biertje. We hebben er nog een hoop uit te proberen, want we zijn nog niet door het assortiment heen.

Sjef, Elly, Inge, Ton: dank voor de interesse in mijn onderzoek en de hulp in ons nieuwe huis!

Pa en ma, jullie weten het wel: onmisbare stabiliteit. Het wordt tijd dat ik weer eens jullie kant op kom in plaats van andersom.

Sander, thanks! Zonder jou was het boekje er niet geweest in de vorm zoals die nu is. Ik ben erg blij met het resultaat, dankzij jouw perfectionisme is het erg mooi geworden. Ik vind het geweldig dat jij en Linda naar Nederland zijn gekomen deze week. Al woon je straks ook een eindje weg, misschien zullen we elkaar wat vaker zien. Ik mis de wandelingetjes samen met jou door de stad.

Bas, met jou samen komt alles goed. We hebben drukke tijden gehad, maar samen kunnen we de wereld aan. Jij had jouw project en ik het mijne, de volgende doen we samen. Ik verheug me eerst eens op een lekker weekendje niet klussen!

Curriculum Vitae

Curriculum vitae

Maaike Schuur was born in Annen, the Netherlands on the 11th of January 1979. After graduating from high school in 1997 at het Maartens College in Haren (Gymnasium), she moved to Groningen to study pharmacy at the University of Groningen. After two years she moved to Rotterdam to start her medical training at Erasmus University.

During her medical training she worked as a student assistant at the Department of plastical surgery at Eramus Medical Center (head of department prof. dr. S.E.R. Hovius) and was student representative in the education committee at Erasmus Medical Center.

In 2003 she was enrolled in a graduation project to study the additional value of follow-up surveillance imaging in children treated for brain tumors at the Department of Children's Oncology at Erasmus Medical Center under the supervision of dr. R.E. Reddingius, dr. C. Catsman-Berrevoets and dr. M.H. Lequin.

After her medical exam in 2005, she worked at the department of Neurology at Erasmus Medical Center (head: prof. dr. P.A.E. Sillevis Smitt). During her clinical work she decided to start a PhD project on the genetic epidemiology of cognitive function and age-related brain changes at the Genetic Epidemiology Unit of the Department of Epidemiology at Erasmus Medical Center under supervision of prof. dr. C.M. van Duijn, prof. dr. B.A. Oostra and dr. J.C. van Swieten, which she started in 2006. In 2008 she graduated from Nihes Master of Science program on Genetic Epidemiology. During her PhD project she was a member of the PhD-committee representing PhD-students enrolled in the Nihes graduation program. She was also involved in the Dutch registration of patients with Creutzfeldt Jakob Disease.

In March 2010 she started her specialty training in Neurology at the Department of Neurology at Erasmus MC Medical Center Rotterdam.

List of Publications

Schuur M, Lequin M.H., Catsman-Berrevoets C.E., Graaf de N., Reddingius R.E. The surplus value of surveillance neuro-imaging in the follow-up of children with a primary brain tumor. Pediatric Blood and Cancer, Vol 45 (4), October 2005, abstract.

Schuur M, Lequin M.H., Catsman-Berrevoets C.E., Graaf de N., Reddingius R.E. Evaluation of progressive disease in pediatric brain tumors: a comparison of one-, two- and three-dimensional measurements. Pediatric Blood and Cancer, Vol 45 (4), October 2005, abstract.

Schuur M, Meerburg B.G., Sanchez-Juan P., Rozemuller A.J.M., Duijn van C.M. Humane spongiforme encefalopathieen in Nederland. Infectieziekten Bulletin, jaargang 18, nummer 10, november 2007.

Liu F, Pardo LM, **Schuur M**, Sanchez-Juan P, Isaacs A, Sleegers K, de Koning I, Zorkoltseva IV, Axenovich TI, Witteman JC, Janssens AC, van Swieten JC, Aulchenko YS, Oostra BA, van Duijn CM. The apolipoprotein E gene and its age-specific effects on cognitive function. Neurobiology of Aging, 2008 Epub Nov 10.

Liu F, Ikram MA, Janssens AC, **Schuur M**, de Koning I, Isaacs A, Struchalin M, Uitterlinden AG, den Dunnen JT, Sleegers K, Bettens K, Van Broeckhoven C, van Swieten J, Hofman A, Oostra BA, Aulchenko YS, Breteler MM, van Duijn CM. A Study of the SORL1 Gene in Alzheimer's Disease and Cognitive Function.. J Alzheimers Dis. 2009 Jul 7.

Kovacs GG, Sanchez-Juan P, Ströbel T, **Schuur M**, Poleggi A, Nocentini S, Giannattasio C, Belay G, Bishop M, Capellari S, Parchi P, Gelpi E, Gal A, Bakos A, Molnar MJ, Heinemann U, Zerr I, Knight RS, Mitrova E, van Duijn C, Budka H. Cathepsin D (C224T) Polymorphism in Sporadic and Genetic Creutzfeldt-Jakob Disease. Alzheimer Dis Assoc Disord. 2009 Epub Jun 30.

Casper Jansen, **Maaike Schuur**, Wim G.M. Spliet, Willem A. van Gool, Cornelia M. van Duijn en Annemieke J.M. Elf jaar obducties wegens de ziekte van Creutzfeldt-Jakob in Nederland. Ned Tijdschr Geneeskd. 2009;153:A172.

Maaike Schuur, Casper Jansen, Annemieke J.M. Rozemuller, Willem A. van Gool en Cornelia M. van Duijn. Humane spongiforme encefalopathieën in Nederland. Ned Tijdschr Geneeskd. 2009;153:A173.

I. Zerr, K. Kallenberg, D.M. Summers, C Romero, A. Taratuto, U. Heinemann, M. Breithaupt, B. Meissner, U. Heinemann, M. Breithaupt, B. Meissner, A. Ladogana, **M. Schuur**, D. Galanaud, S.J. Collins, G.H. Jansen, G.B. Stokin, J, Pimentel, E. Hewer, D. Collie, P. Smith, H. Roberts, J.P. Brandel, C.M. van Duijn, M. Pocchiari, C. Begue, P. Cras, R.G. Will, P. Sanchez-Juan. Updated clinical diagnostic criteria for sporadic Creutzfeldt Jakob disease. Brain 2009;132:2659-68.

M. Schuur, J.C. van Swieten, S. Schol-Gelok, M.A. Ikram, M.W. Vernooij, F. Liu1, A. Isaacs, R. de Boer, I. de Koning, W.J. Niessen, H. Vrooman, B.A. Oostra, A. van der Lugt, M.M.B. Breteler, C.M. van Duijn. Genetic risk factors for cerebral small vessel disease in hypertensive patients from a genetically isolated population. Journal of Neurology, Neurosurgery and Psychiatry, In Press.

Onofre Combarros, Cornelia M van Duijn, Naomi Hammond, Olivia Belbin, Alejandro Arias-Vásquez, Mario Cortina-Borja, Michael G Lehmann, Yurii S Aulchenko, **Maaike Schuur**, Heike Kölsch, Reinhard Heun, Gordon K Wilcock, Kristelle Brown, Patrick G Kehoe, Rachel Harrison, Eliecer Coto, Victoria Alvarez, Panos Deloukas, Ignacio Mateo, Rhian Gwilliam, Kevin Morgan, Donald R Warden, A David Smith and Donald J Lehmann. Replication by the Epistasis Project of the interaction between the genes for IL-6 and IL-10 in the risk of Alzheimer's disease. Journal of Neuroinflammation 2009.

M. Schuur, M.A. Ikram, J.C. van Swieten, A. Isaacs, J.M. Vergeer-Drop, A. Hofman, B.A. Oostra, M.M.B. Breteler, C.M. van Duijn. Cathepsin D gene and the risk of Alzheimer's disease: A population-based study and meta-analysis. Neurobiology of Aging Epub Nov 2009.

Collins SJ, **Schuur M**, Boyd A, Lewis V, Klug GM, McGlade A, van Oosterhout A, Breedveld G, Oostra BA, Masters C, Van Duijn CM. No evidence for prion protein gene locus multiplication in Creutzfeldt-Jakob disease. Neurosci Lett 2009; Epub 472:16-8.

Schuur M, Henneman P, van Swieten JC , M.C. Zillikens, de Koning I, Janssens ACJW, Witteman JCM, Aulchenko YS, Frants RR, Oostra BA, Willems van Dijk K, van Duijn CM. Insulin-resistance and the metabolic syndrome are related to executive function in women in a large family-based study. European Journal of Epidemiology, In press.

Portfolio Summary



Summary of PhD training and teaching

Name PhD student: Maaike Schuur		PhD period: Aug 2006- Febr 2010			
Erasmus MC Departments: Epidemiology &		Promotores: prof. dr. C.M. van Duijn, prof. dr.			
Neurology		B.A. Oostra			
Research School: Nihes		Supervisor: dr. J.C. van Swieten			
1. PhD training					
		Year	Workload (ECTS)		
General courses					
- Biomedical English Writing and Communication		2008	2		
Specific courses (e.g. Research school, Medical					
Training)		2006-2008	35		
- Master in Genetic Epidemiology (Nihes)					
Ser	Seminars and workshops				
-	Weekly scientific seminars Dept. of Epidemiology		2006-2010	5	
Pre	Presentations				
-	Genetic Epidemiology Unit of Dept. of Epide	emiology:	2009	1	
	Evidence of genetic associations in Alzheimer's				
	disease - Considerations in using the Venice criteria"				
-	Genetic Epidemiology Unit of Dept. of Epide	emiology:	2009	1	
	"Cathepsin D gene and the risk of Alzheimer disease: a				
	meta-analysis";				
-	Genetic Epidemiology Unit of Dept. of Epidemiology:		2008	1	
	"Genome-wide association study on cognitive traits"				
-	Genetic Epidemiology Unit of Dept. of Epidemiology:		2008	1	
	"Genetic and epidemiological study on cognitive traits"		0000		
-	Dept. of Epidemiology: "Genome-wide asso	ociation	2008	1	
	Study on cognitive traits	faatawaa	0007	_	
-	Dept. of Neurology: Onderzoek haar risico	lactoren	2007	1	
	Constin Enidemiology Unit of Dent. of Enid	omiology	2007		
-	"Association of KIBPA gone with cognitive	function":	2007	1	
	Genetic Enidemiology Unit of Dept. of Enid	emiology:	2007	1	
-	"Genetic suscentibility of white matter lesio	ne"	2007	1	
(Inter)national conferences					
	NCHA-meeting Delft: oral presentation: "G	enome-wide	2010	1	
	association study of cognitive executive fur	nctions.	2010		
	meta-analysis of CHARGE-consortium"	ionorio.			
_	CHARGE-meeting Washington DC [•] "Prelim	ninarv results	2009	1	
	of genome-wide meta-analysis on executiv	e function"			
_	VasCog conference Singapore: poster "Ge	netic risk	2009	1	
	factors for cerebral small vessel disease in	а			
	genetically isolated population"				

PhD Portfolio

_					
-	VasCog conference Singapore: poster "Cognitive	2009	1		
	function is impaired in people with metabolic syndrome"				
-	AD/PD conference Prague: poster "Cognitive function is	2009	1		
	impaired in people with metabolic syndrome"				
-	Theraprion meeting Milan	2008	0.5		
-	Meeting of European CJD surveillance network Riga	2008	0.5		
-	Meeting of European CJD surveillance network HameIn	2007	0.5		
-	ISAO meeting Antwerpen	2006	0.5		
Other					
-	Coordinator of Dutch Registry of Creutzfeldt Jakob	2006-2009	10		
	Disease	2006-2009	5		
-	PhD member in PhD-committee	2006-2009	5		
-	PhD member in meetings of the Departmental Staff of				
	the Dept. of Epidemiology	2007-2009	5		
-	Research physician in the ERGO research center	2006-2007	1		
-	Organisation of scientific meetings of the Genetic				
	Epidemiology Unit of the Dept. of Epidemiology				
2. Teaching					
S	Supervising practicals and excursions, Tutoring				
-	Principles of Genetic Epidemiology (Nihes)	2008, 2009	20		
Supervising Master's theses					
-	Genome-wide linkage screen on cognitive traits	2009	10		
-	Genetic study of familial Creutzfeldt Jakob Disease	2009	10		
Other					
-	Review of various papers for international journals	2008-2010	5		



-

Rotterdam, May 12th 20|0

<u>Maaike Schuur</u>

.

